THE FEDERAL DEMOCRATIC REPUBLIC OF ETHIOPIA CENTRAL STATISTICAL AGENCY	
STATISTICAL REPORT ON The 2011 URBAN EMPLOYMENT UNEMPLOYMENT SURVEY	
Addis Ababa August 2011	

STATISTICAL BULLETIN

TABLE OF CONTENTS

	ents cal Tables	
Chapter I 1.1 1.2 1.3 1.4 1.5 1.6 1.7 1.8 1.9	Background of the Survey	1 3 4 5 6 7 7 9
Chapter II S	Survey Methodology	11
2.1 2.2 2.3 2.4	Coverage Sample Frame Sample Design Sample Size and Selection Scheme	11 11 11
Chapter III 3.1 3.2	Size and Socio - Demographic Characteristics of Urban Population Introduction Size and Age- Sex Structure of Urban Population and Other Socio - Demographic Data	15
Chapter IV	Economic Activity Status	35
4.1 4.2 4.3 4.4	Introduction Data Collection Approaches of the Economic Activity Status of Urban Population	35 37 37 38
Chapter V 5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9	Employed Population in Urban Areas Introduction Employment to Population Ratio Occupation and Industry Employment Status Number of Hours Worked Earning from Paid Employment Absence from Work Time related Underemployment Formal and Informal Employment Sector	59 59 60 61 61 63 64 64

Chapter V I	Unemployment in Urban Areas	185
6.1 6.2 6.3	Introduction Measurement of Unemployment Size and Rate of Urban Unemployment	186
Annex I Annex II Annex III	Survey Questionnaire Estimation Procedures of Total, Ratio and Sampling Errors Estimates and CV's for Selected Tables	233

References

LIST OF STATISTICAL TABLES

Page

Table 3.1	Estimated Total Population of Urban Areas by Age Group, Sex and Region: 2011
Table 3.2	Distribution of Urban Population, Households and Average Household Size By Region: 2011
Table 3.3	Distribution of Urban Population by Region, Sex and Relationship to the Head of Household: 2011
Table 3.4	Distribution of Urban Population Aged Ten Years and Above by Age Group, Sex, Literacy Status and Educational
Table 3.5	Attainment - COUNTRY TOTAL: 2011
	Above by Region, Sex, Training Status and Types of Training - COUNTRY TOTAL: 2011
Table 3.6	Distribution of Urban Population by Five Years Age Group, Sex and Sex Ratio - COUNTRY TOTAL: 2011
Table 4.1.	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) - COUNTRY TOTAL: 201141
Table 4.1.1	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach) -TIGRAY REGION: 2011
Table 4.1.2	Urban population Aged Ten Years and Above by Age Group, Sex, Activity Status and Rate during the Last Seven Days
Table 4.1.3	(Current Status Approach) - AFFAR REGION: 2011
Table 4.1.4	(Current Status Approach) - AMHARA REGION: 2011
Table 4.1.5	(Current Status Approach) – OROMIA REGION: 2011
Table 4.1.6	(Current Status Approach) - SOMALI REGION: 2011
Table 4.1.7	(Current Status Approach) - BENISHANGUL GUMUZ REGION: 2011
Table 4.1.8	(Current Status Approach) - SNNP REGION: 2011
Table 4.1.9	(Current Status Approach) - GAMBELLA REGION: 2011
Table 4.1.10	(Current Status Approach) - HARARI REGION: 2011
Table 4.1.11	ADDIS ABABA CITY ADMINISTRATION: 2011

Table 4.2	Population of Major Towns Aged Ten Years and Above by Sex, Activity Status and Rate during the Last Seven Days (Current Status Approach): 2011
Table 4.3	Economically Non - Active Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Reasons
	For Not Being Active During the Last Seven Days (Current Status Approach) - COUNTRY TOTAL: 2011
Table 4.4	Economically Non-active Population of Urban Areas
	Aged Ten Years and Above by Region, Sex and Reasons
	For Not Being Active During the Last Seven days
Table 5.1	(Current Status Approach):201157 Urban Population Aged Ten Years and Above by Age Group,
Table J.T	Sex and Employment to Population Ratio
	- COUNTRY TOTAL: 2011
Table 5.1.1	Urban Population Aged Ten Years and Above by Age Group, Sex and Employment to
Table 5.1.2	Population Ratio - TIGRAY REGION: 201167 Urban Population Aged Ten Years and Above by Age Group,
1 able 5.1.2	Sex and Employment to
	Population Ratio - AFFAR REGION: 201167
Table 5.1.3	Urban Population Aged Ten Years and Above by Age Group,
	Sex and Employment to Population Ratio - AMHARA REGION: 201168
Table 5.1.4	Urban Population Aged Ten Years and Above by Age Group,
	Sex and Employment to
	Population Ratio - OROMIA REGION: 2011
Table 5.1.5	Urban Population Aged Ten Years and Above by Age Group,
	Sex and Employment to
Table 5.1.6	Population Ratio - SOMALI REGION: 2011
1 able 5.1.0	Urban Population Aged Ten Years and Above by Age Group, Sex and Employment to
	Population Ratio - BENISHANGUL GUMUZ REGION: 2011
Table 5.1.7	Urban Population Aged Ten Years and Above by Age Group,
	Sex and Employment to
	Population Ratio - SNNP REGION: 2011
Table 5.1.8	Urban Population Aged Ten Years and Above by Age Group, Sex and Employment to
	Population Ratio - GAMBELLA REGION: 2011
Table 5.1.9	Urban Population Aged Ten Years and Above by Age Group,
	Sex and Employment to
	Population Ratio - HARARI REGION: 201170
Table 5.1.10	Urban Population Aged Ten Years and Above by Age Group,
	Sex and Employment to Population Ratio – ADDIS ABABA CITY ADMINISTRATION: 2011
Table 5.1.11	Urban Population Aged Ten Years and Above by Age Group,
	Sex and Employment to Population Ratio –
	DIRE DAWA ADMINISTRATION: 201170
Table 5.2	Urban Population of Major Towns Aged Ten Years and
	Above by Age Group, Sex and Employment to
Table 5.2.1	Population Ratio – TOTAL MAJOR TOWNS: 2011
Table 5.2.1	Urban Population of Major Towns Aged Ten Years and Above by Age Group, Sex and Employment to
	Population Ratio - MEKELE TOWN: 2011
Table 5.2.2	Urban Population of Major Towns Aged Ten Years and
	Above by Age Group, Sex and Employment to
	Population Ratio - ASAYITA TOWN: 201171

Table 5.2.3	Urban Population of Major Towns Aged Ten Years and Above by Age Group, Sex and Employment to
-	Population Ratio - GONDER TOWN: 2011
Table 5.2.4	Urban Population of Major Towns Aged Ten Years and
	Above by Age Group, Sex and Employment to
	Population Ratio - DESSIE TOWN: 2011
Table 5.2.5	Urban Population of Major Towns Aged Ten Years and
	Above by Age Group, Sex and Employment to
	Population Ratio - BAHIR DAR TOWN: 2011
Table 5.2.6	Urban Population of Major Towns Aged Ten Years and
	Above by Age Group, Sex and Employment to
Table 5.2.7	Population Ratio - BISHOFTU TOWN: 2011
1 able 5.2.7	Urban Population of Major Towns Aged Ten Years and
	Above by Age Group, Sex and Employment to
Table 5.2.8	Population Ratio - ADAMA TOWN: 201173 Urban Population of Major Towns Aged Ten Years and
1 able 5.2.0	Above by Age Group, Sex and Employment to
	Population Ratio - JIMMA TOWN: 2011
Table 5.2.9	Urban Population of Major Towns Aged Ten Years and
1 able 5.2.9	Above by Age Group, Sex and Employment to
	Population Ratio - SHASHEMENE TOWN: 2011
Table 5.2.10	Urban Population of Major Towns Aged Ten Years and
Table 5.2.10	Above by Age Group, Sex and Employment to
	Population Ratio - JIJIGA TOWN: 2011
Table 5.2.11	Urban Population of Major Towns Aged Ten Years and
1 abie 5.2.11	Above by Age Group, Sex and Employment to
	Population Ratio - ASOSA TOWN: 2011
Table 5.2.12	Urban Population of Major Towns Aged Ten Years and
1 abie 5.2.12	Above by Age Group, Sex and Employment to
	Population Ratio - HAWASSA TOWN: 2011
Table 5.2.13	Urban Population of Major Towns Aged Ten Years and
1 4010 3.2.13	Above by Age Group, Sex and Employment to
	Population Ratio - GAMBELLA TOWN: 2011
Table 5.2.14	Urban Population of Major Towns Aged Ten Years and
1 able 5.2.14	Above by Age Group, Sex and Employment to
	Population Ratio - HARAR TOWN: 2011
Table 5.2.15	Urban Population of Major Towns Aged Ten Years and
	Above by Age Group, Sex and Employment to
	Population Ratio - ADDIS ABABA CITY: 2011
Table 5.2.16	Urban Population of Major Towns Aged Ten Years and
	Above by Age Group, Sex and Employment to
	Population Ratio - DIRE DAWA TOWN: 2011
Table 5.3	Currently Employed Population of Urban Areas Aged Ten Years
	and Above by Age Group, Sex and
	Major Occupational Groups – COUNTRY TOTAL: 2011
Table 5.4	Currently Employed Population of Urban Areas Aged Ten Years
	and Above by Region, Sex and
	Major Occupational Groups: 2011
Table 5.5	Currently Employed Population of Major Towns Aged Ten Years
	and Above by Sex and
	Major Occupational Groups: 2011
Table 5.6	Currently Employed Population of Urban Areas Aged Ten Years
	and Above by Region, Sex and
	Minor Occupational Groups: 2011
Table 5.7	Currently Employed Population of Urban Areas Aged Ten Years
	and Above by Age Group, Sex and Major Industrial Divisions -
	COUNTRY TOTAL: 2011

Table 5.8	Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Major Industrial Divisions – COUNTRY TOTAL: 2011	
Table 5.9	Currently Employed Population of Major Towns Aged Ten Years and Above by Sex and Major Industrial Divisions: 2011	
Table 5.10	Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Minor Industrial Divisions-	107
Table 5.11	COUNTRY TOTAL: 2011 Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Status in Employment -	109
Table 5.12	COUNTRY TOTAL: 2011 Currently Employed Population of Urban Areas Aged Ten Years	
Table 5.13	and Above by Region, Sex and Status in Employment: 2011 Currently Employed Population of Major Towns Aged Ten Years	
Table 5.14	and Above by Sex and Status in Employment: 2011 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Occupation Groups and	
Table 5.15	Major Industrial Divisions - COUNTRY TOTAL: 2011 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Industrial Divisions and	
Table 5.16	Employment Status – COUNTRY TOTAL: 2011 Currently Employed Population of Urban Areas Aged Ten Years and Above by Major Occupational Groups, Sex and Status in Employment - COUNTRY TOTAL: 2011	
Table 5.17	Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Age Group and Number of Hours Worked Per Week- COUNTRY TOTAL: 2011	
Table 5.18	Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Number of Hours Worked Per Week: 2011	
Table 5.19	Currently Employed Population of Urban Areas Aged Ten Years and Above by Major Industrial Divisions, Sex and Number	
Table 5.20	of Hours Worked Per Week - COUNTRY TOTAL: 2011 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Occupational Groups and Number of Hours Worked Per Week - COUNTRY TOTAL: 2011	
Table 5.21	Currently Employed Population Urban Areas Aged Ten Years and Above by Sex, Status in Employment and Number of Hours Worked Per Week - COUNTRY TOTAL: 2011	
Table 5.22	Paid Employees of Urban Population Aged Ten Years and Above by Region, Sex and Type of Payment: 2011	
Table 5.23	Paid Employees of Urban Population Aged Ten Years and Above by Sex and Period of Payment – COUNTRY TOTAL: 2011	
Table 5.24	Paid Employees of Urban Population Aged Ten Years and Above by Sex, Period of Payment and Amount of Total Payment – COUNTRY TOTAL: 2011	
Table 5. 25	Paid Employees of Urban Population Aged Ten Years and Above by Major Industrial Divisions, Sex and Amount of Payment per month - COUNTRY TOTAL: 2011	
Table 5.26	Total Employed Population Not at Work by Region, Sex, and Reason for Not Working: 2011	
Table 5.27	Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex, Whether Available and Ready to Work Additional Hours and Percent Available and Ready to Work	
	Additional Hours – COUNTRY TOTAL: 2011	

Table 5.28	Currently Employed Population of Urban Areas Aged Ten Years and Above Who Sought or Available to Work Additional Hours and Number of Hours Worked Per Week by Age Group and Sex – COUNTRY TOTAL: 2011
Table 5.29	Currently Employed Population of Urban Areas Aged Ten Years and Above Who Sought or Available to Work Additional Hours and Type of Work Sought for by Region and Sex: 2011
Table 5.30	Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex, Employment Sector and Percentage Share of Informal- COUNTRY TOTAL I: 2011
Table 5.31	Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex, Employment Sector and Percentage Share of Informal: 2011171
Table 5.32	Currently Employed Populations of Major Towns Aged Ten Years and Above by Sex, Employment Sector and Percentage Share of Informal: 2011 172
Table 5.33	Distribution of Urban Population aged ten years and Above Working in the Informal Sector by Major Occupational Group and Region : 2011
Table 5.34	Distribution of Urban Population aged ten years and Above Working in the Informal Sector by Region, Sex and
Table 5.35	Major Branches of Industry: 2011
Table 5.36	Mean Number of Hours Worked Per Week: 2011
Table 5.37	Distribution of Urban Population Aged ten years and Above Working in the Informal Sector by Region and Sex,
Table 5.38	Size of Workers: 2011
Table 5.39	Working Place: 2011
Table 6.1	Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, COUNTRY TOTAL: 2011
Table 6.1.1	Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, TIGRAY REGION: 2011
Table 6.1.2	Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, AFFAR REGION: 2011
Table 6.1.3	Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, AMHARA REGION: 2011
Table 6.1.4	Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, OROMIA REGION: 2011
Table 6.1.5	Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, SOMALI REGION: 2011
Table 6.1.6	Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, BENISHANGUL GUMUZ REGION: 2011

Table 6.1.7	Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex and
-	Unemployment Rate, SNNP REGION: 2011
Table 6.1.8	Currently Unemployed Populations of Urban Areas
	Aged Ten Years and Above by Age Group, Sex and
-	Unemployment Rate, GAMBELLA REGION: 2011
Table 6.1.9	Currently Unemployed Populations of Urban Areas
	Aged Ten Years and Above by Age Group, Sex and
T-1.1. 0 4 40	Unemployment Rate, HARARI REGION: 2011
Table 6.1.10	Currently Unemployed Populations of Urban Areas
	Aged Ten Years and Above by Age Group, Sex and
	Unemployment Rate, ADDIS ABABA CITY ADMINISTRATION: 2011
Table 6.1.11	Currently Unemployed Populations of Urban Areas
	Aged Ten Years and Above by Age Group, Sex and
	Unemployment Rate, DIRE DAWA ADMINISTRATION: 2011
Table 6.2	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate- MAJOR TOWNS TOTAL: 2011
Table 6.2.1	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
Table 6.2.2	Unemployment Rate, MEKELE TOWN: 2011
1 able 6.2.2	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
Table 6.2.3	Unemployment Rate, ASAYITA TOWN: 2011
1 able 0.2.3	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
Table 6.2.4	Unemployment Rate, GONDER TOWN: 2011
Table 6.2.4	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and Unemployment Rate, DESSIE TOWN: 2011
Table 6.2.5	Currently Unemployed Populations of Major Towns
1 able 0.2.5	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, BAHIR DAR TOWN: 2011
Table 6.2.6	Currently Unemployed Populations of Major Towns
1 able 0.2.0	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, BISHOFTU TOWN: 2011
Table 6.2.7	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, ADAMA TOWN: 2011
Table 6.2.8	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, JIMMA TOWN: 2011
Table 6.2.9	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, SHASHEMENE TOWN: 2011
Table 6.2.10	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, JIJIGA TOWN: 2011
Table 6.2.11	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, ASOSSA TOWN: 2011
Table 6.2.12	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, HAWASSA TOWN: 2011
Table 6.2.13	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, GAMBELLA TOWN: 2011

Table 6.2.14	Currently Unemployed Populations of Major Towns Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, HARAR TOWN: 2011198
Table 6.2.15	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, ADDIS ABABA CITY ADMINISTRATION: 2011 199
Table 6.2.16	Currently Unemployed Populations of Major Towns
	Aged Ten Years and Above by Sex, Age Group and
	Unemployment Rate, DIRE DAWA TOWN: 2011 199
Table 6.3	Currently Unemployed Population of Urban Areas Aged Ten Years
	and Above by Age group, Sex, Literacy Status and
	Educational Attainment, COUNTRY TOTAL: 2011200
Table 6.4	Currently Unemployed Populations of Urban Areas Aged Ten Years
	and Above by Region, Sex, Literacy Status and
	Educational Attainment: 2011202
Table 6.5	Currently Unemployed Population of Urban Areas Aged Ten Years
	and Above by Region, Sex, Training Status and
	Types of Training: 2011 204
Table 6.6	Currently Unemployed Population of Urban Areas Aged Ten Years
	and Above by Age Group, Sex and Marital Status,
	COUNTRY TOTAL: 2011 213
Table 6.7	Currently Unemployed Population of Urban Areas Aged Ten Years
	and Above by Region, Sex and Marital Status: 2011214
Table 6.8	Currently Unemployed Population of Urban Areas Aged Ten Years
	and Above Who Were Seeking for Work by Region, Sex and Steps
	Taken to Seek Work: 2011215
Table 6.9	Currently Unemployed Population of Urban Areas Aged ten Years
	and Above by Region, Sex and Type of job looking for: 2011217
Table 6.10	Currently Unemployed Population of Urban Areas Aged Ten Years
	and Above by Age Group, Sex, Previous Work Experience
	and Duration of Unemployment, COUNTRY TOTAL: 2011 218
Table 6.11	Currently Unemployed Population of Urban Areas Aged Ten Years
	and Above by Region, Sex, Previous Work Experience and
	Duration of Unemployment: 2011220
Table 6.12	Currently Unemployed Population of Urban Areas Aged Ten Years
	and Above by Region, Sex, Types of Problem Faced to
	Establish Own Business: 2011

CAHAPTER I

BACKGROUND OF THE SURVEY

1.1 Introduction

Statistical information on all aspects of the population is vital for the design, implementation, monitoring and evaluation of economic and social development plan and policy issues. Labour force survey is one of the most important sources of data for assessing the role of the population of the country in the economic and social development process. It is useful to indicate the extent of available and unutilized human resources that must be absorbed by the national economy to ensure full employment and economic well being of the population. Statistics on the labour force further present the measurement of economic activity status and its relationship to other social and economic characteristics of the population. Seasonal and other variations as well as changes over time in the size and characteristics of employed and unemployed populations that can be monitored using up-to-date information from labour force surveys. It serves as an input for assessing the meeting of the Millennium Development Goals (MDGs). Furthermore, labour force datais also used as a springboard for monitoring and evaluation of the five years growth and transformation plan of a country.

Despite the significance of the labopur force data, the availability of reliable and timely labour force data were inadequate. The lack of reliable and timely data on different aspects of the population hinders the monitoring and evaluation of changes of developmental activities.

In order to fill the gap in data requirement for the purpose of socio-economic development planning, monitoring and evaluation, the Central Statistical Agency (CSA) has been providing labour force and related data at different levels with various contents and details. These include the 1976 Addis Ababa Manpower and Housing Sample Survey, the 1978 Survey on Population and Housing Characteristics of Seventeen Major Towns, the 1980/81 and 1987/88 Rural Labour Force Surveys

(RLFS). Also, the 1984, 1994 and 2007 Population and Housing Censuses and the 1999 and 2005 National Labour Force Surveys provided a comprehensive national labour force data representing both urban and rural areas.

Considering the development process of the country and dynamic nature of the sectors and fulfilling the demands of different data users, the CSA had launched Urban Employment Unemployment Survey program since 2003. The Agency conducted four surveys in four rounds, that is, October 2003, April 2004, April 2006 and May 2009. The results of these surveys were published in statistical and analytical bulletin number 301, 319, 373,469, 477 and 499. The 2011Urban Employment and Unemployment Survey, which was conducted from March 15 - March 30, 2011 is the six series. Unlike the previous surveys, the May 2009, May 2010 and the current survey of March 2011 have collected information from selected major towns with population size 100,000 and above including regional capitals.

These survey results mainly provide data on the main characteristics of employed and unemployed population, that is, the work force engaged or available to be engaged in the production of economic goods and services and its distribution in the various sectors of the economy during a given reference period.

In addition, data on economic activities of children were also collected to measure child labour in urban areas. For this purpose, the former minimum age limit 10 years was lower down to 5 years since May 2009. Therefore, the data in this survey were collected from those persons aged five years and over. However, for the purpose of measuring the economic activity status based on Ethiopian situation, the lower age limit was fixed in to ten years. This is because children in rural and urban areas used to work at their early age such as collection of fire wood, looking after cattle, shoeshine, street vendor, petty trading...etc. Thus, the May 2011 Urban Employment and Unemployment Survey statistical report is mainly aimed at provide information on the economic characteristics of the population aged ten years and over.

Furthermore, the 2011 UEUS provide data on employment on the informal sector, their spatial distribution and problem in the sector. This statistical report presents a brief introduction of each chapter including basic concepts based on ILO standard definitions together with statistical tables.

1.2. Objectives of the Survey

The general objective of the 2011 Urban Employment and Unemployment Survey is to provide statistical data on the characteristics and size of the economic activity status i.e. employed, unemployed population of the country at urban levels on annual basis.

The specific objectives of the survey are to:

- collect statistical data on the potential manpower and those who are available to take part in various socio-economic activities;
- update the data and determine the size and distribution of the labour force participation and the status of economic activity for different sub-groups of the population at different levels of the country; and also to study the socioeconomic and demographic characteristics of these groups;
- identify the size, distribution and characteristics of employed population i.e. working in the formal or informal employment sector of the economy and earnings from paid employees and its distribution by occupation and Industry...etc;
- provide data on the size, characteristics and distribution of unemployed population and rate of unemployment;
- provide data that can be used to assess the situation of women's employment or the participation of women in the labour force;
- > generated time series data to trace changes over time;

1.3. Content and Organization of Survey Questionnaire

The survey questionnaire is organized into six sections;

- Section 1 Area identification of the selected household: this section deals with area identification of respondents such as region, zone, wereda, etc.
- Section 2 Particulars of household members: consists questions on of the general socio-demographic characteristics of the population such as age, sex, relationship to the head of household, literacy status educational levels, types of training and marital status.
- Section 3 Economic activity during the last seven days: this section contains questions to identify whether persons were engaged in productive activities during the last seven days prior to date of interview, the status and characteristics of employed persons such as occupation, industry, employment status, hours of work, employment sector /formal and informal employment sector/ and earnings from paid employment.
- Section 4 Unemployment rate and characteristics of unemployed persons: this section also comprises questions to collect information regarding seeking and availability for work, reasons for inactivity and type of work looking for and duration of unemployment.
- Section 5 Economic activities during the last six months: this section contains questions to gather Information about the economic activity status of the population in the long reference period i.e. during the last six months.
- Section 6 Filtering questions on the informal sector of Employment. It includes whether the enterprise that a person engaged has book account and license etc,.

A structured questionnaire was used to solicit the required data in the survey. The draft questionnaire was tested by undertaking a pretest in selected kebeles (lower administrative unit) in Addis Ababa. Based on the pretest, the content, logical flow, layout and presentation of the questionnaire was amended. The questionnaire used in the field for data collection was prepared in Amharic language. Most questions have pre coded answers and column numbers were assigned for each question. A copy of the questionnaire translated in English language is attached as Annex I of this report for reference.

1.4. Training of Field Staff

The training program of enumerators, supervisors and other field office staff was conducted in two stages. The first stage training of trainers was conducted at the head quarters of the Central Statistical Agency (CSA), in Addis Ababa for five days including half day field practice. The participants were selected from professionals and associate professionals with long time experiences, branch office statisticians and coordinators who were supposed to train enumerators and supervisors during the second stage of training conducted at the branch statistical offices of CSA. The training was guided by an enumerator and supervisor manuals, which consists of detailed explanation of concepts, ideas and instructions on how to fill each entry in the questionnaire.

The second stage training was undertaken at the CSA branch offices. The trainers were those who were trained at the head office and they have conducted similar training for enumerators and field supervisors for ten days. The training consisted of theoretical classroom discussions on concepts, definitions and techniques of completing the questionnaire as well as mock interview. The class discussions were intended to exchange experiences among participants and pinpoint the areas of the survey questions that need more care and attention. There was also one day field practice for interviewing of the households and/or household members. The objectives of mock and practical interviewing of households were twofold. First, it enabled to assess how well the classroom theoretical discussions were understood

by all participants so that they could convey the same message to the enumerators and the supervisors. The second objective was to examine the difficulty, which would likely be encountered during actual fieldwork.

1.5. Organization of the Fieldwork

In order to carry out this survey with the desired level of quality, there was a need to organize a large staff that performs the various survey activities. The 25 Branch Statistical Offices of the Agency carried out the data collection operation. These offices have permanent and contract enumerators stationed in the selected enumeration areas. The data collection operation of the survey involved a total of about 695 enumerators, 48 coordinators and 170 field supervisors with an average supervisor to enumerator ratio of 1: 4.

The interviews have been done by canvassing from house-to-house. Information was collected from the head of the household or from other responsible household member. The enumerators were assigned to one sampled enumeration area, where he/she could easily converse with the respondent's dialect. However, in some cases whenever enumerator faces difficulty to converse with the respondent's dialect interpreters were used. Data collection from the field took place from 15 March – 30 March 2011.

During the survey data collection operation, close and regular supervision was undertaken at various levels. Immediately after the commencement of the data collection exercise the supervisors had made spot checking, close supervision, reinterviewing and a thorough scrutiny of filled-in questionnaires to ensure that the data collection activities are taking place according to the given instruction. The branch statistical office heads, coordinators and statisticians were also made supervision of the data collection operation. Moreover, senior staff members from the CSA head office made field supervision activities in seven branches for a week at different towns.

1.6. Data Processing

The filled-in questionnaires that were retrieved from the field were first subjected to manual editing and coding. During the fieldwork the field supervisors and the heads of branch statistical offices have checked the filled-in questionnaires and carried out some editing. However, the major editing and coding operation was carried out at the head office. All the edited questionnaires were again fully verified and checked for consistency before they were submitted to the data entry by the subject matter experts.

Using the computer edit specifications prepared earlier for this purpose, the entered data were checked for consistencies and then computer editing or data cleaning was made by referring back to the filled-in questionnaire. This is an important part of data processing operation in attaining the required level of data quality. Consistency checks and re-checks were also made based on frequency and tabulation results. This was done by senior programmers using CSPro software in collaboration with the senior subject experts from Manpower Statistics Team of the CSA.

1.7. Definitions of Concepts and Terms

Some of the major operational definitions of basic concepts are given below. Moreover, other concepts and definitions related to the economic activity status are provided under respective chapters.

<u>Urban Center</u>: it is defined as a locality with 2000 or more inhabitants. In this survey, however, for practical purposes an urban center includes the following regardless of the number of inhabitants.

- a) All administrative capitals
 - i) Regional capitals
 - ii) Zonal capitals not included in (i)

- iii) Wereda capitals not included in (i) and (ii)
- iv) Localities with Urban Dweller's Association (UDAs) not included in (I-iii).
- b) Municipal town not included in item (a) above.
- c) All localities which are not included either in item (a) or (b) above, having a population of 1000 or more persons, and whose inhabitants are primarily engaged in non-agricultural activities. Note that localities with population less than 1000 persons should be considered as rural.

Major Urban Centers: For the purpose of this study, major urban center include those selected urban centers with population 100,000 and above. It also includes region capitals irrespective of their population size.

<u>Urban Kebele</u>: is the lowest administrative unit in an urban center with its own jurisdiction. It is an association of urban dwellers (commonly known as kebele) formed by the inhabitants.

Enumeration Area (EA): is a unit of land delineated for the purpose of enumerating population and housing units without omission and duplication. An EA in rural areas usually consists of 150-200 households. On the other hand, an EA in urban areas constitutes 150-200 housing units.

Household: Consists of a person or group of persons, irrespective of whether related or not, who normally live together in the same household and housing units and have common cooking and eating arrangements.

Housing Unit : is defined as a separate and independent place of abode either intended for habitation or not intended for habitation but was occupied as a living quarter by a household during the survey period. Although intended for habitation by one household, a housing unit may, at the survey be occupied by one or more households or may be used partly for living and partly for establishment.

Head of Household: is a person who provides economic supports or manages the households. The head of the household is selected by household members for some reasons of his age or respect regardless of their sex.

Usual Member of a Household: a person is considered as usual members of a household if he or she is:-

- a) a person who continuously live/reside at least for six months and have a common cooking and eating arrangements with the household; or
- b) a person who is temporarily absent from the household at the time of the survey but his absence has not violated the six months criterion.

- c) House maids, guards, baby sitters, etc...with no other dwelling and who were staying with the household at the time of the survey.
- d) Persons who plan to live more than six months due to searching for job and transfer from job ...etc.

1.8. Organization of the Report

This statistical report contains seven chapters. The first chapter covers background to the survey, where the objectives and the contents of the survey questionnaire, training of field staff, organization of field work, concepts and definitions of key variables. Chapter II deals with the survey methodology, where scope, coverage, sample size and response rate. Chapter III deals in brief with the size and socio-demographic characteristics of the urban population such as the distribution of population by age group and sex. Chapter IV presents data on the economic activity status of the population aged ten years and over using the current status approach. Chapter V explores the size, characteristics and distribution of the employed population aged ten years and over. Chapter VI focused on the size, characteristics and distribution of the unemployed population and unemployment rate aged ten years and over. Annex I provides survey questionnaire English version, while estimation procedures of total, ratio and sampling errors; and estimates and CV's for selected tables are presented in Annex II and Annex III, respectively.

1.9 ABBEREVATIONS

CSA	Central Statistical Agency
EA	Enumeration Area
НН	Households
ICSE	International classification of Status in Employment
ILO	International Labour Organization
CSPro	Census and Survey Processing System
ISCO	International Standard Classification of Occupation
ISIC	International Standard Industrial Classification of All Economic Activities
ICLS	International Conference on Labour Statistics
KILM	Key Indicators of Labour Market
MDGs	Millennium Development Goals
NIHSP	National Integrated Household Survey Program
NLFS	National Labour Force Survey
NS	Not Stated
NOIC	National Occupation and Industrial Classification
PASDEP	Plan for Accelerated and Sustained Development to End Poverty
PSU	Primary Sampling Unit
RLFS	Rural Labour Force Survey
SNA	System of National Account
SNNPR	Southern Nations, Nationalities and Peoples Region
TVET	Technical Vocational and Educational Training
UEUS	Urban Employment and Unemployment Survey
UBEUS	Urban Bi-annual Employment and Unemployment Survey

CHAPTER II

SURVEY METHODOLOGY

2.1 COVERAGE

The 2011Urban Employment and Unemployment Survey (UEUS) covered all urban parts of the country except three zones of Afar, Six zones of Somali, where the residents are pastoralists. This survey follows household approach and covers households residing in conventional households, thus population residing in the collective quarters such as universities/colleges, hotel/hostel, monasteries and homeless population etc are not covered by this survey.

It was initially planned to cover 660 EAs and 19,800 households in the survey, but ultimately 100% of EAs and 99.68% of households were successfully covered.

2.2 SAMPLING FRAME

The list of households obtained from the 2007 population and housing census is used to select EAs. A fresh list of households from each EA was prepared at the beginning of the survey period. The list was then used as a frame in order to select 30 households from sample EAs.

2.3 SAMPLE DESIGN

For the purpose of the survey the country was divided into two broad categories. That is major urban center and other urban center categories.

Category I:- Major urban centers:- In this category all regional capitals and five other major urban centers that have a high population size as compared to others were included. Each urban center in this category was considered as a reporting level. The category has a total of 16 reporting levels. In this category, in order to select the sample, a stratified two-stage cluster sample design was implemented. The primary sampling units were EAs of each reporting level. From each sample EA 30 households were then selected as a Second Stage Unit (SSU).

Category II: - Other urban centers: Urban centers in the country other than those under category I were grouped into this category. A domain of other urban centers is formed for each region. Consequently 8 reporting levels were formed in this category. Harari, Addis Ababa and Dire Dawa do not have urban centers other than that grouped in category I. Hence, no domain was formed for these regions under this category.

A stratified three stage cluster sample design was also adopted to select samples from this category. The primary sampling units were urban centers and the second stage sampling units were EAs. From each EA 30 households were selected at the third stage and the survey questionnaires administered for all of them.

2.4 SAMPLE SIZE AND SELECTION SCHEME

Category I: - In this category 393 EAs and 11,790 households were selected from 16 major urban centers. Sample EAs from each reporting level in this category were selected using probability proportional to size /PPS/ with systematic sampling method; size being number of households obtained from the 2007 population and housing census. From the fresh list of households prepared at the beginning of the survey, 30 households per EA were selected using systematic random sampling techniques /SRS/.

Category II:- 86 urban centers, 267 EAs and 8,010 households were selected in this category. Urban centers from each domain and EAs from each urban center were selected using probability proportional to size /PPS/ with systematic sampling method; size being number of households obtained from the 2007 Population and housing census. From the listing of households in each EA then 30 households were selected using systematic random sampling techniques /SRS/ and the survey questionnaire was administered for the 30 sample households.

The distribution of planned and covered EAs and households and the Estimation procedures are given in the appendix.

Appendix I: Number of planned and actually covered sampling units (EAs and households) of the 2011 (2003 E.C) Urban Employment and Unemployment Survey (UEUS).

Region	Stratum	Enumeration Areas		Households	
	Major Urban/	Planned	Covered	Planned	Covered
	Other Urban				
Tigray	Mekele	23	23	690	690
	Tigray other urban	28	28	840	840
Affar	Asayita	15	15	450	450
	Affar other urban	14	14	420	420
Amhara	Bahir dar	23	23	690	688
	Gonder	23	23	690	690
	Dessie	23	23	690	689
	Amhara other urban	54	54	1620	1620
Oromiya	Debrezeit/Bishoftu/	22	22	660	660
	Nazreth/Adama/	23	23	690	690
	Jimma	23	23	690	690
	Shashmene	23	23	690	688
	Oromiya other urban	83	83	2490	2490
Somalie	Jijiga	20	20	600	600
	Somalie other urban	15	15	450	449
Benishangul -	Asosa	15	15	450	450
Gumuz	Benishangul -Gumuz	13	13	390	390
	other urban				
S.N.N.P.	Awassa	23	23	690	687
	S.N.N.P other urban	51	51	1530	1529
Gambela	Gambella	14	14	420	421
	Gambella other urban	9	9	270	270
Harari	Hareri	23	23	690	688
Addis Ababa	Addis ababa	75	75	2250	2247
Dire Dawa	Diredawa	25	25	750	750
Total		660	660	19800	19786

CHAPTER III

SIZE AND SOCIO-DEMOGRAPHIC CHARACTERISTICS OF URBAN POPULATION

3.1 Introduction

The figure in this section refers to the dejure population residing in conventional households. The dejure population comprises all the people who belong to a given area at a given time by virtue of usual residence. Thus, excluded are visitors, persons residing in collective quarters (hotel/hostel, boarding schools, prisons...etc.) as well as homeless persons.

This chapter presents some highlights on the basic socio-demographic characteristics of urban population of the country. The topics covered in this chapter include population size; distribution of the population by age and sex; sex ratio, literacy status, educational level, training status and type of training.

3.2 Size and Age-Sex Structure of Urban Population and Other Socio-Demographic Data

This section portrays a set of tables that depict the main socio-demographic characteristics of the urban population as of 2011. The distribution of estimated total population of urban areas by age group, sex and region is shown in Table 3.1. Table 3.2 presents the distribution of urban population, households and average household size by region. The average household size is computed from the total population to that of the total households. Data on household size and distribution of persons per household can sometimes be used as a proxy measure of crowdedness of population and has a great implication to health and housing problem.

The distribution of urban population by region, sex and the relationship of the respondents to that of the head of household is presented in Table 3.3. The literacy status and educational attainment of urban population at country level by age group and sex is given in Table 3.4. As clearly indicated in this tables secondary education refers to those persons who did not complete high school education, that is, grades 9 - 11 in old curriculum and grade 9 in new curriculum, while secondary education completed covers those persons who completed high school education that means grade 12 in old curriculum and grade 10 in new curriculum.

The 2011 Urban Employment and Unemployment Survey also collected data on status and types of training of urban population aged ten years and above. Table 3.5 contains the status and type of training received classified by region and sex.

Besides, the overall sex ratio for successive five-year age group for urban areas of the country is presented in Table 3.6. Sex ratio is defined as the number of males per 100 females in a given population. The sex composition has an effect on economic activities through changing the relative size of the working population.

	Region													
Age Group	Country					1/6	Benishangul				Addis	Dire		
and Sex	-Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa		
All Ages		3 - 7						-						
Total	13,093,359	911,785	191,483	2,335,662	3,930,354	639,638	123,712	1,618,465	95,914	103,186	2,913,745	229,415		
Male	6,241,603			1,084,376			59,698	794,568	48,374		1,349,045	•		
Female	6,851,757			1,251,286			64,015	823,898	47,540		1,564,700	,		
0-4														
Total	1,347,646	95,184	23,437	222,929	445,841	110,017	13,934	175,729	15,174	10,871	208,966	25,565		
Male	700,305	47,484	12,176	115,906	228,105	58,796	7,426	92,561	8,802	5,666	108,569	14,814		
Female	647,341	47,700	11,261	107,023	217,735	51,221	6,508	83,167	6,373	5,205	100,397	10,751		
5-9														
Total	1,348,633	102,759	22,673	240,246	420,827	97,917	13,899	201,815	14,314	9,968	199,926	24,289		
Male	675,233	49,399	12,011	117,038	212,441	53,950	6,951	99,593	7,887	4,771	98,665	12,526		
Female	673,400	53,360	10,662	123,208	208,386	43,967	6,949	102,222	6,427	5,197	101,261	11,763		
10-14														
Total	1,513,511	112,368	23,339	271,156	471,532	89,658	15,898	218,794	13,753	11,296	258,791	26,926		
Male	713,106	56,401	12,733	129,499	218,748	48,608	7,452	108,585	6,492	5,861	105,438	13,291		
Female	800,405	55,967	10,606	141,657	252,784	41,050	8,446	110,210	7,261	5,435	153,353	13,636		
15-19														
Total	1,735,484		17,048	320,387	517,642	72,294	16,846	227,984	11,233	10,554	389,937	28,694		
Male	751,452	52,721	7,524	137,353	231,607	37,119	6,282	101,769	5,446	5,280	153,156	13,195		
Female	984,032	70,144	9,525	183,034	286,035	35,175	10,564	126,215	5,788	5,273	236,781	15,499		
20-24														
Total	1,556,477		16,340	270,185	480,238	55,276	13,352	178,026	10,037	9,783	393,031	26,815		
Male	679,531	37,711	5,904	111,890	214,739	25,671	5,490	81,545	4,552	4,110	177,419	10,500		
Female	876,947	65,682	10,436	158,295	265,499	29,605	7,863	96,481	5,485	5,672	215,612	16,315		
25-29														
Total	1,470,711	87,570	22,788	258,382	437,867	48,163	13,229	167,408	11,213	12,211	387,359	24,522		
Male	699,615	38,755	9,315	123,400	218,628	21,638	6,311	78,076	4,245	5,896	180,563	12,786		
Female	771,096	48,814	13,473	134,981	219,239	26,525	6,918	89,332	6,967	6,315	206,796	11,736		
30-34														
Total	957,508	62,212	16,743	146,758	293,600	39,467	8,685	122,780	6,211	8,450	239,171	13,431		
Male	498,411	28,797	9,236	72,134	157,134	19,372	4,382	64,878	3,363	3,744	128,616	6,754		
Female	459,097	33,415	7,507	74,624	136,466	20,095	4,304	57,901	2,848	4,705	110,555	6,678		
35-39														
Total	850,679	55,273	13,253	144,231	242,764	31,358	9,264	107,982	5,496	8,073	216,443	16,542		
Male	401,785	24,375	8,090	60,439	119,082	14,072	5,112	52,822	3,592	4,381	102,335	7,485		
Female	448,894	30,898	5,163	83,792	123,682	17,285	4,152	55,161	1,904	3,692	114,108	9,057		

 Table 3.1
 Estimated Total Population of Urban Areas by Age Group, Sex and Region: 2011

Table 3.1 Co	pnt'd											
Age Group						Re	gion					
and Sex	Country -Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
40-44												
Total	542,611	37,346	10,038	101,489	149,371	29,435	4,341	64,209	2,587	4,430	128,656	10,709
Male	298,454	18,607	4,409	55,606	86,072	15,543	2,472	37,428	1,395	2,219	69,369	5,334
Female	244,158	18,739	5,629	45,882	63,299	13,893	1,868	26,782	1,192	2,211	59,287	5,375
45-49												
Total	440,017	26,585	6,988	81,400	126,519	15,929	3,887	47,339	2,264	4,276	115,370	9,461
Male	215,355	12,903	4,153	38,128	65,151	8,347	1,976	24,790	865	2,296	51,776	4,971
Female	224,662	13,682	2,835	43,272	61,368	7,582	1,911	22,549	1,399	1,979	63,594	4,490
50-54												
Total	347,848	24,416	5,782	62,181	86,702	16,747	3,888	33,477	1,075	3,409	104,177	5,994
Male	161,285	10,241	2,029	30,080	36,574	8,511	1,959	17,097	507	1,311	50,208	2,768
Female	186,562	14,174	3,753	32,102	50,128	8,236	1,928	16,379	567	2,099	53,969	3,227
55-59												
Total	245,521	20,522	4,213	48,381	70,617	7,659	1,856	21,843	862	2,933	61,945	4,691
Male	113,124	9,250	3,149	20,384	33,417	4,645	1,065	10,590	343	1,323	26,308	2,650
Female	132,397	11,272	1,064	27,998	37,200	3,014	791	11,253	519	1,610	35,637	2,041
60-64												
Total	234,669	18,321	4,131	52,457	57,881	8,716	1,510	15,206	895	2,441	69,415	3,696
Male	104,252	7,778	2,852	18,953	25,417	4,372	733	9,334	443	1,153	31,775	1,440
Female	130,417	10,543	1,280	33,504	32,463	4,344	777	5,872	452	1,288	37,640	2,257
65+												
Total	502,044	42,971	4,710	115,479	128,952	17,003	3,124	35,874	800	4,493	140,558	8,079
Male	229,695	16,791	2,939	53,565	56,998	11,602	2,087	15,499	443	1,908	64,848	3,015
Female	272,349	26,181	1,771	61,914	71,954	5,401	1,037	20,375	357	2,585	75,711	5,064

Region, Population and Household					Hou	sehold size						Average
size	Total	1	2	3	4	5	6	7	8	9	10 & above	Household Size
COUNTRY TOTAL	Total	ļ	2	5	4	5	0	1	0	3	above	0.20
Household size	3.491.159	552 633	576.634	632.937	602,212	441.386	309,278	173 070	101.567	53.251	47.283	3.8
Population	13,092,792		,	,	,	, = = =	1,855,667	-,	- ,	/ -	,	5.0
TIGRAY REGION	13,032,732	552,055	1,155,200	1,030,012	2,400,040	2,200,951	1,000,007	1,217,001	012,000	479,200	500,909	
Household size	260,014	48,034	50,052	46,700	38,396	33,829	20,770	12,115	4,625	2,304	3,189	3.5
Population	911,785	48,034	100,104	140,100	153,584	169,144	124,619	84,808	36,997	20,739	33,656	5.5
AFFAR REGION	311,703	-0,00-	100,104	140,100	100,004	103,144	124,013	04,000	50,557	20,700	55,050	
Household size	58,039	11,299	10,147	14,355	8,202	6,675	3,836	1,270	1,778	394	84	3.3
Population	191,483	11,299	20,293	43,065	32,807	33,377	23,015	8,891	14,223	3,543	971	0.0
AMHARA REGION	191,400	11,233	20,233	+0,000	52,007	55,577	20,010	0,031	17,220	0,040	571	
Household size	669 409	112,269	129,647	131,084	114.996	79,834	46.146	28,346	14,566	8,086	4.435	3.5
Population	2,335,513		259,295	393,252	459,983	399,169	276,878	198,423		72,772	46,943	0.0
OROMIA REGION	2,000,010	112,203	200,200	000,202	+55,505	555,105	210,010	130,423	110,001	12,112	-0,3-3	
Household size	1,092,401	180 318	182,393	204,732	195,539	126,343	96,341	48.644	28,013	12,180	8,897	3.6
Population	3,930,354		364,786	614,195	782,158	631,714	578,045	-) -			95,900	5.0
SOMALI REGION	0,000,004	103,510	504,700	014,135	702,100	001,714	570,045	040,000	224,100	103,024	35,300	
Household size	129.434	15.485	14.743	15.021	15,358	17,157	15,229	12,117	8.889	6.610	8,824	4.9
Population	639,638	15,485	29,485	45,063	61,431	85,786	91,376	84,821	71,115	59,494	95,584	4.5
BENISHANGUL GUMUZ REGION	000,000	10,400	23,403	+0,000	01,401	00,700	31,570	04,021	71,110	55,757	30,004	
Household size	36,213	7,854	5,899	6,474	6,001	4,603	3,070	835	743	343	389	3.4
Population	123,294	7,854	11,797	19,421	24,006	23,017	18,422	5,848	5,947	3,087	3,895	0.4
SNNP REGION	120,204	7,004	11,757	10,421	24,000	20,017	10,422	0,040	0,047	0,007	0,000	
Household size	423,099	73,613	67,933	59,799	71,522	55,836	41,154	27,198	13,943	7,581	4,522	3.8
Population	1,618,465	73,613	135,865	179,396	286,087	279,180	246,924			68,233	47,239	0.0
GAMBELLA REGION	1,010,400	70,010	100,000	170,000	200,007	275,100	240,024	100,000	111,041	00,200	47,200	
Household size	23,374	3,932	2,648	3,292	4,370	3,536	2,078	1,410	1,119	433	554	4.1
Population	95,914	3,932	5,296	9,877	17,479	17,680	12,471	9,872	8,955	3,900	6,451	
HARARI REGION	00,011	0,002	0,200	0,011	17,170	11,000	12,111	0,012	0,000	0,000	0,101	
Household size	30,939	5,725	6,358	6,012	5,546	3,293	1,634	1,369	569	257	175	3.3
Population	103,186	5,725	12,717	18,035	22,185	16,465	9,807	9,583	4,552	2,313	1,804	0.0
ADDIS ABABA CITY ADMINISTRAT		0,120	12,717	10,000	22,100	10,100	0,001	0,000	1,002	2,010	1,001	
Household size	709.250	75,145	96.584	134,849	132.984	104,318	74,446	37.904	25.736	13.864	13.420	4.1
Population	2,913,745	75,145	193,168	404,546	531,934	521,591	446,676	- ,	-,	- ,	-, -	
DIRE DAWA ADMINISTRATION	_,0.0,1.10	,. 10	100,100	10 1,0 10	001,001	521,001	,	200,020	_00,000	,. 00	,000	
Household size	58,986	9,958	10,231	10,621	9,298	5,962	4,573	2,769	1,586	1,197	2,792	3.9
Population	229,415	9,958	20,462	31,863	37,192	29,808	27,435	19,385	12,684	10,774	29,854	0.0

Table 3.2 Distribution of Urban Population, Households and Average Household Size by Region: 2011

				gion, cox a		ip Head of Hou	sehold	-			
Region and Sex		Llood of		Son/ Daughter of	Son/	Son/	Mother/ Father of	Brother/ Sister of	Other	Paid	Nor
	Total	Head of Household	Spouse	Head and Spouse	Daughter of Head	Daughter of Spouse	Head or Spouse	Head/ Spouse	Other Relatives	Domestic Workers	Non- Relative
COUNTRY TOTAL				•				•			
Total	13,093,359	3,514,618	1,781,819	3,929,761	1,815,732	72,772	95,806	514,174	201,439	1,020,413	146,825
Male	6,241,603	2,269,963	142,157	2,033,927	940,927	38,914	20,502	236,928	30,382	454,688	73,215
Female	6,851,757	1,244,655	1,639,662	1,895,834	874,805	33,858	75,305	277,246	171,057	565,724	73,611
TIGRAY REGION											
Total	911,785	263,769	112,048	266,798	151,902	10,922	7,097	35,037	7,445	53,202	3,565
Male	411,214	149,850	2,246	133,430	76,187	5,983	1,118	14,023	810	25,334	2,232
Female	500,571	113,919	109,801	133,368	75,716	4,938	5,979	21,014	6,635	27,868	1,333
AFFAR REGION											
Total	191,483	58,059	31,681	62,097	21,276	1,129	1,793	4,374	1,563	8,983	527
Male	96,521	43,725	782	33,288	10,119	743	165	2,214	252	4,965	268
Female	94,962	14,335	30,899	28,809	11,157	386	1,628	2,160	1,311	4,018	259
AMHARA REGION											
Total	2,335,662	674,229	324,174	636,542	336,694	16,523	21,094	80,927	37,474	187,632	20,375
Male	1,084,376	405,130	24,356	329,927	171,382	8,207	3,877	39,964	5,351	87,159	9,022
Female	1,251,286	269,098	299,817	306,614	165,312	8,315	17,216	40,963	32,123	100,473	11,354
OROMIA REGION											
Total	3,930,354	1,096,006	570,939	1,181,085	533,200	20,426	23,977	139,549	34,273	280,700	50,199
Male	1,904,116	747,975	31,550	616,675	276,288	11,955	4,675	59,625	7,714	121,460	26,199
Female	2,026,238	348,031	539,389	564,410	256,912	8,471	19,302	79,924	26,559	159,240	24,000
SOMALI REGION											
Total	639,638	131,270	79,656	305,390	70,944	2,118	4,125	16,869	1,473	25,366	2,429
Male	332,245	92,132	7,935	167,902	36,079	1,136	1,474	8,301	491	15,614	1,182
Female	307,393	39,138	71,720	137,488	34,865	982	2,651	8,568	983	9,752	1,247
BENISHANGUL GUN	MUZ REGIOI										
Total	123,712	36,507	20,156	39,444	13,171	2,767	713	3,028	610	6,411	905
Male	59,698	26,017	979	19,594	6,776	1,454	264	1,215	215	2,638	545
Female	64,015	10,490	19,177	19,849	6,395	1,313	449	1,813	395	3,773	360
SNNP REGION											
Total	1,618,465	427,576	222,839	527,820	185,500	4,435	8,755	64,564	19,175	135,328	22,473
Male	794,568	304,983	8,987	275,316	93,516	1,963	1,358	31,769	4,661	59,393	12,622
Female	823,898	122,593	213,852	252,504	91,985	2,472	7,397	32,795	14,514	75,935	9,850

Table 3.3 Distribution of Urban Population by Region, Sex and Relationship to the Head of Household: 2011

Table 3.3 Cont'd

					Relations	nip Head of Hou	usehold				
				Son/			Mother/	Brother/			
Region and Sex				Daughter of	Son/	Son/	Father of	Sister of		Paid	
		Head of		Head and	Daughter of	Daughter of	Head or	Head/	Other	Domestic	Non-
	Total	Household	Spouse	Spouse	Head	Spouse	Spouse	Spouse	Relatives	Workers	Relative
GAMBELLA REGION	N										
Total	95,914	23,407	10,844	23,100	23,629	1,046	997	5,617	316	6,639	318
Male	48,374	14,285	1,041	12,420	12,797	575	212	2,785	98	3,994	167
Female	47,540	9,122	9,804	10,679	10,833	471	785	2,832	218	2,645	150
HARARI REGION											
Total	103,186	31,319	15,300	26,610	15,292	509	906	2,970	1,269	8,131	880
Male	49,920	19,882	1,447	14,406	8,055	321	175	1,508	-	3,495	629
Female	53,266	11,436	13,853	12,204	7,237	188	731	1,461	1,269	4,636	251
ADDIS ABABA CITY	ADMINISTI	RATION									
Total	2,913,745	713,490	365,413	799,050	428,150	11,943	24,246	152,555	94,060	282,273	42,565
Male	1,349,045	428,901	60,787	397,469	230,228	6,142	6,794	71,842	10,716	116,858	19,307
Female	1,564,700	284,589	304,626	401,581	197,922	5,801	17,452	80,713	83,344	165,415	23,257
DIRE DAWA ADMIN	ISTRATION										
Total	229,415	58,986	28,768	61,826	35,974	953	2,105	8,685	3,780	25,747	2,591
Male	111,527	37,082	2,045	33,498	19,501	434	390	3,683	75	13,778	1,041
Female	117,888	21,904	26,724	28,328	16,472	519	1,715	5,002	3,705	11,969	1,550

Alla	inment - C		Jiai. 201										
				1			A	Il Literate	0				
Region and Sex	All Persons	All Illiterate	Non- Formal	Pre-school	Grades 1- 8	Secondary not completed	Secondary Completed	Preparatory	Certificate(TVET, 10+1,20+2 and in Old Cur.) Completed	Diploma Completed	Degree and above Completed	Not Stated	N.S for Literacy Status
COUNTRY TO All Ages	TAL												
Total	10,397,080	2,110,663	121,571	16,884	4,435,034	851,923	1,323,254	374,211	186,638	637,166	337,824	954	958
Male	4,866,065	562,621	72,578	6,668	2,175,276	448,787	705,819	191,962	87,590	347,776	265,834	620	535
Female 10-14	5,531,015	1,548,042	48,993	10,216	2,259,758	403,136	617,435	182,249	99,048	289,390	71,990	335	423
Total	1,513,511	103,780	549	8,696	1,379,335	20,691	-	-	-	-	-	-	460
Male	713,106	36,948	367	2,831	662,647	10,065	-	-	-	-	-	-	248
Female 15-19	800,405	66,832	182	5,865	716,688	10,626	-	-	-	-	-	-	211
Total	1,735,484	118,507	1,352	1,657	903,880	305,761	245,511	112,371	26,215	19,346	883	-	-
Male	751,452	32,172	400	901	400,289	142,242	106,310	48,432	11,774	8,050	883	-	-
Female 20-24	984,032	86,335	952	757	503,591	163,519	139,201	63,939	14,442	11,297	-	-	-
Total	1,556,477	164,022	989	499	509,945	152,574	337,872	136,588	54,042	143,237	56,708	-	-
Male	679,531	37,695	551	49	218,001	78,061	161,228	71,368	19,354	57,603	35,620	-	-
Female	876,947	126,328	438	450	291,944	74,514	176,644	65,219	34,688	85,634	21,088	-	-
25-29													
Total	1,470,711	253,281	7,001	566	474,095	117,754	259,809	61,683	44,043	157,853	94,306	109	211
Male	699,615	59,120	3,579	10	233,299	57,851	138,464	34,962	20,203	80,477	71,650	-	-
Female 30-34	771,096	194,161	3,422	556	240,795	59,903	121,345	26,721	23,840	77,375	22,656	109	211
Total	957,508	194,253	6,505	574	332,952	89,057	158,442	25,491	15,111	86,798	48,325	-	-
Male	498,411	53,714	4,038	574	184,628	50,903	90,458	12,916	9,990	50,531	40,659	-	-
Female 35-39	459,097	140,539	2,466	-	148,325	38,154	67,984	12,575	5,121	36,266	7,666	-	-
Total	850,679	184,290	14,521	998	276,727	67,472	135,549	18,667	14,317	86,809	51,042	-	287
Male	401,785	42,579	8,566	517	136,380	38,152	74,458	10,303	5,793	45,539	39,211	-	287
Female 40-44	448,894	141,711	5,955	481	140,347	29,320	61,091	8,363	8,524	41,269	11,831	-	-
Total	542,611	153,500	13,467	567	161,899	36,635	73,557	7,521	7,693	50,769	37,004	-	-
Male	298,454	41,687	8,013	281	92,093	26,517	49,948	5,502	3,856	36,791	33,766	-	-
Female	244,158	111,813	, -	286	69,806	10,118	23,609	2,019	3,836	13,978	3,238	-	-

Table 3.4Distribution of Urban Population Aged 10 Years and Over by Age group, Sex, Literacy Status and Educational
Attainment - Country Total: 2011

Table 3.4 Cont'd

			All Literate												
Region and Sex	All Persons	All Illiterate	Non- Formal	Pre-school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate (TVET, 10+1,20+2 and in Old Cur.) Completed	Diploma Completed	Degree and above Completed	Not Stated	N.S for Literacy Status		
45-49	•	•		•				• • •	·						
Total	440,017	156,021	12,651	555	129,354	19,627	50,423	6,648	9,382	32,161	23,196	-	-		
Male	215,355	34,538	5,953	417	74,186	12,052	34,796	4,004	5,267	23,284	20,859	-	-		
Female	224,662	121,484	6,698	138	55,168	7,575	15,627	2,644	4,115	8,877	2,338	-	-		
50-54															
Total	347,848	160,448	14,645	970	83,375	13,273	24,656	2,798	6,475	26,443	14,764	-	-		
Male	161,285	40,493	8,410	136	46,286	9,174	19,644	2,136	4,493	17,872	12,642	-	-		
Female	186,562	119,955	6,234	835	37,090	4,099	5,013	663	1,982	8,571	2,122	-	-		
55-59															
Total	245,521	119,808	10,393	293	62,529	9,785	15,583	1,534	5,651	15,396	4,324	226	-		
Male	113,124	29,016	4,229	50	38,185	6,886	12,657	1,534	4,437	12,272	3,857	-	-		
Female	132,397	90,791	6,164	243	24,343	2,898	2,926	-	1,214	3,124	467	226	-		
60-64															
Total	234,669	141,166	10,657	840	47,987	8,243	8,810	222	1,791	10,498	4,146	310	-		
Male	104,252	37,605	7,164	281	32,200	6,310	7,180	222	1,157	8,263	3,562	310	-		
Female	130,417	103,561	3,493	559	15,787	1,933	1,630	-	634	2,236	584	-	-		
65+															
Total	502,044	361,586	28,842	668	72,956	11,050	13,042	689	1,918	7,856	3,127	310	-		
Male	229,695	117,054	21,307	622	57,082	10,572	10,677	585	1,267	7,093	3,127	310	-		
Female	272,349	244,532	7,535	46	15,873	478	2,365	105	652	763	-	-	-		

Training status and						Re	gion					
Types of Training							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNP	Gambela	Harari	Ababa	Dawa
All Persons												
Total	10,397,080	•				,		1,240,922	,		2,504,853	•
Male	4,866,065				1,463,569	,	45,321	602,413	•		1,141,811	84,187
Female	5,531,015	399,512	73,040	1,021,055	1,600,116	212,205	50,558	638,508	34,740	42,864	1,363,042	95,374
Not Trained												
Total				1,551,406	2,533,297	392,943	,	1,016,917	•		1,842,109	142,682
Male	3,617,151	•	62,221	654,441	1,110,644	188,765	34,271	462,531	25,309		751,712	60,148
Female	4,791,216	357,411	67,835	896,965	1,422,653	204,178	45,770	554,387	32,560	36,525	1,090,397	82,534
Trained												
Total	1,987,645	•	15,317	321,080	530,074	38,762	15,838	223,826	8,557	18,441	662,169	36,878
Male	1,248,599	74,602	10,113	196,991	352,610	30,734	11,050	139,883	6,377	12,102	390,099	24,039
Female	739,046	42,101	5,205	124,089	177,464	8,027	4,788	83,944	2,180	6,339	272,070	12,839
Not Stated												
Total	1,068	-	-	-	315	-	-	178	-	-	574	-
Male	315	-	-	-	315	-	-	-	-	-	-	-
Female	753	-	-	-	-	-	-	178	-	-	574	-
Types of Training												
Agriculture												
Total	42,109	2,695	1,717	9,000	14,597	2,248	362	7,129	173	434	3,523	230
Male	34,192	2,125	1,294	7,305	11,536	1,233	316	7,084	79	348	2,642	230
Female	7,917	569	423	1,694	3,061	1,015	47	45	94	87	882	-
Water and Soil Conserv	ation											
Total	4,486	432	602	889	907	-	75	1,406	-	95	-	80
Male	2,549	432	308	486	272	-	54	822	-	95	-	80
Female	1,937	-	294	403	635	-	21	583	-	-	-	-
forestry Science												
Total	3,112	324	-	-	1,216	-	112	542	-	38	879	-
Male	1,893	324	-	-	589	-	112	542	-	38	287	-
Female	1,219	-	-	-	627	-	-	-	-	-	592	-
General medicine, Opht	halmology F	Pathology,	Internal r	nedicine, G	ynecology,	Surgery, A	nesthesiology					
Total	6,351	281	-	838	1,458	-	18	395	127	181	2,755	298
Male	4,879	197	-	532	1,034	-	18	200	127	80	2,468	223
Female	1,471	84	-	306	424	-	-	195	-	101	287	75
Medical Laboratory/Rad												
Total	10,250	-	155	1,117	2,778	217	313	1,173	32	83	4,382	-
Male	6,330	-	19	632	1,761	75	100	719	32	83	2,909	-
Female	3,921	-	136	485	1,017	142	213	454	-	-	1,473	-

Table 3.5 Distribution of Urban Population Aged Ten Years and Above by Region, Sex, Training Status and Types of Training COUNTRY TOTAL: 2011

Table 3.5 cont'd						Ro	gion					
Training status and							Benishangu				Addis	Dire
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	I-Gumuz	SNNP	Gambela	Harari	Ababa	Dawa
Pharmacy/Pharmacy Teo	chnician											
Total	18,718	589	315	2,406	3,825	828	89	2,638	23	211	7,492	301
Male	11,857	389	139	1,110	3,349	788	89	1,991	23	172	3,578	228
Female	6,860	199	176	1,296	476	40	-	647	-	39	3,914	73
Sanitary Science												
Total	2,475	-	-	50	546	42	134	683	-	41	907	72
Male	858	-	-	50	216	-	134	58	-	41	287	72
Female	1,617	-	-	-	330	42	-	625	-	-	620	-
Clinical Science/Nursing												
Total	61,533	3,462	410	10,156	21,717	1,282	321	11,828	131	969	10,368	888
Male	26,912	1,715	332	4,918	11,734	169	70	5,220	98	369	2,048	239
Female	34,621	1,747	79	5,238	9,983	1,113	251	6,608	33	600	8,320	650
Public health Nurse, Hea	alth officer/ Sa	anitarian/F	Post Healt	th Officer								
Total	15,798	1,674	57	2,649	3,248	1,501	252	2,268	-	401	3,603	146
Male	10,388	1,518	57	1,996	1,482	982	182	1,337	-	310	2,449	75
Female	5,410	156	-	653	1,767	520	70	931	-	91	1,154	71
Personal business mana	igement											
Total	71,112	4,297	478	10,298	22,336	656	1,259	10,293	237	443	19,933	882
Male	50,235	2,592	420	6,181	17,190	497	946	7,010	213	302	14,214	669
Female	20,877	1,705	57	4,117	5,145	159	313	3,283	25	142	5,718	213
Economics												
Total	20,873	1,161	195	3,524	2,683	88	243	2,928	70	136	9,395	450
Male	15,827	677	195	3,210	1,305	38	243	2,286	70	136	7,292	375
Female	5,046	484	-	314	1,378	50	-	642	-	-	2,103	75
Book Keeping/Accountin	g											
Total	177,393	10,014	567	22,359	48,175	1,940	1,339	23,702	527	821	65,428	2,521
Male	92,431	5,676	265	11,784	31,079	1,544	953	11,393	405	469	27,150	1,714
Female	84,962	4,339	302	10,575	17,096	396	386	12,310	122	352	38,277	808
Secretarial Science												
Total	47,552	3,524	523	6,043	11,767	234	289	3,836	48	468	20,246	575
Male	2,747	324	64	277	966	-	-	466	17	-	633	-
Female	44,806	3,200	460	5,766	10,801	234	289	3,369	31	468	19,614	575
Banking and Finance												
Total	12,338	1,248	-	2,484	2,212	302	45	1,714	59	-	4,202	72
Male	7,081	716	-	1,938	1,149	302	45	489	35	-	2,336	72
Female	5,257	533	-	546	1,063	-	-	1,225	24	-	1,866	-
Law												
Total	41,487	1,928	264	5,919	15,951	593	402	7,154	419	371	7,874	612
Male	31,312	1,740	80	4,360	11,262	593	364	6,458	320	292	5,380	462
Female	10,175	188	184	1,559	4,688	-	38	696	99	79	2,494	150

Table 3.5 cont'd

Training status and						Re	gion					
Types of Training							Benishangu				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	I-Gumuz	SNNP	Gambela	Harari	Ababa	Dawa
Language and Literature		nd Foreigi	n)									
Total	32,437	263	-	9,620	8,933	354	899	6,358	113	162	5,664	73
Male	17,178	263	-	4,624	4,577	104	550	3,527	59	162	3,240	7
Female	15,259	-	-	4,996	4,356	250	350	2,831	54	-	2,424	
Theatrical art												
Total	5,390	197	9	851	781	246	20	173	-	39	3,074	
Male	3,721	197	9	452	469	246	-	173	-	-	2,175	
Female	1,668	-	-	399	312	-	20	-	-	39	899	
Educational Pedagogy,	Teaching cur	riculum ins	struction E	Educational	administrat	ion, Adults	education					
Total	129,732	7,979	1,343	19,127	46,880	4,347	237	17,594	1,194	2,128	27,664	1,239
Male	74,777	3,876	921	9,394	30,572	3,325	42	10,641	872	1,207	13,156	77
Female	54,955	4,103	422	9,732	16,308	1,021	195	6,953	323	921	14,508	46
Geography	- ,	,		-, -	-,	, -		-,		-	,	-
Total	22,592	1,115	10	6,746	5,410	1,429	338	4,776	94	254	2,195	22
Male	15,962	750	10	4,961	4,320	1,122	294	2,863	94	123	1,199	22
Female	6,631	365	-	1,784	1,090	307	45	1,913	-	130	995	
History	0,001	000		1,101	1,000	001	10	1,010		100	000	
Total	10,944	_	-	2,982	2,080	482	246	2,279	-	-	2,799	7
Male	8,841	-	-	2,640	2,080	482	182	1,990	_	-	1,392	75
Female	2,102	-	-	342	2,000 -		64	289	-	-	1,407	1
Political science	2,102	_	-	042	-	_	04	203	_	_	1,407	
Total	2,844		_	321	728	-	48	1,051	37	85	574	
Male	2,044 2,289	-	-	44	490	-	48	1,051	37	44	574	
Female	2,209	-	-	277	238	-	40	1,051	- 57	44	574	
Sociology and Anthropo		-	-	211	200	-	-	-	-	41	-	
Total	5,267	89		234	2,985	43		407	12	41	1,456	
Male	4,027	89 89	-	234 84	2,965	43	-	407	- 12	41	584	
Female	4,027	- 09	-	04 150	2,822	- 43	-	407	- 12		564 871	
	1,240	-	-	150	163	43	-	-	12	-	871	
Psychology				040	4 4 9 4	4.40		4 445			000	
Total	4,454	-	-	816	1,121	142	-	1,415	-	-	960	
Male	3,266	-	-	538	963	142	-	661	-	-	960	
Female	1,189	-	-	277	158	-	-	754	-	-	-	
Civil Engineering/Buildir	•											
Total	17,811	463	59	1,149	2,701	331	54	1,421	-	53	11,127	453
Male	15,056	358	59	927	2,195	331	54	1,084	-	53	9,621	37
Female	2,755	105	-	222	507	-	-	337	-	-	1,506	7
Mechanical Engineering												
Total	9,986	366	59	971	1,204	-	-	1,837	37	-	5,512	
Male	8,356	366	-	193	714	-	-	1,837	37	-	5,210	
Female	1,630	-	59	778	490	-	-	-	-	-	302	

Training status and						Re	gion					
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangu I-Gumuz	SNNP	Gambela	Harari	Addis Ababa	Dire Dawa
Electrical Engineering											•	
Total	9,346	524	9	1,186	2,646	-	21	-	12	41	4,908	
Male	7,100	524	9	807	1,218	-	21	-	12	-	4,510	
Female	2,246	-	-	379	1,428	-	-	-	-	41	399	
Industrial Engineering												
Total	3,303	346	-	472	1,780	63	-	63	-	-	579	
Male	1,462	346	-	154	611	63	-	-	-	-	287	
Female	1,842	-	-	318	1,169	-	-	63	-	-	292	
lunicipality Engineering/					,							
Total	4,628	-	186	723	879	-	-	722	-	38	2,080	
Male	3,202	-	186	272	776	-	-	722	-	38	1,207	
Female	1,427	-	-	451	103	-	-	-	-	-	873	
Other types of Engineeri		,Bio-divers	sity, Hydra	aulic Engine	ering Irriga	tion Engine	eering					
Total	7,605	83	64	1,330	1,614	653	63	1,302	-	-	2,497	
Male	5,977	83	64	961	961	653	31	1,302	-	-	1,922	
Female	1,628	-	-	369	653	-	31	-	-	-	574	
Automotive technology	,						-				-	
Total	14,909	3,057	336	2,201	4,343	-	-	984	240	53	3,615	-
Male	10,889	2,126	136	1,748	2,842	-	-	984	191	53	2,730	
Female	4,020	931	200	453	1,502	-	-	-	49	-	885	
Construction technology	,				,				-			
Total	15,515	1,846	-	2,624	2,669	-	180	1,410	23	-	6,521	24
Male	10,894	1,522	-	1,679	1,350	-	39	1,410	23	-	4,706	1
Female	4,621	324	-	945	1,319	-	141	-	-	-	1,815	
Drafting technology	, -	-			,						,	
Total	7,415	687	-	819	660	-	-	901	-	79	4,130	1
Male	3,222	578	-	181	430	-	-	711	-	39	1,213	
Female	4,193	109	-	638	230	-	-	189	-	41	2,917	
Manufacturing technolog					_00						_,• · ·	
Total	2,537	539	-	742	969	-	-	-	-	-	287	
Male	1,617	539	-	359	718	-	-	-	-	-	-	
Female	920	-	-	382	250	-	-	-	-	-	287	
Metal technology	020			002	200						201	
Total	2,490	1,126	-	-	513	-	-	126	-	82	644	
Male	2,490	1,126	-	-	513	-	-	126	-	82	644	
Female	2,400	-, 120	-	-	-	_	-	- 120	-	- 02		
Veterinary Science, Anin	nal husbandr	V										
Total	5,902	, 396	68	949	1,258	900	252	1,398	47	_	635	
Male	4,801		68	727	1,230	900	156	1,398	47	-	287	
Maic	4,801	- 396	00	222	40	900	96	1,590	4/	-	348	

Training status and						Reg						
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangu I-Gumuz	SNNP	Gambela	Harari	Addis Ababa	Dire Dawa
Communication												
Total	2,600	90	-	65	276	-	-	575	35	-	1,559	
Male	1,508	-	-	65	276	-	-	210	35	-	923	
Female	1,092	90	-	-		-	-	365	-	-	636	
Power, Energy and other			Electrical	Power Med	chanical and	d Manufact	urina Power	000			000	
Total	6,271	278	128	975	1,644	43		1,072	-	184	1,796	15
Male	5,079	278	13	878	1,606	43	-	462	-	146	1,504	15
Female	1,192		115	97	38	-	_	610	-	39	292	
Biology	1,102		110	01	00			010		00	202	
Total	13,621	422	-	3,468	3,374	-	431	2,668	59	118	2,922	16
Male	9,789	263	-	2,789	3,257	-	199	1,829	59	41	1,194	16
Female	3,833	160	-	679	117	-	233	839	-	78	1,728	
Chemistry	0,000	100		010			200	000			1,120	
Total	9,891	818	-	1,665	3,063	52	335	1,801	37	198	1,766	15
Male	8,229	445	-	1,665	2,433	52	311	1,537	37	120	1,474	15
Female	1,662	373	-	1,000	631		24	264	-	78	292	
Geology	1,002	010			001		21	201		10	202	
Total	4,437	622	-	424	2,089	246	_	471		_	584	
Male	3,040	83	-	272	2,000	246	-	120		-	297	
Female	1,396	539	-	152	68	- 240	_	350	_	_	287	
Physics	1,000	000		102	00			000			201	
Total	9,426	545	10	1,481	2,333	179	149	2,098	_	41	2,506	8
Male	8,044	434	10	1,115	1,779	179	149	2,098	_	41	2,000	6
Female	1,382	111	-	366	553	-	-	2,000		-	352	
Mathematics	1,002			000	000						002	
Total	19,888	1,345	115	2,904	5,453	654	329	5,592	10	124	3,272	ç
Male	16,733	1,256	115	2,504	5,068	654	246	3,787	10	124	2,698	ç
Female	3,155	89	-	2,004	385	- 00	83	1,804	-	- 124	2,030 574	
Statistics	5,155	05		220	000		00	1,004			5/4	
Total	4,879	811	20	1,135	911	-	19	1,258	31	96	598	
Male	3,257	279	20	466	896	_	19	853	31	96	598	
Female	1,622	532	- 20	669	15	_	-	405	-	50	000	
Demography	1,022	002	_	003	10	_	_	-00	_	-	-	
Total	1,997	573		_	723		62				640	
Male	1,997	286	-	-	684	-	62 62	-	-	-	640 640	
Female	326	286	-	-	39	-	- 02	-	-	-	040	
Computer Science/Inform			-	-	39	-	-	-	-	-	-	
Total	41,929	2,244	90	2,471	8,566	136	346	5,397	206	174	21,631	66
Male	41,929	2,244 742	90 12	2,471 729	0,566 1,646	92	340 112	5,397 1,655	206 52		21,631 9,599	59
IVIDIC	13,291	142	12	1,742	6,920	92	234	3,742	52 155	55	9,599 12,032	55

Table 3.5 cont'd						RA	gion					
Training status and							Benishangu				Addis	Dire
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	I-Gumuz	SNNP	Gambela	Harari	Ababa	Dawa
Health and Physical edu	ication											
Total	15,197	931	23	3,707	4,018	119	71	951	-	247	4,694	435
Male	8,158	301	10	1,337	2,543	49	23	670	-	41	2,971	214
Female	7,039	631	13	2,370	1,475	70	48	280	-	206	1,723	221
Meteorological Science												
Total	572	286	-	-	121	38	23	104	-	-	-	-
Male	446	286	-	-	40	38	23	59	-	-	-	-
Female	126	-	-	-	81	-	-	45	-	-	-	-
Chemical Engineering												
Total	1,043	-	-	74	-	-	-	-	32	-	938	-
Male	969	-	-	-	-	-	-	-	32	-	938	-
Female	74	-	-	74	-	-	-	-	-	-	-	-
Textile Engineering												
Total	782	160	-	44	-	-	-	-	-	-	578	-
Male	737	160	-	-	-	-	-	-	-	-	578	-
Female	44	-	-	44	-	-	-	-	-	-	-	-
Purchasing supply												
Total	7,582	427	-	1,300	2,522	61	256	671	23	41	2,131	150
Male	3,565	123	-	448	756	61	235	671	23	-	1,179	69
Female	4,017	304	-	853	1,766	-	21	-	-	41	952	80
Software Engineering	·											
Total	3,865	765	-	451	586	-	24	-	-	81	1,959	-
Male	1,971	324	-	222	500	-	-	-	-	-	925	-
Female	1,894	441	-	229	86	-	24	-	-	81	1,033	-
Home economics												
Total	13,340	246	23	3,085	2,668	307	129	974	-	101	5,593	214
Male	1,445	-	-	295	130	-	76	62	-	-	882	-
Female	11,895	246	23	2,789	2,539	307	54	912	-	101	4,711	214
Leather handicraft techn	ology											
Total	3,252	1,013	403	285	-	-	-	-	-	-	1,476	75
Male	1,208	216	403	-	-	-	-	-	-	-	589	-
Female	2,044	798	-	285	-	-	-	-	-	-	887	75
Decorative work training												
Total	36,713	4,103	9	3,961	6,345	201	45	3,134	-	232	17,214	1,469
Male	6,702	388	-	347	1,224	158	-	307	-	53	4,153	72
Female	30,011	3,715	9	3,614	5,121	44	45	2,827	-	179	13,061	1,397
Video photographic trair								•				
Total	5,122	587	84	78	1,152	37	54	59	-	53	2,877	142
Male	3,163	324	74	78	619	37	54	59	-	53	1,723	142
Female	1,960	263	10	-	533	-	-	-	-	-	1,154	-

Table 3.5 cont'd		Region											
Training status and							Benishangu				Addis	Dire	
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	I-Gumuz	SNNP	Gambela	Harari	Ababa	Dawa	
Pottery													
Total	1,331	188	-	743	-	-	-	60	-	53	287	-	
Male	242	188	-	-	-	-	-	-	-	53	-	-	
Female	1,090	-	-	743	-	-	-	60	-	-	287	-	
Weavery													
Total	4,075	466	-	1,449	798	-	-	299	-	-	972	91	
Male	2,867	377	-	1,388	420	-	-	62	-	-	620	-	
Female	1,208	90	-	62	378	-	-	236	-	-	352	91	
Wood work/carpentry													
Total	26,585	1,051	220	4,606	7,608	237	164	1,499	23	313	10,100	765	
Male	24,644	736	220	4,384	7,236	237	164	1,375	23	313	9,514	442	
Female	1,942	315	-	222	372	-	-	124	-	-	585	323	
Electrical work													
Total	40,536	2,555	169	5,723	9,578	37	239	4,256	56	252	15,999	1,672	
Male	33,559	2,068	169	4,305	7,944	-	222	2,829	56	252	14,415	1,299	
Female	6,977	487	-	1,418	1,634	37	17	1,427	-	-	1,584	374	
Metal welding and bindir	ng												
Total	19,752	1,532	217	5,119	4,977	63	43	1,226	109	229	5,866	370	
Male	18,024	1,532	217	4,303	4,887	63	43	1,083	109	229	5,186	370	
Female	1,728	-	-	816	89	-	-	142	-	-	680	-	
Auto Mechanics includin	ig motorcycle	, Aircraft n	naintenar	nce									
Total	49,949	2,037	194	4,535	8,620	1,303	277	3,902	55	562	27,994	469	
Male	45,988	2,037	182	4,028	7,740	1,303	277	3,425	55	516	25,956	469	
Female	3,961	-	13	507	880	-	-	477	-	46	2,038	-	
Mason													
Total	32,343	2,526	621	6,195	6,931	1,178	139	3,609	123	144	10,269	608	
Male	28,425	1,956	486	4,805	6,432	1,178	79	3,001	123	144	9,684	537	
Female	3,918	569	136	1,390	499	-	60	608	-	-	585	71	
Typing													
Total	31,965	525	265	4,782	10,542	352	253	4,380	169	252	9,601	845	
Male	1,955	-	-	276	781	161	-	121	-	41	574	-	
Female	30,010	525	265	4,506	9,761	191	253	4,258	169	211	9,026	845	
Tailoring													
Total	45,515	2,416	19	8,566	11,660	157	90	1,527	34	257	19,404	1,386	
Male	12,193	460	-	3,051	4,460	157	-	216	-	84	3,680	85	
Female	33,322	1,955	19	5,515	7,200	-	90	1,311	34	173	15,724	1,301	
Driver													
Total	210,283	13,315	1,903	33,351	59,235	4,257	1,125	13,871	481	2,046	73,140	7,558	
Male	203,372	13,274	1,894	32,889	56,420	4,257	1,125	13,407	458	1,964	70,195	7,489	
Female	6,911	41	9	462	2,815	-	-	464	23	82	2,945	69	

Table 3.5 cont'd												
Training status and						Re	gion		1	1		D:
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangu I-Gumuz	SNNP	Gambela	Harari	Addis Ababa	Dire Dawa
Development Agent	Total	rigidy	7 (10)	7 innara	oronna	Comano	1 Outline2	CININ	Cambola	Tiaran	710000	Dana
Total	8,312	123	-	2,148	2,224	50	291	2,747	401	41	287	-
Male	6,698	-	-	1,926	2,184	50	247	1,671	333	-	287	-
Female	1,614	123	-	222	40	-	44	1,077	68	41		-
Traditional Birth Attendar	,							.,				
Total	3,508	160	-	258	1,337	-	-	1,155	25	-	574	-
Male	1,499	-	-	222	965	-	-	287	25	-	-	-
Female	2,010	160	-	36	372	-	-	868	-	-	574	-
Computer Applications a					• • -							
Total	50,722	2,372	501	11,802	13,042	1,184	19	3,532	222	892	14,603	2,554
Male	11,553	188	186	2,768	2,787	319	-	716		194	4,013	381
Female	39,168	2,184	315	9,034	10,255	865	19	2,817	222	697	10,589	2,172
Lower Scale training of A								,			,	
Total	6,773	683	443	1,502	523	1,261	96	397	523	-	1,195	149
Male	5,761	263	367	1,407	436	1,261	96	397	476	-	908	149
Female	1,011	420	76	94	87	-	-	-	47	-	287	-
Hotels and Tourism	,											
Total	21,253	307	13	1,953	1,990	119	-	342	-	93	16,212	224
Male	4,290	207	-	392	795	75	-	64	-	40	2,716	-
Female	16,963	100	13	1,561	1,195	44	-	278	-	53	13,495	224
Police	,										,	
Total	44,546	3,127	192	10,301	10,640	2,660	407	5,595	651	829	9,181	962
Male	39,011	2,681	20	8,825	10,415	2,624	365	4,967	572	562	7,100	882
Female	5,535	446	172	1,476	226	37	42	627	80	267	2,081	80
Military												
Total	25,616	2,267	115	2,498	7,098	449	520	2,574	253	1,154	7,491	1,197
Male	24,442	2,104	115	2,221	7,058	449	520	2,516	253	1,113	6,897	1,197
Female	1,174	163	-	277	39	-	-	59	-	41	595	-
Cobblestone												
Total	31,069	3,467	-	9,978	5,936	74	-	865	-	644	9,150	954
Male	18,734	2,348	-	5,522	3,418	37	-	783	-	505	5,471	651
Female	12,335	1,119	-	4,457	2,518	37	-	82	-	139	3,679	304
Embroidery												
Total	11,954	359	97	2,358	3,152	37	193	842	-	141	4,775	-
Male	1,395	-	-	-	389	-	-	-	-	-	1,006	-
Female	10,560	359	97	2,358	2,763	37	193	842	-	141	3,769	-
Plumbery												
Total	10,476	1,663	10	2,015	1,474	1,190	-	1,593	27	41	2,394	69
Male	8,693	1,663	10	1,274	992	874	-	1,348	27	41	2,394	69
Female	1,783	-	-	741	482	315	-	245	-	-	-	-

Table 3.5 cont'd													
Training status and		Region											
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangu I-Gumuz	SNNP	Gambela	Harari	Addis Ababa	Dire Dawa	
Other Senior and Junior	Professions	/Specified											
Total	72,492	1,669	-	10,951	23,150	1,676	1,050	6,357	295	297	26,173	874	
Male	47,966	888	-	6,864	14,636	1,457	715	3,961	221	177	18,541	505	
Female	24,526	781	-	4,087	8,513	219	335	2,396	74	120	7,632	369	
Other Skilled and Techn	ical Profession	ons/Specit	ied										
Total	185,934	6,928	2,000	30,939	50,273	1,483	1,049	14,644	1,021	764	74,147	2,684	
Male	101,432	4,279	952	17,153	32,102	1,034	646	6,308	578	528	36,344	1,508	
Female	84,501	2,649	1,048	13,786	18,172	449	403	8,336	443	236	37,803	1,176	
Not Stated for type of tra	aining												
Total	622	-	-	146	189	-	-	-	-	-	287	-	
Male	550	-	-	74	189	-	-	-	-	-	287	-	
Female	72	-	-	72	-	-	-	-	-	-	-	-	

Grou	101AL: 2011			
Age Group		Sex		Sex Ratio
Age Gloup	Total	Male	Female	Sex Ralio
All Ages	13,093,359	6,241,603	6,851,757	91.1
0-4	1,347,646	700,305	647,341	108.2
5-9	1,348,633	675,233	673,400	100.3
10-14	1,513,511	713,106	800,405	89.1
15-19	1,735,484	751,452	984,032	76.4
20-24	1,556,477	679,531	876,947	77.5
25-29	1,470,711	699,615	771,096	90.7
30-34	957,508	498,411	459,097	108.6
35-39	850,679	401,785	448,894	89.5
40-44	542,611	298,454	244,158	122.2
45-49	440,017	215,355	224,662	95.9
50-54	347,848	161,285	186,562	86.5
55-59	245,521	113,124	132,397	85.4
60-64	234,669	104,252	130,417	79.9
65+	502,044	229,695	272,349	84.3

Table 3.6Distribution of Urban Population by Five Years AgeGroup, Sex and Sex Ratio - COUNTRY TOTAL: 2011

CHAPTER IV

ECONOMIC ACTIVITY STATUS

4.1 Introduction

Statistical data on the economic activity status of a population is essential for planning and formulating policies on the utilization of human resources of a country. The economic activity status indicates the size and distribution of the work forces of economically active and non-active population during a given reference period at a specified area and also its distribution in the various sectors of the economy.

In this survey, information was collected on economic activity status of all persons aged five years and above. For the purpose of gathering information about the economic activities of children, the lower age limit was fixed at five years and hence data were collected from persons aged five years and above. However, for measuring the economic activity statuses of population, those persons aged ten years and above were included in the analysis.

The 2011 Urban Employment and Unemployment Survey administered detailed labour force questions to measure the economic activity status of persons in urban areas of the country. These includes:-

- a) whether persons were engaged in economic / productive activities in the last seven days prior to the survey date or reasons for not being engaged;
- b) whether persons were looking for work or will be available for work in the coming one month or reason for not being available during the reference period;
- c) whether persons were engaged in economic / productive activities during most of the last six months or reasons for not being engaged.

Every person aged ten years and above in the sampled household was asked if he/she was engaged in any productive activities during the last seven days, and in the last six months prior to the survey date. Based on the response obtained from respondents, the economic activity status of the population was determined by using the standard definitions of economic and non economic activities.

The 2011 Urban Employment and Unemployment Survey followed the ILO international standard definitions of economic and non-economic activities. The concept of economic activity as adopted by the 13th International Conference on Labour Statistics (ICLS,1982) is defined as the production of goods and/or services that falls within the United Nations System of National Accounts (SNA) production boundary (ILO, 1990). In this survey, economic/productive activity is defined as work, which involves the production of goods and /or services for sale or exchange. In addition, production of goods and services for own consumption or own uses are also considered as economic activities. These include production of primary products (agriculture, hunting, fishing, forestry and logging, mining and guarrying), for own consumption; processing of primary products by the producers themselves; production of other commodities where part of it is sold on the market; and own account construction and fixed asset formation (expected life use of one year or more). Such economic activities could be performed for an individual, family or private enterprise, government establishment or social organization. The remuneration may be on daily, weekly, monthly, yearly or contract basis. The practical activities of apprentices are also considered as economic activities. On the other hand, unpaid household chores such as preparing food, cleaning the house, taking care of children are not considered as economic activities. Similarly, unpaid community and volunteer services are classified as non-economic activities.

Based on the ILO definitions of economic and non economic activities, data on economic activity status of persons aged ten years and above was collected in the survey. Persons who were engaged in productive work /employed/ or available to be engaged /unemployed/ during the reference period were considered as economically active persons. While those who were neither engaged nor available to e engaged in

the production of economic goods and services during the reference period were considered as economically non- active or inactive population.

4.2 Data Collection Approaches of the Economic Activity Status of Urban Population

In the 2011 Urban Employment and Unemployment Survey, two approaches were used in the collection of data about the economic activity status of urban population. These are current and usual activity status approaches. The main difference between these two approaches is the length of the reference period.

4.2.1 Current Activity Status Approach

The current activity status approach measures the economic activity status of a person based on a short reference period of seven days before the date of interview. In this approach, all persons aged ten years and above who were engaged in productive work or available to be engaged during the reference week were considered as currently economically active while those who were neither engaged nor available to be engaged in productive work were taken as currently economically non active.

4.2.2 Usual Activity Status Approach

The usual activity status approach measures the economic activity status of a person based on a long reference period of the six months prior to the date of interview. In the usual activity status approach, all persons aged ten years and above who were engaged in productive works or available to be engaged in the production of economic goods and/or services during most of the six months were classified as usually economically active. Those who were neither employed nor unemployed in most of the six months categorized as economically non-active. The choice of appropriate economic activity status measurement approach depends on the extent of variability of economic activities of the population during a specified period of time in the study area. In the rural areas seasonal variation of activities is common. In order to collect economic activity status of the rural population including the slack and peak periods of agricultural activities, a longer reference period (usual activity status approach) is found to be more appropriate. On the other hand, in urban areas, activities are relatively less affected by seasonal variations and hence shorter reference period or current activity status approach is found to be more convenient. But in this survey, both approaches were used to collect data on economic activity status of urban population.

This chapter presents the economic activity status, size and distribution of economically active and non-active population, activity rate, and reasons for being inactive of urban population aged ten years and above using only the current status approach at national, regional and major town level.

4.3 Current Activity Status

As mentioned above, current activity status approach measures the economic activity status of a person in a short reference period, that is, the seven days prior to the date of the interview. In this approach a series of inquiries were administered to determine the economic activity status of the population during the reference week such as engagement in economic activity, attachment to jobs, seeking and availability to work, reason for not seeking or available to work, etc., These questions were presented to those persons aged five years and above. But the responses of persons aged ten years and above were used to classify the population into three mutually exclusive categories: employed, unemployed, and not in the labour force (non-active). The employed and the unemployed population together constitute the labour force or the currently economically active population. The third category, those who were not in the labour force are called currently non- active population.

As part of the economically active population, employed population consists of persons aged ten years and above who engaged in a productive activity or work at least for four

hours during the seven days prior to the date of the interview. In addition, persons who had regular jobs, or business, or holdings to return to but who were temporarily absent from work (not at work or worked less than four hours) for various reasons such as illness or injury, holiday or vacation, strike or lockout, and seasonality of work, annual leave, temporary closure of establishment were also considered as employed. For a person to be considered as absent from work, he/she must have formal attachment to the job.

The currently unemployed population also comprises those persons without work but looking for work or available and ready to work if any job is found during the reference period of the coming one month. The reference period of the coming one month refers to the survey week plus the next three weeks. If a person who is looking for work but engaged in productive activity during the reference week is recorded as employed but not as unemployed.

4.4 Economically Active and Non-active Population and Activity Rate

Based on the above classification of economic activity status, the number of economically active population including employed and unemployed persons, non active population and activity rates of urban population aged ten years and above at national and regional level is presented by sex and age group in Table 4.1 and Table 4.1.1- 4.1.11 using the current status approach, respectively. Besides, the economic activity status and activity rate of selected major towns' population aged ten years and above is also presented in Table 4.2.

The extent of the economic participation of the population is measured using economic activity rates. Economic activity rate or economic participation rate is computed as the percentage of the economically active population over the total of active plus non-active population aged ten years and above. The national estimate shows that 60.3 percent of the population aged ten years and above were economically active (Table 4.1). With respect to sex, the activity rate of male is 67.9 percent and the corresponding female activity rate is 53.7 percent. The age distributions of activity rate illustrates that it is

lower below age 20, but steadily rise and reaches its peak in age 40, and then after declines (Table 4.1).

As far as the economic activity status of major towns' population is concerned, 60.9 percent of the populations were economically active while the remaining 39.1 percent were found economically non- active. In terms of sex, the activity rate of male was higher by 15.9 percent than female activity rate. Among the major towns, the highest activity rate was recorded in Bahir Dar town (72.5 percent) followed by Harar town, Adama town, Asayita town, Bishoftu town, Hawassa town, Jimma town, and Addis Ababa, by 68.1 percent, 63.3 percent, 62.8 percent, 62.6 percent, 61.8 percent, 61.5 percent, 61.2 percent, respectively. The lowest activity rate was recorded in Gambella town (48.7 percent) and Jijiga town (47.6 percent).

The size of economically active and non-active population has its own implication on the development of the local as well as the national economy of a region. The high activity rate of a population implies the prevalence of huge potential of the human resource or human labour that foster sustainable economic development. On the other hand, the low economic activity rate also implies the lack of adequate and trained human labour or the existence of large number of economically non-productive population which has no contribution for development at the time. Hence, the economic activity rate of urban population of the Ethiopia can be viewed from these perspectives.

The distribution of economically non-active population of urban areas at national and regional level by age group, sex and reason for not being economically active (for those who were neither engaged nor available and/ or seeking work) during the last seven days are presented in Table 4.3 and Table 4.4,respectively. The reasons for not looking for or available to work include; student/training, home making / domestic work, pregnancy, injury / disability, illness, too young, old age /pensioned... etc. The result in Table 4.3 shows that the lion share of the non-active population were students (67.8%). The remaining non- active population (32.2%) comprises home makers, old aged, youngsters, injured / disabled, remitted, and other persons.

	Approach	<u>)- COUNTR</u>				
	All		Activit	y Status		A otivity
Age Group and Sex	Persons	E	conomically Ac	tive	Economically	Activity Rate
		Employed	Unemployed	Total Active	Non- Active	. tute
All Ages						
Total	10,396,615	5,139,831	1,125,536	6,265,367	4,131,249	60.3
Male	4,865,855	2,928,953	375,585	3,304,538	1,561,317	67.9
Female	5,530,760	2,210,879	749,950	2,960,829	2,569,931	53.5
10-14						
Total	1,513,511	144,811	18,226	163,037	1,350,474	10.8
Male	713,106	71,614	7,394	79,007	634,099	11.1
Female 15-19	800,405	73,197	10,832	84,029	716,375	10.
Total	1,735,484	429,517	157,489	587,006	1,148,479	33.
Male	751,452	192,665	57,681	250,346	501,106	33.
Female 20-24	984,032	236,852	99,808	336,660	647,372	34.
Total	1,556,316	797,478	308,809	1,106,287	450,029	71.
Male	679,472	404,296	106,010	510,306	169,166	75.
Female 25-29	876,845	393,182	202,800	595,982	280,863	68.
Total	1,470,711	1,009,623	226,773	1,236,396	234,315	84.
Male	699,615	579,802	68,133	647,935	51,680	92.
Female 30-34	771,096	429,821	158,640	588,461	182,634	76.
Total	957,508	719,297	126,267	845,564	111,944	88.
Male	498,411	439,566	42,168	481,734	16,677	96.
Female 35-39	459,097	279,731	84,099	363,830	95,267	79.
Total	850,679	656,750	94,558	751,307	99,372	88.
Male	401,785	363,332	26,782	390,114	11,672	97.
Female 40-44	448,894	293,418	67,776	361,194	87,700	80.
Total	542,568	409,957	57,218	467,174	75,394	86.
Male	298,410	269,401	17,180	286,581	11,830	96.
Female 45-49	244,158	140,556	40,038	180,594	63,564	74.
Total	440,017	326,320	44,271	370,591	69,426	84.
Male	215,355	196,881	10,464	207,346	8,009	96.
Female 50-54	224,662	129,439	33,807	163,246	61,416	72.
Total	347,802	232,556	34,995	267,551	80,250	76.
Male	161,285	138,821	10,971	149,792	11,493	92.
Female 55-59	186,516	93,735	24,024	117,760	68,757	63.
Total	245,414	152,901	19,201	172,102	73,312	70.
Male	113,124	92,826	6,794	99,621	13,503	88.
Female 60-64	132,290	60,074	12,407	72,481	59,809	54.
Total	234,669	108,220	16,508	124,728	109,941	53.
Male	104,252	72,874	7,265	80,140	24,112	76.
Female 65+	130,417	35,346	9,243	44,589	85,829	34.
Total	501,937	152,402	21,220	173,622	328,315	34.
Male	229,588	106,875	14,743	121,618	107,970	53.
Female	272,349	45,527	6,477	52,004	220,345	19.

Table 4.1Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days (Current Status
Approach)- COUNTRY TOTAL: 2011

(Current Status Approach)- TIGRAY REGION: 2011										
	All		Activity Status							
Age Group and Sex	Persons	Ec	conomically Ac	tive	Economically	Activity Rate				
		Employed	Unemployed	Total Active	Non- Active					
All Ages										
Total	713,842	321,569	74,758	396,327	317,515	55				
Male	314,330	173,501	21,057	194,559	119,772	61				
Female	399,512	148,068	53,701	201,769	197,743	50				
10-14										
Total	112,368	3,594	2,909	6,503	105,865	5				
Male	56,401	1,955	1,269	3,224	53,177	5				
Female	55,967	1,640	1,640	3,279	52,688	5				
15-19										
Total	122,865	23,755	13,080	36,835	86,031	30				
Male	52,721	10,722	4,035	14,757	37,964	28				
Female	70,144	13,033	9,045	22,078	48,067	31				
20-24	-,	-,	-,	,	-,					
Total	103,393	51,651	24,267	75,919	27,475	73				
Male	37,711	21,960	7,266	29,227	8,485	77				
Female	65,682	29,691	17,001	46,692	18,990	71				
25-29	,		,	,						
Total	87,570	60,085	13,299	73,384	14,186	83				
Male	38,755	31,638	3,426	35,064	3,692	90				
Female	48,814	28,447	9,873	38,321	10,494	78				
30-34	10,011	20,117	0,070	00,021	10,101					
Total	62,212	46,661	6,911	53,572	8,640	86				
Male	28,797	26,896	1,353	28,249	548	98				
Female	33,415	19,765	5,558	25,323	8,092	75				
35-39	55,415	19,705	5,550	20,020	0,092	1.				
Total	55,273	42,296	5,361	47,657	7,616	86				
Male	24,375	22,156	1,444	23,600	775	96				
Female	30,898	20,141	3,917	23,000	6,841	77				
40-44	50,090	20,141	5,917	24,007	0,041					
	37,346	28,020	2 945	30,865	6,481	82				
Total		26,020	2,845			02 92				
Male Female	18,607	10,203	849	17,112	1,495	92 73				
45-49	18,739	11,757	1,996	13,753	4,986	13				
	00 505	10.000	4 0 4 0	20 5 4 9	C 007					
Total	26,585	19,200	1,348	20,548	6,037	77				
Male	12,903	12,116	61	12,177	726	94				
Female 50-54	13,682	7,084	1,287	8,371	5,311	61				
	04.440	44.044	0.050	40.000	7 540					
Total	24,416	14,644	2,253	16,898	7,518	69				
Male	10,241	8,295	490	8,785	1,456	85				
Female	14,175	6,349	1,763	8,112	6,062	57				
55-59	00 500	40 407	000	40.400	7 050	~				
Total	20,522	12,487	682	13,169	7,352	64				
Male	9,250	8,180	-	8,180	1,070	88				
Female	11,272	4,307	682	4,990	6,282	44				
60-64										
Total	18,321	9,221	1,049	10,270	8,051	56				
Male	7,778	6,178	366	6,544	1,234	84				
Female	10,543	3,042	683	3,726	6,817	35				
65+										
Total	42,971	9,956	752	10,708	32,263	24				
Male	16,791	7,143	497	7,640	9,150	45				
Female	26,181	2,813	255	3,068	23,113	11				

Table 4.1.1Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days
(Current Status Approach)- TIGRAY REGION: 2011

(Current Status Approach)- AFFAR REGION: 2011								
			Activit	ty Status				
Age Group and Sex	All Persons	E	conomically Ac	tive	Economically	Activity Rate		
		Employed	Unemployed	Total Active	Non- Active	riato		
All Ages								
Total	145,374	71,536	15,066	86,602	58,772	59.6		
Male	72,334	45,023	4,378	49,402	22,932	68.3		
Female	73,040	26,512	10,688	37,200	35,840	50.9		
10-14								
Total	23,339	106	231	337	23,001	1.4		
Male	12,733	57	115	173	12,560	1.4		
Female	10,606	49	115	164	10,441	1.		
15-19								
Total	17,048	2,527	1,025	3,553	13,496	20.		
Male	7,524	1,554	217	1,770	5,753	23.		
Female	9,525	974	809	1,782	7,742	18.		
20-24								
Total	16,340	6,777	4,105	10,882	5,458	66.		
Male	5,904	2,923	1,194	4,118	1,786	69.		
Female	10,436	3,853	2,911	6,764	3,672	64.		
25-29								
Total	22,788	13,818	3,392	17,210	5,578	75.		
Male	9,315	7,777	942	8,719	596	93.		
Female	13,473	6,041	2,450	8,491	4,983	63.		
30-34								
Total	16,743	12,970	1,397	14,366	2,377	85.		
Male	9,236	8,811	293	9,104	131	98.		
Female 35-39	7,507	4,159	1,103	5,262	2,245	70.		
Total	13,253	10,506	930	11,436	1,818	86.		
Male	8,090	7,660	133	7,793	297	96.		
Female	5,163	2,846	797	3,643	1,520	90. 70.		
40-44	5,105	2,040	191	3,043	1,520	70.		
Total	10,038	7,761	1,059	8,820	1,219	87.		
Male	4,409	3,990	247	4,237	172	96.		
Female	5,629	3,771	811	4,582	1,046	81.		
45-49								
Total	6,988	5,090	1,441	6,531	457	93.		
Male	4,153	3,798	355	4,153	-	100.		
Female 50-54	2,835	1,292	1,086	2,378	457	83.		
Total	5,782	3,781	757	4,537	1,244	78.		
Male	2,029	1,682	267	1,949	80	96.		
Female	3,753	2,098	490	2,588	1,165	69.		
55-59	0,100	_,		_,000	.,			
Total	4,213	3,024	-	3,024	1,189	71.		
Male	3,149	2,682	-	2,682	467	85.		
Female	1,064	342	-	342	722	32.		
60-64	,							
Total	4,131	2,816	699	3,515	616	85.		
Male	2,852	2,258	584	2,842	10	99.		
Female	1,280	558	115	674	606	52.		
65+								
Total	4,710	2,360	31	2,390	2,319	50.		
Male	2,939	1,830	31	1,860	1,078	63.		
Female	1,771	530	-	530	1,241	29.		

Table 4.1.2Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days
(Current Status Approach)- AFFAR REGION: 2011

(Current Status Approach)- AMHARA REGION: 2011 Activity Status									
Age Group and Sex	All			-		Activity			
Age Group and Sex	Persons	Employed	conomically Ac Unemployed		Economically Non- Active	Rate			
All Ages	•								
Total	1,872,111	921,246	220,336	1,141,582	730,530	61			
Male	851,266	505,459	73,128	578,586	272,679	68			
Female	1,020,846	415,787	147,208	562,996	457,850	55			
10-14									
Total	271,156	31,510	4,297	35,807	235,350	13			
Male	129,499	15,570	2,066	17,636	111,863	13			
Female	141,657	15,940	2,231	18,171	123,486	12			
15-19									
Total	320,387	84,970	36,333	121,303	199,084	37			
Male	137,353	39,463	14,151	53,615	83,738	39			
Female	183,034	45,506	22,181	67,688	115,346	37			
20-24									
Total	270,024	135,617	63,143	198,760	71,264	73			
Male	111,831	64,520	19,912	84,432	27,399	75			
Female	158,194	71,098	43,230	114,328	43,865	72			
25-29									
Total	258,382	173,454	47,422	220,876	37,506	85			
Male	123,400	98,997	15,751	114,748	8,653	93			
Female	134,981	74,457	31,671	106,128	28,853	78			
30-34									
Total	146,758	109,255	19,129	128,384	18,374	87			
Male	72,134	64,521	5,274	69,794	2,340	96			
Female	74,624	44,735	13,855	58,590	16,034	78			
35-39									
Total	144,231	105,957	18,434	124,390	19,841	86			
Male	60,439	52,515	5,679	58,194	2,246	96			
Female	83,792	53,442	12,755	66,197	17,596	79			
40-44									
Total	101,489	80,832	9,062	89,893	11,595	88			
Male	55,606	51,619	2,276	53,895	1,711	96			
Female	45,882	29,213	6,785	35,998	9,884	78			
45-49	04 400	00.440	0.000	00.047	10 750	•			
Total	81,400	62,410	6,238	68,647	12,752	84			
Male	38,128	35,272	1,375	36,647	1,480	96			
Female 50-54	43,272	27,138	4,862	32,000	11,272	74			
Total	62,181	42,461	4,833	47,293	14,888	76			
Male	30,080	26,983	4,000	28,177	1,903	93			
Female	30,000	20,983 15,477	3,639	19,116	12,986	59			
55-59	52,102	13,477	5,059	19,110	12,900	0.			
Total	48,274	31,240	3,898	35,138	13,137	72			
Male	20,384	16,462	1,424	17,886	2,498	87			
Female	27,891	14,778	2,474	17,252	10,639	61			
60-64	27,001	11,770	2,171	17,202	10,000	Ũ			
Total	52,457	26,036	2,174	28,210	24,247	53			
Male	18,953	14,119	941	15,060	3,893	79			
Female	33,504	11,917	1,233	13,150	20,354	39			
65+	,	,011	.,200	,	_0,001				
Total	115,372	37,505	5,375	42,880	72,492	37			
Male	53,458	25,419	3,084	28,503	24,956	53			
Female	61,914	12,086	2,292	14,377	47,537	23			

Table 4.1.3Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days
(Current Status Approach)- AMHARA REGION: 2011

	Current		oach)- ORO	/ Status		
Age Croup and Say	All					Activity
Age Group and Sex All Ages Total Male Female 10-14 Total Male Female 15-19 Total Male Female 20-24 Total Male Female 25-29 Total Male Female 30-34 Total Male Female 30-34 Total Male Female 35-39 Total Male Female 35-39 Total Male Female 40-44 Total Male Female 40-44 Total Male Female 45-49 Total Male Female 45-49 Total Male Female 45-59 Total Male Female 50-54 T	Persons		onomically Act		Economically Non- Active	Rate
		Employed	Unemployed	Total Active		
	2 062 506	1,609,473	264 760	1 074 044	1,189,355	61.
	3,063,596		264,769	1,874,241		
	1,463,526	930,772	84,133	1,014,905	448,621	69.
	1,600,070	678,701	180,636	859,336	740,734	53
	471,532	51,677	3,015	54,693	416,839	11
	218,748	25,757	629	26,387	192,361	12
Female	252,784	25,920	2,386	28,306	224,478	11
	517,642	138,518	43,070	181,588	336,054	35
	231,608	64,716	15,952	80,668	150,940	34
	286,035	73,802	27,118	100,921	185,114	35
	200,033	75,002	27,110	100,921	105,114	
Total	480,238	269,442	87,181	356,623	123,615	74
	214,739	145,253	26,166	171,418	43,321	79
	265,499	124,189	61,015	185,205	80,294	69
Total	437,867	315,407	48,246	363,653	74,214	83
Male	218,628	189,872	13,020	202,892	15,736	92
Female	219,239	125,535	35,226	160,761	58,478	73
Total	293,600	227,235	34,358	261,593	32,007	89
	157,134	140,988	11,707	152,695	4,440	97
Female	136,466	86,247	22,651	108,898	27,568	79
Total	242,764	204,030	16,427	220,456	22,308	90
Male	119,082	113,431	4,225	117,656	1,426	98
Female	123,682	90,599	12,202	102,800	20,882	83
Total	149,328	122,229	7,762	129,991	19,337	87
	86,029	81,913	2,366	84,279	1,749	98
Female	63,299	40,316	5,396	45,712	17,587	72
Total	126,519	101,218	7,024	108,242	18,277	85
	65,151	61,255	2,226	63,481	1,671	97
Female	61,368	39,963	4,798	44,761	16,606	72
Total	86,656	58,320	6,190	64,510	22,146	74
	36,574	31,758	1,739	33,496	3,078	91
Female	50,082	26,562	4,451	31,013	19,069	61
	70,617	47,538	4,002	51,539	19,078	73
	33,417	27,973	1,414	29,387	4,030	87
Female	37,200	19,564	2,588	22,152	15,047	59
	57,881	28,477	3,233	31,710	26,170	54
	25,417	19,188	1,260	20,448	4,970	80
Female	32,463	9,289	1,974	11,263	21,201	34
65+						
Total	128,952	45,381	4,261	49,642	79,310	38
Male	56,998	28,667	3,431	32,098	24,900	56
Female	71,954	16,714	830	17,544	54,410	24

Table 4.1.4Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days
(Current Status Approach)- OROMIA REGION: 2011

	(Current	Status App	roach)- SOM	ALI REGIO	N: 2011	
			Activity	/ Status		A <i>1</i> C
Age Group and Sex	All Persons	Ec	onomically Act	ive	Economically	Activity Rate
		Employed	Unemployed	Total Active	Non- Active	Rato
All Ages						
Total	431,705	167,435	34,678	202,113	229,592	46.8
Male	219,499	98,864	13,735	112,599	106,901	51.3
Female	212,205	68,571	20,943	89,514	122,691	42.2
10-14						
Total	89,658	7,268	1,164	8,432	81,226	9.4
Male	48,608	3,501	141	3,642	44,965	7.5
Female 15-19	41,050	3,767	1,023	4,790	36,260	11.7
Total	72,294	10,173	5,034	15,207	57,087	21.0
Male	37,119	4,655	1,822	6,477	30,642	17.4
Female 20-24	35,175	5,518	3,212	8,730	26,445	24.8
Total	55,276	18,958	6,744	25,702	29,575	46.5
Male	25,671	9,426	2,080	11,507	14,164	44.8
Female 25-29	29,605	9,532	4,664	14,195	15,410	47.9
Total	48,163	26,124	6,469	32,593	15,570	67.7
Male	21,638	14,009	2,682	16,690	4,948	77.1
Female 30-34	26,525	12,115	3,787	15,902	10,622	60.0
Total	39,467	26,927	4,408	31,335	8,132	79.4
Male	19,372	16,630	1,443	18,072	1,300	93.3
Female 35-39	20,094	10,298	2,965	13,263	6,832	66.0
Total	31,358	22,971	3,143	26,114	5,244	83.3
Male	14,072	12,345	1,518	13,864	209	98.5
Female 40-44	17,285	10,626	1,625	12,250	5,035	70.9
Total	29,435	17,157	3,647	20,803	8,632	70.7
Male	15,543	11,738	1,888	13,627	1,916	87.7
Female 45-49	13,893	5,419	1,758	7,177	6,716	51.7
Total	15,929	10,720	1,850	12,570	3,358	78.9
Male	8,347	7,289	472	7,761	586	93.0
Female 50-54	7,582	3,432	1,378	4,810	2,772	63.4
Total	16,747	10,647	1,238	11,885	4,862	71.0
Male	8,511	6,433	857	7,289	1,222	85.6
Female 55-59	8,236	4,214	382	4,596	3,640	55.8
Total	7,659	4,486	379	4,865	2,794	63.5
Male	4,645	2,996	229	3,225	1,420	69.4
Female 60-64	3,014	1,490	150	1,640	1,374	54.4
Total	8,716	3,141	50	3,191	5,525	36.6
Male	4,372	2,543	50	2,593	1,780	59.3
Female 65+	4,344	599	-	599	3,745	13.8
Total	17,003	8,862	553	9,415	7,588	55.4
Male	11,602	7,299	553	7,852	3,749	67.7
Female	5,401	1,563	-	1,563	3,839	28.9

Table 4.1.5Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days
(Current Status Approach)- SOMALI REGION: 2011

		Status Approach)- BENISHANGUL GUMUZ REGION: 20 Activity Status							
Age Group and Sex	All		onomically Act		Feenemiaelly	Activity			
ige croup and bex	Persons	Employed	Unemployed	0	Economically Non- Active	Rate			
All Ages		Employed	Chempleyea						
Total	95,879	53,221	5,470	58,691	37,188	61.			
Male	45,321	30,493	1,326	31,818	13,503	70.			
Female	50,558	22,729	4,144	26,873	23,685	53			
10-14									
Total	15,898	1,386	170	1,556	14,342	9			
Male	7,452	662	100	763	6,689	10			
Female 15-19	8,446	724	70	793	7,653	9			
Total	16,846	3,929	1,169	5,098	11,748	30			
Male	6,282	1,715	239	1,954	4,329	31			
Female 20-24	10,564	2,214	930	3,144	7,420	29			
Total	13,352	8,196	1,502	9,698	3,655	72			
Male	5,490	3,613	492	4,105	1,385	74			
Female 25-29	7,863	4,583	1,010	5,593	2,270	71			
Total	13,229	10,598	964	11,562	1,666	87			
Male	6,311	5,978	27	6,005	306	95			
Female 30-34	6,918	4,620	937	5,557	1,361	80			
Total	8,685	7,540	581	8,121	564	93			
Male	4,382	4,107	225	4,332	50	98			
Female 35-39	4,304	3,433	356	3,789	515	88			
Total	9,264	8,307	387	8,694	569	93			
Male	5,112	4,999	113	5,112	-	100			
Female 40-44	4,152	3,309	274	3,583	569	86			
Total	4,341	3,777	86	3,863	478	89			
Male	2,472	2,472	-	2,472	-	100			
Female 45-49	1,869	1,304	86	1,390	478	74			
Total	3,887	2,873	441	3,314	573	85			
Male	1,976	1,769	108	1,878	98	95			
Female 50-54	1,911	1,104	333	1,436	475	75			
Total	3,888	2,969	27	2,996	891	77			
Male	1,959	1,959	-	1,959	-	100			
Female 55-59	1,928	1,010	27	1,037	891	53			
Total	1,856	1,195	21	1,215	641	65			
Male	1,065	1,003	-	1,003	62	94			
Female 60-64	791	192	21	213	579	26			
Total	1,510	874	100	975	535	64			
Male	733	733	-	733	-	100			
Female 65+	777	141	100	241	535	31			
Total	3,124	1,578	21	1,599	1,526	51			
Male	2,087	1,481	21	1,502	585	72			
Female	1,037	97	-	97	940	9			

Table 4.1.6Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days
(Current Status Approach)- BENISHANGUL GUMUZ REGION: 2011

Approach)- SNNP REGION: 2011 Activity Status											
	All		Activity								
Age Group and Sex	Persons	Ec	onomically Act	ive	Economically	Rate					
		Employed	Unemployed	Total Active	Non- Active						
All Ages											
Total	1,240,922	680,560	90,711	771,271	469,651	62.2					
Male	602,413	387,610	27,660	415,270	187,143	68.9					
Female	638,508	292,950	63,050	356,001	282,508	55.8					
10-14											
Total	218,794	33,356	2,614	35,971	182,823	16.4					
Male	108,585	19,799	1,678	21,476	87,109	19.8					
Female	110,210	13,558	937	14,495	95,715	13.2					
15-19 Total	227 094	60.000	10 100	00 000	120.076	20 (
Total Mala	227,984	68,880	19,128	88,008	139,976	38.6					
Male	101,769	33,936	6,895	40,831	60,938	40.2					
Female 20-24	126,215	34,944	12,233	47,177	79,038	37.4					
Total	178,026	100,499	24,587	125,087	52,940	70.3					
Male	81,545	51,610	7,405	59,015	22,530	72.4					
Female	96,481	48,889	17,183	66,072	30,409	68.5					
25-29		,	,	,							
Total	167,408	128,085	16,798	144,883	22,525	86.5					
Male	78,076	70,736	3,894	74,630	3,446	95.6					
Female	89,332	57,349	12,904	70,253	19,079	78.6					
30-34											
Total	122,780	101,579	8,705	110,284	12,495	89.8					
Male	64,878	60,530	1,948	62,478	2,400	96.3					
Female	57,901	41,049	6,757	47,806	10,095	82.6					
35-39											
Total	107,982	89,417	7,167	96,584	11,398	89.4					
Male	52,822	50,467	1,184	51,651	1,171	97.8					
Female	55,161	38,950	5,984	44,933	10,227	81.					
40-44	04.000	55 404	0.004	50.045	5 00 4						
Total	64,209	55,431	2,884	58,315	5,894	90.8					
Male	37,428	36,120	908	37,028	400	98.					
Female 45-49	26,782	19,311	1,976	21,288	5,494	79.					
Total	47,339	37,581	3,903	41,484	5,855	87.0					
Male	24,790	23,492	701	24,193	597	97.0					
Female	22,549	14,089	3,202	17,292	5,257	76.					
50-54	22,010	1,000	0,202	,202	0,201						
Total	33,477	26,464	1,573	28,037	5,439	83.8					
Male	17,097	16,122	918	17,040	58	99.					
Female	16,379	10,343	655	10,997	5,382	67.					
55-59											
Total	21,843	15,482	1,331	16,813	5,030	77.					
Male	10,590	8,873	877	9,750	840	92.					
Female	11,253	6,609	454	7,063	4,190	62.					
60-64											
Total			1,061	11,285	3,922	74.					
Male			297	7,760	1,575	83.					
Female	5,872	2,760	765	3,525	2,347	60.					
65+	05.05.										
Total	35,874	13,562	958	14,520	21,354	40.					
Male	15,499	8,462	958	9,420	6,079	60.8					
Female	20,375	5,100	-	5,100	15,275	25.					

Table 4.1.7Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days (Current Status
Approach)- SNNP REGION: 2011

	••		LLA REGION Activit	y Status		
Age Group and Sex	All	E	conomically Act	,	Economically	Activity
rige croup and cox	Persons	Employed	Unemployed		Non- Active	Rate
All Ages		Linpioyeu	Onemployed	Total Active		
Total	66,426	33,342	2,785	36,127	30,299	54.
Male	31,685	16,628	1,368	17,995	13,690	56.
Female	34,740	16,715	1,417	18,132	16,609	52.
10-14	0+,1+0	10,710	1,417	10,102	10,000	02.
Total	13,753	769	157	926	12,827	6.
Male	6,492	167	61	228	6,263	3.
Female 15-19	7,261	601	96	697	6,564	9.
Total	11,233	2,891	349	3,239	7,994	28.
Male	5,446	1,073	165	1,238	4,207	22.
Female	5,788	1,817	184	2,001	3,786	34.
20-24						
Total	10,037	5,665	820	6,485	3,552	64.
Male	4,552	2,641	380	3,022	1,531	66.
Female 25-29	5,485	3,023	440	3,463	2,021	63.
Total	11,213	8,424	784	9,208	2,004	82.
Male	4,245	3,176	408	3,585	661	84.
Female 30-34	6,967	5,248	376	5,624	1,344	80.
Total	6,211	5,043	226	5,269	942	84.
Male	3,363	3,051	34	3,084	279	91.
Female 35-39	2,848	1,993	192	2,185	663	76.
Total	5,496	4,891	94	4,985	511	90.
Male	3,592	3,457	46	3,503	89	97.
Female 40-44	1,904	1,434	48	1,482	422	77.
Total	2,587	2,240	95	2,335	252	90.
Male	1,395	1,298	70	1,368	27	98.
Female 45-49	1,192	942	26	968	225	81.
Total	2,264	1,751	80	1,831	433	80.
Male	865	643	56	699	166	80.
Female 50-54	1,399	1,108	25	1,132	267	80.
Total	1,075	598	37	635	440	59.
Male	507	438	37	474	33	93.
Female 55-59	567	161	-	161	407	28.
Total	862	323	31	354	508	41.
Male	343	237		237	106	69.
Female 60-64	519	86	31	117	402	22.
Total	895	563	56	619	275	69.
Male	443	262	56	319	124	72.
Female 65+	452	301	-	301	151	66.
Total	800	184	55	239	561	29.
Male	800 443	184	55	239	204	29. 54.
Female	443 357	104	55	239	204 357	54.

Table 4.1.8Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days (Current Status
Approach)- GAMBELLA REGION: 2011

	Approach)- HARARI REGION: 2011											
			Activit	y Status		A						
Age Group and Sex	All Persons	Ec	conomically Ac	tive	Economically	Activity Rate						
		Employed	Unemployed	Total Active	Non- Active	Rato						
All Ages	•			•								
Total	82,347	48,450	7,624	56,074	26,273	68.1						
Male	39,483	25,806	2,267	28,073	11,410	71.1						
Female	42,864	22,644	5,357	28,001	14,863	65.3						
10-14												
Total	11,296	1,275	70	1,345	9,950	11.9						
Male	5,861	677	70	747	5,114	12.7						
Female	5,435	599	-	599	4,837	11.0						
15-19												
Total	10,554	3,285	745	4,029	6,525	38.2						
Male	5,280	1,560	347	1,906	3,374	36.1						
Female 20-24	5,273	1,725	398	2,123	3,151	40.3						
Total	,		1,673	7,724	2,059	79.0						
Male	4,110		378	3,132	978	76.2						
Female 25-29	5,672	3,297	1,295	4,592	1,080	81.0						
Total	12,211	9,500	1,521	11,021	1,190	90.3						
Male	5,896	5,123	417	5,540	356	94.0						
Female	6,315	4,376	1,104	5,481	834	86.8						
30-34												
Total	8,450	6,764	997	7,761	689	91.8						
Male	3,744	3,439	226	3,665	79	97.9						
Female 35-39	4,705	3,324	771	4,096	610	87.0						
Total	8,073	6,856	595	7,451	621	92.3						
Male	4,381	4,159	78	4,238	143	96.7						
Female	3,692	2,697	517	3,214	478	87.0						
40-44												
Total	4,430	3,505	423	3,929	501	88.7						
Male	2,219	2,044	80	2,124	95	95.7						
Female 45-49	2,211	1,462	343	1,805	406	81.6						
Total	4,276	3,447	568	4,016	260	93.9						
Male	2,296	2,065	231	2,296	-	100.0						
Female 50-54	1,979	1,382	337	1,719	260	86.8						
Total	3,409	2,897	217	3,115	295	91.4						
Male	1,311	1,311	-	1,311	-	100.0						
Female 55-59	2,099	1,587	217	1,804	295	86.0						
Total	2,933	2,329	292	2,620	312	89.4						
Male	1,323	1,196	85	1,281	42	96.8						
Female	1,610	1,133	207	1,340	270	83.2						
60-64												
Total	2,441	1,343	354	1,697	744	69.5						
Male	1,153	660	274	934	219	81.0						
Female	1,288	682	80	763	525	59.2						
65+												
Total	4,493	1,199	167	1,366	3,127	30.4						
Male	1,908	818	81	899	1,009	47.1						
Female	2,585	381	86	467	2,118	18.1						

Table 4.1.9Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days (Current Status
Approach)- HARARI REGION: 2011

				/ Status	RATION: 2011	1
Age Croup and Say	All					Activity
Age Group and Sex	Persons		nomically Act		Economically Non- Active	Rate
All Ages		Employed I	Jnemployed	Total Active		
Total	2,504,853	1,148,974	384,317	1,533,291	971,562	61.
Male	1,141,811	667,973	139,334	807,307	334,504	70.
Female	1,363,042	481,001	244,982	725,984	637,058	70. 53.
10-14	1,303,042	401,001	244,902	725,904	037,030	55.
Total	258,791	11,315	3,524	14,839	243,952	5.
Male	105,438	2,372	1,189	3,561	101,877	3.
Female	153,353	8,943	2,335	11,278	142,075	7.
15-19	100,000	0,010	2,000	11,210	1 12,010	•
Total	389,937	84,047	34,626	118,673	271,265	30.
Male	153,156	30,378	13,491	43,870	109,286	28.
Female	236,781	53,669	21,134	74,803	161,979	31
20-24	200,101	00,000	21,101	. 1,000	101,010	01.
Total	393,031	181,644	88,017	269,660	123,371	68.
Male	177,419	93,626	39,177	132,803	44,616	74
Female	215,612	88,018	48,839	136,857	78,755	63
25-29						
Total	387,359	248,032	83,267	331,299	56,060	85
Male	180,563	142,071	26,336	168,407	12,156	93
Female	206,796	105,961	56,931	162,892	43,904	78
30-34						
Total	239,171	165,921	47,111	213,032	26,139	89
Male	128,616	104,895	19,058	123,953	4,664	96
Female	110,555	61,025	28,054	89,079	21,475	80
35-39						
Total	216,443	150,185	39,242	189,427	27,016	87.
Male	102,335	85,898	11,533	97,431	4,904	95
Female	114,108	64,288	27,709	91,996	22,111	80
40-44						
Total	128,656	81,553	27,618	109,171	19,484	84
Male	69,369	58,003	7,491	65,493	3,875	94
Female	59,287	23,550	20,128	43,678	15,609	73
45-49						
Total	115,370	74,669	20,348	95,017	20,353	82
Male	51,776	44,742	4,500	49,242	2,533	95
Female	63,594	29,926	15,848	45,774	17,820	72
50-54						
Total	104,177	65,727	16,663	82,390	21,787	79
Male	50,208	41,529	5,155	46,684	3,524	93
Female	53,969	24,197	11,508	35,705	18,264	66
55-59						
Total	61,945	32,026	7,754	39,780	22,165	64
Male	26,308	21,243	2,400	23,643	2,665	89
Female	35,637	10,784	5,353	16,137	19,500	45
60-64						
Total	69,415	23,960	7,415	31,375	38,040	45
Male	31,775	18,868	3,123	21,992	9,783	69
Female	37,640	5,091	4,292	9,383	28,257	24
65+						
Total	140,558	29,896	8,732	38,628	101,930	27
Male	64,848	24,347	5,881	30,228	34,620	46
Female	75,711	5,549	2,851	8,400	67,310	11

Table 4.1.10Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days (Current Status
Approach) - ADDIS ABABA CITY ADMINISTRATION: 2011

	Approach) - DIRE DAWA ADMINISTRATION: 2011 Activity Status										
Ago Croup and	All		Activity								
Age Group and Sex	Persons	E	conomically Ac	Economically	Rate						
		Employed	Unemployed	Total Active	Non- Active						
All Ages											
Total	179,561	84,025	25,024	109,048	70,512	60.7					
Male	84,187	46,825	7,200	54,024	30,162	64.2					
Female	95,374	37,200	17,824	55,024	40,350	57.7					
10-14											
Total	26,926	2,554		2,628	24,298	9.8					
Male	13,291	1,096		1,171	12,120	8.					
Female	13,636	1,457	-	1,457	12,179	10.					
15-19				- <i>i</i> = <i>i</i>							
Total	28,694	6,543	2,931	9,474	19,220	33.					
Male	13,195	2,892		3,260	9,935	24.					
Female 20-24	15,499	3,651	2,563	6,214	9,285	40.					
Total	26,815	12,978	6,770	19,747	7,068	73.					
Male	10,500	5,969	1,559	7,528	2,972	71.					
Female	16,315	7,009	5,211	12,219	4,096	74.					
25-29 Total	24,522	16,097	4,610	20,707	3,815	84.					
Male	12,786	10,424	1,231	11,655	1,132	91.					
Female	11,736	5,673	3,379	9,052	2,684	77.					
30-34	,	0,010	0,010	0,002	_,						
Total	13,431	9,404	2,443	11,847	1,585	88.					
Male	6,754	5,700		6,307	447	93.					
Female	6,678	3,704		5,540	1,138	83.					
35-39											
Total	16,542	11,333		14,111	2,431	85.					
Male	7,485	6,244		7,074	412	94.					
Female	9,057	5,089	1,949	7,038	2,019	77.					
40-44											
Total	10,709	7,452	1,736	9,189	1,520	85.					
Male	5,334	3,941	1,005	4,946	388	92.					
Female 45-49	5,375	3,511	732	4,243	1,132	78.					
Total	9,461	7,360	1,030	8,390	1,071	88.					
Male	9,401 4,971	4,440	379	4,818		96.					
Female	4,490	2,921	651	3,572	918	50. 79.					
50-54	4,400	2,021	001	0,012	510	70.					
Total	5,994	4,048	1,208	5,256	738	87.					
Male	2,768	2,311	315	2,626	141	94.					
Female	3,227	1,736	893	2,630	597	81.					
55-59		·		·							
Total	4,691	2,771	813	3,584	1,107	76.					
Male	2,650	1,982	366	2,348	302	88.					
Female	2,041	789	447	1,236	805	60.					
60-64											
Total	3,696	1,566	315	1,881	1,815	50.					
Male	1,440	600	315	915	524	63.					
Female	2,257	966	-	966	1,291	42.					
65+											
Total	8,079	1,920	314	2,234	5,844	27.					
Male	3,015	1,224	152	1,376	1,638	45.					
Female	5,064	696	162	858	4,206	16.					

Table 4.1.11Urban population Aged Ten Years and above by Age Group, Sex,
Activity Status and Rate during the Last Seven Days Current Status
Approach) - DIRE DAWA ADMINISTRATION: 2011

	urrent Stat	us Approach	1): 2011 Activity	-		
			Activity			
Town	All	Ec	onomically Acti	ve	Economically	Rate
	Persons	Employed	Unemployed	Total Active	Non- Active	itato
MAJOR TOWNS TOT	AL					
Total	4,179,322	1,965,425	580,108	2,545,533	1,633,789	60.9
Male	1,920,953	1,134,199	200,434	1,334,633	586,320	69.5
Female	2,258,369	831,226	379,674	1,210,901	1,047,469	53.6
MEKELE TOWN		,			, ,	
Total	190,825	87,546	22,400	109,946	80,879	57.6
Male	85,504	49,393	7,085	56,478	29,026	66.1
Female	105,321	38,153	15,315	53,468	51,853	50.8
ASAYITA TOWN	,	,	,	,	,	
Total	11,591	6,472	810	7,283	4,309	62.8
Male	5,525	3,810	198	4,007	1,517	72.5
Female	6,067	2,663	613	3,276	2,791	54.0
GONDER TOWN	0,001	_,	0.0	0,210	_,	0.110
Total	177,487	82,743	18,426	101,170	76,318	57.0
Male	78,724	45,705	6,867	52,572	26,152	66.8
Female	98,764	37,038	11,559	48,598	50,166	49.2
DESSIE TOWN	00,701	07,000	11,000	10,000	00,100	10.2
Total	105,306	46,017	10,363	56,380	48,926	53.5
Male	49,121	25,986	3,311	29,297	19,823	59.6
Female	56,185	20,032	7,051	27,083	29,103	48.2
BAHIR DAR TOWN	00,100	20,002	7,001	21,000	20,100	10.2
Total	161,116	89,993	26,835	116,828	44,288	72.5
Male	69,452	46,674	8,876	55,550	13,903	80.0
Female	91,664	43,319	17,959	61,278	30,386	66.9
BISHOFTU TOWN	51,004	40,010	17,000	01,270	00,000	00.0
Total	80,796	40,673	9,915	50,588	30,208	62.6
Male	37,510	22,604	2,978	25,583	11,927	68.2
Female	43,286	18,069	6,936	25,005	18,281	57.8
ADAMA TOWN	40,200	10,003	0,350	23,005	10,201	57.0
Total	202,760	101,178	27,097	128,275	74,485	63.3
Male	93,992	60,623	7,972	68,596	25,397	73.0
Female	108,768	40,555	19,125	59,679	49,089	54.9
JIMMA TOWN	100,700	+0,000	10,120	00,070	40,000	04.0
Total	106,713	55,260	10,374	65,634	41,079	61.5
Male	50,940	33,297	3,048	36,345	14,595	71.3
Female	55,774	21,963	7,326	29,289	26,485	52.5
SHASHEMENE TOWN		21,303	7,520	23,203	20,400	02.0
Total	114,686	54,907	10,390	65,297	49,389	56.9
Male	56,346	32,752	3,135	35,887	20,459	63.7
Female						50.4
JIJIGA TOWN	58,340	22,155	7,255	29,410	28,930	50.4
Total	80 420	20.080	9,216	38,304	12 124	47.6
Male	80,439 40,616	29,089 20,272	3,384		42,134 16,961	47.6 58.2
				23,655		
Female	39,822	8,817	5,832	14,649	25,174	36.8
ASOSA TOWN	04 700	40.040	4 400	4 4 757	0.000	E0 7
Total Mala	24,739	13,319	1,438	14,757	9,982	59.7
Male	11,848	7,830	324	8,154	3,694	68.8
Female	12,891	5,489	1,115	6,603	6,288	51.2

Table 4.2Populations of Major Towns Aged Ten Years and Above by Sex,
Activity Status and Rate during the Last Seven Days
(Current Status Approach): 2011

Table 4.2 cont'd

			Activity	Status		
Town	All	Ec	onomically Acti	ve	Economically	Activity Rate
	Persons	Employed	Unemployed	Total Active	Non- Active	Nate
HAWASSA TOWN						
Total	127,209	64,568	14,017	78,586	48,624	61.8
Male	61,125	37,212	3,521	40,733	20,392	66.6
Female	66,085	27,356	10,496	37,852	28,232	57.3
GAMBELLA TOWN						
Total	28,893	12,209	1,864	14,073	14,820	48.7
Male	14,770	7,437	935	8,372	6,398	56.7
Female	14,124	4,772	929	5,701	8,422	40.4
HARAR TOWN						
Total	82,347	48,450	7,624	56,074	26,273	68.1
Male	39,483	25,806	2,267	28,073	11,410	71.1
Female	42,864	22,644	5,357	28,001	14,863	65.3
ADDIS ABABA CITY						
ADMINISTRATION						
Total	2,504,853	1,148,974	384,317	1,533,291	971,562	61.2
Male	1,141,811	667,973	139,334	807,307	334,504	70.7
Female	1,363,042	481,001	244,982	725,984	637,058	53.3
DIRE DAWA TOWN						
Total	179,561	84,025	25,024	109,048	70,512	60.7
Male	84,187	46,825	7,200	54,024	30,162	64.2
Female	95,374	37,200	17,824	55,024	40,350	57.7

		o During		on Days (O		for Non -		JUNIRTIUL			
Age Group and	Total Non-	Home					Too		Old ago/		Not
Sex		Maker	Dragnanav	Student	Injury/	Illnooo		Domittonoo	Old age/	Othoro	
	Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Remittance	Pensioned	Others	Stated
All Ages											
Total	4,131,249	403,760	128,393	2,800,699	24,597	210,567	55,639	68,289	359,382	79,645	277
Male	1,561,317	11,567		1,300,405	11,263	73,312		12,030	106,900	22,716	277
Female	2,569,931	392,193	128,393	1,500,294	13,334	137,255	32,792	56,259	252,483	56,929	
10-14	2,000,001	002,100	120,000	1,000,201	10,001	101,200	02,102	00,200	202,100	00,020	
Total	1,350,474	10,236	-	1,270,774	1,408	4,580	55,639	772	-	6,787	277
Male	634,099	1,313	-	603,825	933	2,663	22,847	289	-	1,952	277
Female	716,375	8,924	-	666,949	476	1,917	32,792	483	-	4,835	
15-19	,	-,		,		.,	,			,	
Total	1,148,479	29,793	14,627	1,081,618	2,089	10,012	-	2,242	-	8,098	-
Male	501,106	1,251	-	491,524	640	4,356	-	567	-	2,769	-
Female	647,372	28,542	14,627	590,095	1,449	5,656	-	1,675	-	5,329	-
20-24	- ,-	- , -	7 -	,	, -	-,		,		-,	
Total	450,029	49,774	41,446	322,999	1,780	16,428	-	5,031	-	12,572	-
Male	169,166	1,300	-	154,913	697	7,492	-	713	-	4,051	-
Female	280,863	48,473	41,446	168,086	1,082	8,936	-	4,318	-	8,521	-
25-29			,			,					
Total	234,315	76,012	39,834	78,352	2,220	19,402	-	8,424	-	10,070	-
Male	51,680	3,199	-	35,512	692	7,539	-	880	-	3,859	-
Female	182,634	72,813	39,834	42,840	1,529	11,863	-	7,545	-	6,211	-
30-34											
Total	111,944	51,012	15,315	15,323	2,091	16,561	-	2,956	-	8,686	-
Male	16,677	637	-	4,978	1,004	7,666	-	577	-	1,815	-
Female	95,267	50,374	15,315	10,345	1,087	8,895	-	2,380	-	6,871	-
35-39											
Total	99,372	51,340	12,292	7,135	809	15,520	-	3,447	-	8,829	-
Male	11,672	79	-	2,907	392	4,944	-	637	-	2,712	-
Female	87,700	51,261	12,292	4,228	417	10,576	-	2,809	-	6,117	-
40-44											
Total	75,394	36,371	3,610	2,942	2,074	18,255	-	5,119	-	7,022	-
Male	11,830	662	-	811	1,266	6,455	-	409	-	2,227	-
Female	63,564	35,709	3,610	2,131	808	11,801	-	4,710	-	4,795	-

Table 4.3Economically Non - Active Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Reasons for
Not Being Active During the Last Seven Days (Current Status Approach) - COUNTRY TOTAL: 2011

Table 4.3 cont'd

Age Group and					Reasons	for Non -	Active				
Sex	Total Non- Active	Home Maker	Pregnancy	Student	Injury/ Disabled	Illness	Too Young	Remittance	Old age/ Pensioned	Others	Not Stated
45-49											
Total	69,426	35,747	1,269	4,522	1,457	16,469	-	4,189	-	5,772	
Male	8,009	764	-	821	1,279	3,482	-	227	-	1,437	
Female	61,416	34,983	1,269	3,701	179	12,987	-	3,962	-	4,335	
50-54		,				,				,	
Total	80,250	29,330	-	3,137	1,979	20,702	-	7,329	15,117	2,656	
Male	11,493	892	-	312	425	5,006	-	1,436	3,211	211	
Female	68,757	28,438	-	2,825	1,554	15,696	-	5,893	11,906	2,445	
55-59	, -	-,		,	,	-,		-,	,	, -	
Total	73,312	14,292	-	1,440	1,586	18,454	-	5,602	29,810	2,129	
Male	13,503	211	-	520	358	5,002	-	538	6,620	254	
Female	59,809	14,081	-	919	1,228	13,452	-	5,065	23,190	1,875	
60-64		,				,			,	,	
Total	109,941	11,169	-	3,801	1,673	17,704	-	7,421	64,156	4,018	
Male	24,112	336	-	1,329	583	3,935	-	1,654	15,388	886	
Female	85,829	10,833	-	2,472	1,090	13,768	-	5,767	48,768	3,131	
65+	,	,		,	,	,		,	,	,	
Total	328,315	8,684	-	8,656	5,430	36,480	-	15,757	250,300	3,007	-
Male	107,970	⁹²¹	-	2,954	2,993	14,772	-	4,105	81,680	544	
Female	220,345	7,763	-	5,703	2,436	21,708	-	11,653	168,619	2,463	

	Reasons for Non - Active										
Region and Sex	Total Non-	Home			Injury/		Тоо		Old age/		Not
	Active	Maker	Pregnancy	Student	Disabled	Illness	Young	Remittance	Pensioned	Others	Stated
COUNTRY - TOTAL											
Total	4,131,249	403,760	128,393	2,800,699	,	210,567	55,639	68,289	359,382	79,645	277
Male	1,561,317	11,567	-	1,300,405	11,263	73,312	22,847	12,030	106,900	22,716	277
Female	2,569,931	392,193	128,393	1,500,294	13,334	137,255	32,792	56,259	252,483	56,929	-
TIGRAY REGION											
Total	317,515	31,481	9,379	199,747	2,023	20,564	12,843	3,207	35,067	3,202	-
Male	119,772	732	-	95,567	1,109	6,393	6,009	452	9,198	311	-
Female	197,743	30,749	9,379	104,180	914	14,172	6,834	2,755	25,869	2,891	-
AFFAR REGION											
Total	58,772	8,768	2,926	39,542	270	3,306	160	403	2,620	779	-
Male	22,932	524	-	19,928	68	890	20	31	1,173	298	-
Female	35,840	8,244	2,926	19,614	202	2,416	139	372	1,447	481	-
AMHARA REGION											
Total	730,530	71,931	22,136	486,153	5,614	45,420	4,756	16,940	69,939	7,364	277
Male	272,679	958	-	223,809	1,868	18,102	2,602	2,033	21,794	1,237	277
Female	457,850	70,973	22,136	262,344	3,746	27,318	2,154	14,907	48,145	6,128	-
OROMIA REGION											
Total	1,189,355	119,944	43,574	835,416	7,336	53,944	12,045	17,368	86,051	13,677	-
Male	448,621	2,184	-	390,264	4,402	15,940	4,601	2,052	24,880	4,298	-
Female	740,734	117,759	43,574	445,152	2,934	38,005	7,444	15,316	61,171	9,379	-
SOMALI REGION											
Total	229,592	36,706	7,327	143,214	897	8,310	4,911	8,855	8,044	11,326	-
Male	106,901	2,931	-	84,705	617	5,155	1,945	2,817	2,577	6,152	-
Female	122,691	33,775	7,327	58,509	280	3,155	2,966	6,038	5,468	5,175	-
BENISHANGUL GUN	IUZ REGION										
Total	37,188	2,338	953	28,761	-	1,347	346	419	2,236	790	-
Male	13,503	31	-	12,464	-	104	106	19	627	153	-
Female	23,685	2,307	953	16,297	-	1,243	239	400	1,609	638	-
SNNP REGION											
Total	469,651	47,177	9,990	357,228	2,418	13,999	7,672	2,815	23,924	4,428	-
Male	187,143	1,274	-	169,040	673	5,901	3,147	765	6,148	195	-
Female	282,508	45,903	9,990	188,188	1,745	8,098	4,525	2,050	17,776	4,232	-

Table 4.4Economically Non - Active Population of Urban Areas Aged Ten Years and Above by Region, Sex and Reasons
for Not Being Active during the Last Seven days (Current Status Approach): 2011

Table 4.4 cont'd											
	Reasons for Non - Active										
Region and Sex	Total Non- Active	Home Maker	Pregnancy	Student	Injury/ Disabled	Illness	Too Young	Remittance	Old age/ Pensioned	Others	Not Stated
GAMBELLA REGION	7101170	Marci	rieghaney	Olddeni	Disablea	1111000	roung	Tterrittariee	1 choinea	Others	Olaica
Total	30,299	2,842	739	21,245	155	907	2,506	482	925	499	-
Male	13,690	258	-	11,157	81	456	1,075	274	202	188	-
Female	16,609	2,584	739	10,088	74	451	1,431	209	723	311	-
HARARI REGION											
Total	26,273	1,593	526	17,664	275	2,305	215	693	2,342	661	-
Male	11,410	41	-	9,268	84	951	-	290	593	184	-
Female ADDIS ABABA CITY ADMINISTRATION	14,863	1,553	526	8,397	191	1,354	215	403	1,748	477	-
Total	971,562	75,694	28,714	624,454	5,276	55,534	9,271	14,410	122,591	35,617	-
Male	334,504	2,546	-	259,804	2,117	17,172	2,963	2,483	38,265	9,154	-
Female DIRE DAWA ADMINISTRATION	637,058	73,148	28,714	364,650	3,158	38,362	6,308	11,927	84,327	26,464	-
Total	70,512	5,286	2,129	47,275	334	4,931	915	2,698	5,644	1,301	-
Male	30,162	87	-	24,400	243	2,249	378	815	1,443	546	-
Female	40,350	5,199	2,129	22,875	91	2,682	536	1,882	4,201	755	-

CHAPTER V

EMPLOYED POPULATION IN URBAN AREAS

5.1. Introduction

The economically active population has defined in detail in the previous chapter. As a part of the economically active population, employed population are those who were engaged in productive activity at least four hours during seven days prior to the date of interview. It also includes persons who had regular jobs or business to return to but who were temporarily absent from work.

The 2011 UEUS has collected the following information regarding the characteristics of the currently employed population:

- a) Main type of occupation;
- b) Main product or service of the establishment or industry;
- c) Employment status for main activity and terms of employment;
- d) Reasons for absence from work
- e) Amount of earnings for paid employees;
- f) Hours worked per week and
- g) Time related under related Under employment.
- h) Formal or informal employment sectors;

This chapter presents statistical data on the size, distribution and characteristics of the currently employed population in the consecutive sub-sections.

5.2. Employment to Population Ratio

According to the 18 Key Indicators of Labour Market (KILM) used by the ILO, employment to population ratio is calculated as a percentage of the total employed persons to the working age population (ILO, 1999). The distribution of employment to population ratio for urban areas of the country by age group and sex is shown in Table 5.1. The corresponding figures for each region are presented in Table 5.1.1 - 5.1.11. The data for employment to population ratio for the total major towns and each

selected major towns by age group and sex are also presented in Table 5.2 and 5.2.1 - 5.2.16, respectively.

5.3. Occupation and Industry

In this survey, the currently employed persons were asked about the type of main activity such as occupation and major product or service of the establishment or industry in which they were engaged during the survey reference period. The questions used to identify the type of occupation and industry, were left open ended so that they will be filled in with the fullest description and its code in the field. For those employed persons engaged in multiple activities, the activity that took the highest share of the respondents' time was taken as the main type of activity.

The type of occupation and industry filled in by enumerators were further checked and edited at the head quarters using the National Occupational and Industrial Classifications (NOIC) code-book. The NOIC codes were adopted from the International Standard Classification of Occupation ISCO-88 (ILO, 1990) and International Standard Industrial Classification (ISIC, 1993) by taking into account the prevailing national socio-economic conditions.

The NOIC uses a 3-digit coding systems corresponding to 3 levels classifications in successively finer details. In the cases of occupation and industry, the levels are ordered as major divisions, sub major divisions, and minor divisions. The occupational classification was categorized into 9 major groups, 28 sub major groups and 113 minor groups. The industrial classification contains, 13, 60 and 159 major, sub major and minor divisions, respectively.

The distribution of employed population of urban areas by major occupational groups, age group and sex for the country total and each region is presented in Table 5.3 and 5.4, respectively. Table 5.5 presents the distribution of employed population by major occupation, sex and for each selected major towns. The distribution of employed population by minor occupational groups for each region classified by sex appears in Table 5.6. One should note that in Table 5.6 and 5.10 major and sub major occupation

and industrial classification are written in capital letter, while minor occupation and industrial classification are written in small letter, respectively. Analogous information that deals with the industrial divisions are also shown in Table 5.7 to 5.10.

5.4. Employment Status

Employment status of a person indicates the level of involvement and degree of decision-making in his main activity. In this survey employed persons were asked about the employment status of their main activity which was classified into: employer, paid employees, self-employed, unpaid family worker, apprentice, member's of producer's, cooperative and others categories. Paid employees were further disaggregated into employees-government, employees government parastatal, employees NGO's, employees private organization, employees domestic and other paid employees.

Table 5.11 and Table 5.12 present the distribution of status in employment for urban areas of country total and each region by age group and sex. The distribution of status in employment of the employed population of major towns classified by sex is presented in Tables 5.13. Table 5.14 contains data on major occupational groups classified by major industrial divisions and sex for urban areas of the country total. The cross classification of status in employment by major industrial divisions and sex for a country total is shown in Table 5.15. Also, the figures for status in employment by major occupation and sex for country total are presented in Table 5.16.

5.5. Number of Hours worked

The absence of standard working hours and irregular nature of working days in the informal sectors or in self employment, data collection as well as getting accurate data on hours of work were found to be difficult. In addition, the reliability of the data collection is likely to be affected due to memory lapse and lack of knowledge or information about the concepts of time on the part of the respondent. Thus, considering the inherent data collection problem on hours of work in developing

countries, the figures presented in these tables should be regarded as indicative and not the true levels of intensity of work.

The 2011 UEUS included questions on the number of hours worked for two purposes. First, the response on the number of hours worked during the seven days prior to the date of the interview is used to classify the respondents as employed or not on the basis of the given criteria of the minimum required hours of work. As described in Chapter IV persons who worked at least four hours or more, and those who were not worked or worked less than 4 hours but had a job to return to were considered as employed. The rest of persons were then subjected to other filtering questions for unemployment and inactivity. The second objective of including questions on number of hours worked was to estimate the average number of working hours of the employed population.

In this survey, all persons who were engaged in productive activities during the seven days prior to the date of interview were asked to provide the number of hours they worked in all of their productive activities. The response was recorded for each of the days in a separate notebook so as to get the total actual hours of work during the reference week. Then the hours so obtained were added together for the week and finally filled in the main questionnaire.

The 20011 UEUS define hours actually worked is based on the resolution provided by the tenth ICLS in 1962 (ILO, 1976). The statistics on the hours actually worked includes:

- a) Hours actually worked during normal periods of work and time spent for waiting for the market;
- b) Time worked in addition to hours worked during normal periods of work, and generally paid at higher rates than normal rates (overtime);
- c) Time spent at place of work on activities such as the preparation of the work place; repairing maintenance, preparing and cleaning of tools and others;
- d) Time spent at the place of work waiting or standing by for such reasons as lack of supply of work, break down of machinery, or accidents, or time spent at the

place of wok during which no work is done but for which payment is made under a guaranteed employment contract and

e) Time corresponding to short rest periods at the work place including tea and coffee breaks;

According to the resolution in recording the number of hours worked, care was taken to exclude hours paid for but not worked, such as paid annual leave, paid public holidays or paid sick leave. Time spent on meal breaks, travel time from home to work and vice versa /for those who have specified place of work/, and hours spent on domestic activities that were not considered as productive are also excluded. For employed persons who were not at work during the seven days prior to the date of interview, the number of hours of work is recorded as zero. On the other hand, any time that employed persons have spent in productive activity in places other than the work site is considered as working hour.

Table 5.17 and Table 5.18 present the distribution employed persons by the number of hours worked, age group and sex during the last seven at country urban and regional levels, respectively. The distribution of employed persons by the number of hours worked, major industrial divisions, major occupation, and status in employment for country total disaggregated by sex are shown in Table 5.19, 5.20 and 5.21, in that order.

5.6. Earning from Paid Employment

Paid employment jobs are those jobs where the employees hold explicit (written or oral) or implicit employment contracts, which give them a basic remuneration. Some or all of the tools, capitals equipment, information systems and/or premises used by the employees may be owned by others, and the employees may work under direct supervision or according to the strict guidelines set by the owner(s) or persons in the owners' employment. Persons in "paid employment" jobs are typically remunerated by wages and salaries, but may be paid by commission from sales, by piece rates, bonuses or in kind payments such as food, housing or clothing. In this survey,

earnings for paid employees refer to gross remuneration and include bonus, overtime, allowances and other benefits that are obtained only from the main job.

Table 5.22 shows the distribution of paid employees for urban areas the country and each region disaggregated by sex. The distribution of paid employees in urban areas by sex and period of payment at country level is presented in Table 5.23. The distribution of paid employees of urban areas of the country classified by amount and period of payment and sex is presented in table 5.24. Paid employees population of urban areas of the country by major industrial divisions, sex and amount of payment per month has also shown in Table 5.25

5.7. Absence from Work

Employed persons that were considered to be absent from work were those who were not engaged in any productive activity or who work less than four hours during the seven days prior to the date of the interview but have formal job attachment. These persons were asked their main reason for being absent from work. Reasons for absence from work was classified as illness, holiday /vacation, Training/education, offseason, reduction in economic activity and "others" categories. The data presented in Table 5.26 indicates the distribution of employed population for their reasons for being absent from work by region and sex.

5.8. Time Related Under Employment

Time related under employment reflects underutilization of the productive capacity of the employed population including those which arise from deficient national or local economic system. It is defined as those persons who were willing and available to work additional hours, regardless of the hours they actually worked during the reference period (ILO, 2000).

In the survey time related under employment data was collected. Those employed persons who were willing and available to work additional hours were categorized as

time related under employed. Besides, underemployed persons state the type of work they sought to work additional hours like more hours at present job, additional hours in other job in addition to the present job and replacing the current job by another job with more hours of work.

The distribution of employed population whether they were willing and available to work additional hours by age group and sex is shown in Table 5.27. Table 5.28 shows distribution of employed persons who were willing and available to work additional hours together with the actual number of hours worked by age group and sex for the total urban areas of the country. Table 5.29 presents the distribution of urban employed population of each region who were willing and available to work additional hours of work classified by type of work and sex.

5.9. Formal and Informal Employment Sector

For statistical purposes, the informal sector is regarded as a group of production units, which according to the definitions and classification provided in the United Nation System of National Accounts SNA Rev.4, form part of the household sector as household enterprises or, equivalently, unincorporated enterprises owned by households. The informal sector is defined irrespective of the kind of work place where the productive activities are carried out, the extent of fixed capital used, the duration of the enterprise and its operation as main or secondary activity of the owner (ILO, 2000).

The surveys administer questions to identify whether the employed persons are engaged in the formal or the informal employment sector. The questions were not asked to the whole employed population but rather to those persons whose employment status were employer, employee of private organization, self employed, unpaid family workers and apprentice, were asked whether the business /enterprise they were engaged in has;

- a) book of account that show the monthly income statement and balance sheet; or
- b) business/enterprise license.

Based on the response to these two questions, classification on employment sector of the employed population was categorized as: formal, informal or "not-identified". Employed persons who satisfy at least one of the above conditions were considered as working in the formal sector. For those who respond "no" for all of the two questions, the activity of the employed person was taken as informal. For those who do not know the situation about their main activity/business/enterprise with respect to the above questions was labeled as "not-identified".

Based on recommendations given in the 15th ICLS, those employed persons who were engaged in subsistence farming and working in private households were exempted from the analysis of the formal and informal employment sectors of the economic activity. Besides, those employed persons whose employment status were classified as employee-government, employee-government parasitatal, employee-non-governmental organization, or members of producers' cooperative were not asked such questions because they were considered as formal and later classified as being working in the formal sector.

Table 5.30 and Table 5.31 present the distribution of employed population of urban areas according to the employment sector classified by age group and sex at country and regional levels, respectively. Similar information for selected major town is shown in Table 5.32. Table 5.33-5.38 presents the cross classification of the informal sector by sex and region with some of the related variables, such as major occupation and employment sector, mean number of hours of worked, educational attainment, size of workers and type of working place. The distribution of informal sector by type of difficulties they faced during their work at national level is also given in Table 5.39.

					Employment to					
	Total Population			Total Er	Population Ratio					
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	10,397,080	4,866,065	5,531,015	5,139,831	2,928,953	2,210,879	49.4	60.2	40.0	
10-14	1,513,511	713,106	800,405	144,811	71,614	73,197	9.6	10.0	9.1	
15-19	1,735,484	751,452	984,032	429,517	192,665	236,852	24.7	25.6	24.1	
20-24	1,556,477	679,531	876,947	797,478	404,296	393,182	51.2	59.5	44.8	
25-29	1,470,711	699,615	771,096	1,009,623	579,802	429,821	68.6	82.9	55.7	
30-34	957,508	498,411	459,097	719,297	439,566	279,731	75.1	88.2	60.9	
35-39	850,679	401,785	448,894	656,750	363,332	293,418	77.2	90.4	65.4	
40-44	542,611	298,454	244,158	409,957	269,401	140,556	75.6	90.3	57.6	
45-49	440,017	215,355	224,662	326,320	196,881	129,439	74.2	91.4	57.6	
50-54	347,848	161,285	186,562	232,556	138,821	93,735	66.9	86.1	50.2	
55-59	245,521	113,124	132,397	152,901	92,826	60,074	62.3	82.1	45.4	
60-64	234,669	104,252	130,417	108,220	72,874	35,346	46.1	69.9	27.1	
65+	502,044	229,695	272,349	152,402	106,875	45,527	30.4	46.5	16.7	
65+	502,044	229,695	272,349	152,402	106,875	45,527	30.4	46.5	16	

Table 5.1Urban Population Aged Ten Years and Above by Age Group, Sex and
Employment to Population Ratio - COUNTRY TOTAL: 2011

Table 5.1.1	Urban Population Aged Ten Years and Above by Age Group, Sex and
	Employment to Population Ratio - TIGRAY REGION: 2011

	Ŧ,			-	1.4	Employment to			
	l ota	al Populatio	n	Total En	Population Ratio				
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	713,842	314,330	399,512	321,569	173,501	148,068	45.0	55.2	37.1
10-14	112,368	56,401	55,967	3,594	1,955	1,640	3.2	3.5	2.9
15-19	122,865	52,721	70,144	23,755	10,722	13,033	19.3	20.3	18.6
20-24	103,393	37,711	65,682	51,651	21,960	29,691	50.0	58.2	45.2
25-29	87,570	38,755	48,814	60,085	31,638	28,447	68.6	81.6	58.3
30-34	62,212	28,797	33,415	46,661	26,896	19,765	75.0	93.4	59.1
35-39	55,273	24,375	30,898	42,296	22,156	20,141	76.5	90.9	65.2
40-44	37,346	18,607	18,739	28,020	16,263	11,757	75.0	87.4	62.7
45-49	26,585	12,903	13,682	19,200	12,116	7,084	72.2	93.9	51.8
50-54	24,416	10,241	14,174	14,644	8,295	6,349	60.0	81.0	44.8
55-59	20,522	9,250	11,272	12,487	8,180	4,307	60.8	88.4	38.2
60-64	18,321	7,778	10,543	9,221	6,178	3,042	50.3	79.4	28.9
65+	42,971	16,791	26,181	9,956	7,143	2,813	23.2	42.5	10.7

Та	ble 5.1.2	Years and Above by Age Gro Ratio - AFFAR REGION: 2011	up, S	ex	and	
			_			

	Total Population			Total Err	Employment to Population Ratio				
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	145,374	72,334	73,040	71,536	45,023	26,512	49.2	62.2	36.3
10-14	23,339	12,733	10,606	106	57	49	0.5	0.5	0.5
15-19	17,048	7,524	9,525	2,527	1,554	974	14.8	20.7	10.2
20-24	16,340	5,904	10,436	6,777	2,923	3,853	41.5	49.5	36.9
25-29	22,788	9,315	13,473	13,818	7,777	6,041	60.6	83.5	44.8
30-34	16,743	9,236	7,507	12,970	8,811	4,159	77.5	95.4	55.4
35-39	13,253	8,090	5,163	10,506	7,660	2,846	79.3	94.7	55.1
40-44	10,038	4,409	5,629	7,761	3,990	3,771	77.3	90.5	67.0
45-49	6,988	4,153	2,835	5,090	3,798	1,292	72.8	91.4	45.6
50-54	5,782	2,029	3,753	3,781	1,682	2,098	65.4	82.9	55.9
55-59	4,213	3,149	1,064	3,024	2,682	342	71.8	85.2	32.1
60-64	4,131	2,852	1,280	2,816	2,258	558	68.2	79.2	43.6
65+	4,710	2,939	1,771	2,360	1,830	530	50.1	62.3	29.9

							Em	ploymen	it to	
	Total Population			Total Employed Population			Population Ratio			
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	1,872,486	851,432	1,021,055	921,246	505,459	415,787	49.2	59.4	40.7	
10-14	271,156	129,499	141,657	31,510	15,570	15,940	11.6	12.0	11.3	
15-19	320,387	137,353	183,034	84,970	39,463	45,506	26.5	28.7	24.9	
20-24	270,185	111,890	158,295	135,617	64,520	71,098	50.2	57.7	44.9	
25-29	258,382	123,400	134,981	173,454	98,997	74,457	67.1	80.2	55.2	
30-34	146,758	72,134	74,624	109,255	64,521	44,735	74.4	89.4	59.9	
35-39	144,231	60,439	83,792	105,957	52,515	53,442	73.5	86.9	63.8	
40-44	101,489	55,606	45,882	80,832	51,619	29,213	79.6	92.8	63.7	
45-49	81,400	38,128	43,272	62,410	35,272	27,138	76.7	92.5	62.7	
50-54	62,181	30,080	32,102	42,461	26,983	15,477	68.3	89.7	48.2	
55-59	48,381	20,384	27,998	31,240	16,462	14,778	64.6	80.8	52.8	
60-64	52,457	18,953	33,504	26,036	14,119	11,917	49.6	74.5	35.6	
65+	115,479	53,565	61,914	37,505	25,419	12,086	32.5	47.5	19.5	

Table 5.1.3	Urban Population Aged Ten Years and Above by Age Group, Sex and
	Employment to Population Ratio - AMHARA REGION: 2011

Table 5.1.4Urban Population Aged Ten Years and Above by Age Group, Sex and
Employment to Population Ratio - OROMIA REGION: 2011

	Total Population		Total Employed Population			Employment to Population Ratio			
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	3,063,686	1,463,569	1,600,116	1,609,473	930,772	678,701	52.5	63.6	42.4
10-14	471,532	218,748	252,784	51,677	25,757	25,920	11.0	11.8	10.3
15-19	517,642	231,607	286,035	138,518	64,716	73,802	26.8	27.9	25.8
20-24	480,238	214,739	265,499	269,442	145,253	124,189	56.1	67.6	46.8
25-29	437,867	218,628	219,239	315,407	189,872	125,535	72.0	86.8	57.3
30-34	293,600	157,134	136,466	227,235	140,988	86,247	77.4	89.7	63.2
35-39	242,764	119,082	123,682	204,030	113,431	90,599	84.0	95.3	73.3
40-44	149,371	86,072	63,299	122,229	81,913	40,316	81.8	95.2	63.7
45-49	126,519	65,151	61,368	101,218	61,255	39,963	80.0	94.0	65.1
50-54	86,702	36,574	50,128	58,320	31,758	26,562	67.3	86.8	53.0
55-59	70,617	33,417	37,200	47,538	27,973	19,564	67.3	83.7	52.6
60-64	57,881	25,417	32,463	28,477	19,188	9,289	49.2	75.5	28.6
65+	128,952	56,998	71,954	45,381	28,667	16,714	35.2	50.3	23.2

Table 5.1.5 Urban Population Aged Ten Years and Above by Age Group, Sex and Employment to Population Ratio - SOMALI REGION: 2011

							Employment to		
	Tota	al Populatio	n	Total Employed Population			Population Ratio		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	431,705	219,499	212,205	167,435	98,864	68,571	38.8	45.0	32.3
10-14	89,658	48,608	41,050	7,268	3,501	3,767	8.1	7.2	9.2
15-19	72,294	37,119	35,175	10,173	4,655	5,518	14.1	12.5	15.7
20-24	55,276	25,671	29,605	18,958	9,426	9,532	34.3	36.7	32.2
25-29	48,163	21,638	26,525	26,124	14,009	12,115	54.2	64.7	45.7
30-34	39,467	19,372	20,095	26,927	16,630	10,298	68.2	85.8	51.2
35-39	31,358	14,072	17,285	22,971	12,345	10,626	73.3	87.7	61.5
40-44	29,435	15,543	13,893	17,157	11,738	5,419	58.3	75.5	39.0
45-49	15,929	8,347	7,582	10,720	7,289	3,432	67.3	87.3	45.3
50-54	16,747	8,511	8,236	10,647	6,433	4,214	63.6	75.6	51.2
55-59	7,659	4,645	3,014	4,486	2,996	1,490	58.6	64.5	49.4
60-64	8,716	4,372	4,344	3,141	2,543	599	36.0	58.2	13.8
65+	17,003	11,602	5,401	8,862	7,299	1,563	52.1	62.9	28.9

					Employment to				
	Total Population			Total Employed Population			Population Ratio		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	95,879	45,321	50,558	53,221	30,493	22,729	55.5	67.3	45.0
10-14	15,898	7,452	8,446	1,386	662	724	8.7	8.9	8.6
15-19	16,846	6,282	10,564	3,929	1,715	2,214	23.3	27.3	21.0
20-24	13,352	5,490	7,863	8,196	3,613	4,583	61.4	65.8	58.3
25-29	13,229	6,311	6,918	10,598	5,978	4,620	80.1	94.7	66.8
30-34	8,685	4,382	4,304	7,540	4,107	3,433	86.8	93.7	79.8
35-39	9,264	5,112	4,152	8,307	4,999	3,309	89.7	97.8	79.7
40-44	4,341	2,472	1,868	3,777	2,472	1,304	87.0	100.0	69.8
45-49	3,887	1,976	1,911	2,873	1,769	1,104	73.9	89.6	57.7
50-54	3,888	1,959	1,928	2,969	1,959	1,010	76.4	100.0	52.4
55-59	1,856	1,065	791	1,195	1,003	192	64.4	94.2	24.2
60-64	1,510	733	777	874	733	141	57.9	100.0	18.1
65+	3,124	2,087	1,037	1,578	1,481	97	50.5	71.0	9.3

Table 5.1.6	Urban Population Aged Ten Years and Above by Age Group, Sex and
	Employment to Population Ratio - BENISHANGUL GUMUZ REGION: 2011

 Table 5.1.7
 Urban Population Aged Ten Years and Above by Age Group, Sex and Employment to Population Ratio - SNNP REGION: 2011

	.			TALE	Employment to				
	l Ota	al Populatio	n	Total En	Population Ratio				
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	1,240,922	602,413	638,508	680,560	387,610	292,950	54.8	64.3	45.9
10-14	218,794	108,585	110,210	33,356	19,799	13,558	15.2	18.2	12.3
15-19	227,984	101,769	126,215	68,880	33,936	34,944	30.2	33.3	27.7
20-24	178,026	81,545	96,481	100,499	51,610	48,889	56.5	63.3	50.7
25-29	167,408	78,076	89,332	128,085	70,736	57,349	76.5	90.6	64.2
30-34	122,780	64,878	57,901	101,579	60,530	41,049	82.7	93.3	70.9
35-39	107,982	52,822	55,161	89,417	50,467	38,950	82.8	95.5	70.6
40-44	64,209	37,428	26,782	55,431	36,120	19,311	86.3	96.5	72.1
45-49	47,339	24,790	22,549	37,581	23,492	14,089	79.4	94.8	62.5
50-54	33,477	17,097	16,379	26,464	16,122	10,343	79.1	94.3	63.1
55-59	21,843	10,590	11,253	15,482	8,873	6,609	70.9	83.8	58.7
60-64	15,206	9,334	5,872	10,223	7,463	2,760	67.2	80.0	47.0
65+	35,874	15,499	20,375	13,562	8,462	5,100	37.8	54.6	25.0

Table 5.1.8 Urban Population Aged Ten Years and Above by Age Group, Sex and Employment to Population Ratio - GAMBELLA REGION: 2011

									Employment to		
	Tota	al Populatio	n	Total Err	oulation	Population Ratio					
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	66,426	31,685	34,740	33,342	16,628	16,715	50.2	52.5	48.1		
10-14	13,753	6,492	7,261	769	167	601	5.6	2.6	8.3		
15-19	11,233	5,446	5,788	2,891	1,073	1,817	25.7	19.7	31.4		
20-24	10,037	4,552	5,485	5,665	2,641	3,023	56.4	58.0	55.1		
25-29	11,213	4,245	6,967	8,424	3,176	5,248	75.1	74.8	75.3		
30-34	6,211	3,363	2,848	5,043	3,051	1,993	81.2	90.7	70.0		
35-39	5,496	3,592	1,904	4,891	3,457	1,434	89.0	96.3	75.3		
40-44	2,587	1,395	1,192	2,240	1,298	942	86.6	93.1	79.0		
45-49	2,264	865	1,399	1,751	643	1,108	77.3	74.4	79.2		
50-54	1,075	507	567	598	438	161	55.7	86.2	28.3		
55-59	862	343	519	323	237	86	37.5	69.0	16.6		
60-64	895	443	452	563	262	301	62.9	59.2	66.6		
65+	800	443	357	184	184	-	23.0	41.5	-		

							Employment to		
	Tota	al Populatic	n	Total En	nployed Pop	oulation	Population Ratio		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	82,347	39,483	42,864	48,450	25,806	22,644	58.8	65.4	52.8
10-14	11,296	5,861	5,435	1,275	677	599	11.3	11.5	11.0
15-19	10,554	5,280	5,273	3,285	1,560	1,725	31.1	29.5	32.7
20-24	9,783	4,110	5,672	6,051	2,754	3,297	61.9	67.0	58.1
25-29	12,211	5,896	6,315	9,500	5,123	4,376	77.8	86.9	69.3
30-34	8,450	3,744	4,705	6,764	3,439	3,324	80.0	91.8	70.7
35-39	8,073	4,381	3,692	6,856	4,159	2,697	84.9	94.9	73.0
40-44	4,430	2,219	2,211	3,505	2,044	1,462	79.1	92.1	66.1
45-49	4,276	2,296	1,979	3,447	2,065	1,382	80.6	89.9	69.8
50-54	3,409	1,311	2,099	2,897	1,311	1,587	85.0	100.0	75.6
55-59	2,933	1,323	1,610	2,329	1,196	1,133	79.4	90.4	70.4
60-64	2,441	1,153	1,288	1,343	660	682	55.0	57.2	53.0
65+	4,493	1,908	2,585	1,199	818	381	26.7	42.9	14.7

Table 5.1.9Urban Population Aged Ten Years and Above by Age Group, Sex and
Employment to Population Ratio - HARARI REGION: 2011

Table 5.1.10Urban Population Aged Ten Years and Above by Age Group, Sex and
Employment to Population Ratio - ADDIS ABABA CITY ADMINISTRATION:
2011

	2011								
	Tot	al Populatio	on	Total En	Employment to Population Ratio				
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	2,504,853	1,141,811	1,363,042	1,148,974	667,973	481,001	45.9	58.5	35.3
10-14	258,791	105,438	153,353	11,315	2,372	8,943	4.4	2.2	5.8
15-19	389,937	153,156	236,781	84,047	30,378	53,669	21.6	19.8	22.7
20-24	393,031	177,419	215,612	181,644	93,626	88,018	46.2	52.8	40.8
25-29	387,359	180,563	206,796	248,032	142,071	105,961	64.0	78.7	51.2
30-34	239,171	128,616	110,555	165,921	104,895	61,025	69.4	81.6	55.2
35-39	216,443	102,335	114,108	150,185	85,898	64,288	69.4	83.9	56.3
40-44	128,656	69,369	59,287	81,553	58,003	23,550	63.4	83.6	39.7
45-49	115,370	51,776	63,594	74,669	44,742	29,926	64.7	86.4	47.1
50-54	104,177	50,208	53,969	65,727	41,529	24,197	63.1	82.7	44.8
55-59	61,945	26,308	35,637	32,026	21,243	10,784	51.7	80.7	30.3
60-64	69,415	31,775	37,640	23,960	18,868	5,091	34.5	59.4	13.5
65+	140,558	64,848	75,711	29,896	24,347	5,549	21.3	37.5	7.3

 Table 5.1.11
 Urban Population Aged Ten Years and Above by Age Group, Sex and Employment to Population Ratio - DIRE DAWA ADMINISTRATION: 2011

							Employment to		
	Tota	al Populatio	n	Total Employed Population			Population Ratio		
Age Group	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	179,561	84,187	95,374	84,025	46,825	37,200	46.8	55.6	39.0
10-14	26,926	13,291	13,636	2,554	1,096	1,457	9.5	8.2	10.7
15-19	28,694	13,195	15,499	6,543	2,892	3,651	22.8	21.9	23.6
20-24	26,815	10,500	16,315	12,978	5,969	7,009	48.4	56.8	43.0
25-29	24,522	12,786	11,736	16,097	10,424	5,673	65.6	81.5	48.3
30-34	13,431	6,754	6,678	9,404	5,700	3,704	70.0	84.4	55.5
35-39	16,542	7,485	9,057	11,333	6,244	5,089	68.5	83.4	56.2
40-44	10,709	5,334	5,375	7,452	3,941	3,511	69.6	73.9	65.3
45-49	9,461	4,971	4,490	7,360	4,440	2,921	77.8	89.3	65.0
50-54	5,994	2,768	3,227	4,048	2,311	1,736	67.5	83.5	53.8
55-59	4,691	2,650	2,041	2,771	1,982	789	59.1	74.8	38.7
60-64	3,696	1,440	2,257	1,566	600	966	42.4	41.7	42.8
65+	8,079	3,015	5,064	1,920	1,224	696	23.8	40.6	13.7

Age Group	Total Population			Total Em	ployed Pop	ulation	Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	4,179,787	1,921,162	2,258,624	1,965,425	1,134,199	831,226	47.0	59.0	36.8	
10-14	490,525	212,125	278,401	31,787	11,539	20,248	6.5	5.4	7.3	
15-19	666,828	270,238	396,590	155,181	59,089	96,092	23.3	21.9	24.2	
20-24	674,257	297,917	376,340	321,615	162,994	158,622	47.7	54.7	42.1	
25-29	636,369	301,499	334,869	416,793	240,556	176,237	65.5	79.8	52.6	
30-34	390,952	208,609	182,343	278,238	175,663	102,575	71.2	84.2	56.3	
35-39	346,927	167,455	179,472	249,818	145,396	104,422	72.0	86.8	58.2	
40-44	213,757	114,686	99,071	145,339	98,787	46,551	68.0	86.1	47.0	
45-49	187,831	88,679	99,152	126,554	78,088	48,466	67.4	88.1	48.9	
50-54	155,147	73,205	81,942	98,591	61,085	37,506	63.5	83.4	45.8	
55-59	101,241	43,464	57,777	54,410	34,910	19,500	53.7	80.3	33.8	
60-64	103,549	47,151	56,398	38,883	28,526	10,357	37.6	60.5	18.4	
65+	212,403	96,134	116,270	48,216	37,565	10,650	22.7	39.1	9.2	

Table 5.2Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - TOTAL MAJOR TOWNS: 2011

Table 5.2.1Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - MEKELE TOWN: 2011

Age Group	Total Population			Total Em	Total Employed Population			Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	190,825	85,504	105,321	87,546	49,393	38,153	45.9	57.8	36.2		
10-14	24,365	12,372	11,993	798	609	190	3.3	4.9	1.6		
15-19	33,798	13,389	20,409	8,159	3,150	5,009	24.1	23.5	24.5		
20-24	31,517	12,671	18,846	14,071	6,079	7,992	44.6	48.0	42.4		
25-29	26,782	12,091	14,691	17,718	9,815	7,903	66.2	81.2	53.8		
30-34	17,172	8,610	8,563	12,877	7,821	5,056	75.0	90.8	59.0		
35-39	16,062	7,626	8,436	11,715	7,074	4,640	72.9	92.8	55.0		
40-44	9,996	6,026	3,970	8,224	5,631	2,593	82.3	93.4	65.3		
45-49	8,081	3,704	4,377	5,538	3,534	2,004	68.5	95.4	45.8		
50-54	6,284	2,088	4,195	3,079	1,890	1,189	49.0	90.5	28.3		
55-59	4,696	1,966	2,730	2,067	1,337	730	44.0	68.0	26.8		
60-64	4,155	1,569	2,587	1,772	1,188	584	42.6	75.7	22.6		
65+	7,916	3,392	4,524	1,529	1,266	263	19.3	37.3	5.8		

Table 5.2.2	Urban Population of Major Towns Aged Ten Years and Above by Age Group,
	Sex and Employment to Population Ratio - ASAYITA TOWN: 2011

Age Group	Tot	Total Population			Total Employed Population				Employment to Population Ratio		
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	11,591	5,525	6,067	6,472	3,810	2,663	55.8	69.0	43.9		
10-14	1,332	595	737	49	-	49	3.7	-	6.6		
15-19	1,651	694	957	413	146	267	25.0	21.0	27.9		
20-24	1,551	610	941	743	374	369	47.9	61.3	39.3		
25-29	1,951	898	1,052	1,409	772	637	72.2	85.9	60.5		
30-34	1,261	701	560	1,025	691	334	81.3	98.6	59.7		
35-39	1,152	649	503	911	637	274	79.1	98.2	54.5		
40-44	746	331	415	612	331	281	82.1	100.0	67.8		
45-49	610	242	368	437	231	206	71.6	95.5	55.9		
50-54	382	189	192	257	158	98	67.3	83.8	51.1		
55-59	314	160	153	243	160	82	77.4	100.0	53.8		
60-64	186	134	52	166	124	42	89.2	92.6	80.4		
65+	457	321	136	208	186	22	45.4	57.9	16.1		

Age Group	Total Population			Total Em	nployed Pop	Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	177,487	78,724	98,764	82,743	45,705	37,038	46.6	58.1	37.5
10-14	28,556	12,717	15,839	7,401	3,340	4,061	25.9	26.3	25.6
15-19	30,009	13,540	16,469	9,958	4,699	5,258	33.2	34.7	31.9
20-24	29,374	13,049	16,325	13,727	7,611	6,116	46.7	58.3	37.5
25-29	23,501	11,149	12,352	13,863	8,384	5,479	59.0	75.2	44.4
30-34	12,435	5,810	6,624	8,580	4,721	3,859	69.0	81.3	58.3
35-39	11,972	4,556	7,416	7,515	4,125	3,390	62.8	90.6	45.7
40-44	8,141	3,965	4,176	5,720	3,338	2,382	70.3	84.2	57.0
45-49	8,860	3,957	4,903	6,138	3,528	2,610	69.3	89.2	53.2
50-54	5,456	2,441	3,016	3,047	1,905	1,142	55.8	78.1	37.9
55-59	4,126	924	3,202	2,053	770	1,282	49.8	83.4	40.1
60-64	4,987	2,112	2,875	1,987	1,301	686	39.8	61.6	23.9
65+	10,072	4,505	5,567	2,755	1,983	772	27.4	44.0	13.9

Table 5.2.3Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - GONDER TOWN: 2011

Table 5.2.4Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - DESSIE TOWN: 2011

Age Group	Total Population			Total Em	ployed Pop	ulation	Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	105,306	49,121	56,185	46,017	25,986	20,032	43.7	52.9	35.7	
10-14	13,789	6,838	6,951	342	100	242	2.5	1.5	3.5	
15-19	18,767	8,884	9,882	3,679	1,686	1,993	19.6	19.0	20.2	
20-24	14,844	6,757	8,087	6,515	3,544	2,971	43.9	52.5	36.7	
25-29	14,118	6,161	7,957	8,676	4,335	4,341	61.5	70.4	54.6	
30-34	8,318	3,397	4,921	5,701	2,981	2,720	68.5	87.8	55.3	
35-39	8,151	4,213	3,937	5,991	3,722	2,269	73.5	88.3	57.6	
40-44	5,834	2,897	2,937	4,103	2,737	1,366	70.3	94.5	46.5	
45-49	4,553	2,313	2,240	3,303	2,149	1,154	72.5	92.9	51.5	
50-54	3,927	2,073	1,854	2,770	1,751	1,019	70.5	84.5	54.9	
55-59	3,040	1,171	1,869	1,891	999	891	62.2	85.3	47.7	
60-64	2,805	1,605	1,200	1,486	968	518	53.0	60.3	43.2	
65+	7,161	2,811	4,350	1,561	1,014	547	21.8	36.1	12.6	

	oup,
Sex and Employment to Population Ratio - BAHIR DAR TOWN: 2011	

Age Group	Total Population			Total Em	Total Employed Population			Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	161,491	69,618	91,873	89,993	46,674	43,319	55.7	67.0	47.2		
10-14	18,196	6,033	12,163	3,215	786	2,430	17.7	13.0	20.0		
15-19	25,653	9,329	16,324	9,738	3,167	6,571	38.0	33.9	40.3		
20-24	33,144	12,514	20,630	17,862	7,471	10,391	53.9	59.7	50.4		
25-29	27,932	13,425	14,507	20,148	10,807	9,342	72.1	80.5	64.4		
30-34	17,203	8,946	8,256	13,040	8,088	4,952	75.8	90.4	60.0		
35-39	10,756	5,865	4,892	9,070	5,485	3,584	84.3	93.5	73.3		
40-44	7,187	4,205	2,982	5,901	3,991	1,910	82.1	94.9	64.0		
45-49	4,594	2,625	1,969	3,508	2,566	942	76.4	97.7	47.8		
50-54	3,878	1,236	2,643	2,791	1,236	1,555	72.0	100.0	58.9		
55-59	4,115	1,320	2,795	1,741	946	795	42.3	71.7	28.4		
60-64	3,120	1,257	1,863	1,219	691	528	39.1	54.9	28.3		
65+	5,712	2,864	2,848	1,761	1,442	319	30.8	50.3	11.2		

Age Group	Tot	Total Population			Total Employed Population			Employment to Population Ratio		
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	80,840	37,554	43,286	40,673	22,604	18,069	50.3	60.2	41.7	
10-14	8,469	4,573	3,895	266	172	94	3.1	3.8	2.4	
15-19	13,405	4,951	8,454	2,613	832	1,781	19.5	16.8	21.1	
20-24	12,847	5,767	7,081	7,441	3,630	3,811	57.9	63.0	53.8	
25-29	11,083	4,942	6,140	7,564	3,992	3,572	68.2	80.8	58.2	
30-34	8,452	4,485	3,967	6,454	3,801	2,653	76.4	84.8	66.9	
35-39	5,978	3,022	2,957	4,662	2,903	1,758	78.0	96.1	59.5	
40-44	3,773	1,770	2,003	2,793	1,555	1,238	74.0	87.8	61.8	
45-49	4,341	1,934	2,407	3,095	1,759	1,336	71.3	90.9	55.5	
50-54	3,354	1,790	1,564	2,420	1,624	796	72.2	90.8	50.9	
55-59	2,104	992	1,113	1,240	739	501	58.9	74.5	45.0	
60-64	1,774	975	799	898	740	158	50.6	75.8	19.8	
_65+	5,259	2,353	2,907	1,229	857	372	23.4	36.4	12.8	

Table 5.2.6Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - BISHOFTU TOWN: 2011

Table 5.2.7	Urban Population of Major Towns Aged Ten Years and Above by Age Group,
	Sex and Employment to Population Ratio - ADAMA TOWN: 2011

Age Group	Tot	Total Population			Total Employed Population			Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	202,760	93,992	108,768	101,178	60,623	40,555	49.9	64.5	37.3		
10-14	24,900	9,218	15,683	542	270	271	2.2	2.9	1.7		
15-19	33,757	13,229	20,529	8,408	2,890	5,518	24.9	21.8	26.9		
20-24	37,163	15,806	21,357	19,920	10,081	9,839	53.6	63.8	46.1		
25-29	31,768	16,213	15,555	22,163	13,808	8,356	69.8	85.2	53.7		
30-34	19,210	11,701	7,509	14,591	10,403	4,188	76.0	88.9	55.8		
35-39	16,610	8,946	7,664	13,754	7,988	5,766	82.8	89.3	75.2		
40-44	10,491	5,247	5,244	7,441	5,058	2,383	70.9	96.4	45.4		
45-49	7,245	4,277	2,968	5,050	3,720	1,330	69.7	87.0	44.8		
50-54	5,373	2,831	2,542	3,209	2,202	1,007	59.7	77.8	39.6		
55-59	4,757	2,373	2,384	3,030	2,087	943	63.7	87.9	39.6		
60-64	3,409	1,195	2,214	884	614	271	25.9	51.3	12.2		
65+	8,076	2,956	5,119	2,185	1,503	681	27.1	50.8	13.3		

Table 5.2.8	Urban Population of Major Towns Aged Ten Years and Above by Age Group,
	Sex and Employment to Population Ratio - JIMMA TOWN: 2011

Age Group	To	Total Population			Total Employed Population			Employment to Population Ratio		
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	106,713	50,940	55,774	55,260	33,297	21,963	51.8	65.4	39.4	
10-14	15,096	6,765	8,331	959	243	716	6.4	3.6	8.6	
15-19	16,859	5,773	11,086	4,855	1,910	2,944	28.8	33.1	26.6	
20-24	19,293	9,153	10,141	10,172	5,732	4,440	52.7	62.6	43.8	
25-29	18,195	9,806	8,389	13,195	8,636	4,559	72.5	88.1	54.3	
30-34	8,933	5,125	3,808	7,162	4,744	2,418	80.2	92.6	63.5	
35-39	7,870	4,177	3,693	6,662	3,985	2,676	84.6	95.4	72.5	
40-44	4,870	2,567	2,304	3,916	2,376	1,540	80.4	92.6	66.8	
45-49	3,740	1,770	1,970	2,427	1,605	822	64.9	90.7	41.7	
50-54	3,291	1,639	1,652	2,361	1,501	860	71.7	91.6	52.1	
55-59	2,748	1,437	1,312	1,680	1,199	481	61.1	83.5	36.7	
60-64	2,223	884	1,338	808	703	106	36.4	79.5	7.9	
65+	3,594	1,844	1,750	1,063	663	400	29.6	35.9	22.9	

Age Group	Total Population			Total Employed Population			Employment to Population Ratio		
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	114,732	56,346	58,386	54,907	32,752	22,155	47.9	58.1	37.9
10-14	17,992	8,706	9,285	894	692	202	5.0	7.9	2.2
15-19	21,816	10,415	11,402	5,809	2,575	3,234	26.6	24.7	28.4
20-24	20,338	9,293	11,045	10,206	6,070	4,137	50.2	65.3	37.5
25-29	15,603	8,912	6,691	11,381	7,333	4,048	72.9	82.3	60.5
30-34	10,117	5,555	4,562	7,954	5,131	2,823	78.6	92.4	61.9
35-39	7,259	3,334	3,926	5,863	3,152	2,711	80.8	94.6	69.1
40-44	4,954	2,220	2,733	3,180	1,933	1,247	64.2	87.1	45.6
45-49	5,102	2,395	2,707	3,619	2,054	1,565	70.9	85.8	57.8
50-54	2,987	1,279	1,708	1,798	1,032	766	60.2	80.6	44.8
55-59	2,122	1,148	974	1,530	969	561	72.1	84.4	57.6
60-64	1,829	1,055	774	1,187	857	330	64.9	81.2	42.7
65+	4,612	2,034	2,579	1,484	953	531	32.2	46.9	20.6

Table 5.2.9Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - SHASHEMENE TOWN: 2011

Table 5.2.10Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - JIJIGA TOWN: 2011

Age Group	То	Total Population			Total Employed Population			Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	80,439	40,616	39,822	29,089	20,272	8,817	36.2	49.9	22.1		
10-14	12,843	6,744	6,098	403	221	182	3.1	3.3	3.0		
15-19	11,309	5,066	6,244	990	510	481	8.8	10.1	7.7		
20-24	12,402	5,666	6,736	3,734	2,012	1,722	30.1	35.5	25.6		
25-29	12,776	6,084	6,693	6,924	4,540	2,384	54.2	74.6	35.6		
30-34	8,924	4,732	4,192	4,733	3,950	783	53.0	83.5	18.7		
35-39	6,296	3,280	3,016	3,758	2,674	1,083	59.7	81.5	35.9		
40-44	4,822	3,324	1,498	3,607	2,870	737	74.8	86.3	49.2		
45-49	3,944	2,179	1,765	2,487	1,822	665	63.1	83.6	37.7		
50-54	2,395	1,230	1,165	1,067	601	466	44.6	48.8	40.0		
55-59	1,086	624	462	486	384	102	44.7	61.6	22.1		
60-64	1,238	567	671	478	341	137	38.6	60.1	20.5		
65+	2,404	1,121	1,283	422	348	75	17.6	31.0	5.8		

Table 5.2.11	Urban Population of Major Towns Aged Ten Years and Above by Age Group,
	Sex and Employment to Population Ratio - ASOSA TOWN: 2011

Age Group	Tot	Total Population			Total Employed Population			Employment to Population Ratio		
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	24,739	11,848	12,891	13,319	7,830	5,489	53.8	66.1	42.6	
10-14	3,545	1,764	1,781	119	61	58	3.4	3.4	3.3	
15-19	4,161	1,501	2,660	837	222	616	20.1	14.8	23.1	
20-24	4,240	1,594	2,646	2,405	1,040	1,365	56.7	65.2	51.6	
25-29	3,650	1,685	1,965	2,745	1,536	1,209	75.2	91.2	61.5	
30-34	2,724	1,511	1,212	2,261	1,399	862	83.0	92.5	71.1	
35-39	2,364	1,424	940	2,035	1,424	612	86.1	100.0	65.1	
40-44	1,103	678	425	981	678	303	88.9	100.0	71.3	
45-49	1,203	695	508	876	627	249	72.8	90.2	49.1	
50-54	458	291	167	411	291	119	89.6	100.0	71.5	
55-59	471	193	278	247	193	54	52.4	100.0	19.4	
60-64	258	175	83	216	175	40	83.6	100.0	48.9	
65+	561	336	225	186	186	-	33.2	55.3	-	

Age Group	Tota	Total Population			Total Employed Population			Employment to Population Ratio		
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	127,209	61,125	66,085	64,568	37,212	27,356	50.8	60.9	41.4	
10-14	19,210	8,729	10,481	1,534	859	676	8.0	9.8	6.4	
15-19	21,942	9,649	12,293	5,484	2,339	3,145	25.0	24.2	25.6	
20-24	23,229	10,789	12,440	12,236	6,043	6,193	52.7	56.0	49.8	
25-29	20,240	8,681	11,559	14,532	7,473	7,059	71.8	86.1	61.1	
30-34	12,356	7,288	5,069	9,674	6,490	3,185	78.3	89.1	62.8	
35-39	8,850	4,369	4,481	7,342	4,229	3,113	83.0	96.8	69.5	
40-44	6,709	3,844	2,864	5,237	3,674	1,563	78.1	95.6	54.6	
45-49	5,436	3,000	2,436	3,921	2,803	1,118	72.1	93.4	45.9	
50-54	3,155	1,577	1,578	2,350	1,520	831	74.5	96.4	52.7	
55-59	1,609	620	990	873	557	315	54.2	89.9	31.9	
60-64	1,473	966	507	658	540	117	44.6	55.9	23.2	
65+	3,001	1,613	1,387	726	684	42	24.2	42.4	3.0	

Table 5.2.12Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - HAWASSA TOWN: 2011

Table 5.2.13Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio - GAMBELLA TOWN: 2011

Age Group	To	Total Population			Total Employed Population			Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	28,893	14,770	14,124	12,209	7,437	4,772	42.3	50.4	33.8		
10-14	5,221	2,481	2,740	121	42	78	2.3	1.7	2.9		
15-19	4,515	2,186	2,329	364	133	231	8.1	6.1	9.9		
20-24	4,685	2,219	2,466	1,910	958	953	40.8	43.1	38.6		
25-29	4,678	2,207	2,471	2,846	1,508	1,338	60.8	68.3	54.2		
30-34	2,796	1,634	1,162	2,099	1,410	688	75.1	86.3	59.2		
35-39	2,549	1,794	754	2,165	1,694	471	84.9	94.4	62.4		
40-44	1,337	690	647	1,114	629	485	83.3	91.2	74.9		
45-49	1,014	545	469	680	443	237	67.0	81.2	50.5		
50-54	626	255	371	358	222	136	57.2	87.0	36.7		
55-59	483	256	228	204	149	55	42.2	58.4	24.1		
60-64	540	288	252	258	157	101	47.7	54.4	40.0		
65+	448	213	235	91	91	-	20.4	42.8	-		

Table 5.2.14	Urban Population of Major Towns Aged Ten Years and Above by Age Group,
	Sex and Employment to Population Ratio - HARAR TOWN: 2011

Age Group	To	tal Populatio	on	Total Em	ployed Pop	ulation	Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	82,347	39,483	42,864	48,450	25,806	22,644	58.8	65.4	52.8	
10-14	11,296	5,861	5,435	1,275	677	599	11.3	11.5	11.0	
15-19	10,554	5,280	5,273	3,285	1,560	1,725	31.1	29.5	32.7	
20-24	9,783	4,110	5,672	6,051	2,754	3,297	61.9	67.0	58.1	
25-29	12,211	5,896	6,315	9,500	5,123	4,376	77.8	86.9	69.3	
30-34	8,450	3,744	4,705	6,764	3,439	3,324	80.0	91.8	70.7	
35-39	8,073	4,381	3,692	6,856	4,159	2,697	84.9	94.9	73.0	
40-44	4,430	2,219	2,211	3,505	2,044	1,462	79.1	92.1	66.1	
45-49	4,276	2,296	1,979	3,447	2,065	1,382	80.6	89.9	69.8	
50-54	3,409	1,311	2,099	2,897	1,311	1,587	85.0	100.0	75.6	
55-59	2,933	1,323	1,610	2,329	1,196	1,133	79.4	90.4	70.4	
60-64	2,441	1,153	1,288	1,343	660	682	55.0	57.2	53.0	
65+	4,493	1,908	2,585	1,199	818	381	26.7	42.9	14.7	

Age Group	Тс	otal Populatio	on	Total Em	ployed Pop	ulation	Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	2,504,853	1,141,811	1,363,042	1,148,974	667,973	481,001	45.9	58.5	35.3	
10-14	258,791	105,438	153,353	11,315	2,372	8,943	4.4	2.2	5.8	
15-19	389,937	153,156	236,781	84,047	30,378	53,669	21.6	19.8	22.7	
20-24	393,031	177,419	215,612	181,644	93,626	88,018	46.2	52.8	40.8	
25-29	387,359	180,563	206,796	248,032	142,071	105,961	64.0	78.7	51.2	
30-34	239,171	128,616	110,555	165,921	104,895	61,025	69.4	81.6	55.2	
35-39	216,443	102,335	114,108	150,185	85,898	64,288	69.4	83.9	56.3	
40-44	128,656	69,369	59,287	81,553	58,003	23,550	63.4	83.6	39.7	
45-49	115,370	51,776	63,594	74,669	44,742	29,926	64.7	86.4	47.1	
50-54	104,177	50,208	53,969	65,727	41,529	24,197	63.1	82.7	44.8	
55-59	61,945	26,308	35,637	32,026	21,243	10,784	51.7	80.7	30.3	
60-64	69,415	31,775	37,640	23,960	18,868	5,091	34.5	59.4	13.5	
65+	140,558	64,848	75,711	29,896	24,347	5,549	21.3	37.5	7.3	

Table 5.2.15Urban Population of Major Towns Aged Ten Years and Above by Age Group,
Sex and Employment to Population Ratio – ADDIS ABABA CITY: 2011

Table 5.2.16	Urban Population of Major T	owns Aged Ten Years and A	bove by Age Group,
	Sex and Employment to Pop	ulation Ratio - DIRE DAWA TO	OWN: 2011

	Total Population			Total Em	ployed Pop	ulation	Employment to Population Ratio			
Age Group 10	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	179,561	84,187	95,374	84,025	46,825	37,200	46.8	55.6	39.0	
10-14	26,926	13,291	13,636	2,554	1,096	1,457	9.5	8.2	10.7	
15-19	28,694	13,195	15,499	6,543	2,892	3,651	22.8	21.9	23.6	
20-24	26,815	10,500	16,315	12,978	5,969	7,009	48.4	56.8	43.0	
25-29	24,522	12,786	11,736	16,097	10,424	5,673	65.6	81.5	48.3	
30-34	13,431	6,754	6,678	9,404	5,700	3,704	70.0	84.4	55.5	
35-39	16,542	7,485	9,057	11,333	6,244	5,089	68.5	83.4	56.2	
40-44	10,709	5,334	5,375	7,452	3,941	3,511	69.6	73.9	65.3	
45-49	9,461	4,971	4,490	7,360	4,440	2,921	77.8	89.3	65.0	
50-54	5,994	2,768	3,227	4,048	2,311	1,736	67.5	83.5	53.8	
55-59	4,691	2,650	2,041	2,771	1,982	789	59.1	74.8	38.7	
60-64	3,696	1,440	2,257	1,566	600	966	42.4	41.7	42.8	
65+	8,079	3,015	5,064	1,920	1,224	696	23.8	40.6	13.7	

		oroups = (JOUNIRT IO			upational G					
						Service,	loups				
Age Group		Legislator				Service, Shop &	Skilled	Crafts	Plant		
and Sex	Total	Senior		Technical and		Market	agricultural	and	,Machine		
	Employed	Officials &		Associate		sales	and	related	Operators &	Elementary	Not
	Person	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
All Ages											
Total	5,139,831	153,155	336,011	330,286	289,991	1,275,292	394,017	958,971	268,268	1,132,631	1,209
Male	2,928,953	123,323	226,454	218,506	119,703	576,401	259,469	576,176	246,870	581,582	470
Female	2,210,879	29,832	109,557	111,781	170,288	698,891	134,548	382,795	21,398	551,049	740
10-14											
Total	144,811	-	-	-	760	29,461	27,502	11,388	813	74,887	-
Male	71,614	-	-	-	327	14,607	18,281	4,979	813	32,607	-
Female	73,197	-	-	-	433	14,854	9,221	6,409	-	42,280	-
15-19											
Total	429,517	287	2,068	6,961	10,107	114,860	35,467	66,406	8,018	185,342	-
Male	192,665	-	1,097	3,767	2,616	50,802	23,101	36,267	6,290	68,726	-
Female	236,852	287	971	3,194	7,492	64,058	12,367	30,139	1,728	116,616	-
20-24											
Total	797,478	11,926	51,367	53,748	61,207	208,343	34,631		39,378	198,823	405
Male	404,296	7,370	25,935	26,171	19,646	83,145	21,727	95,467	33,658	91,177	-
Female	393,182	4,556	25,432	27,577	41,562	125,198	12,904	42,181	5,721	107,646	405
25-29											
Total	1,009,623	28,087	84,813	85,477	70,767	253,770	46,291	187,257	65,301	187,115	745
Male	579,802	22,670	55,948	52,787	25,703	112,231	27,560	125,087	62,281	95,125	410
Female	429,821	5,417	28,865	32,690	45,064	141,539	18,731	62,171	3,020	91,990	335
30-34											
Total	719,297	22,304	51,362	51,116	44,546	184,558	45,759		50,639	127,208	59
Male	439,566	18,062	35,673	36,266	16,884	93,327	28,971	91,643	45,588	73,093	59
Female	279,731	4,242	15,689	14,850	27,662	91,230	16,788	50,104	5,051	54,115	-
35-39											
Total	656,750	29,241	51,405	50,090	37,468	170,722	,	126,644	35,945	109,860	-
Male	363,332	21,901	32,322	30,827	15,799	74,717	26,295	68,398	34,121	58,953	-
Female	293,418	7,340	19,084	19,263	21,669	96,005	19,079	58,246	1,824	50,907	-
40-44											
Total	409,957	23,061	30,679	29,807	25,134	104,391	34,866	78,915	22,545	60,561	-
Male	269,401	19,809	24,612	24,227	14,963	55,917	27,541	44,094	21,756	36,482	-
Female	140,556	3,252	6,066	5,580	10,171	48,473	7,325	34,821	789	24,079	-

 Table 5.3
 Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Major

 Occupational Groups – COUNTRY TOTAL: 2011

					Major Occ	upational G	roups				
Age Group		Legislator				Service, Shop &	Skilled	Crafts	Plant		
and Sex	Total	Senior		Technical and		Market	agricultural	and	,Machine		
	Employed	Officials &		Associate		sales	and	related	Operators &	Elementary	Not
	Person	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
45-49											
Total	326,320	15,428	27,223	16,789	17,653	76,731	28,413	65,889	19,939	58,255	-
Male	196,881	13,321	21,671	13,397	9,445	31,354	15,951	36,461	18,345	36,936	-
Female	129,439	2,107	5,552	3,392	8,209	45,376	12,461	29,428	1,594	21,319	-
50-54											
Total	232,556	11,036	20,600	14,280	13,100	45,586	24,221	47,489	10,753	45,490	-
Male	138,821	9,541	14,725	12,201	7,049	17,824	16,408	23,964	10,043	27,065	-
Female	93,735	1,495	5,874	2,079	6,051	27,762	7,813	23,526	710	18,425	-
55-59											
Total	152,901	6,212	9,346	12,280	5,944	32,559	18,455	32,084	6,458	29,563	-
Male	92,826	5,783	8,305	9,727	4,316	11,927	12,614	16,155	6,236	17,764	-
Female	60,074	429	1,041	2,553	1,628	20,631	5,841	15,929	222	11,800	-
60-64											
Total	108,220	3,123	4,477	6,583	1,748	21,128	19,167	26,618	4,385	20,990	-
Male	72,874	2,458	3,792	5,983	1,748	11,966	13,771	13,766	3,999	15,390	-
Female	35,346	665	685	601	-	9,162	5,396	12,852	385	5,600	-
65+											
Total	152,402	2,450	2,672	3,154	1,556	33,185	33,870	36,884	4,094	34,536	-
Male	106,875	2,408	2,374	3,154	1,208	18,582	27,248	19,896	3,739	28,264	-
Female	45,527	42	298	-	348	14,603	6,622	16,988	355	6,271	-

Groups: 2				N		upational G	roupe				
		Logialator	[ľ		•		1	Plant		
Region		Legislator Senior		Technical		Service, Shop &	Skilled	Crafts	,Machine		
And Sex	Total	Officials		and		Market	agricultural		Operators		
	Employed	&		Associate		sales	and	related	&	Elementary	Not
	Person		Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers		
Country Total	1	U U					, , , , , , , , , , , , , , , , , , ,				11
Total	5,139,831	153,155	336,011	330,286	289.991	1,275,292	394.017	958,971	268,268	1,132,631	1,209
Male	2,928,953	123,323	226,454		119,703	576,401		576,176	246,870	581,582	470
Female	2,210,879	29,832	109,557		170,288	698,891		382,795	21,398	551,049	740
Tigray Region	, ,		,	,	,	,	,	,	,	,	
Total	321,569	7,355	24,228	14,059	18,866	90,842	18,127	64,824	23,654	59,274	341
Male	173,501	5,875	14,600	8,854	7,559	34,533	12,061	40,982	18,380	30,317	341
Female	148,068	1,480	9,628	5,205	11,307	56,309	6,066	23,843	5,274	28,956	-
Affar Region	ŗ	·			·				·		
Total	71,536	1,556	2,172	3,403	4,832	13,613	6,976	10,308	1,802	26,875	-
Male	45,023	1,515	1,477	2,909	2,656	6,741	6,131	5,750	1,553	16,292	-
Female	26,512	41	695	494	2,175	6,872	845	4,557	249	10,583	-
Amhara Region											
Total	921,246	24,785	47,594	59,287	43,220	221,444	95,723	184,328	35,729	208,672	464
Male	505,459	20,424	34,298	34,597	16,867	91,579	71,277	97,300	33,870	105,188	59
Female	415,787	4,361	13,297	24,690	26,353	129,865	24,445	87,028	1,859	103,485	405
Oromia Region											
Total	1,609,473	49,754	89,595	100,392	88,007	407,073	147,029	324,930	78,186	324,471	35
Male	930,772	38,838	63,584	71,736	43,571	183,533	91,788	188,685	70,581	178,456	-
Female	678,701	10,917	26,011	28,656	44,435	223,540	55,241	136,245	7,605	146,015	35
Somali Region											
Total	167,435	6,194	7,971	12,028	5,043	53,260	25,292	16,817	4,096	36,735	-
Male	98,864	5,497	5,884	9,448	3,273	20,822	18,283	12,784	3,792	19,079	-
Female	68,571	697	2,086	2,580	1,770	32,437	7,008	4,033	304	17,656	-
Benishangul Gumuz											
Region											
Total	53,221	1,523	3,378	4,851	3,150	9,756	10,504	7,579	1,530	10,949	-
Male	30,493	1,523	2,757	3,090	1,499	3,365	7,271	3,542	1,510	5,935	-
Female	22,729	-	621	1,761	1,651	6,392	3,233	4,037	20	5,013	-
Snnp Region											
Total	680,560	20,010	47,890	54,385	35,818	180,295	61,238	116,016	23,342	141,266	299
Male	387,610	16,152	32,613	34,269	15,538	90,253	36,454	60,035	22,030	80,266	-
Female	292,950	3,858	15,276	20,116	20,280	90,042	24,784	55,982	1,312	61,000	299

Table 5.4 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Major Occupational Groups: 2011

	Tak	ble	5.4	Cont'd
--	-----	-----	-----	--------

				Ma	ajor Occu	upational (Groups				
		Legislator				Service,			Plant		
Region		Senior		Technical		Shop &	Skilled	Crafts	,Machine		
And Sex	Total	Officials		and		Market	agricultural	and	Operators		
	Employed	&		Associate		sales		related			
	Person	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
Gambella Region											
Total	33,342	1,130	1,571	2,188	1,772	5,480	11,247	4,539	755	4,659	-
Male	16,628	913	1,203	1,660	1,162	2,422	4,367	1,406	755	2,740	-
Female	16,715	216	369	529	611	3,058	6,880	3,133	-	1,919	-
Harari Region											
Total	48,450	1,046	2,490	3,931	3,152	15,663	650	8,363	2,189	10,966	-
Male	25,806	952	1,854	1,746	1,114	5,518	550	5,540	2,107	6,426	-
Female	22,644	94	636	2,185	2,039	10,145	101	2,823	82	4,540	-
Addis Ababa City											
Administration											
Total	1,148,974	37,488	105,769	71,444	80,810	255,957	15,180	207,245	90,454	284,627	-
Male	667,973	29,831	65,721	47,245	23,090	128,787	10,068	151,792	86,137	125,301	-
Female	481,001	7,656	40,048	24,200	57,720	127,170	5,112	55,453	4,317	159,326	-
Dire Dawa											
Administration											
Total	84,025	2,314	3,353	4,318	5,320	21,908	2,051	14,022	6,531	24,138	69
Male	46,825	1,804	2,463	2,953	3,373	8,848	1,219	8,360	6,155	11,581	69
Female	37,200	511	891	1,365	1,947	13,060	832	5,662	376	12,557	-

Groups. 20					Major Oco	cupational	Groups				
					.,	Service,					
		Legislator				Shop &	Skilled				
Region and	Total	Senior		Technical and		Market	agricultura	Crafts and	Plant ,Machine		
Sex	Employed	Officials &		Associate		sales	I and	related	Operators &	Elementary	Not
	Persons	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
COUNTRY TOTAL											
Total	1,965,425	59,951	153,698	123,409	•	445,240	54,546	362,315	144,939	488,808	200
Male	1,134,199	48,085	97,651	80,337	44,374		39,006	252,676	135,898	227,514	128
Female	831,226	11,866	56,047	43,073	87,946	236,709	15,540	109,639	9,041	261,294	71
MEKELE TOWN											
Total	87,546	2,254	6,673	5,177	7,282	21,865	637	22,246	6,668	14,744	-
Male	49,393	1,856	3,998	3,276	3,319	8,110	547	15,577	6,151	6,558	-
Female	38,153	398	2,676	1,901	3,962	13,754	90	6,669	517	8,186	-
ASAYITA TOWN											
Total	6,472	274	246	326	423	1,632	762	1,304	135	1,371	-
Male	3,810	233	193	220	200	512	658	964	124	703	-
Female	2,663	41	53	106	222	1,120	103	340	10	668	-
GONDER TOWN											
Total	82,743	1,570	2,595	4,036	3,096	12,431	15,105	13,106	3,319	27,487	-
Male	45,705	1,278	1,612	2,458	901	5,663	13,744	6,026	2,857	11,167	-
Female	37,038	292	982	1,578	2,195	6,768	1,361	7,080	461	16,320	-
DESSIE TOWN											
Total	46,017	1,086	3,394	3,176	2,466	10,885	1,426	9,833	2,048	11,668	36
Male	25,986	910	2,175	1,781	1,022	5,131	714	6,516	1,869	5,867	-
Female	20,032	176	1,218	1,395	1,444	5,754	713	3,317	179	5,800	36
BAHIR DAR TOWN											
Total	89,993	2,556	5,279	6,705	6,841	16,565	2,445	18,109	6,063	25,372	59
Male	46,674	2,282	3,245	4,074	1,918	5,512	1,574	10,987	5,787	11,236	59
Female	43,319	274	2,033	2,631	4,924	11,053	870	7,122	276	14,136	-
BISHOFTU TOWN											
Total	40,673	1,859	2,227	2,073	2,419	5,995	3,864	9,539	3,624	9,038	35
Male	22,604	1,311	1,503	1,612	746	2,171	2,087	5,226	2,907	5,041	-
Female	18,069	548	724	461	1,673	3,824	1,777	4,313	717	3,997	35
ADAMA TOWN											
Total	101,178	2,120	3,811	6,092	4,968	22,638	4,195	20,503	9,835	27,015	-
Male	60,623	1,614	2,282	4,418	1,758	11,738	2,315	14,384	8,520	13,595	-
Female	40,555	506	1,530	1,674	3,210	10,900	1,881	6,119	1,315	13,421	-

Table 5.5Currently Employed Population of Major Towns Aged Ten Years and above by Sex and Major Occupational
Groups: 2011

Table 5.5 Cont'd				
Region and Sex	Total Employed Persons	Legislator Senior Officials & Managers	Professional	Tech As Prof
JIMMA TOWN				
Total	55,260	1,244	5,406	
Male	33,297	1,103	3,990	
Female	21,963	141	1,417	
SHASHEMENE TOWN	l			
Total	E4 007	1 064	2 001	

					Major Oco		Groups				
Region and Sex	Total Employed Persons	Legislator Senior Officials & Managers	Professional	Technical and Associate Professionals	Clerks	Service, Shop & Market sales Workers	Skilled agricultura I and Fishery	Crafts and related trades	Plant ,Machine Operators & Assemblers	Elementary Occupation	Not Stated
JIMMA TOWN	•		•			•	•			•	
Total	55,260	1,244	5,406	4,179	4,052	13,376	775	10,519	2,618	13,091	
Male	33,297	1,103	3,990	2,511	1,788	6,016	725	8,130	2,425	6,610	
Female SHASHEMENE TOWI	21,963 N	141	1,417	1,668	2,264	7,360	50	2,389	194	6,481	
Total	54,907	1,064	2,001	2,060	2,080	19,009	4,843	8,005	3,796	12,048	
Male	32,752	693	1,300	1,386	1,505	7,773	3,093	5,350	3,462	8,190	
Female JIJIGA TOWN	22,155	372	701	675	574	11,236	1,751	2,656	334	3,857	
Total	29,089	464	2,266	2,231	1,261	8,030	720	4,260	1,955	7,902	
Male	20,272	464	1,659	1,874	610	4,546	598	3,578	1,955	4,988	
Female ASOSA TOWN	8,817	-	607	357	651	3,485	122	682	-	2,914	
Total	13,319	550	1,004	1,796	1,421	3,061	635	5 1,228	926	2,699	
Male	7,830	550	793	1,080	690	1,214	426	859	906	1,314	
Female HAWASSA TOWN	5,489	-	211	716	731	1,847	209	368	20	1,385	
Total	64,568	3,298	6,254	4,533	5,669	13,390	910	12,267	4,249	13,998	
Male	37,212	2,657	4,252	2,765	1,648	5,431	516	8,588	4,006	7,348	
Female GAMBELLA TOWN	27,356	642	2,001	1,768	4,021	7,959	394	3,679	243	6,650	
Total	12,209	765	929	1,332	1,060	2,835	347	1,767	528	2,645	
Male	7,437	549	610	939	692	1,560	172	799	528	1,589	
Female HARAR TOWN	4,772	216	319	394	368	1,275	175	968	-	1,056	
Total	48,450	1,046	2,490	3,931	3,152	15,663	650	8,363	2,189	10,966	
Male	25,806	952	1,854	1,746	1,114	5,518	550	5,540	2,107	6,426	
Female DIRE DAWA TOWN	22,644	94	636	2,185	2,039	10,145	101	2,823	82	4,540	
Total	84,025	2,314	3,353	4,318	5,320	21,908	2,051	14,022	6,531	24,138	
Male	46,825	1,804	2,463	2,953	3,373	8,848	1,219	,	6,155	,	
Female	37,200	511	891	1,365	1,947	13,060	832		376	12,557	

010005.2011												
Minor Occupational						Re	gion					
Group and Sex							Benishangul				Addis	Dire
•	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Total Employed Persons												
Total					1,609,473	167,435		680,560	33,342	48,450	1,148,974	84,025
Male	2,928,953	173,501	45,023	505,459	930,772	98,864	30,493	387,610	16,628	25,806	667,973	46,825
Female	2,210,879				678,701	68,571	22,729	292,950	16,715	22,644	481,001	37,200
MAJOR GROUP 1: LEGISLA	TORS, SENI	OR OFFI	CIALS A	ND MANA	AGERS							
Total	153,155	7,355	1,556	24,785	49,754	6,194	1,523	20,010	1,130	1,046	37,488	2,314
Male	123,323	5,875	1,515	20,424	38,838	5,497	1,523	16,152	913	952	29,831	1,804
Female	29,832	1,480	41	4,361	10,917	697	-	3,858	216	94	7,656	511
LEGISLATORS AND SENIO	OR OFFICIAL	S										
Total	41,188	1,737	727	6,005	11,655	2,751	362	7,193	454	315	9,467	523
Male	31,216	1,433	686	4,745	6,519	2,540	362	5,918	295	274	7,996	448
Female	9,972	304	41	1,260	5,135	211	-	1,276	159	41	1,471	75
Legislators												
Total	684	-	11	-	117	44	-	115	69	41	287	-
Male	451	-	-	-	-	44	-	45	35	41	287	-
Female	232	-	11	-	117	-	-	70	35	-	-	-
Senior government officials												
Total	17,799	869	493	3,108	4,277	249	335	4,143	244	190	3,585	307
Male	14,338	869	483	2,708	2,406	249	335	3,288	205	190	3,298	307
Female	3,461	-	9	399	1,872	-	-	855	39	-	287	-
Traditional chiefs and heads	of villages											
Total	13,075	564	21	1,673	4,298	2,408	27	443	-	-	3,569	72
Male	10,457	564	21	1,581	2,591	2,197	27	443	-	-	2,960	72
Female	2,619	-	-	92	1,706	211	-	-	-	-	609	-
Senior officials of special-inte	erest organiza	ations										
Total	9,630	304	203	1,224	2,963	50	-	2,493	141	84	2,025	144
Male	5,970	-	182	455	1,523	50	-	2,142	56	42	1,451	69
Female	3,660	304	21	769	1,440	-	-	350	85	41	574	75
CORPORATE MANAGERS												
Total	49,785	1,404	246	8,841	15,836	1,050	430	7,168	451	266	13,587	507
Male	41,521	1,362	246	7,850	13,297	1,050	430	6,108	424	212	10,104	437
Female	8,263	41	-	992	2,539	-	-	1,060	27	53	3,483	69

Minor Occupational						Re	gion					
Group and Sex							Benishangul				Addis	Dire
·	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Directors and chief executives	3											
Total	8,847	311	32	1,606	1,535	-	75	709	44	53	4,330	154
Male	7,887	311	32	1,606	1,535	-	75	664	44	-	3,468	154
Female	960	-	-	-	-	-	-	45	-	53	862	-
Production and operations c	department m	nanagers										
Total	5,295	41	-	552	2,501	-	-	562	35	41	1,564	-
Male	3,996	-	-	552	1,833	-	-	562	35	41	973	-
Female	1,300	41	-	-	668	-	-	-	-	-	591	-
Other depatmet managers												
Total	35,642	1,051	214	6,683	11,800	1,050	355	5,897	373	172	7,694	353
Male	29,638	1,051	214	5,692	9,930	1,050	355	4,882	346	172	5,663	284
Female	6,004	-	-	992	1,871	-	-	1,015	27	-	2,030	69
GENERAL MANAGERS												
Total	62,183	4,214	583	9,939	22,264	2,393	732	5,650	225	466	14,433	1,285
Male	50,586	3,079	583	7,830	19,021	1,907	732	4,126	194	466	11,731	918
Female	11,596	1,135	-	2,109	3,243	486	-	1,523	31	-	2,703	367
General managers												
Total	62,183	4,214	583	9,939	22,264	2,393	732	5,650	225	466	14,433	1,285
Male	50,586	3,079	583	7,830	19,021	1,907	732	4,126	194	466	11,731	918
Female	11,596	1,135	-	2,109	3,243	486	-	1,523	31	-	2,703	367
MAJOR GROUP 2:PROFESS	IONALS											
Total	336,011	24,228	2,172	47,594	89,595	7,971	3,378	47,890	1,571	2,490	105,769	3,353
Male	226,454	14,600	1,477	34,298	63,584	5,884	2,757	32,613	1,203	1,854	65,721	2,463
Female	109,557	9,628	695	13,297	26,011	2,086	621	15,276	369	636	40,048	891
PHYSICAL, MATHEMATICA	L AND ENG	INEERING	SCIEN	CE PROF	ESSIONAL	S						
Total	30,823	1,191	256	4,194	3,578	258	87	3,203	85	243	17,507	223
Male	24,421	800	256	2,844	2,168	258	39	2,322	23	203	15,357	151
Female	6,402	391	-	1,350	1,410	-	48	880	62	41	2,150	72
Physicists, chemists and relat	ed professio	nals										
Total	2,433	-	176	757	270	111	-	188	-	-	932	-
Male	2,357	-	176	681	270	111	-	188	-	-	932	-
Female	76	-	-	76	-	-	-	-	-	-	-	-

						Re	gion					
Minor Occupational							Benishangul				Addis	Dire
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Mathematicians, statisticians a	nd related pr	ofessiona	ls							· · · · ·		
Total	2,583	-	20	604	511	-	-	166	-	-	1,282	-
Male	2,171	-	20	604	98	-	-	166	-	-	1,282	-
Female	413	-	-	-	413	-	-	-	-	-	-	-
Computing professionals												
Total	4,781	286	-	222	180	40	24	880	62	79	2,936	72
Male	2,257	-	-	-	142	40	-	-	-	79	1,995	-
Female	2,525	286	-	222	38	-	24	880	62	-	941	72
Architects, engineers and rela	ated profession	onals										
Total	21,025	905	59	2,611	2,617	107	63	1,968	23	164	12,357	151
Male	17,636	800	59	1,559	1,658	107	39	1,968	23	123	11,148	151
Female	3,389	105	-	1,052	959	-	24	-	-	41	1,209	-
LIFE SCIENCE AND HEALT	H PROFESS											
Total	46,857	3,937	1,039	5,938	12,846	1,368	195	6,592	151	426	13,559	806
Male	27,875	2,112	805	4,188	7,828	1,080	173	4,407	118	210	6,509	445
Female	18,982	1,825	234	1,750	5,018	287	23	2,184	33	216	7,050	361
Life science professionals												
Total	6,311	286	-	1,474	2,040	356	43	1,480	-	48	584	-
Male	5,773	286	-	1,382	1,663	356	43	1,410	-	48	584	-
Female	538	-	-	92	376	-	-	70	-	-	-	-
Health professionals/except r	•											
Total	20,839	1,939	318	2,438	3,678	643	112	2,127	108	173	8,641	662
Male	15,106	1,150	141	1,954	3,066	643	112	1,859	108	120	5,508	445
Female	5,732	789	176	483	613	-	-	268	-	53	3,133	217
Nursing and midwifery profes	sionals											
Total	19,707	1,712	721	2,026	7,128	369	41	2,984	44	205	4,334	144
Male	6,996	675	664	852	3,098	81	18	1,139	10	42	416	-
Female	12,711	1,037	57	1,175	4,030	287	23	1,846	33	162	3,918	144
TEACHING PROFESSIONAL	_S											
Total	160,585	12,658	335	26,735	51,708	4,435	1,833	27,328	851	1,243	31,885	1,574
Male	111,086	7,182	198	19,201	37,875	3,022	1,446	19,124	617	944	20,141	1,335
Female	49,499	5,475	137	7,534	13,832	1,413	388	8,204	234	299	11,744	239
College, university and highe	r education te		ofessior	nals								
Total	21,443	510	19	3,146	9,056	303	81	4,636	-	427	3,023	242
Male	18,693	510	19	2,551	7,492	261	81	4,167	-	347	3,023	242
Female	2,750	-	-	594	1,564	42	-	469	-	80	-	-

Minor Occupational						Re	gion					
Group and Sex							Benishangul				Addis	Dire
•	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Secondary education teachin	• ·											
Total	74,342	4,859	108	15,808	25,313	1,483	1,206	10,428	597	598	13,169	775
Male	58,615	3,722	39	13,010	20,699	774	1,137	7,870	398	434	9,758	775
Female	15,728	1,137	69	2,798	4,614	709	69	2,558	199	164	3,411	-
Primary and pre-primary ed		•.										
Total	54,382	5,496	135	6,458	15,459	2,318	522	10,348	141	119	12,917	470
Male	26,587	1,949	67	2,943	7,805	1,656	203	5,442	107	119	6,065	231
Female	27,795	3,547	67	3,514	7,654	662	319	4,906	35	-	6,852	239
Special education teaching p	rofessionals											
Total	3,467	1,793	73	359	250	-	-	993	-	-	-	-
Male	2,273	1,001	73	-	250	-	-	948	-	-	-	-
Female	1,195	791	-	359	-	-	-	45	-	-	-	-
Other teaching professionals												
Total	6,950	-	-	966	1,630	332	24	924	112	99	2,777	87
Male	4,919	-	-	697	1,630	332	24	697	112	44	1,296	87
Female	2,031	-	-	269	-	-	-	227	-	55	1,481	-
OTHER PROFESSIONALS												
Total	97,746	6,442	543	10,727	21,464	1,909	1,262	10,767	485	579	42,818	750
Male	63,072	4,506	218	8,064	15,713	1,524	1,099	6,759	445	497	23,715	532
Female	34,674	1,936	325	2,663	5,751	386	163	4,007	40	81	19,104	219
Business professionals												
Total	50,315	1,825	142	5,372	9,977	616	732	5,045	212	293	25,664	438
Male	27,984	1,501	-	3,383	6,052	465	607	2,833	172	253	12,423	295
Female	22,332	324	142	1,989	3,925	150	125	2,212	40	41	13,241	143
Legal professionals												
Total	14,273	1,485	169	2,324	3,988	493	199	1,463	153	158	3,613	227
Male	11,486	1,077	159	1,871	3,462	493	162	1,463	153	158	2,335	152
Female	2,787	407	11	452	526	-	38	-	-	-	1,278	75
Archivists, librarians and relation	ted informatio		ionals									
Total	6,724	1,114	-	396	1,337	142	-	1,167	49	-	2,520	-
Male	1,736	-	-	396	835	-	-	58	49	-	399	-
Female	4,988	1,114	-	-	502	142	-	1,110	-	-	2,121	-
Social science and related pr	ofessionals											
Total	9,621	89	231	1,580	503	235	308	1,605	-	46	4,938	85
Male	6,655	89	59	1,359	97	142	308	1,356	-	46	3,114	85
Female	2,966	-	172	222	406	93	-	248	-	-	1,824	-

(Regular) Group and Sex Total Tigray Afar Amhara Oronia Somalle Genishangul Gambela Hara Addis Dire Writers and creative or performing artiss Total 9,137 455 - 591 2,031 - 23 1,339 71 - 4,626 - Male 7,575 364 - 591 1,639 - 23 1,339 71 - 4,626 - Religious professionals 1,561 91 - 392 - 438 - 640 - 640 - 640 - - - - 438 - 640 - - - - - - 41 - - - - - - - 41 - - - - - - - - - - - - - - - - 1 45							Re	gion					
Order Tigray Atar Amhara Oromia Somalie -Gumuž SNNPR Gambela Harari Abbaba Dawa Writers and creative or performing artists 9,137 455 591 2,031 - 23 9,01 71 - 4,626 - Male 7,575 364 - 591 1,639 - 23 901 71 - 4,626 - Female 1,561 91 - 382 - 148 - 81 1,457 - 464 3,628 423 - 148 - 81 1,457 - <td></td> <td>Addis</td> <td>Dire</td>												Addis	Dire
Writers and creative or performing artists Total 9,137 455 - 591 1,639 - 23 1,39 71 - 4,266 - Male 7,575 3,64 - 591 1,639 - 23 9,01 71 - 4,307 - Female 1,561 91 - - 332 - - 23 9,01 71 - 6,40 - Religious professionals 7,676 1,475 - 464 3,628 423 - 1.48 - 81 1.457 -<	Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie			Gambela	Harari		Dawa
Male 7,575 364 - 591 1,639 - 23 901 71 - 3,987 - Female 1,561 91 - - 392 - - 438 - - 640 . Religious professionals 7,676 1,475 - 464 3,628 423 - 148 - 811 1,457 . Male 7,636 1,475 - 464 3,628 423 - 148 - 811 1,457 . MALOG GROUP 3:TECHNICIANS AND ASSOCITE PROFESSIONALS - - - 41 - - . - - - - 41 4,745 2,953 . 10,303 68 10,553 2,186 3,936 10,761 20,116 52,92 2,185 2,420 1,366 PhysicAL AND ENGINEERING SCIENCE ASSOCIATE PROFESSIONALS - 168 3,647 2,787 3,336 1,71 2,517 <t< td=""><td>Writers and creative or perfor</td><td>ming artists</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	Writers and creative or perfor	ming artists											
Female 1,561 91 - - 392 - - 438 - - 640 Religious professionals 7,676 1,475 - 464 3,628 423 - 148 - 80 1,457 - Male 7,636 1,475 - 464 3,628 423 - 148 - 40 1,457 - Male 7,636 1,475 - 464 3,628 423 - 148 - 40 1,457 - 441 - - - - - 41 - 40 1,457 - 441 - 40 1,457 - - - 40 1,457 - - - 41 - 40 1,457 - - - - 41 40 1,457 - - - - - - - - - - - <	Total	9,137	455	-	591	2,031	-	23	1,339	71	-	4,626	-
Religious professionals Total 7,676 1,475 - 464 3,628 423 - 148 - 401 1,457 - Male 7,636 1,475 - 464 3,628 423 - 148 - 400 1,457 - MAJOR GROUP 3: TECHNICIANS AND ASSOCIATE PROFESSIONALS Total 30,026 1,405 3,403 59,287 100,392 12,028 4,851 54,385 2,188 3,931 71,444 4,318 Male 218,506 8,654 2,909 34,597 71,736 9,448 3,090 3,2269 1,761 2,116 529 2,185 2,4200 1,355 PHYSICAL AND ENGINEERING SCIENCE ASSOCIATE PROFESSIONALS 0 1,112 - 66 764 81 88 4,704 2,284 4690 Male 2,824 2,109 2,362 3,374 544 74 1,554 148 8,825 308 Female 8,225	Male	7,575	364	-	591	1,639	-	23	901	71	-	3,987	-
Total 7,676 1,475 - 464 3,628 423 - 148 - 81 1,457 - Male 7,636 1,475 - - - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - - 41 - 41 4,518 - 41 4,518 - 41 - - 44 4,318 - 41 4,51 - 44 4,318 - - 41 4,718 - 418 4,52 - 436 - 1,518 - 418 1,420 - 1,458 - 1,458 - 1,458 - 1,551 - - 6 763 1,11 <td></td> <td>1,561</td> <td>91</td> <td>-</td> <td>-</td> <td>392</td> <td>-</td> <td>-</td> <td>438</td> <td>-</td> <td>-</td> <td>640</td> <td>-</td>		1,561	91	-	-	392	-	-	438	-	-	640	-
Male 7,636 1,475 - 464 3,628 423 - 148 - 40 1,457 Female 41 - - - - - - - 41 - - - - - 41 - - - - 41 - - - - 41 - - - - - 41 - - - - 41 - - - 41 - - - 41 - - - 41 - - - 41 - - - 41 - - - 41 - - - 41 431 431 - - - 61 64 31 65 2,501 2,501 2,501 2,501 2,66 2,501 1,48 1,80 2,71 1,112 - 66 764 81 88 4,704 2,27 - Female 3,253 333 - 890 1,112 -	Religious professionals												
Female 41 - </td <td></td> <td>,</td> <td>,</td> <td>-</td> <td>464</td> <td></td> <td></td> <td>-</td> <td></td> <td>-</td> <td>-</td> <td></td> <td>-</td>		,	,	-	464			-		-	-		-
MAJOR GROUP 3: TECHNICIANS AND ASSOCIATE PROFESSIONALS Total 330,286 14,059 3,403 59,287 100,392 12,028 4,851 54,385 2,188 3,931 71,444 4,318 Male 218,506 8,854 2,099 34,597 71,736 9,448 3,090 34,269 1,660 1,746 47,245 2,953 Female 111,781 5,205 494 24,690 2,866 2,580 1,761 20,116 529 2,120 1,366 PHYSICAL AND ENGINEERING SCIENCE ASSOCIATE PROFESSIONALS 763 737 3,336 126 237 17,587 696 Male 2,294 2,109 258 3,069 5,799 763 171 2,571 45 148 12,884 4690 Female 8,325 393 - 890 1,112 - 66 764 81 88 4,704 227 Physical and engineering science technicians . 416 480 -	Male	7,636	1,475	-	464	3,628	423	-	148	-	40	1,457	-
Total 330,286 14,059 3,403 59,287 100,392 12,028 4,851 54,385 2,188 3,931 71,444 4,318 Male 218,506 8,854 2,909 34,597 71,736 9,448 3,000 34,269 1,660 1,746 47,245 2,933 PHYSICAL AND ENGINEERING SCIENCE ASSOCIATE PROFESSIONLS 2,500 763 2,37 3,336 126 2,37 1,7,87 696 Male 28,294 2,109 268 3,969 5,799 763 171 2,571 450 148 12,884 469 Female 8,325 393 - 890 1,112 - 66 764 81 88 4,704 227 Physical and engineering science techniciars - 416 483 - 18 82 23 48 2,038 161 Male 3,263 - - 416 483 - 18 82 23 48			-	-	-	-	-	-	-	-	41	-	-
Male 218,506 8,854 2,909 34,597 71,736 9,448 3,090 34,269 1,660 1,746 47,245 2,953 Female 111,781 5,205 494 24,690 28,656 2,560 1,761 20,116 529 2,185 24,200 1,365 PHYSICAL AND ENGINEERING SCIENCE 2,501 268 3,956 6,910 763 277 3,336 126 237 17,587 696 Male 28,294 2,109 268 3,069 5,799 763 171 2,571 45 148 12,884 469 Female 8,325 393 - 800 1,12 - 66 764 81 88 4,004 22,07 Physical and engineering science technicians 7148 3,856 544 93 1,636 23 196 10,863 464 Male 1,874 1,484 180 2,302 3,374 544 74 1,554	MAJOR GROUP 3:TECHNICI	ANS AND A	SSOCIAT	E PROF	ESSIONA	LS							
Female 111,781 5,205 494 24,690 28,656 2,580 1,761 20,116 529 2,185 24,200 1,365 PHYSICAL AND ENGINEERING SCIENCE XSSOLATE PROFESSIONALS Total 36,618 2,501 268 3,958 6,910 763 237 3,336 126 237 17,587 696 Male 28,294 2,109 268 3,056 5,799 763 171 2,571 45 148 12,884 469 Female 8,325 393 - 890 1,112 - 66 764 81 88 4,704 2276 Physical and engineering science techniciars - - 800 1,112 - 66 764 81 82 23 48 4,69 Male 1,704 1,484 180 2,718 3,856 544 74 1,554 - 148 8,825 308 Female 3,066 1,224 72 721		330,286		3,403	59,287		12,028	4,851		2,188	3,931		4,318
PHYSICAL AND ENGINEERING SCIENCE ASSOCIATE PROFESSIONALS Total 36.618 2,501 268 3,958 6,910 763 237 3,336 126 237 17,587 696 Male 28,294 2,109 268 3,069 5,799 763 171 2,571 45 148 12,884 469 Female 8,325 393 - 890 1,112 - 66 764 81 88 4,704 227 Physical and engineering science technicians 3,263 3,336 544 93 1,636 23 196 10,863 464 Male 18,794 1,484 180 2,302 3,374 544 74 1,554 - 148 8,825 308 Female 3,263 - - 416 483 - 18 82 233 161 Male 3,006 123 - 721 721 - - 565 35 41 3,039 161 Male 3,006 123	Male		,	2,909	34,597		9,448	3,090	34,269	1,660	1,746	47,245	2,953
Total 36,618 2,501 268 3,958 6,910 763 237 3,336 126 237 17,587 696 Male 28,294 2,109 268 3,069 5,799 763 171 2,571 45 148 12,884 469 Female 8,325 393 - 800 1,112 - 66 764 81 88 4,704 227 Physical and engineering science technicaus 1,484 180 2,718 3,856 544 93 1,636 23 196 10,863 464 Male 18,794 1,484 180 2,302 3,374 544 74 1,554 - 148 8,825 308 Female 3,203 - - 416 483 - 18 82 23 48 2,039 161 Male 3,006 123 - 172 721 - - - 1,830 161 Female 2,099 302 - 164 494 -							2,580	1,761	20,116	529	2,185	24,200	1,365
Male 28,294 2,109 268 3,069 5,799 763 171 2,571 45 148 12,884 469 Female 8,325 393 - 890 1,112 - 66 764 81 88 4,704 227 Physical and engineering science technicias 22,057 1,484 180 2,718 3,856 544 93 1,636 23 196 10,863 464 Male 18,794 1,484 180 2,718 3,856 544 93 1,636 23 196 10,863 464 Male 18,794 1,484 180 2,718 3,856 544 74 1,554 - 148 8,825 308 157 Computer associate professionals - - 416 483 - 18 82 23 48 2,038 157 Computer associate professionals - - 565 35 41 3,039 161 Male 3,006 123 - 172 721	PHYSICAL AND ENGINEER												
Female 8,325 393 - 890 1,112 - 66 764 81 88 4,704 227 Physical and engineering science technicians 22,057 1,484 180 2,718 3,856 544 93 1,636 23 196 10,863 464 Male 18,794 1,484 180 2,302 3,374 544 74 1,554 - 148 8,825 308 Female 3,263 - - 416 483 - 18 82 23 48 2,038 157 Computer associate professionals - - 416 483 - 18 82 23 48 2,038 157 Computer associate professionals - - 416 483 - - 565 35 41 3,039 161 Male 3,006 123 - 172 721 - - - 1,830 161 Male 3,461 502 - 376 1,204 400			,	268		6,910		237		126	237		696
Physical and engineering science technicians View of the technicians View of technicians	Male			268		,	763						
Total 22,057 1,484 180 2,718 3,856 544 93 1,636 23 196 10,863 464 Male 18,794 1,484 180 2,302 3,374 544 74 1,554 - 148 8,825 308 Female 3,263 - - 416 483 - 18 82 23 48 2,038 157 Computer associate professionals - - 416 483 - - 565 35 41 3,039 161 Male 3,006 123 - 172 721 - - - - 1,830 161 Female 2,809 302 - 164 494 - - 565 35 41 1,209 - Optical and electronic equipment operators - 164 494 - - 565 35 41 1,209 - Total 4,397 593 - 686 1,204 40 72 648				-	890	1,112	-	66	764	81	88	4,704	227
Male 18,794 1,484 180 2,302 3,374 544 74 1,554 - 148 8,825 308 Female 3,263 - - 416 483 - 18 82 23 48 2,038 157 Computer associate professionals - - 336 1,215 - - 565 35 41 3,039 161 Male 3,006 123 - 172 721 - - - - 1,830 161 Female 2,809 302 - 164 494 - - 565 35 41 1,209 - Optical and electronic equipment operators - 1686 1,204 40 121 766 - 917 71 Male 3,461 502 - 376 1,204 40 72 648 - 298 71 Ship and aircraft controllers and technicians - 122 128 179 - 305 10 - 1,77	Physical and engineering scie	ence technici											
Female3,263-+416483-188223482,038157Computer associate professionalsTotal5,815424-3361,21556535413,039161Male3,006123-1727211,830161Female2,809302-16449456535411,209-Optical and electronic equipment operators-1681,20440012176691771Male3,461502-3761,20440072648619-Female93691-3104811820971Ship and aircraft controllers and technicians-122238179-30510-1,778-Male2,019-12122198179-30510-1,193-Female62440584-584-Safety and quality inspectors-1707-7697398-246458-991-Male1,014-7697302-246435-416-			1,484			3,856				23		10,863	
Computer associate professionals Total 5,815 424 - 336 1,215 - - 565 35 41 3,039 161 Male 3,006 123 - 172 721 - - - - 1,830 161 Female 2,809 302 - 164 494 - - 565 35 41 1,209 - Optical and electronic equipment operators - - 565 35 41 1,209 - Total 4,397 593 - 686 1,204 400 121 766 - - 917 71 Male 3,461 502 - 376 1,204 400 72 648 - - 208 71 Male 3,461 502 - 376 1,204 400 72 648 - - 208 71 Ship a			1,484	180			544						
Total 5,815 424 - 336 1,215 - - 565 35 41 3,039 161 Male 3,006 123 - 172 721 - - - - 1,830 161 Female 2,809 302 - 164 494 - - 565 35 41 1,209 - Optical and electronic equipment operators - 164 494 - - 565 35 41 1,209 - Optical and electronic equipment operators - - 686 1,204 400 121 766 - 917 71 Male 3,461 502 - 376 1,204 400 72 648 - 619 - Female 936 91 - 310 - - 48 118 - - 298 71 Ship and aircraft controllers and technicians - 12 122 128 179 305 10 - 1,778			-	-	416	483	-	18	82	23	48	2,038	157
Male 3,006 123 - 172 721 - - - - - 1,830 161 Female 2,809 302 - 164 494 - - 565 35 41 1,209 - Optical and electronic equipment operators - 4,397 593 - 686 1,204 40 121 766 - - 917 71 Male 3,461 502 - 376 1,204 40 72 648 - - 619 - Female 936 91 - 310 - - 48 118 - - 298 71 Ship and aircraft controllers and technicians - 12 122 238 179 - 305 10 - 1,778 - Male 2,019 - 12 122 198 179 - 305 10 - 1,193 - Female 624 - - 40 - -													
Female2,809302-16449456535411,209-Optical and electronic equipment operators-6861,2044012176691771Male3,461502-3761,2044072648-619-Female93691-31048118-29871Ship and aircraft controllers and technicians-12122238179-30510-1,778-Male2,019-12122198179-30510-1,193-Female62440584-Safety and quality inspectors-7697398-246458-991-Male1,014-7697302-246435-416-				-			-	-	565	35	41		
Optical and electronic equipment operators Total 4,397 593 - 686 1,204 40 121 766 - - 917 71 Male 3,461 502 - 376 1,204 40 72 648 - - 619 - Female 936 91 - 370 - - 48 118 - - 298 71 Ship and aircraft controllers and technicians 12 122 238 179 - 305 10 - 1,778 - Male 2,019 - 12 122 238 179 - 305 10 - 1,778 - Male 2,019 - 12 122 198 179 - 305 10 - 1,193 - Female 624 - - 40 - - - 584 - Safety and quality inspectors - 76 97 398 - 24 64				-			-	-	-	-	-		161
Total4,397593-6861,2044012176691771Male3,461502-3761,2044072648619-Female93691-3104811829871Ship and aircraft controllers and technicians-12122238179-30510-1,778-Total2,643-12122198179-30510-1,193-Male2,019-12122198179-30510-1,193-Female62440584-Safety and quality inspectors-7697398-246458-991-Male1,014-7697302-246435-416-				-	164	494	-	-	565	35	41	1,209	-
Male 3,461 502 - 376 1,204 40 72 648 - - 619 - Female 936 91 - 310 - - 48 118 - - 298 71 Ship and aircraft controllers and technicians 71 7 48 118 - - 298 71 Ship and aircraft controllers and technicians 12 122 238 179 - 305 10 - 1,778 - Male 2,019 - 12 122 198 179 - 305 10 - 1,193 - Female 624 - - 40 - 40 - - 584 - Safety and quality inspectors - - 40 - 24 64 58 - 991 - Male 1,014 - 76 97 392 - 24 64 35 - 416 -		-											
Female93691-3104811829871Ship and aircraft controllers and techniciansTotal2,643-12122238179-30510-1,778-Male2,019-12122198179-30510-1,193-Female62440584-Safety and quality inspectorsTotal1,707-7697398-246458-991-Male1,014-7697302-246435-416-				-		•				-	-	-	71
Ship and aircraft controllers and technicians Total 2,643 - 12 122 238 179 - 305 10 - 1,778 - Male 2,019 - 12 122 198 179 - 305 10 - 1,193 - Female 624 - - 40 - - - 584 - Safety and quality inspectors - - 76 97 398 - 24 64 58 - 991 - Male 1,014 - 76 97 302 - 24 64 35 - 416 -				-		1,204	40			-	-		-
Total2,643-12122238179-30510-1,778-Male2,019-12122198179-30510-1,193-Female62440584-Safety and quality inspectorsTotal1,707-7697398-246458-991-Male1,014-7697302-246435-416-				-	310	-	-	48	118	-	-	298	71
Male 2,019 - 12 122 198 179 - 305 10 - 1,193 - Female 624 - - - 40 - - - - 584 - Safety and quality inspectors - - 76 97 398 - 24 64 58 - 991 - Male 1,014 - 76 97 302 - 24 64 35 - 416 -	•		าร										
Female 624 - - 40 - - - - 584 - Safety and quality inspectors - - 76 97 398 - 24 64 58 - 991 - Male 1,014 - 76 97 302 - 24 64 35 - 416 -			-					-			-		-
Safety and quality inspectors Total 1,707 - 76 97 398 - 24 64 58 - 991 - Male 1,014 - 76 97 302 - 24 64 35 - 416 -			-	12	122		179	-	305	10	-		-
Total1,707-7697398-246458-991-Male1,014-7697302-246435-416-		624	-	-	-	40	-	-	-	-	-	584	-
Male 1,014 - 76 97 302 - 24 64 35 - 416 -	• • • •												
			-		-		-		-		-		-
Female 693 95 23 - 574 -			-	76	97		-	24	64		-		-
	Female	693	-	-	-	95	-	-	-	23	-	574	-

						Re	gion					
Minor Occupational							Benishangul				Addis	Dire
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
LIFE SCIENCE AND HEALT	H ASSOCIAT	E PROFE	SSION	ALS					•			
Total	84,739	4,935	1,414	17,465	28,703	5,163	2,227	14,711	910	1,052	7,350	810
Male	53,052	2,721	1,362	10,414	19,012	3,489	1,322	9,892	740	370	3,207	523
Female	31,688	2,213	53	7,050	9,691	1,674	905	4,819	170	682	4,142	287
Life science technicians and	related assoc	ciate profe	ssionals									
Total	22,562	1,027	788	4,277	9,526	948	640	4,189	629	86	302	149
Male	19,731	936	765	4,101	7,625	905	545	3,909	494	-	302	149
Female	2,831	91	23	176	1,901	43	95	280	135	86	-	-
Modern health associate prof		cept nurs	•									
Total	36,442	1,149	552	8,932	9,275	2,350	942	6,612	218	462	5,588	361
Male	20,241	453	533	4,497	4,657	1,989	525	4,313	195	245	2,618	217
Female	16,201	696	19	4,435	4,618	361	417	2,299	23	218	2,970	145
Nursing and midwifery assoc	•											
Total	23,799	2,650	75	3,478	9,257	1,789	490	3,848	63	462	1,459	228
Male	12,359	1,332	64	1,817	6,231	557	139	1,669	51	125	287	85
Female	11,441	1,317	11	1,661	3,027	1,232	350	2,178	12	337	1,172	143
Traditional medicine practition												
Total	1,937	109	-	778	645	75	155	62	-	41	-	72
Male	721	-	-	-	499	38	112	-	-	-	-	72
Female	1,216	109	-	778	145	38	43	62	-	41	-	-
TEACHING ASSOCIATE PR												
Total	100,801	2,486	1,528	21,439	33,289	4,392	1,314	18,035	353	1,881	15,144	939
Male	54,202	1,136	1,096	8,419	20,116	3,865	720	10,015	256	687	7,275	618
Female	46,599	1,351	432	13,020	13,173	527	594	8,020	97	1,194	7,869	321
Primary education teaching a												
Total	80,721	1,563	1,360	19,488	27,925	3,911	999	13,712	279	1,351	9,448	686
Male	44,899	593	1,074	7,401	17,715	3,384	720	7,426	207	603	5,231	546
Female	35,823	970	285	12,087	10,210	527	279	6,286	72	748	4,217	140
Pre-primary education teachi	•											
Total	11,987	159	158	971	4,617	161	91	1,367	25	530	3,657	253
Male	3,627	-	11	140	1,745	161	-	520	-	84	895	72
Female	8,360	159	147	831	2,872	-	91	847	25	446	2,762	181
Special education teaching a		essionals										
Total	3,621	-	-	-	553	-	-	1,565	49	-	1,454	-
Male	2,867	-	-	-	461	-	-	1,495	49	-	862	-
Female	753	-	-	-	91	-	-	70	-	-	592	-

Minor Occupational						Re	gion					
Minor Occupational Group and Sex					a .		Benishangul				Addis	Dire
•	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Other teaching Associate Pro					405							
Total	4,472	764	11	980	195	321	225	1,391	-	-	585	-
Male	2,809	543	11	878	195	321	-	574	-	-	287	-
Female	1,663	222	-	102	-	-	225	817	-	-	298	-
OTHER ASSOCIATE PROF												
Total	108,129	4,136	193	16,425	31,490	1,711	1,073	18,303	800	762	31,364	1,872
Male	82,959	2,888	184	12,695	26,809	1,332	878	11,791	619	541	23,879	1,343
Female	25,169	1,248	9	3,729	4,680	379	195	6,512	181	221	7,485	529
Finance and sales associate	•											
Total	13,931	687	-	2,973	3,204	61	109	1,553	119	-	4,853	373
Male	9,852	246	-	2,367	2,919	61	109	518	119	-	3,372	142
Female	4,079	441	-	606	285	-	-	1,035	-	-	1,481	231
Business services agents and												
Total	15,341	353	-	987	5,890	317	21	495	-	-	7,128	149
Male	13,280	353	-	912	5,279	156	21	495	-	-	5,914	149
Female	2,061	-	-	75	611	161	-	-	-	-	1,214	-
Administrative associate prof	essionals											
Total	36,424	1,250	28	6,818	8,906	341	676	9,696	417	437	7,459	396
Male	24,326	705	28	4,745	7,382	341	502	4,435	236	216	5,414	323
Female	12,098	546	-	2,073	1,525	-	174	5,261	181	221	2,045	73
Customs, tax and related gov	ernment ass	ociate pro	fessiona	als								
Total	7,923	160	-	298	3,792	141	-	1,293	82	53	1,951	154
Male	5,673	-	-	298	2,796	141	-	1,293	82	53	1,011	-
Female	2,249	160	-	-	996	-	-	-	-	-	940	154
Police inspectors and detective	ves											
Total	11,390	1,107	29	1,463	3,862	604	143	1,409	66	142	2,424	141
Male	10,593	1,107	19	1,242	3,639	604	121	1,409	66	142	2,103	141
Female	797	-	9	222	223	-	21	-	-	-	322	-
Social work associate profess	sionals											
Total	1,284	-	-	1,241	43	-	-	-	-	-	-	-
Male	638	-	-	, 595	43	-	-	-	-	-	-	-
Female	646	-	-	646	-	-	-	-	-	-	-	-
Artistic, entertainment and sp		e professi	ionals									
Total	6,174	216	-	559	1,523	29	27	178	-	92	3,406	144
Male	4,669	216	-	452	1,294	29	27	178	-	92	2,238	144
Female	1,505		-	107	230	-		-	-	-	1,169	-
. emaio	1,000				200						1,100	

Minor Occupational						Re	gion					
Minor Occupational Group and Sex							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Religious associate professio	onals											
Total	15,246	262	-	2,085	4,270	219	98	3,502	116	38	4,142	514
Male	13,613	262	-	2,085	3,459	-	98	3,285	116	38	3,828	443
Female	1,633	-	-	-	812	219	-	217	-	-	315	71
Associat professionals not el		sified.										
Total	416	101	136	-	-	-	-	178	-	-	-	-
Male	315	-	136	-	-	-	-	178	-	-	-	-
Female	101	101	-	-	-	-	-	-	-	-	-	-
MAJOR GROUP 4: CLERKS												
Total	289,991	18,866	4,832	43,220	88,007	5,043	3,150	35,818	1,772	3,152	80,810	5,320
Male	119,703	7,559	2,656	16,867	43,571	3,273	1,499	15,538	1,162	1,114	23,090	3,373
Female	170,288	11,307	2,175	26,353	44,435	1,770	1,651	20,280	611	2,039	57,720	1,947
OFFICE CLERKS												
Total	197,215	13,304	4,111	30,361	54,150	3,903	2,552	27,688	1,478	2,053	54,214	3,400
Male	70,489	4,530	2,187	8,517	23,856	2,239	1,203	10,605	933	567	13,735	2,117
Female	126,726	8,775	1,924	21,845	30,294	1,663	1,350	17,083	546	1,486	40,479	1,282
Secretaries and keyboard-op	erating clerks	5										
Total	90,176	7,105	1,163	13,589	24,505	1,498	444	9,056	461	924	30,010	1,421
Male	12,506	1,921	370	1,142	3,086	490	-	854	85	55	3,850	653
Female	77,670	5,184	792	12,447	21,418	1,009	444	8,202	376	870	26,160	768
Numerical clerks												
Total	51,972	3,001	1,445	6,521	16,720	1,340	509	9,170	432	224	11,815	795
Male	27,508	1,177	788	2,829	11,192	842	387	4,544	362	131	4,760	496
Female	24,464	1,824	658	3,692	5,527	498	122	4,627	70	93	7,055	299
Material-recording and transp	oort clerks											
Total	24,472	1,425	543	3,946	6,319	337	788	3,991	178	499	5,802	644
Male	13,344	508	335	1,486	4,440	260	509	1,864	154	297	3,062	429
Female	11,128	918	207	2,460	1,879	77	279	2,127	25	202	2,740	215
Library, maill and related cler	ks											
Total	18,433	1,530	327	4,606	2,415	123	516	3,510	383	326	4,392	304
Male	8,777	680	147	1,871	1,867	44	118	1,578	308	85	1,775	304
Female	9,657	849	179	2,735	548	80	399	1,932	75	241	2,618	-
Other office clerks												
Total	12,162	243	634	1,699	4,192	605	296	1,961	23	79	2,194	235
Male	8,354	243	546	1,189	3,270	605	190	1,765	23	-	287	235
Female	3,808	-	88	511	922	-	106	195	-	79	1,907	-
	, -										,	

						Re	gion					
Minor Occupational							Benishangul				Addis	Dire
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
CUSTOMER SERVICES CLI	ERKS						•					
Total	92,776	5,562	721	12,859	33,856	1,140	598	8,130	294	1,099	26,596	1,921
Male	49,214	3,029	469	8,351	19,715	1,034	296	4,933	229	546	9,355	1,256
Female	43,562	2,533	251	4,508	14,141	106	301	3,197	65	553	17,241	665
Cashiers, tellers and related	clerks											
Total	83,861	5,085	721	11,653	31,467	907	555	7,333	294	1,055	22,948	1,842
Male	46,213	2,644	469	8,168	18,065	827	296	4,813	229	546	8,978	1,177
Female	37,648	2,442	251	3,484	13,402	80	258	2,520	65	509	13,970	665
Client information clerks												
Total	8,915	477	-	1,206	2,389	233	43	796	-	44	3,648	78
Male	3,001	385	-	183	1,651	206	-	120	-	-	378	78
Female	5,914	91	-	1,023	739	26	43	676	-	44	3,271	-
MAJOR GROUP 5: SERVIC	E WORKERS	S AND SH		D MARKE	T SALES V	VORKES						
Total	1,275,292	90,842	13,613	221,444	407,073	53,260	9,756	180,295	5,480	15,663	255,957	21,908
Male	576,401	34,533	6,741	91,579	183,533	20,822	3,365	90,253	2,422	5,518	128,787	8,848
Female	698,891	56,309		129,865	223,540	32,437	6,392	90,042	3,058	10,145	127,170	13,060
PERSONAL AND PROTECT	IVE SERVIC		KERS									
Total	432,485	31,026	2,901	69,609	122,292	13,756	4,071	71,060	2,291	3,302	106,442	5,734
Male	186,080	9,245	1,177	27,657	54,552	7,092	1,252	33,686	991	1,628	45,829	2,972
Female	246,405	21,781	1,723	41,953	67,741	6,664	2,819	37,374	1,300	1,674	60,614	2,763
Travel attendants and related	d workers											
Total	13,330	762	20	1,348	2,373	762	-	3,547	35	40	4,140	303
Male	10,961	762	20	1,277	2,032	480	-	3,393	-	40	2,654	303
Female	2,369	-	-	72	342	282	-	154	35	-	1,486	-
Housekeeping and restauran												
Total	259,999	16,597	1,681	43,360	76,151	5,582	2,749	45,355	1,203	888	64,741	1,691
Male	82,283	2,143	335	9,117	27,118	1,456	494	16,774	157	324	23,903	460
Female	177,716	14,454	1,346	34,243	49,032	4,126	2,255	28,581	1,046	564	40,838	1,231
Personal care and related wo	orkers											
Total	33,441	2,435	-	4,083	7,681	1,038	346	5,203	37	652	11,051	916
Male	9,406	1,420	-	978	1,682	26	173	2,558	37	167	2,215	149
Female	24,035	1,015	-	3,105	5,999	1,011	173	2,645	-	485	8,836	766
Other personal services work												
Total	69,202	5,429	380	8,723	22,671	2,206	342	7,797	121	786	18,746	2,002
Male	41,728	1,230	259	6,929	14,544	1,534	215	3,878	49	428	11,105	1,557
Female	27,474	4,199	121	1,794	8,127	672	127	3,919	72	358	7,641	445

						Re	gion					
Minor Occupational							Benishangul				Addis	Dire
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz		Gambela	Harari	Ababa	Dawa
Astrologers, fortune-Tellers a	nd Related W											
Total	2,036	243	-	92	677	-	-	70	-	40	914	-
Male	866	243	-	92	186	-	-	-	-	40	305	-
Female	1,170	-	-	-	491	-	-	70	-	-	609	-
Protective services workers												
Total	46,043	3,975	736	11,146	9,302	3,633	391	7,772	885	896	6,564	744
Male	40,550	3,446	564	9,264	8,990	3,596	370	7,083	748	629	5,360	502
Female	5,493	530	172	1,882	312	37	21	689	137	267	1,204	242
Prostitutes.												
Total	8,148	1,584	84	857	3,438	536	244	1,316	10	-	-	78
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	8,148	1,584	84	857	3,438	536	244	1,316	10	-	-	78
Protective services/body guar	rd, patrol not	elsewhere	e classifi	ed.								
Total	287	-	-	-	-	-	-	-	-	-	287	-
Male	287	-	-	-	-	-	-	-	-	-	287	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
PERSONAL AND PROTECT												
Total	842,807	59,817	10,713	151,835	284,781	39,504	5,685	109,235		12,361	149,514	,
Male	390,321	25,288	5,564	63,922	128,981	13,730	2,113	56,567	1,431	3,891	82,958	5,876
Female	452,487	34,528	5,149	87,912	155,800	25,774	3,572	52,669	1,758	8,471	66,557	10,297
Fashion and other models												
Total	6,949	101	-	1,863	1,681	218	51	966	23	179	1,866	-
Male	4,998	101	-	602	1,274	174	51	907	23	-	1,866	-
Female	1,951	-	-	1,262	407	44	-	59	-	179	-	-
Shop salespersons and demo												
Total	395,035	30,056	6,789	61,862	116,584	17,159	2,215	44,616	931	4,760	103,414	6,648
Male	231,234	15,017	3,993	34,747	72,715	7,923	1,167	29,245	672	2,530	60,172	3,053
Female	163,801	15,039	2,796	27,115	43,869	9,236	1,049	15,371	260	2,230	43,242	3,595
Sellers in gulit & open marke	•	•										
Total	291,606	20,964	1,570	68,009	100,070	13,725	2,425	45,025	1,438	3,930	30,221	4,229
Male	100,491	7,614	894	20,227	33,046	2,654	700	18,609	403	729	14,290	1,325
Female	191,114	13,349	676	47,782	67,024	11,072	1,724	26,416	1,035	3,201	15,931	2,903
"Katikala" retailers												
Total	31,665	3,513	176	3,773	15,201	-	399	5,541	464	256	2,171	170
Male	4,728	1,432	10	-	738	-	-	1,296	111	175	966	-
Female	26,937	2,081	166	3,773	14,463	-	399	4,245	353	81	1,205	170

						D-	aion					
Minor Occupational		[]		, ı		Re	gion Boniohongul			<u> </u>	۸ ما دا : م	Dire
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Related workers not stated el		5-7						-				
Total	117,553	5,183	2,178	16,326	51,245	8,402	594	13,087	333	3,236	11,842	5,127
Male	48,869	1,124	667	8,346	21,208	2,980	194	6,510	222	457	5,663	1,499
Female	68,684	4,059	1,511	7,980	30,037	5,422	400	6,577	111	2,779	6,179	3,628
MAJOR GROUP 6:SKILLED	AGRICULTU	RAL AND	FISHE	RY WORK	ERS							
Total	394,017	18,127	6,976	95,723	147,029	25,292	10,504	61,238	11,247	650	15,180	2,051
Male	259,469	12,061	6,131	71,277	91,788	18,283	7,271	36,454	4,367	550	10,068	1,219
Female	134,548	6,066	845	24,445	55,241	7,008	3,233	24,784	6,880	101	5,112	832
MARKET-ORIENTED SKILL	ED AGRICUL	TURAL A	ND FIS	HERY WO	RKERS							
Total	170,253	12,691	6,502	30,935	59,940	19,520	2,460	18,542	3,979	361	13,272	2,051
Male	111,700	7,973	5,693	22,593	35,856	13,052	1,565	12,324	2,254	306	8,866	1,219
Female	58,552	4,718	810	8,341	24,084	6,468	896	6,218	1,725	55	4,406	832
Market gardeners and crop g	rowers											
Total	82,591	4,918	1,464	9,755	34,069	13,800	2,246	8,176	989	221	6,547	406
Male	57,347	3,976	1,357	8,666	21,344	9,077	1,410	6,896	776	166	3,364	316
Female	25,244	942	107	1,089	12,725	4,723	836	1,280	214	55	3,183	91
Market-oriented animal produ												
Total	56,514	5,423	685	15,221	20,850	4,078	60	5,039	478	85	3,323	1,272
Male	29,704	2,438	395	8,558	11,615	2,383	-	1,153	85	85	2,461	530
Female	26,810	2,985	290	6,664	9,235	1,695	60	3,886	393	-	862	742
Market-oriented crop and ani	•											
Total	22,366	1,163	721	5,497	3,857	731	103	4,588	2,276	55	3,082	293
Male	17,174	767	444	4,908	2,160	682	103	3,735	1,305	55	2,721	293
Female	5,192	396	277	589	1,697	49	-	852	972	-	361	-
Forestry and related workers												
Total	7,573	1,187	3,633	462	737	911	51	276	236	-	-	80
Male	6,694	791	3,497	462	737	911	51	76	88	-	-	80
Female	879	396	136	-	-	-	-	200	148	-	-	-
Fishery workers, hunters and												
Total	1,209	-	-	-	427	-	-	463	-	-	319	-
Male	782	-	-	-	-	-	-	463	-	-	319	-
Female	427	-	-	-	427	-	-	-	-	-	-	-
SUBSISTENCE AGRICULTU												
Total	223,765	5,436	473		87,089	5,772	8,044	42,696	7,268	290	1,908	-
Male	147,768	4,088	438	48,684	55,932	5,231	5,706	24,130	2,113	244	1,202	-
Female	75,996	1,348	35	16,104	31,157	541	2,338	18,566	5,155	46	706	-

Minor Occupational						Re	gion					
Minor Occupational Group and Sex							Benishangul				Addis	Dire
•	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Subsistence agricultural-crop												
Total	90,888	4,278	447	13,172	37,848	2,290	4,509	24,623	2,684	146	892	-
Male	67,138	3,095	421	11,216	31,502	1,749	3,064	13,234	1,819	146	892	-
Female	23,751	1,183	26	1,956	6,346	541	1,445	11,389	866	-	-	-
Subsistence livestock produc												
Total	34,832	330	9	4,453	19,356	142	48	9,415	63	-	1,016	-
Male	15,263	165	-	3,074	6,106	142	-	5,466	-	-	310	-
Female	19,569	165	9	1,379	13,250	-	48	3,949	63	-	706	-
Knowledgeable subsistence of	crop & anima	l producer	S									
Total	87,042	828	17	47,163	23,317	3,340	3,486	8,659	88	144	-	-
Male	62,954	828	17	34,394	16,205	3,340	2,642	5,431	-	98	-	-
Female	24,088	-	-	12,769	7,112	-	845	3,228	88	46	-	-
Subsistence forestry and rela	•	vorkers										
Total	11,002	-	-	-	6,569	-	-	-	4,433	-	-	-
Male	2,414	-	-	-	2,119	-	-	-	295	-	-	-
Female	8,588	-	-	-	4,450	-	-	-	4,138	-	-	-
Fishery workers and hunters	for own cons	sumption										
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
MAJOR GROUP 7: CRAF	FAND RELA	TED TRA	DES W	ORKERS								
Total	958,971	64,824	10,308	184,328	324,930	16,817	7,579	116,016	4,539	8,363	207,245	14,022
Male	576,176	40,982	5,750	97,300	188,685	12,784	3,542	60,035	1,406	5,540	151,792	8,360
Female	382,795	23,843	4,557	87,028	136,245	4,033	4,037	55,982	3,133	2,823	55,453	5,662
EXTRACTION AND BUILDIN	IG TRADES \	NORKER	S									
Total	276,299	20,813	3,511	44,530	88,070	6,694	1,950	33,976	1,035	3,671	66,457	5,593
Male	246,317	18,889	3,350	37,283	80,827	6,496	1,929	32,033	908	3,208	56,505	4,889
Female	29,982	1,924	161	7,247	7,243	198	21	1,943	127	462	9,951	704
Miners, shotfirers, stone cutte	ers and carve	rs										
Total	33,441	2,043	408	7,484	10,127	90	139	2,889	31	1,185	8,692	352
Male	24,152	1,470	408	5,248	7,836	90	139	2,586	31	905	5,158	283
Female	9,289	573	-	2,236	2,292	-	-	303	-	280	3,534	69
Building frame and related tra	ades workers											
Total	178,204	12,467	2,745	26,753	63,338	4,413	1,157	25,347	855	1,766	35,051	4,312
Male	166,437	11,896	2,584	23,366	59,673	4,373	1,157	24,276	727	1,584	32,942	3,858
Female	11,766	571	161	3,387	3,665	40	-	1,071	127	182	2,108	454

Minor Occupational	Region											
Group and Sex							Benishangul				Addis	Dire
•	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Building finiishers and related	I trades work	ers										
Total	48,489	5,001	212	8,813	10,975	2,124	475	3,992	92	598	15,579	626
Male	44,029	4,576	212	7,565	10,252	1,967	454	3,525	92	598	14,343	445
Female	4,460	426	-	1,248	723	158	21	467	-	-	1,236	181
Painters, building structure cl				kers								
Total	16,165	1,301	146	1,480	3,629	66	179	1,747	57	122	7,135	303
Male	11,699	948	146	1,104	3,066	66	179	1,646	57	122	4,062	303
Female	4,466	354	-	376	563	-	-	100	-	-	3,073	-
METAL, MACHINERY AND F	RELATED TR	RADES W	ORKER	S								
Total	119,095	6,684	765	16,532	31,725	2,847	566	10,905	133	993	45,845	2,100
Male	113,499	6,684	765	16,218	29,985	2,770	549	10,040	133	993	43,353	2,010
Female	5,596	-	-	314	1,740	77	17	865	-	-	2,492	91
Metal moulders, weldeers, sh	eetmetal wor	kers, stru	ctural m	etal								
Total	37,829	2,485	308	5,540	10,367	152	173	3,498	17	378	14,221	691
Male	36,365	2,485	308	5,540	10,005	112	173	3,439	17	378	13,219	691
Female	1,464	-	-	-	362	40	-	59	-	-	1,003	-
Blacksmiths, tool-makers and	related trade	es workers	5									
Total	7,217	288	81	2,119	2,820	653	54	1,048	-	80	-	75
Male	6,579	288	81	1,898	2,790	653	54	661	-	80	-	75
Female	638	-	-	222	30	-	-	386	-	-	-	-
Machinery mechanics and fitt	ers											
Total	45,438	1,845	271	5,271	10,408	1,115	166	3,104	116	384	22,026	733
Male	43,103	1,845	271	5,271	9,562	1,115	166	3,104	116	384	20,536	733
Female	2,335	-	-	-	846	-	-	-	-	-	1,489	-
Electrical and electronic equi	oment mecha	anics and f	fitters									
Total	28,611	2,066	106	3,602	8,131	927	173	3,255	-	151	9,598	602
Male	27,452	2,066	106	3,509	7,628	890	157	2,836	-	151	9,598	512
Female	1,159	-	-	93	503	37	17	419	-	-	-	91
PRECISION, HANDICRAF	T, PRINTING	AND REL	ATED 1	FRADES W	VORKERS							
Total	69,918	5,522	269	16,533	18,481	1,043	465	5,445	377	955	20,392	436
Male	39,354	3,225	75	10,331	7,915	903	121	4,035	25	279	12,232	213
Female	30,563	2,297	194	6,202	10,566	140	344	1,410	352	676	8,160	223
Precision workers in metal ar	nd related ma	terials										
Total	6,639	1,418	-	3,248	980	38	-	845	-	40	-	69
Male	6,208	987	-	3,248	980	38	-	845	-	40	-	69
Female	431	431	-	-	-	-	-	-	-	-	-	-

	Region											
Minor Occupational							Benishangul				Addis	Dire
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Potters, glass-makers and re								•				
Total	28,262	1,792	204	7,052	9,885	73	344	2,107	327	630	5,776	72
Male	4,916	188	11	1,168	971	-	-	820	-	-	1,758	-
Female	23,346	1,603	194	5,883	8,914	73	344	1,287	327	630	4,018	72
Handicraft workers in wood, t	extile, leather	r and relat	ed mate	rials								
Total	24,141	1,527	-	5,146	6,035	865	103	2,059	49	149	8,136	73
Male	21,260	1,527	-	5,146	5,069	865	103	1,995	25	149	6,382	-
Female	2,881	-	-	-	967	-	-	64	25	-	1,754	73
Printing and related trades we	orkers											
Total	10,876	784	64	1,087	1,581	67	18	435	-	136	6,481	222
Male	6,970	522	64	769	895	-	18	375	-	91	4,092	144
Female	3,906	263	-	319	686	67	-	59	-	46	2,389	78
OTHER CRAFT AND RELAT	ED TRADES	WORKE	RS									
Total	493,622	31,806	5,762	106,733	186,653	6,234	4,598	65,691	2,994	2,706	74,551	5,893
Male	177,006	12,184	1,560	33,468	69,957	2,615	943	13,927	341	1,060	39,702	1,248
Female	316,616	19,622	4,202	73,265	116,696	3,619	3,655	51,764	2,653	1,646	34,849	4,645
Food processing and related	trades worke	rs										
Total	270,671	9,153	4,961	59,077	108,182	5,007	3,432	48,723	2,725	2,081	23,131	4,197
Male	36,767	1,326	1,287	6,975	13,824	1,438	227	3,706	178	641	6,577	587
Female	233,904	7,827	3,674	52,102	94,358	3,570	3,206	45,017	2,547	1,440	16,554	3,610
Wood treaters, cabinet-make	ers and relate											
Total	21,267	500	21	4,851	6,247	342	128	3,441	32	85	5,477	143
Male	19,040	-	10	4,657	5,793	342	128	3,322	32	42	4,570	143
Female	2,228	500	11	194	454	-	-	119	-	42	907	-
Textile, garment and related	trades worker											
Total	182,494	20,922	707	40,177	68,922	793	1,037	11,932	238	500	35,943	1,324
Male	106,171	9,903	189	20,334	47,079	744	588	5,304	132	336	21,273	290
Female	76,323	11,019	518	19,843	21,843	49	449	6,627	106	164	14,669	1,034
Pelt, leather and shoemaking	trades worke	ers										
Total	19,190	1,230	74	2,628	3,301	92	-	1,594	-	41	10,001	229
Male	15,028	954	74	1,501	3,261	92	-	1,594	-	41	7,282	229
Female	4,162	276	-	1,127	40	-	-	-	-	-	2,718	-
MAJOR GROUP 8:PLANT A			TORS	AND ASSE								
Total	268,268	23,654	1,802	35,729	78,186	4,096	1,530	23,342	755	2,189	90,454	6,531
Male	246,870	18,380	1,553	33,870	70,581	3,792	1,510	22,030	755	2,107	86,137	6,155
Female	21,398	5,274	249	1,859	7,605	304	20	1,312	-	82	4,317	376

	Region											
Minor Occupational							Benishangul				Addis	Dire
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
STATIONARY-PLANT AND											-	
Total	15,102	587	380	1,587	3,810	-	209	2,396	-	94	5,797	241
Male	12,700	476	257	1,304	2,852	-	209	1,853	-	94	5,486	169
Female	2,402	111	123	283	959	-	-	543	-	-	311	72
Mining- and mineral-process	ing-plant oper	ators										
Total	2,095	77	230	277	-	-	-	340	-	-	1,172	-
Male	1,900	77	230	277	-	-	-	144	-	-	1,172	-
Female	195	-	-	-	-	-	-	195	-	-	-	-
Metal-processing-plant opera	ators											
Total	4,559	89	-	315	2,476	-	-	491	-	-	1,188	-
Male	3,998	89	-	315	1,915	-	-	491	-	-	1,188	-
Female	561	-	-	-	561	-	-	-	-	-	-	-
Glass, ceramics and related	plant operato	rs										
Total	522	-	-	-	-	-	-	219	-	-	304	-
Male	522	-	-	-	-	-	-	219	-	-	304	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Wood-processing- and paper	rmaking-plant	operators	6									
Total	2,741	249	59	-	689	-	-	117	-	-	1,627	-
Male	1,973	249	-	-	291	-	-	117	-	-	1,316	-
Female	767	-	59	-	398	-	-	-	-	-	311	-
Chemical-processing-plant o	perators											
Total	1,396	-	-	303	54	-	-	344	-	-	610	85
Male	1,396	-	-	303	54	-	-	344	-	-	610	85
Female	-	-	-	-	-	-	-	-	-	-	-	-
Power-production and relate	d plant operat	ors										
Total	3,763	172	91	692	592	-	183	887	-	94	896	155
Male	2,885	61	27	409	592	-	183	539	-	94	896	84
Female	878	111	64	283	-	-	-	348	-	-	-	72
Automated-assembly-line an	d industrial-ro	bot opera	tors									
Total	26	-	-	-	-	-	26	-	-	-	-	-
Male	26	-	-	-	-	-	26	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
MACHINE OPERATORS AN	ID ASSEMBL	ERS										
Total	67,286	10,943	205	7,157	21,768	830	268	6,616	231	365	17,545	1,359
Male	50,883	6,070	79	5,732	15,555	527	247	5,949	231	324	15,045	1,125
Female	16,403	4,873	126	1,424	6,213	304	20	667	_	41	2,500	235

Minor Occupational						Re	gion					
Group and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Metal- and mineral-products		0,	7 (10)	/ unnara	oronna	Comano	Ounidz		Cambola	Tharan	710000	Dawa
Total	4,660	889	125	645	736	-	-	519	-	-	1,570	176
Male	3,980	889	10	368	736	-	-	519	-	-	1,283	176
Female	680	-	115	277	-	-	-	-	-	-	287	_
Chemical-prpdicts machine o	perators											
Total	2,129	-	-	-	808	-	-	675	-	-	574	72
Male	2,129	-	-	-	808	-	-	675	-	-	574	72
Female	-	-	-	-	-	-	-	-	-	-	-	-
Rubber- and plastic-products	machine ope	erators										
Total	2,374	-	-	-	748	-	-	376	10	-	1,239	-
Male	1,818	-	-	-	193	-	-	376	10	-	1,239	-
Female	556	-	-	-	556	-	-	-	-	-	-	-
Wood-products machine ope	rators											
Total	1,563	281	-	198	64	-	-	1,021	-	-	-	-
Male	1,488	281	-	123	64	-	-	1,021	-	-	-	-
Female	75	-	-	75	-	-	-	-	-	-	-	-
Printing- binding- and paper-	products mac	hine oper	ators									
Total	2,688	292	-	92	38	-	-	-	-	-	2,266	-
Male	1,452	-	-	92	-	-	-	-	-	-	1,360	-
Female	1,236	292	-	-	38	-	-	-	-	-	906	-
Textile- fur- and leather-produ	ucts machine	operators	6									
Total	10,535	3,899	-	761	3,025	-	-	428	-	-	1,838	584
Male	4,548	665	-	585	1,754	-	-	286	-	-	908	349
Female	5,987	3,234	-	176	1,271	-	-	142	-	-	929	235
Food and related products ma	achine opera	tors										
Total	25,189	2,708	-	4,349	13,476	674	38	2,573	159	168	665	379
Male	17,568	1,362	-	3,453	9,345	371	38	2,048	159	127	287	379
Female	7,621	1,347	-	896	4,131	304	-	526	-	41	378	-
Assemblers												
Total	16,448	2,625	69	1,112	2,689	156	209	846	62	197	8,484	-
Male	16,256	2,625	59	1,112	2,527	156	188	846	62	197	8,484	-
Female	192	-	10	-	162	-	20	-	-	-	-	-
Machines products assemble	ers not elsewh	nere class	ified.									
Total	1,700	249	10	-	183	-	21	179	-	-	908	149
Male	1,645	249	10	-	129	-	21	179	-	-	908	149
Female	55	-	-	-	55	-	-	-	-	-	-	-

Minor Occupational						Re	gion					
Minor Occupational Group and Sex							Benishangul				Addis	Dire
•	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
DRIVERS AND MOBILE-PLA												
Total	185,880	12,124	1,217	26,985	52,608	3,266	1,054	14,329	524	1,730	67,112	4,931
Male	183,287	11,833	1,217	26,834	52,175	3,266	1,054	14,228	524	1,689	65,606	4,862
Female	2,593	291	-	151	433	-	-	101	-	41	1,506	69
Locomotive-engine drivers ar		rkers										
Total	1,588	-	-	75	458	-	-	59	-	134	862	-
Male	1,529	-	-	75	458	-	-	-	-	134	862	-
Female	59	-	-	-	-	-	-	59	-	-	-	-
Motor-vehicle drivers												
Total	175,112	11,833	1,102	26,217	48,285	3,266	788	12,732	524	1,556	64,121	4,689
Male	172,877	11,792	1,102	26,065	47,901	3,266	788	12,690	524	1,515	62,615	4,620
Female	2,235	41	-	151	384	-	-	42	-	41	1,506	69
Agricultural and other mobile	• •											
Total	8,579	291	-	495	3,865	-	265	1,538	-	41	1,843	242
Male	8,281	41	-	495	3,816	-	265	1,538	-	41	1,843	242
Female	298	249	-	-	49	-	-	-	-	-	-	-
Ship's deck crews and related	d workers											
Total	600	-	115	198	-	-	-	-	-	-	287	-
Male	600	-	115	198	-	-	-	-	-	-	287	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
MAJOR GROUP 9: ELEMEN	FARY OCCU	PATIONS	5									
Total	1,132,631	59,274	26,875	208,672	324,471	36,735	10,949	141,266	4,659	10,966	284,627	24,138
Male	581,582	30,317	16,292	105,188	178,456	19,079	5,935	80,266	2,740	6,426	125,301	11,581
Female	551,049	28,956	10,583	103,485	146,015	17,656	5,013	61,000	1,919	4,540	159,326	12,557
SALES AND SERVICES ELE	EMENTARY (DCCUPA ⁻	TIONS									
Total	835,106	42,282	20,318	134,307	227,640	23,998	6,385	102,075	3,220	9,882	245,282	19,717
Male	375,803	18,191	11,730	60,640	111,864	10,121	2,950	51,283	1,906	5,522	93,208	8,387
Female	459,303	24,091	8,588	73,667	115,775	13,878	3,435	50,793	1,314	4,360	152,073	11,329
Street vendors and related we	orkers											
Total	75,151	4,781	1,120	16,828	19,067	3,342	381	8,728	281	986	16,686	2,950
Male	37,300	2,314	246	7,621	9,167	939	144	5,838	119	587	9,751	576
Female	37,851	2,467	875	9,208	9,900	2,404	237	2,890	162	399	6,934	2,374
Shoe cleaning and other stree	et services el	ementary	occupat	ions								
Total	28,034	1,537	-	6,887	9,087	615	106	4,346	116	81	4,728	531
Male	22,497	672	-	5,969	8,205	615	106	4,346	116	81	1,856	531
Female	5,537	865	-	918	882	-	-	-	-	-	2,872	-

Minor Occupational Group and Sex Total Tigray Afar Amhara Oromia Somalie Benishangul -Gumuz SNNPR Gambela Harari Addis Ababa Dire Daws Domestic and related helpers, cleaners and launderes 299,843 14,601 4,343 46,418 77,712 3,092 1,517 25,212 678 3,116 116,420 6,73 Male 20,157 1,305 295 4,024 6,677 100 168 1,456 50 238 5,379 46 Female 279,687 13,296 4,048 42,394 71,035 2,992 1,349 23,756 629 2,878 111,041 6,269 Building caretakers, helpers, cleaners and launderers Total 12,787 1,100 157 567 1,618 135 40 572 47 41 8,218 29 Male 2,505 - 19 355 633 - 17 383 - - 878 22
Choop and Sex Total Tigray Afar Amhara Oromia Somalie -Gumuz SNNPR Gambela Harari Ababa Dawa Domestic and related helpers, cleaners and launderes 299,843 14,601 4,343 46,418 77,712 3,092 1,517 25,212 678 3,116 116,420 6,73 Male 201,157 1,305 295 4,024 6,677 100 168 1,456 50 238 5,379 46 Female 279,687 13,296 4,048 42,394 71,035 2,992 1,349 23,756 629 2,878 111,041 6,26 Building caretakers, helpers, cleaners and launderers 12,787 1,100 157 567 1,618 135 40 572 47 41 8,218 29 Male 2,505 - 19 355 633 - 17 383 - - 878 22 Female 10,282 1,100
Domestic and related helpers, cleaners and launderes Total 299,843 14,601 4,343 46,418 77,712 3,092 1,517 25,212 678 3,116 116,420 6,73 Male 20,157 1,305 295 4,024 6,677 100 168 1,456 50 238 5,379 46 Female 279,687 13,296 4,048 42,394 71,035 2,992 1,349 23,756 629 2,878 111,041 6,26 Building caretakers, helpers, cleaners and launderers Total 12,787 1,100 157 567 1,618 135 40 572 47 41 8,218 29 Male 2,505 19 355 633 - 17 383 - - 878 22 Female 10,282 1,100 138 212 985 135 23 189 47 41 7,340 7 Mesenegers, porters, doorkeepers and related workers
Male 20,157 1,305 295 4,024 6,677 100 168 1,456 50 238 5,379 466 Female 279,687 13,296 4,048 42,394 71,035 2,992 1,349 23,756 629 2,878 111,041 6,269 Building caretakers, helpers, cleaners and launderers 12,787 1,100 157 567 1,618 135 40 572 47 41 8,218 29. Male 2,505 - 19 355 633 - 17 383 - - 878 220 Female 10,282 1,100 138 212 985 135 23 189 47 41 8,218 29. Messengers, porters, doorkeepers and related workers - 10,282 11,009 44,577 61,943 118,453 16,603 4,319 61,535 2,086 5,658 95,716 9,21 Male 289,983 13,525 11,049 41,856 86,750 8,468 2,514 38,565 1,621 4,6
Female 279,687 13,296 4,048 42,394 71,035 2,992 1,349 23,756 629 2,878 111,041 6,266 Building caretakers, helpers, cleaners and launderers 12,787 1,100 157 567 1,618 135 40 572 47 41 8,218 29 Male 2,505 - 19 355 633 - 17 383 - - 878 22 Female 10,282 1,100 138 212 985 135 23 189 47 41 7,340 7 Messengers, porters, doorkeepers and related workers 7 61,943 118,453 16,603 4,319 61,535 2,086 5,658 95,716 9,21 Male 289,983 13,525 11,049 41,856 86,750 8,468 2,514 38,565 1,621 4,617 74,421 6,599 Female 119,122 5,478 3,528 20,086 31,704 8,135 1,805 22,969 464 1,042 21,295 2,618
Building caretakers, helpers, cleaners and launderers Total 12,787 1,100 157 567 1,618 135 40 572 47 41 8,218 29 Male 2,505 - 19 355 633 - 17 383 - - 878 22 Female 10,282 1,100 138 212 985 135 23 189 47 41 7,340 7 Messengers, porters, doorkeepers and related workers - 61,943 118,453 16,603 4,319 61,535 2,086 5,658 95,716 9,21 Male 289,983 13,525 11,049 41,856 86,750 8,468 2,514 38,565 1,621 4,617 74,421 6,599 Female 119,122 5,478 3,528 20,086 31,704 8,135 1,805 22,969 464 1,042 21,295 2,618 Garbage collectors and related labourers - - 694 - - 923 Female 6,825
Total 12,787 1,100 157 567 1,618 135 40 572 47 41 8,218 29 Male 2,505 - 19 355 633 - 17 383 - - 878 22 Female 10,282 1,100 138 212 985 135 23 189 47 41 7,340 77 Messengers, porters, doorkeepers and related workers 409,105 19,004 14,577 61,943 118,453 16,603 4,319 61,535 2,086 5,658 95,716 9,21 Male 289,983 13,525 11,049 41,856 86,750 8,468 2,514 38,565 1,621 4,617 74,421 6,599 Female 119,122 5,478 3,528 20,086 31,704 8,135 1,805 22,969 464 1,042 21,295 2,618 Garbage collectors and related labourers 10,186 1,259 121 1,663 1,702 211 21 1,682 12 - 3,514
Male 2,505 - 19 355 633 - 17 383 - - 878 222 Female 10,282 1,100 138 212 985 135 23 189 47 41 7,340 77 Messengers, porters, doorkeepers and related workers - 409,105 19,004 14,577 61,943 118,453 16,603 4,319 61,535 2,086 5,658 95,716 9,21 Male 289,983 13,525 11,049 41,856 86,750 8,468 2,514 38,565 1,621 4,617 74,421 6,594 Female 119,122 5,478 3,528 20,086 31,704 8,135 1,805 22,969 464 1,042 21,295 2,618 Garbage collectors and related labourers - - 694 - - 923 - - 694 - 923 - - 923 - - 923 - - 923 - - 923 - - 923 -
Female10,2821,1001382129851352318947417,34077Messengers, porters, doorkeepers and related workersTotal409,10519,00414,57761,943118,45316,6034,31961,5352,0865,65895,7169,21Male289,98313,52511,04941,85686,7508,4682,51438,5651,6214,61774,4216,594Female119,1225,4783,52820,08631,7048,1351,80522,9694641,04221,2952,618Garbage collectors and related labourersTotal10,1861,2591211,6631,702211211,68212-3,514Male3,361374121815433694923Female6,825884-8481,2702112198812-2,591AGRICULTURAL, FISHERY AND RELATED LABOURERSTotal109,0461,3644,38540,23729,9657,9273,04118,0876922072,24689
Messengers, porters, doorkeepers and related workers Total 409,105 19,004 14,577 61,943 118,453 16,603 4,319 61,535 2,086 5,658 95,716 9,21 Male 289,983 13,525 11,049 41,856 86,750 8,468 2,514 38,565 1,621 4,617 74,421 6,599 Female 119,122 5,478 3,528 20,086 31,704 8,135 1,805 22,969 464 1,042 21,295 2,619 Garbage collectors and related labourers 74 121 1,663 1,702 211 21 1,682 12 - 3,514 Male 3,361 374 121 815 433 - - 694 - - 923 Female 6,825 884 - 848 1,270 211 21 988 12 - 2,591 AGRICULTURAL, FISHERY AND RELATED LABOURERS Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 <td< td=""></td<>
Total 409,105 19,004 14,577 61,943 118,453 16,603 4,319 61,535 2,086 5,658 95,716 9,21 Male 289,983 13,525 11,049 41,856 86,750 8,468 2,514 38,565 1,621 4,617 74,421 6,599 Female 119,122 5,478 3,528 20,086 31,704 8,135 1,805 22,969 464 1,042 21,295 2,619 Garbage collectors and related labourers 1 10,186 1,259 121 1,663 1,702 211 21 1,682 12 - 3,514 Male 3,361 374 121 815 433 - - 694 - - 923 Female 6,825 884 - 848 1,270 211 21 988 12 - 2,591 AGRICULTURAL, FISHERY AND RELATED LABOURERS Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 894<
Male 289,983 13,525 11,049 41,856 86,750 8,468 2,514 38,565 1,621 4,617 74,421 6,599 Female 119,122 5,478 3,528 20,086 31,704 8,135 1,805 22,969 464 1,042 21,295 2,619 Garbage collectors and related labourers Total 10,186 1,259 121 1,663 1,702 211 21 1,682 12 - 3,514 Male 3,361 374 121 815 433 - - 694 - - 923 Female 6,825 884 - 848 1,270 211 21 988 12 - 2,591 AGRICULTURAL, FISHERY AND RELATED LABOURERS Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 89
Female 119,122 5,478 3,528 20,086 31,704 8,135 1,805 22,969 464 1,042 21,295 2,618 Garbage collectors and related labourers 10,186 1,259 121 1,663 1,702 211 21 1,682 12 - 3,514 Male 3,361 374 121 815 433 - - 694 - - 923 Female 6,825 884 - 848 1,270 211 21 988 12 - 2,591 AGRICULTURAL, FISHERY AND RELATED LABOURERS Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 894
Garbage collectors and related labourers Total 10,186 1,259 121 1,663 1,702 211 21 1,682 12 - 3,514 Male 3,361 374 121 815 433 - - 694 - - 923 Female 6,825 884 - 848 1,270 211 21 988 12 - 2,591 AGRICULTURAL, FISHERY AND RELATED LABOURERS Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 894
Total 10,186 1,259 121 1,663 1,702 211 21 1,682 12 - 3,514 Male 3,361 374 121 815 433 - - 694 - - 923 Female 6,825 884 - 848 1,270 211 21 988 12 - 2,591 AGRICULTURAL, FISHERY AND RELATED LABOURERS Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 894
Male 3,361 374 121 815 433 - - 694 - - 923 Female 6,825 884 - 848 1,270 211 21 988 12 - 2,591 AGRICULTURAL, FISHERY AND RELATED LABOURERS - 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 89
Female 6,825 884 - 848 1,270 211 21 988 12 - 2,591 AGRICULTURAL, FISHERY AND RELATED LABOURERS Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 89
AGRICULTURAL, FISHERY AND RELATED LABOURERS Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 89
Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 89
Male 54,233 1,038 2,523 19,177 13,342 4,281 1,515 10,256 315 207 993 58
Female 54,813 326 1,862 21,060 16,624 3,645 1,525 7,830 376 - 1,253 312
Agricultural, fishery and related labourers
Total 109,046 1,364 4,385 40,237 29,965 7,927 3,041 18,087 692 207 2,246 89
Male 54,233 1,038 2,523 19,177 13,342 4,281 1,515 10,256 315 207 993 58
Female 54,813 326 1,862 21,060 16,624 3,645 1,525 7,830 376 - 1,253 312
LABOURERS IN MINING, MANUFACTURING AND TRANSPORT
Total 188,479 15,628 2,172 34,129 66,866 4,810 1,523 21,105 747 876 37,099 3,524
Male 151,547 11,089 2,039 25,371 53,250 4,677 1,470 18,727 519 697 31,099 2,600
Female 36,933 4,540 133 8,758 13,616 133 53 2,377 229 179 6,000 919
Mining and construction labourers
Total 77,671 7,134 1,014 15,011 22,537 1,983 454 9,589 500 742 17,141 1,56
Male 55,828 4,939 941 9,088 16,089 1,850 402 7,490 271 562 13,220 97
Female 21,843 2,195 74 5,923 6,448 133 53 2,099 229 179 3,922 58
Manufacturing labourers
Total 36,787 4,660 327 4,313 19,653 182 60 1,029 - 88 5,885 58
Male 21,849 2,316 268 1,575 12,540 182 60 752 - 88 3,807 26
Female 14,938 2,344 59 2,739 7,113 278 2,078 32

Minor Occupational						Re	gion					
Group and Sex	Title	T :			0		Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Transport labourers and frieg	ht handlers											
Total	74,021	3,834	830	14,805	24,676	2,645	1,009	10,486	248	46	14,073	1,370
Male	73,870	3,834	830	14,708	24,621	2,645	1,009	10,486	248	46	14,073	1,370
Female	151	-	-	96	55	-	-	-	-	-	-	-
NOT STATED												
Total	1,209	341	-	464	35	-	-	299	-	-	-	69
Male	470	341	-	59	-	-	-	-	-	-	-	69
Female	740	-	-	405	35	-	-	299	-	-	-	-

	1	Soundry	Total:	2011		Λ.							
Major Industrial Divisions		40.44		00.04			e Group	40.44	4 - 40			00.04	
and Sex	All Ages	10-14	15-19	20-24	25-29	30-34	35-39	40-44	45-49	50-54	55-59	60-64	65+
Total Employed Population													
Total					1,009,623								
Male	2,928,953	,	192,665	,	,	,	363,332	,	,	,	92,826	,	106,875
Female	2,210,879	,	236,852	393,182	429,821	279,731	293,418	140,556	129,439	93,735	60,074	35,346	45,527
Agriculture, Hunting, Fore													
Total	486,402	,	48,472	40,869	60,222	52,660	54,124	40,900	34,189	29,615	21,853	22,428	35,949
Male	311,909	30,922	32,345	23,452	34,204	32,892	31,401	30,471	19,774	17,957	14,609	16,169	27,713
Female	174,493	14,199	16,128	17,417	26,018	19,768	22,723	10,428	14,415	11,659	7,244	6,259	8,236
Mining & Quarrying													
Total	18,178	80	655	2,260	4,954	2,317	3,493	804	1,690	351	710	351	513
Male	15,790	80	585	1,979	4,662	1,324	2,842	804	1,690	351	609	351	513
Female	2,388	-	70	281	292	993	651	-	-	-	101	-	-
Manufacturing													
Total	675,481	8,813	49,080	98,926	125,082	99,063	79,293	53,871	50,308	36,034	27,801	18,227	28,983
Male	370,099	4,774	25,023	54,646	75,694	56,092	41,607	28,790	26,113	16,305	13,650	10,670	16,734
Female	305,382	4,039	24,057	44,280	49,388	42,971	37,686	25,082	24,194	19,729	14,151	7,557	12,249
Electricity, Gas and Wate	r Supply												
Total	59,456	1,641	4,453	10,571	10,351	8,058	5,900	8,106	4,232	3,781	1,368	538	458
Male	38,832	319	1,723	5,997	8,098	5,607	4,587	5,019	3,077	2,617	875	538	376
Female	20,624	1,322	2,730	4,574	2,254	2,451	1,313	3,087	1,155	1,164	492	-	83
Construction													
Total	380,581	582	21,420	74,321	89,934	61,792	48,901	30,668	21,604	14,074	6,696	4,999	5,590
Male	318,433	217	13,439	57,719	74,691	53,600	42,784	27,248	19,672	11,932	6,696	4,845	5,590
Female	62,148	365	7,981	16,603	15,244	8,192	6,117	3,420	1,932	2,142	-	154	-
Whole Sale and Retail Tra	ade												
Total	1,099,758	26,987	81,601	155,120	208,469	165,770	155,801	97,004	70,435	46,647	30,384	25,481	36,059
Male	580,722	15,379	47,222	79,444	106,693	94,329	74,078	55,223	34,734	23,080	12,492	15,586	22,463
Female	519,036	11,608	34,379	75,676	101,776	71,441	81,723	41,781	35,702	23,566	17,892	9,895	13,596
Hotels and Restaurants													
Total	408,076	9,167	49,027	76,120	75,024	50,680	50,594	27,338	23,970	14,503	9,653	9,961	12,039
Male	105,545	2,081	15,578	18,385	20,255	14,462	9,349	7,401	3,460	4,766	2,037	3,149	4,623
Female	302,531	7,086	33,449	57,735	54,770	36,218	41,245	19,937	20,510	9,737	7,616	6,812	7,416
Transport, Storage a	and Commu	nications											
Total	242,410	1,772	12,943	36,814	61,180	41,396	30,598	19,548	15,221	9,998	5,590	3,056	4,293
Male	226,779	1,772	12,547	34,461	57,198	39,861	27,944	18,025	13,474	8,557	5,590	3,056	4,293
Female	15,631	-	396	2,353	3,982	1,534	2,654	1,523	1,747	1,441	-	-	-

Table 5.7 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Age Group and MajorIndustrial Divisions - Country Total: 2011

Major Industrial Divisions						Age	e Group						
, and Sex	All Ages	10-14	15-19	20-24	25-29	30-34	35-39	40-44	45-49	50-54	55-59	60-64	65+
Financial Intermediation													
Total	91,143	-	1,233	11,840	22,820	14,382	15,591	7,870	8,703	4,415	2,907	1,004	380
Male	51,635	-	69	3,275	10,111	8,397	8,300	7,180	7,352	3,289	2,279	1,004	380
Female	39,508	-	1,164	8,564	12,709	5,985	7,291	689	1,351	1,126	628	-	-
Real Estate, Renting and	d Business A	Activities											
Total	116,708	445	4,296	22,590	23,182	16,150	18,604	9,245	9,520	5,326	3,783	1,543	2,024
Male	76,784	445	2,860	12,951	13,338	10,845	11,457	7,084	6,983	4,998	2,621	1,256	1,945
Female	39,924	-	1,436	9,639	9,844	5,305	7,147	2,160	2,537	328	1,162	287	79
PUBLIC ADMINISTRATIO	ON AND DE	FENCE											
Total	324,536	-	3,720	40,324	69,015	55,029	59,047	41,517	26,208	15,150	8,073	1,614	4,838
Male	222,301	-	1,078	21,250	45,655	35,552	40,443	34,875	20,178	11,567	6,128	1,410	4,165
Female	102,234	-	2,642	19,073	23,361	19,477	18,604	6,642	6,030	3,583	1,945	204	673
Education													
Total	346,352	48	7,376	55,537	88,633	47,959	51,318	29,126	23,948	22,454	12,661	4,257	3,036
Male	201,522	-	3,081	22,117	49,993	28,385	28,983	19,897	16,853	14,983	10,369	4,113	2,749
Female	144,831	48	4,295	33,420	38,640	19,574	22,335	9,229	7,095	7,471	2,292	144	287
Health and Social Work													
Total	161,675	-	2,711	30,717	39,286	26,304	21,751	11,625	10,947	5,972	7,171	4,220	971
Male	80,565	-	580	9,521	19,148	13,551	7,987	7,650	8,328	4,169	4,888	3,771	971
Female	81,110	-	2,131	21,196	20,138	12,753	13,764	3,975	2,619	1,803	2,283	449	-
Other Community, Social a	and Persona	al Service	Activities										
Total	459,078	28,655	58,641	83,590	91,793	55,811	47,796	23,477	20,994	16,985	10,629	8,439	12,268
Male	276,244	14,161	32,554	51,451	52,867	36,025	27,604	14,977	13,044	10,555	7,789	5,262	9,953
Female	182,834	14,493	26,087	32,139	38,926	19,786	20,191	8,500	7,950	6,431	2,839	3,177	2,315
Private Households with	Employed	Persons											
Total	230,362	21,501	82,691	51,113	32,502	13,555	8,858	5,271	2,703	5,081	2,056	1,623	3,409
Male	24,981	1,463	3,981	4,083	3,907	2,250	464	1,459	809	1,596	938	1,215	2,817
Female	205,381	20,038	78,710	47,030	28,595	11,305	8,394	3,811	1,894	3,485	1,119	408	593
Extra-Territorial Orgar	nizations an	d NGOs											
Total	37,425	-	1,198	5,489	6,384	8,312	4,999	3,587	1,648	2,171	1,566	478	1,591
Male	25,493	-	-	2,738	2,880	6,335	3,478	3,296	1,340	2,101	1,256	478	1,591
Female	11,931	-	1,198	2,751	3,504	1,977	1,521	291	308	70	310	-	-
Not Stated													
Total	2,188	-	-	1,278	791	59	60	-	-	-	-	-	-
Male	1,296	-	-	827	410	59	-	-	-	-	-	-	-
Female	892	-	-	451	380	-	60	-	-	-	-	-	-

Majar Industrial Distance						Re	gion					
Major Industrial Divisions and Sex	All						Benishangul-					
and Sex	Persons	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Total Employed Population												
Total	5,139,831	321,569	71,536	921,246	1,609,473	167,435	53,221	680,560	33,342	48,450	1,148,974	84,025
Male	2,928,953	173,501	45,023	505,459	930,772	98,864	30,493	387,610	16,628	25,806	667,973	46,825
Female	2,210,879	148,068	26,512	415,787	678,701	68,571	22,729	292,950	16,715	22,644	481,001	37,200
Agriculture, Hunting Forestry & Fishing												
Total	486,402	20,487	20,521	132,819	171,136	34,385	13,788	67,642	8,370	1,007	13,872	2,374
Male	311,909	12,995	14,957	88,804	104,319	22,924	9,000	43,593	4,812	826	8,277	1,403
Female	174,493	7,493	5,564	44,015	66,817	11,461	4,788	24,049	3,558	182	5,596	971
Mining & Quarrying												
Total	18,178	1,871	1,027	3,891	4,188	179	182	1,327	129	1,070	3,882	432
Male	15,790	1,608	1,027	3,448	3,900	179	182	1,267	99	707	2,941	432
Female	2,388	263	-	443	288	-	-	59	31	363	941	-
Manufacturing												
Total	675,481	48,452	8,577	109,738	245,362	8,925	3,066	68,660	2,163	6,449	164,482	9,606
Male	370,099	24,349	4,648	60,241	132,619	4,914	1,215	28,153	1,028	3,502	104,772	4,658
Female	305,382	24,103	3,929	49,497	112,742	4,011	1,851	40,507	1,136	2,947	59,710	4,948
Electricity, Gas and Water Supply												
Total	59,456	3,457	2,036	9,131	19,229	4,905	730	3,692	332	373	14,018	1,551
Male	38,832	2,692	1,701	5,156	12,944	1,065	598	2,531	284	251	10,682	928
Female	20,624	765	335	3,975	6,285	3,840	131	1,161	48	122	3,336	624
Construction												
Total	380,581	28,593	5,645	57,826	108,345	7,571	3,002	48,403	1,542	2,987	108,645	8,021
Male	318,433	23,478	4,683	43,665	93,928	7,284	2,697	42,894	1,138	2,590	89,610	6,465
Female	62,148	5,115	962	14,161	14,417	287	305	5,509	404	397	19,035	1,556
Wholesale and Retail Trade												
Total	1,099,758	72,440	12,562	183,331	361,317	44,906	5,747	132,204	7,263	14,576	239,267	26,145
Male	580,722	35,622	6,744	86,599	189,849	17,262	2,566	76,778	1,982	5,651	146,720	10,949
Female	519,036	36,817	5,818	96,732	171,468	27,643	3,181	55,426	5,281	8,925	92,548	15,196
Hotels and Restaurants												
Total	408,076	27,482	4,427	79,505	135,683	6,78	31 5,43	67,49	2 3,52	.6 1,2	51 73,30	0 3,190
Male	105,545	3,595	601	13,527	34,593	3 1,83	88 46	3 22,06	3 38	32 49	94 27,01	16 973
Female	302,531	23,887	3,826	65,977	101,091	4,94	3 4,97	7 45,42	9 3,14	4 7!	57 46,28	34 2,217

Major Industrial Divisions						Regi						
and Sex	All						Benishangul					Dire
	Persons	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dawa
ransport, Storage and Communications												
Total	242,410		1,195		72,106			21,452	701			6,30
Male	226,779	13,080	1,195		70,452			20,774	667			5,94
Female	15,631	1,442		- 2,450	1,654	370	118	678	35)	- 8,523	36
inancial Intermediation												
Total	91,143	9,545	471	10,610	32,909	1,645	943	10,059	493		3 22,508	1,67
Male	51,635	5,465	458	6,897	18,307	1,370	526	6,286	348	8 87	7 10,581	1,31
Female	39,508	4,080	13	3,713	14,602	275	417	3,773	145	5 201	l 11,927	36
Real Estate, Renting and Business Activitie	2S											
Total	116,708	3,982	962	2 15,486	27,678	1,089	849	15,898	393	3 572	48,175	1,62
Male	76,784	1,972	709	9,988	16,434	894	740	10,595	292	2 355	5 33,703	1,10
Female	39,924	2,011	253	5,499	11,243	195	109	5,303	101	217	7 14,472	52
ublic administration and defense												
Total	324,536	20,544	4,269	64,545	95,175	16,919	5,321	49,938	4,192	4,468	3 55,106	4,05
Male	222,301	13,782	2,383	47,150	66,388	14,977	3,727	34,769	2,658	3 2,817	30,759	2,89
Female	102,234	6,762	1,885	5 17,395	28,787	1,942	1,594	15,169	1,534	1,650	24,348	1,16
ducation												
Total	346,352	23,384	2,284	64,073	103,827	8,815	4,314	62,262	1,678	4,658	3 66,991	4,06
Male	201,522	11,138	1,771	35,247	66,183	6,778	2,887	35,911	1,172	2,380	35,147	2,90
Female	144,831	12,246	513	8 28,825	37,644	2,038	1,427	26,352	506	5 2,278	3 31,844	1,15
lealth and Social Work												
Total	161,675	11,191	3,086	24,170	45,905	6,024	2,934	23,984	894	2,528	38,753	2,20
Male	80,565	6,024	1,938	11,581	24,876	3,779	1,498	12,418	669	9 1,016	5 15,719	1,04
Female	81,110		1,148		21,028			11,566	225			1,15
Other Community, Social and Personal Se												
Total	459,078	24,267	2,454	76,847	136,213	13,199	4,750	88,303	1,078	3 4,388	3 100,719	6,86
Male	276,244	14,842	1,816					45,222	730			4,99
Female	182,834	9,424	638		49,664			43,081	348			1,86
rivate Households with Employed Person		.,		-,,		-,	,			,		,
Total	230,362	8,910	1,672	42,292	45,463	1,826	618	14,848	104	1,663	3 107,665	5,30
Male	24,981	966	66		5,057			1,561		4.05		44
Female	205,381	7,943	1,606		40,406			13,287	104			4,86

Major Industrial Divisions						Reg	ion					
and Sex	All Persons	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Extra-Territorial Organizations and NGOs												
Total	37,425	2,102	346	7,532	4,383	3,091	I 72	4,096	46	0 42	9 14,369	543
Male	25,493	1,552	326	3,731	3,915	5 1,844	4 51	2,795	34	3 31	3 10,315	309
Female	11,931	550	21	3,801	468	3 1,247	7 21	1,302	11	7 11	6 4,054	234
Not Stated												
Total	2,188	341	-	926	553	}		299		-		69
Male	1,296	341	-	429	457	1		-		-		69
Female	892	-	-	497	95)		299		-		-

									Major Inc	lustrial Divis	ions							
Major Industrial Divisions and Sex	Total Employed Population	Agriculture, Hunting, Forestry & Fishing	Mining & Quarrying	Manufa- cturing	Electricity, Gas and Water Supply	Constru ction	Whole Sale and Retail Trade	Hotels And Restaur ants	Transport, Storage and Communi cations	Financial Intermedi ation	Real Estate, Renting and Business Activities	Public Administ ration and Defence	Education	Health and Social Work	Other Community , Social and Personal Service Activities	Private Households with Employed Persons	Extra- Territorial Organizati ons and NGOs	Not Stated
TOTAL MAJOR TO	OWNS									•	•							
Total	1,965,425	67,947	8,976	267,512	26,004	188,443	410,650	131,567	124,465	35,619	65,479	109,552	129,198	65,633	161,154	151,778	21,074	35
Male	1,134,199	44,121	7,319	160,031	17,268	154,342	234,925	42,517	112,354	18,552	45,878	67,571	69,694	27,805	99,373	17,595	14,702	12
Female MEKELE TOWN	831,226	23,826	1,657	107,481	8,735	34,101	175,726	89,049	12,111	17,067	19,601	41,980	59,504	37,828	61,781	134,183	6,373	22
Total	87,546	986	1,051	13,713	1,472	9,643	16,697	6,889	4,749	3,596	1,882	5,522	6,525	2,636	6,603	4,674	910	
Male	49,393	802	788	7,522	1,109	8,636	9,538	1,326	4,270	2,004	1,423	3,847	3,248	1,034	2,529	801	515	
Female ASAYITA TOWN	38,153	184	263	6,191	363	1,007	7,158	5,563	479	1,592	459	1,675	3,277	1,603	4,074	3,873	394	
Total	6,472	781	20	523	130	1,114	1,129	648	231	77	53	731	394	163	211	247	21	
Male	3,810	657	20	246	111	935	433	179	231	64	31	413	249	103	130	9	-	
Female	2,663	124	-	277	19	179	696	470	-	13	22	317	146	61	81	238	21	
GONDER TOWN																		
Total	82,743	24,296	644	8,641	3,480	6,736	9,154	5,934	2,860	475	907	3,997	4,923	1,875	5,529	3,216	76	
Male	45,705	16,660	644	3,218	313	4,846	5,862	849	2,708	245	659	2,538	2,791	538	3,635	123	76	
Female	37,038	7,635	-	5,423	3,167	1,889	3,293	5,085	152	230	248	1,458	2,133	1,338	1,894	3,093	-	
DESSIE TOWN																		
Total	46,017	1,581	52		708	2,859	9,927	2,970	2,254	616	1,008	2,921	3,965	2,299	4,142	4,152		12
Male	25,986	815	52		600	2,347	5,192	686	1,977	337	701	2,207	2,423	956	3,077	189	700	
Female	20,032	766	-	1,832	107	512	4,735	2,284	277	279	307	713	1,541	1,344	1,065	3,963	178	12
BAHIR DAR TOWN																		
Total	89,993	4,195	702	,	1,228	13,174	13,884	8,324	4,964	1,076	2,937	5,529	6,788	2,579	6,281	7,621	894	5
Male	46,674	3,104	702	5,040	971	8,819	7,436	1,548	4,154	532	2,094	3,488	3,113	996	3,786	168	664	5
Female	43,319	1,091	-	4,718	257	4,355	6,448	6,777	810	545	843	2,041	3,675	1,584	2,495	7,453	230	
BISHOFTU TOWN		4 000		0.047			5 000	0.074	0 (00	F 40	700	0.000	4 04 4	010	0.4/7	4 004	070	
Total	40,673	4,890	38		230	4,557	5,999	2,271	2,608	548	739	3,808	1,814	813	2,467	1,331	278	3
Male	22,604	2,838	38	3,940	230	3,432	2,723	651	2,444	278	358	2,180	1,240	328	1,586	100	238	~
Female	18,069	2,053	-	4,308	-	1,125	3,275	1,620	164	270	381	1,628	574	485	881	1,230	40	3
ADAMA TOWN	101 170	4 101	00	17 704	(00	10.000	24.221	0.000	0.400	010	1 740	F 000	(000	2 (02	F 707	4 700		
Total	101,178	4,181	88		602	10,898		8,293	8,493	813	1,742	5,008	6,008	2,603	5,787	4,720	-	
Male	60,623	2,289	88	9,452	383	9,939		3,493	7,989	497	1,563	3,500	3,633	1,352	3,773	488	-	
Female	40,555	1,892	-	8,268	219	959	12,038	4,800	505	316	179	1,508	2,375	1,250	2,013	4,232	-	

Table 5.9 Currently Employed Population of Major Towns Aged Ten Years and Above by Sex and Major Industrial Divisions: 2011

									Maior Ind	ustrial Divis	ions							
								r			Real			1	Other	1	1	
Major Industrial Divisions and Sex	Total Employed Population	Agriculture, Hunting, Forestry & Fishing	Mining & Quarrying		Electricity, Gas and Water Supply	Constru ction	Whole Sale and Retail Trade	Hotels And Restaur ants	Transport, Storage and Communi cations	Financial Intermedi ation	Real Estate, Renting and Business Activities	Public Administ ration and Defence	Education	Health and Social Work	Other Community, Social and Personal Service Activities	Private Households with Employed Persons	Extra-Territorial Organizations and NGOs	Not Stated
JIMMA TOWN							•		•						•	•	•	
Total	55,260	988	286	5,156	456	5,158	11,245	3,989	2,865	1,283	1,315	2,054	10,642	2,844	3,190	3,346	444	-
Male	33,297	838	286	3,677	361	3,970	6,250	1,551	2,621	826	924	1,554	5,881	1,540	2,618	203	197	-
Female	21,963	150	-	1,479	95	1,187	4,994	2,439	244	457	390	499	4,762	1,304	573	3,144	247	-
SHASHEMENE T	OWN																	
Total	54,907	5,083	300	6,093	61	2,454	15,602	6,044	3,257	213	1,195	1,686	2,000	1,255	6,661	2,912	30	60
Male	32,752	3,433	300	3,417	46	2,306	8,444	1,212	3,257	123	896	1,281	1,021	705	5,860	422	30	-
Female	22,155	1,650	-	2,676	15	149	7,158	4,832	-	91	299	405	979	550	801	2,490	-	60
JIJIGA TOWN																		
Total	29,089	741	-	1,365	443	3,196	7,550	1,845	2,420	560	529	3,518	1,812	1,096	2,625	908	481	-
Male	20,272	497	-	954	351	2,910	4,740	684	2,208	427	334	2,973	1,291	490	2,017	100	297	-
Female	8,817	244	-	411	92	287	2,811	1,161	213	133	195	545	521	606	608	808	184	-
ASOSA TOWN																		
Total	13,319	699	-	520	177	968	2,144	952	664	217	260	2,726	1,514	755	1,408	295	21	-
Male	7,830	469	-	240	107	801	1,030	204	643	217	151	1,704	1,049	322	872	23	-	-
Female HAWASSA TOWN	5,489	230		280	69	167	1,113	749	21	-	109	1,022	465	433	536	273	21	
Total	64,568	1,699	317	8,512	919	7,049	11,751	4,550	3,428	1,319	2,185	5,191	6,293	2,580	3,844	3,693	1,240	-
Male	37,212	934	258	5,050	693	5,980	6,737	1,504	3,100	811	1,329	3,342	2,936	1,203	2,265	369	705	-
Female	27,356	765	59	3,463	226	1,068	5,014	3,046	328	508	857	1,850	3,357	1,377	1,579	3,324	535	-
GAMBELLA TOW	N																	
Total	12,209	574	95	1,170	156	983	1,360	1,115	404	356	356	3,228	806	647	441	35	460	-
Male	7,437	280	64	619	133	755	1,037	148	369	211	256	2,077	385	457	280	-	343	-
Female	4,772	294	31	551	23	229	323	966	35	145	101	1,151	420	190	161	35	117	-
HARAR TOWN																		
Total	48,450	1,007	1,070	6,449	373	2,987	14,576	1,251	1,744	288	572	4,468	4,658	2,528	4,388	1,663	429	-
Male	25,806	826	707	3,502	251	2,590	5,651	494	1,744	87	355	2,817	2,380	1,016	2,939	135	313	-
Female	22,644	182	363	2,947	122	397	8,925	757	-	201	217	1,650	2,278	1,512	1,449	1,528	116	-
ADDIS ABABA CI	ΤY																	
Total	1,148,974	13,872	3,882	164,482	14,018	108,645	239,267	73,300	77,222	22,508	48,175	55,106	66,991	38,753	100,719	107,665	14,369	-
Male	667,973	8,277	2,941	104,772	10,682	89,610	146,720	27,016	68,699	10,581	33,703	30,759	35,147	15,719	59,007	14,026	10,315	-
Female	481,001	5,596	941	59,710	3,336	19,035	92,548	46,284	8,523	11,927	14,472	24,348	31,844	23,034	41,712	93,639	4,054	-
DIRE DAWA TOW																		
Total	84,025	2,374	432	9,606	1,551	8,021	26,145	3,190	6,303	1,674	1,623	4,059	4,066	2,207	6,860	5,302	543	69
Male	46,825	1,403	432	4,658	928	6,465	10,949	973	5,940	1,312	1,101	2,890	2,908	1,048	4,999	440	309	69
Female	37,200	971	-	4,948	624	1,556	15,196	2,217	363	362	522	1,169	1,158	1,159	1,861	4,861	234	-

Minor Industrial Divisions	-					R	egion					
and Sex							Benishangul-				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
TOTAL EMPLOYED POPULATION												
Total	5,139,831	321,569	71,536	921,246	1,609,473	167,435	53,221	680,560	33,342	48,450	1,148,974	84,025
Male	2,928,953	173,501	45,023	505,459	930,772	98,864	30,493	387,610	16,628	25,806	667,973	46,825
Female	2,210,879				678,701	68,571	22,729	292,950	16,715	22,644	481,001	37,200
MAJOR GROUP 1:AGRICULTURE: -	HUNTING ,	FOREST	RY AND	FISHING								
Total	486,402	20,487	20,521	132,819	171,136	34,385	13,788	67,642	8,370	1,007	13,872	2,374
Male	311,909	12,995	14,957	88,804	104,319	22,924	9,000	43,593	4,812	826	8,277	1,403
Female	174,493	7,493		44,015	66,817	11,461	4,788	24,049	3,558	182	5,596	971
AGRICULTURE, HUNTING AND REL	ATED SER\	ICE ACT	IVITIES									
Total	460,349	18,062	17,192	130,247	160,981	31,641	12,638	65,636	7,629	937	13,252	2,134
Male	298,593	12,318	11,628	87,676	100,323	21,402	8,323	41,786	4,781	756	8,277	1,323
Female	161,756	5,744	5,564	42,571	60,658	10,239	4,315	23,850	2,848	182	4,975	811
Growing of crops; market gardening; h	orticulture											
Total	218,150	11,147	15,147	36,381	81,524	18,067	8,241	35,201	3,945	405	7,701	390
Male	156,803	8,463	10,185	31,566	58,084	11,833	5,348	23,316	2,900	351	4,368	390
Female	61,346	2,684	4,962	4,815	23,440	6,234	2,893	11,885	1,045	55	3,334	-
Farming of animals												
Total	106,671	5,796	738	24,042	45,892	9,932	412	13,837	395	129	4,033	1,466
Male	54,721	2,736	554	13,136	21,710	5,927	253	7,016	145	129	2,391	725
Female	51,949	3,061	184	10,905	24,182	4,005	158	6,822	250	-	1,642	741
Growing of crops combined with farmin	ng of animals	s (mixed fa	arming)									
Total	107,765	1,120	36	65,242	19,370	3,643	3,278	10,885	2,866	312	944	69
Male	70,483	1,120	27	39,188	13,485	3,643	2,074	8,178	1,570	185	944	69
Female	37,282	· -	9	26,054	5,885	-	1,204	2,707	1,295	127	-	-
Agricultural and animal husbandry service		s, except v	veterinar				,	,	,			
Total	27,109	-	1,135	4,508	13,750	-	707	5,713	424	91	574	208
Male	16,512	-	863	3,712	7,044	-	647	3,277	165	91	574	139
Female	10,597	-	272	796	6,705	-	60	2,436	258	-	-	69
Hunting, trapping and game propagation		related se			0,100			_,				
Total	655	-	136	73	446	-	-	-	-	-	-	-
Male	73	-	-	73	-	-	-	-	-	-	-	-
Female	582	-	136	-	446	-	-	-	-	-	-	-
FORESTRY, LOGGING AND RELATE												
Total	24,866	2,425	3,329	2,496	9,674	2,743	1,150	1,687	742	70	310	240
								,				-
Male	12,439	676	3,329	1,052	3,515	1,521	676	1,487	32	70	-	80

Major Industrial Divisions						Re	gion					
and Sex							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Forestry, logging and related s												
Total	24,866	2,425	3,329	2,496	9,674	2,743	1,150	1,687	742	70	310	240
Male	12,439	676	3,329	1,052	3,515	1,521	676	1,487	32	70	-	80
Female	12,427	1,748	-	1,444	6,159	1,222	474	200	710	-	310	160
FISHING												
Total	1,188	-	-	76	482	-	-	320	-	-	310	-
Male	877	-	-	76	482	-	-	320	-	-	-	-
Female	310	-	-	-	-	-	-	-	-	-	310	-
Fishing, operation of fish hatcl		sh farms; s	service a	activities in		Fishing						
Total	1,188	-	-	76	482	-	-	320	-	-	310	-
Male	877	-	-	76	482	-	-	320	-	-	-	-
Female	310	-	-	-	-	-	-	-	-	-	310	-
MAJOR GROUP 2: MINING A	AND QUARR	YING										
Total	18,178	1,871	1,027	3,891	4,188	179	182	1,327	129	1,070	3,882	432
Male	15,790	1,608	1,027	3,448	3,900	179	182	1,267	99	707	2,941	432
Female	2,388	263	-	443	288	-	-	59	31	363	941	-
MINING OF COAL AND LIGN	IITE; EXTRA	CTION OF	F PEAT									
Total	582	-	542	-	40	-	-	-	-	-	-	-
Male	582	-	542	-	40	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Mining and agglomeration of h	nard coal											
Total	582	-	542	-	40	-	-	-	-	-	-	-
Male	582	-	542	-	40	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Mining and agglomeration of li	ignite											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Extraction and agglomeration	of peat											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
EXTRACTION OF CRUDE	PETROLEU	IM AND N	ATURA	L GAS; SE	RVICE AC	TIVITIES						
Total	1,173	-	-	-	40	-	-	241	-	-	892	-
Male	1,173	-	-	-	40	-	-	241	-	-	892	-
Female	-	-	-	-	-	-	-	-	-	-	-	

Major Industrial Divisions						Re	gion					
and Sex	T ()				. .	o	Benishangul	0.0.0			Addis	Dire
Foto of smale materia	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Hararı	Ababa	Dawa
Extraction of crude petroleum	-	jas			10						007	
Total	337	-	-	-	40	-	-	-	-	-	297	-
Male	337	-	-	-	40	-	-	-	-	-	297	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Service activities incidental to		extraction	excludin	g surveyin	g							
Total	835	-	-	-	-	-	-	241	-	-	595	-
Male	835	-	-	-	-	-	-	241	-	-	595	-
Female		-	-			-	-	-	-	-	-	-
MINING OF GOLD, COPPER			KANIUM	AND THC		-5						
Total	788	788	-	-	-	-	-	-	-	-	-	-
Male	525	525	-	-	-	-	-	-	-	-	-	-
Female	263	263	-	-	-	-	-	-	-	-	-	-
Mining of gold,copper and as v			orium or	es								
Total	788	788	-	-	-	-	-	-	-	-	-	-
Male	525	525	-	-	-	-	-	-	-	-	-	-
Female	263	263	-	-	-	-	-	-	-	-	-	-
MINING OF METAL ORES												
Total	3,672	263	10	651	1,046	179	-	264	31	-	1,228	-
Male	2,641	263	10	651	1,046	179	-	204	-	-	287	-
Female	1,031	-	-	-	-	-	-	59	31	-	941	-
Mining of iron ores												
Total	2,405	263	10	578	185	179	-	264	31	-	896	-
Male	1,706	263	10	578	185	179	-	204	-	-	287	-
Female	699	-	-	-	-	-	-	59	31	-	609	-
Mining of non-ferrous metal or	es, except ur	ranium an	d thoriur	n ores								
Total	1,266	-	-	74	861	-	-	-	-	-	332	-
Male	935	-	-	74	861	-	-	-	-	-	-	-
Female	332	-	-	-	-	-	-	-	-	-	332	-
OTHER MINING AND QUARF	RYING											
Total	11,964	820	475	3,240	3,062	-	182	822	99	1,070	1,762	432
Male	10,869	820	475	2,797	2,774	-	182	822	99	707	1,762	432
Female	1,095	-	-	443	288	-	-	-	-	363	-	-
Quarrying of stone, sand and												
Total	10,748	820	402	2,963	2,832	-	182	760	99	1,070	1,188	432
Male	9,653	820	402	2,520	2,544	-	182	760	99	707	1,188	432
Iviale												

						Re	gion					
Major Industrial Divisions and Sex							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Mining and quarrying n.e.c.												
Total	1,216	-	73	277	230	-	-	62	-	-	574	-
Male	1,216	-	73	277	230	-	-	62	-	-	574	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
MAJOR GROUP 3: MANUFA	CTURING											
Total	675,481	48,452	8,577	109,738	245,362	8,925	3,066	68,660	2,163	6,449	164,482	9,606
Male	370,099	24,349	4,648	60,241	132,619	4,914	1,215	28,153	1,028	3,502	104,772	4,658
Female	305,382	24,103	3,929	49,497	112,742	4,011	1,851	40,507	1,136	2,947	59,710	4,948
MANUFACTURE OF FOOD F	PRODUCTS A	AND BEV	ERAGE	S								
Total	200,757	6,300	6,670	26,353	91,066	5,263	561	25,928	637	3,478	30,424	4,077
Male	73,006	2,049	3,185	11,506	34,366	1,914	251	5,413	205	1,627	11,456	1,036
Female	127,751	4,251	3,486	14,848	56,699	3,349	311	20,515	433	1,852	18,968	3,041
Production, processing and pr		f meat, fis	h, fruit,v	-								
Total	17,357	-	67	1,961	6,776	142	161	1,960	-	264	5,367	659
Male	11,638	-	67	1,218	5,735	-	161	1,740	-	133	2,513	71
Female	5,719	-	-	743	1,041	142	-	220	-	130	2,854	589
Manufacture of dairy products												
Total	3,198	90	-	378	1,047	481	-	328	-	124	574	176
Male	728	-	-	-	-	-	-	154	-	-	574	-
Female	2,470	90	-	378	1,047	481	-	175	-	124	-	176
Manufacture of grain mill prod	ucts, starche	s and star	ch prod	ucts, and p	prepared ar	imal feeds	3					
Total	59,629	4,569	38	9,403	31,662	1,186	37	4,275	204	503	6,488	1,262
Male	34,155	1,965	18	6,288	19,996	608	21	2,612	181	255	1,771	441
Female	25,473	2,604	20	3,115	11,666	578	17	1,664	23	248	4,717	821
Manufacture of other food pro	ducts											
Total	98,313	1,031	6,202	11,513	43,165	3,454	312	18,705	433	1,390	10,490	1,618
Male	14,773	-	2,979	2,193	4,612	1,306	69	776	23	377	2,202	235
Female	83,540	1,031	3,223	9,320	38,552	2,148	243	17,929	410	1,013	8,289	1,383
Manufacture of beverages												
Total	22,261	610	363	3,098	8,416	-	51	659	-	1,197	7,504	362
Male	11,713	84	121	1,806	4,023	-	-	132	-	861	4,396	290
Female	10,549	527	242	1,292	4,393	-	51	527	-	336	3,108	72
MANUFACTURE OF TOBAC	CO AND LOO	CAL BEVE	RAGE	PRODUCT	S							
Total	39,195	1,592	-	6,611	17,349	-	740	9,541	226	-	3,136	-
Male	5,777	561	-	369	3,305	-	-	-	58	-	1,483	-
Female	33,417	1,030	-	6,242	14,043	-	740	9,541	168	-	1,653	-

Major Industrial Divisions						Re	gion					
and Sex							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of tobacco produ												
Total	4,791	-	-	2,667	-	-	-	963	12	-	1,149	-
Male	956	-	-	369	-	-	-	-	12	-	574	-
Female	3,835	-	-	2,298	-	-	-	963	-	-	574	-
Manufacture and Wholesale o	f tela and bo	rde										
Total	6,014	1,592	-	185	3,142	-	-	-	17	-	1,078	-
Male	561	561	-	-	-	-	-	-	-	-	-	-
Female	5,452	1,030	-	185	3,142	-	-	-	17	-	1,078	-
Manufacture and Wholesale o	f teji											
Total	451	-	-	451	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	451	-	-	451	-	-	-	-	-	-	-	-
Manufacture and Wholesale o	f Areke/katike	ela/										
Total	27,939	-	-	3,308	14,207	-	740	8,579	197	-	908	-
Male	4,260	-	-	-	3,305	-	-	-	46	-	908	-
Female	23,679	-	-	3,308	10,902	-	740	8,579	151	-	-	-
MANUFACTURE OF TEXTILE	ES											
Total	149,123	23,451	263	33,300	49,991	505	727	8,258	131	92	30,774	1,631
Male	76,047	9,148	186	14,412	31,723	249	277	2,080	25	92	17,265	589
Female	73,075	14,303	76	18,888	18,268	255	449	6,178	106	-	13,508	1,042
Spinning, weaving and finishir	ng of textiles											
Total	123,113	20,517	186	27,631	43,201	255	337	6,151	25	46	23,755	1,008
Male	66,112	8,268	186	12,063	29,964	-	277	1,440	25	46	13,565	277
Female	57,001	12,248	-	15,569	13,237	255	60	4,711	-	-	10,190	730
Manufacture of other textiles												
Total	19,136	2,197	-	4,382	3,737	249	60	1,920	106	46	5,815	624
Male	9,788	780	-	2,302	1,759	249	-	640	-	46	3,700	312
Female	9,348	1,417	-	2,081	1,978	-	60	1,280	106	-	2,115	312
Manufacture of knitted and cro	ocheted fabrio	cs and arti	icles									
Total	6,873	738	76	1,287	3,053	-	330	187	-	-	1,203	-
Male	148	100	-	48	-	-	-	-	-	-	-	-
Female	6,725	638	76	1,239	3,053	-	330	187	-	-	1,203	-
MANUFACTURE OF WEARI												
Total	38,542	2,737	209	7,394	14,642	494	291	2,012	156	369	9,953	283
Male	31,403	2,296	189	6,477	12,856	494	291	1,767	132	328	6,434	139
Female	7,139	441	21	917	1,786	-		245	25	41	3,519	144
	.,.00			0.1	.,. 00			=.0	_0	••	0,010	

Major Industrial Divisions						Re	gion			-		-
and Sex	T ()				<u> </u>		Benishangul	0.00	<u> </u>		Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of wearing appar	-											
Total	37,950	2,645	196	7,240	14,642	494	291	2,012	156	369	9,621	283
Male	30,904	2,296	176	6,323	12,856	494	291	1,767	132	328	6,102	139
Female	7,046	349	21	917	1,786	-	-	245	25	41	3,519	144
Dressing and dyeing of fur; ma												
Total	592	93	13	154	-	-	-	-	-	-	332	-
Male	499	-	13	154	-	-	-	-	-	-	332	-
Female	93	93	-	-	-	-	-	-	-	-	-	-
TANNING AND DRESSING O	F LEATHER	; MANUF	ACTURE	E OF LUG	GAGE, HAI	NDBAGS,	SADDLER, HA		AND FOOT	WEAR		
Total	21,709	502	-	2,288	4,979	-	-	1,166	-	41	12,573	160
Male	13,414	-	-	1,330	2,013	-	-	1,166	-	41	8,706	160
Female	8,294	502	-	958	2,966	-	-	-	-	-	3,867	-
Tanning and dressing of leat	her; manufac	cture of lug	ggage, h	andbags,s	addler and	Harness						
Total	8,540	-	-	2,288	1,295	-	-	954	-	-	3,843	160
Male	6,029	-	-	1,330	911	-	-	954	-	-	2,674	160
Female	2,511	-	-	958	384	-	-	-	-	-	1,169	-
Manufacture of footwear												
Total	13,169	502	-	-	3,684	-	-	212	-	41	8,730	-
Male	7,386	-	-	-	1,102	-	-	212	-	41	6,032	-
Female	5,783	502	-	-	2,582	-	-	-	-	-	2,699	-
MANUFACTURE OF WOOD / MANUFACTURE OF ARTICLI					ORK, EXC	EPT FURI	NITURE;					
Total	36,488	2,892	-	4,195	9,650	817	198	4,740	270	289	13,116	322
Male	33,151	2,538	-	3,902	8,800	817	198	4,561	208	206	11,599	322
Female	3,338	354	-	293	850	-	-	178	62	83	1,517	
Saw milling and planning of we	,	004		235	000			170	02	00	1,017	
Total	4,469	-	-	214	1,494	-	-	121	216	-	2,425	-
Male	3,413	-	-	214	1,097	-	-	121	154	-	1,827	-
Female	1,057	-	-		398	-	_		62	-	597	-
Manufacture of products of wo		aw and pla	aitina ma	terials in f					02		001	
Total	32,019	2,892	-	3,981	8,155	817	198	4.619	54	289	10,691	322
Male	29,738	2,538	-	3,688	7,703	817	198	4,440	54	206	9,771	322
Female	2,281	354	-	293	452	-	-	178		83	920	
MANUFACTURE OF PAPER			CTS	200	-102	_	_	170		00	520	_
Total	2,702	93	59	_	305	_	_	896	47	_	1,303	-
Male	2,702	93 93	59	-	193	-	-	686	47		1,016	-
Female	2,034	93	- 59	-	193	-	-	210	47	-	287	-
FEIIIdle	000	-	59	-	112	-	-	210	-	-	201	-

Major Industrial Divisions and Sex Manufacture of paper and paper Total Male	Total r products f 2,702 2,034		Afar	Amhara	Oromia		Benishangul				Addis	Dire
Total	r products I 2,702	Manufactu		Amnara			<u> </u>					
Total	2,702					Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
				iel allicies		nu papero	Uaru	000	47		4 000	
	2 0.37	93	59	-	305	-	-	896	47	-	1,303	
		93	-	-	193	-	-	686	47	-	1,016	
Female	668	-	59	-	112	-	-	210	-	-	287	
PUBLISHING, PRINTING AND												
Total	14,316	446	76	1,057	1,685	40	-	472	-	234	10,226	79
Male	6,633	446	-	466	566	-	-	430	-	91	4,635	
Female	7,683	-	76	591	1,119	40	-	42	-	144	5,592	79
Publishing												
Total	10,121	105	-	835	659	-	-	42	-	92	8,388	
Male	4,664	105	-	466	55	-	-	-	-	46	3,992	
Female	5,457	-	-	369	604	-	-	42	-	46	4,396	
Printing and service activities rel	lated to prir	nting										
Total	3,446	341	76	222	966	40	-	73	-	142	1,506	79
Male	1,220	341	-	-	451	-	-	73	-	44	311	
Female	2,226	-	76	222	515	40	-	-	-	98	1,196	79
Reproduction of recorded media	a											
Total	749	-	-	-	60	-	-	357	-	-	332	
Male	749	-	-	-	60	-	-	357	-	-	332	
Female	-	-	-	-	-	-	-	-	-	-	-	
MANUFACTURE OF COKE, RE	EFINED PE	TROLEUI	M PROD	UCTS AN	ID NUCLEA	R FUEL						
Total	827	85	-	-	44	-	-	-	54	-	644	
Male	420	-	-	-	44	-	-	-	54	-	322	
Female	407	85	-	-	-	-	-	-	-	-	322	
Manufacture of coke oven produ	ucts											
Total	129	85	-	-	44	-	-	-	-	-	-	
Male	44	-	-	-	44	-	-	-	-	-	-	
Female	85	85	-	-	-	-	-	-	-	-	-	
Manufacture of refined petroleur												
Total	54	-	-	-	-	-	-	-	54	-	-	
Male	54	-	-	-	-	-	-	-	54	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Processing of nuclear fuel												
Total	_	-	_	-	-	_	_	_	_	_	-	
Male	_	-	-	-	_	-	_	_	-	-	_	
Female	-	-	-	-	-	-	-	-	-	-	-	

Major Industrial Divisions						Re	gion	1				
and Sex	Tatal	T :	A. 6	A	0	0	Benishangul		Ormhalt	Llaure	Addis	Dire
MANUFACTURE OF CHEMIC		Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
		373	. FRODU 10		F F04			534	00		F 000	75
Total Male	11,917		10	78 78	5,534	-	-		90	-	5,222	75 75
	8,155	-		-	3,691 1,843	-	-	534	66	-	3,700	-
Female Manufacture of basic chemica	3,762 Is Manufactu	373 re of basi	- c chemic	- als evcer		- and nitroc	-	-	23	-	1,522	-
compounds	is manufactu					anu muog	Jen					
Total	1,898	-	-	-	1,123	-	-	400	23	-	352	-
Male	1,551	-	-	-	799	-	-	400	- 20	-	352	-
Female	347	-	-	-	324	-	-	-	23	-	- 00	-
Manufacture of other chemical		d of plasti	cs in prii	merv form					20			
Total	10,008	373	-	78	4,411	-	-	135	66	-	4,870	75
Male	6,593	-	-	78	2,891	-	-	135	66	-	3,348	75
Female	3,415	373	-	-	1,520	-	-	-	-	-	1,522	-
Manufacture of man-made fibr		0.0			.,•=•						.,•==	
Total	10	-	10	-	-	-	-	-	-	-	-	-
Male	10	-	10	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
MANUFACTURE OF RUBBER	R AND PLAS	TICS PRO	ODUCTS	5								
Total	9,849	264	8	774	4,090	211	-	42	-	-	4,303	156
Male	6,613	61	8	774	3,024	-	-	42	-	-	2,704	-
Female	3,235	203	-	-	1,067	211	-	-	-	-	1,599	156
Manufacture of rubber product					,						,	
Total	5,017	-	8	74	2,506	211	-	-	-	-	2,218	-
Male	3,684	-	8	74	2,101	-	-	-	-	-	1,501	-
Female	1,332	-	-	-	404	211	-	-	-	-	717	-
Manufacture of plastic product												
Total	4,832	264	-	700	1,585	-	-	42	-	-	2,085	156
Male	2,929	61	-	700	922	-	-	42	-	-	1,203	-
Female	1,903	203	-	-	662	-	-	-	-	-	882	156
MANUFACTURE OF OTHER	NON-METAL	LIC MINE	ERAL PF	RODUCTS	;							
Total	30,573	2,638	589	4,512	6,320	-	-	5,037	89	677	9,005	1,706
Male	20,493	2,045	468	3,156	4,763	-	-	3,265	67	631	4,734	1,364
Female	10,079	593	121	1,356	1,557	-	-	1,772	23	46	4,270	341
Manufacture of other non-meta	allic mineral p	oroducts										
Total	30,573	2,638	589	4,512	6,320	-	-	5,037	89	677	9,005	1,706
Male	20,493	2,045	468	3,156	4,763	-	-	3,265	67	631	4,734	1,364
Female	10,079	593	121	1,356	1,557	_	-	1,772	23	46	4,270	341
	10,079	093	121	1,550	1,007	-	-	1,112	23	40	4,270	541

Major Industrial Divisions and Sex Total Tigray Afar Amhara Oromia Benishar Somalie Benishar -Gumu MANUFACTURE OF BASIC METALS 16,209 933 21 3,406 6,123 607 Male 15,594 754 21 3,406 5,728 567 Female 615 179 - - 395 40 Manufacture of basic iron and steel - - 395 40 Male 3,749 346 - 777 2,158 - Male 3,749 346 - 777 1,854 - Female 393 89 - - 304 - Manufacture of basic precious and non-ferrous metals - - - - - Total 2,839 90 - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - -			I	Addis Ababa	Dire Dawa
Iotal Tigray Afar Amnara Oromia Somalie -Gumu MANUFACTURE OF BASIC METALS Total 16,209 933 21 3,406 6,123 607 Male 15,594 754 21 3,406 5,728 567 Female 615 179 - - 395 40 Manufacture of basic iron and steel Total 4,142 435 - 777 2,158 - Male 3,749 346 - 777 1,854 - - Manufacture of basic precious and non-ferrous metals - 303 89 - - 304 - Manufacture of basic precious and non-ferrous metals - - 304 - - Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - -	70 1,517	23	I	Ababa	Dawa
Total 16,209 933 21 3,406 6,123 607 Male 15,594 754 21 3,406 5,728 567 Female 615 179 - - 395 40 Manufacture of basic iron and steel - 777 2,158 - Total 4,142 435 - 777 1,854 - Male 3,749 346 - 777 1,854 - Female 393 89 - - 304 - Manufacture of basic precious and non-ferrous metals - - 304 - Manufacture of basic precious and non-ferrous metals - - 304 - Total 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - - Total			160		
Male 15,594 754 21 3,406 5,728 567 Female 615 179 - 395 40 Manufacture of basic iron and steel - 395 40 Total 4,142 435 - 777 2,158 - Male 3,749 346 - 777 1,854 - Female 393 89 - - 304 - Manufacture of basic precious and non-ferrous metals - 304 - Male 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - - Male 9,028 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91			160		
Female 615 179 - - 395 40 Manufacture of basic iron and steel Total 4,142 435 - 777 2,158 - Male 3,749 346 - 777 1,854 - Female 393 89 - - 304 - Manufacture of basic precious and non-ferrous metals - 304 - Total 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - - Male 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT	70 1,517	ົ່	-	3,122	224
Manufacture of basic iron and steel Total 4,142 435 - 777 2,158 - Male 3,749 346 - 777 1,854 - Female 393 89 - - 304 - Manufacture of basic precious and non-ferrous metals - 304 - Total 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - - Male 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT		23	162	3,122	224
Total 4,142 435 - 777 2,158 - Male 3,749 346 - 777 1,854 - Female 393 89 - - 304 - Manufacture of basic precious and non-ferrous metals - 304 - Total 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - - Male 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT		-	-	-	-
Male 3,749 346 - 777 1,854 - Female 393 89 - - 304 - Manufacture of basic precious and non-ferrous metals - 304 - Total 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - - Male 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT					
Female 393 89 - - 304 - Manufacture of basic precious and non-ferrous metals - 1,734 181 38 Total 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - - Male 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT	- 62		-	686	-
Manufacture of basic precious and non-ferrous metals Total 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - Total 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT	- 62	23	-	686	-
Total 2,839 90 - 1,734 181 38 Male 2,749 - - 1,734 181 38 Female 90 90 - - - - Manufacture of molded/shaped metals - - - - - Male 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT		-	-	-	-
Male 2,749 - - 1,734 181 38 Female 90 90 - - - - - Manufacture of molded/shaped metals - - - - - - Total 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT - - 91 40					
Female 90 90 90 -	- 672		55	-	69
Manufacture of molded/shaped metalsTotal9,228407218953,784570Male9,096407218953,693529Female1319140MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT	- 672		55	-	69
Total 9,228 407 21 895 3,784 570 Male 9,096 407 21 895 3,693 529 Female 131 - - 91 40 MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT		-	-	-	-
Male9,096407218953,693529Female1319140MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT					
Female1319140MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT	70 782		107	2,437	155
MANUFACTURE OF FABRICATED METAL PRODUCTS, EXCEPT MACHINERY AND EQUIPMENT	70 782	. –	107	2,437	155
		-	-	-	-
Total 21,360 1,159 212 4,619 6,243 197	- 1,370			6,911	455
Male 19,426 1,159 212 4,397 5,174 197	- 1,044	. 33	162	6,593	455
Female 1,934 222 1,069 -	- 326		-	318	-
Manufacture of structural metal products, tanks, reservoirs and steam generator boilers					
Total 16,042 1,054 121 3,944 3,907 37	- 647	33	82	5,762	455
Male 14,478 1,054 121 3,723 2,883 37	- 647	33	82	5,444	455
Female 1,564 222 1,025 -		· -	-	318	-
Manufacture of other fabricated metal products; metal working service activities					
Total 5,229 105 91 674 2,246 161	- 723		80	1,149	-
Male 4,859 105 91 674 2,203 161	- 397	-	80	1,149	-
Female 370 44 -	- 326		-	-	-
MANUFACTURE OF MACHINERY AND EQUIPMENT N.E.C.					
Total 3,799 385 - 94 1,362 -	- 64		-	1,823	72
Male 3,180 123 - 94 1,362 -	- 64	. -	-	1,466	72
Female 620 263		-	-	357	-
Manufacture of general purpose machinery					
Total 2,134 123 - 94 607 -					
Male 1,777 123 - 94 607 -		-	-	1,239	72
Female 357			-	1,239 881	72 72

Major Industrial Divisions						Re	gion					
and Sex							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Manufacture of special purpos												
Total	1,403	-	-	-	755	-	-	64	-	-	584	-
Male	1,403	-	-	-	755	-	-	64	-	-	584	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of domestic appli	ances n.e.c.											
Total	263	263	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	263	263	-	-	-	-	-	-	-	-	-	-
MANUFACTURE OF OFFICE	, ACCOUNTI	NG AND	COMPL	JTING MAG	CHINERY							
Total	372	85	-	-	-	-	-	-	-	-	287	-
Male	372	85	-	-	-	-	-	-	-	-	287	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of office, account	ting and comp	outing ma	chinery									
Total	372	85	-	-	-	-	-	-	-	-	287	-
Male	372	85	-	-	-	-	-	-	-	-	287	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
MANUFACTURE OF ELECTR	RICAL MACH	INERY AI	ND APP	ARATUS I	N.E.C.							
Total	4,002	324	-	-	780	-	-	1,282	-	41	1,505	72
Male	2,587	324	-	-	291	-	-	686	-	-	1,214	72
Female	1,416	-	-	-	489	-	-	596	-	41	291	-
Manufacture of electric motors	s, generators	and trans	formers									
Total	701	-	-	-	701	-	-	-	-	-	-	-
Male	250	-	-	-	250	-	-	-	-	-	-	-
Female	451	-	-	-	451	-	-	-	-	-	-	-
Manufacture of electricity distri	ibution and c	ontrol app	aratus									
Total	1,695	263	-	-	-	-	-	1,029	-	41	291	72
Male	1,021	263	-	-	-	-	-	686	-	-	-	72
Female	674	-	-	-	-	-	-	343	-	41	291	-
Manufacture of insulted wire a	ind cable											
Total	40	-	-	-	40	-	-	-	-	-	-	-
Male	40	-	-	-	40	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of accumulators,	primary cells	and prim	ary batte	eries								
Total	-	· -	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-

Major Industrial Divisions						Re	gion					
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Manufacture of electric lamps	and lighting	equipment	t									
Total	58	-	-	-	-	-	-	58	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	58	-	-	-	-	-	-	58	-	-	-	-
Manufacture of other electrica	l equipment i	n.e.c.										
Total	1,509	61	-	-	38	-	-	195	-	-	1,214	-
Male	1,275	61	-	-	-	-	-	-	-	-	1,214	-
Female	233	-	-	-	38	-	-	195	-	-	-	-
MANUFACTURE OF RADIO, EQUIPMENT AND APPARAT		N AND CC	MMUN	ICATION								
Total	1,882	91	-	94	563	-	-	231	-	-	902	-
Male	1,775	91	-	46	563	-	-	173	-	-	902	-
Female	107	-	-	48	-	-	-	59	-	-	-	-
Manufacture of electronic valv	es and tubes	and othe	r electro	nic compo	nents							
Total	563	-	-	-	563	-	-	-	-	-	-	-
Male	563	-	-	-	563	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of television and	radio transm	itters and	apparat	us for line	telephonya	nd linetele	graphy					
Total	752	91	-	46	-	-	-	-	-	-	615	-
Male	752	91	-	46	-	-	-	-	-	-	615	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of television and apparatus, and associated go		ers, sound	or video	o recording) or reprodu	icing						
Total	567	-	-	48	-	-	-	231	-	-	287	-
Male	460	-	-	-	-	-	-	173	-	-	287	-
Female	107	-	-	48	-	-	-	59	-	-	-	-
MANUFACTURE OF MEDIC.	AL, PRECISI	ON AND	OPTICA	LINSTRU	MENTS, W	ATCHES /	AND CLOCKS					
Total	1,823	61	-	75	421	-	-	-	-	41	1,224	-
Male	1,167	61	-	75	421	-	-	-	-	-	609	-
Female	656	-	-	-	-	-	-	-	-	41	615	-
Manufacture of medical applia navigating and other purposes	ances and ins s, except opti	truments cal Instrur	and app ment	liances for	measuring	, checking	ı, testing,					
Total	834	61	-	-	421	-	-	-	-	41	311	-
Male	483	61	-	-	421	-	-	-	-	-	-	-
Female	351	-	-	-	-	-	-	-	-	41	311	-
Manufacture of optical instrum		otographic	equipm	nent								
	609			-	_	_	_	_	-	-	609	-
Total	009	-	-			_					003	
l otal Male	609	-	-	-	-	-	-	-	-	-	609	-

Major Industrial Divisions						Re	gion					
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Manufacture of watches and o			7	7	0.0.1.10	•••••••	Canaz	•••••	e al libera		710404	Dana
Total	380	-	-	75	-	-	-	-	-	-	305	-
Male	75	-	-	75	-	-	-	-	-	-	-	-
Female	305	-	-	-	-	-	-	-	-	-	305	-
MANUFACTURE OF MOTOR	R VEHICLES,	TRAILER	S AND	SEMI-TRA	ILERS							
Total	9,886	829	346	403	1,682	40	-	334	23	-	6,227	-
Male	9,702	645	346	403	1,682	40	-	334	23	-	6,227	-
Female	184	184	-	-	-	-	-	-	-	-	-	-
Manufacture of motor vehicles	S											
Total	1,527	302	-	-	628	-	-	-	23	-	574	-
Male	1,435	209	-	-	628	-	-	-	23	-	574	-
Female	93	93	-	-	-	-	-	-	-	-	-	-
Manufacture of bodies (coach	work) for mot	tor vehicle	s; manu	facture of	trailers and	l semi- trai	lers					
Total	8,358	527	346	403	1,054	40	-	334	-	-	5,653	-
Male	8,267	436	346	403	1,054	40	-	334	-	-	5,653	-
Female	91	91	-	-	-	-	-	-	-	-	-	-
MANUFACTURE OF OTHER	TRANSPOR	T EQUIPI	MENT									
Total	2,579	173	-	462	598	-	-	165	-	-	1,181	-
Male	2,230	173	-	462	559	-	-	165	-	-	872	-
Female	349	-	-	-	39	-	-	-	-	-	309	-
Building and repairing of ships	s and boats/re	epairing of	f railway	, aircraft,m	anufacture	of bicycle	/					
Total	331	-	-	-	43	-	-	-	-	-	287	-
Male	331	-	-	-	43	-	-	-	-	-	287	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of railway and tra	amway locom	otives and	d rolling	stock								
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Manufacture of aircraft and sp												
Total	502	-	-	-	39	-	-	165	-	-	297	-
Male	462	-	-	-	-	-	-	165	-	-	297	-
Female	39	-	-	-	39	-	-	-	-	-	-	-
Manufacture of transport equi	-											
Total	1,746	173	-	462	515	-	-	-	-	-	597	-
Male	1,437	173	-	462	515	-	-	-	-	-	287	-
Female	309	-	-	-	-	-	-	-	-	-	309	-

Major Industrial Divisions						Re	gion					
and Sex							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
MANUFACTURE OF FURNIT	URE; MANU	FACTURI	NG N.E.	C.								
Total	53,568	2,526	114	13,325	20,278	749	480	4,508	383	864	10,048	294
Male	32,915	1,185	23	8,191	9,838	634	130	3,663	86	163	8,852	150
Female	20,654	1,341	91	5,134	10,440	115	350	845	297	700	1,196	144
Manufacture of household and	d office furnitu	ure										
Total	53,568	2,526	114	13,325	20,278	749	480	4,508	383	864	10,048	294
Male	32,915	1,185	23	8,191	9,838	634	130	3,663	86	163	8,852	150
Female	20,654	1,341	91	5,134	10,440	115	350	845	297	700	1,196	144
RICYCLING												
Total	3,719	226	-	696	1,658	-	-	564	-	-	574	-
Male	3,719	226	-	696	1,658	-	-	564	-	-	574	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Recycling of metal waste and												
Total	3,312	226	-	419	1,658	-	-	434	-	-	574	-
Male	3,312	226	-	419	1,658	-	-	434	-	-	574	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Recycling of non-metal waste	and scrap											
Total	407	-	-	277	-	-	-	130	-	-	-	-
Male	407	-	-	277	-	-	-	130	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
MAJOR GROUP 4: ELECTRI	-											
Total	59,456	3,457	2,036	9,131	19,229	4,905	730	3,692	332	373	14,018	1,551
Male	38,832	2,692	1,701	5,156	12,944	1,065	598	2,531	284	251	10,682	928
Female	20,624	765	335	3,975	6,285	3,840	131	1,161	48	122	3,336	624
ELECTRICITY, GAS, STEAM	AND HOT W	ATER SU	JPPLY									
Total	25,031	2,297	393	2,834	7,243	124	283	1,756	122	-	9,215	764
Male	21,752	2,053	272	2,760	6,399	124	266	986	122	-	8,005	764
Female	3,279	243	121	74	844	-	17	770	-	-	1,210	-
Production, collection and dist		•										
Total	23,515	2,297	272	2,663	6,746	98	283	1,756	98	-	8,609	693
Male	20,539	2,053	151	2,590	5,903	98	266	986	98	-	7,702	693
Female	2,975	243	121	74	844	-	17	770	-	-	907	-
Manufacture of gas; distribution	•	s fuels thro	ough ma	ins								
Total	1,070	-	121	171	51	26	-	-	25	-	607	71
Male	767	-	121	171	51	26	-	-	25	-	303	71
Female	303	-	-	-	-	-	-	-	-	-	303	-

Major Industrial Divisions				<u>г</u>		Re	gion					
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul	SNNPR	Gambela	Harari	Addis Ababa	Dire
Steam and hot water supply	TOTAL	rigiay	Alar	Annara	Oromia	Somalle	-Gumuz	SININFR	Gampela	Haran	ADADA	Dawa
Total	446				446							
Male	440	-	-	-	440	-	-	-	-	-	-	-
Female	440	-	-	-	440	-	-	-	-	-	-	-
COLLECTION, PURIFICATIO	- דפוח חואג וא	- וחודו ופוס		- ATED	-	-	-	-	-	-	-	-
Total	34,425	1,161	1,643	6,298	11,986	4,781	447	1,936	210	373	4,803	787
Male	17,080	638	1,043	2,396	6,545	4,781 941	333	1,545	162	251	4,803	164
Female	17,000	522	214	2,390	5,442	3,840	114	392	48	122	2,077	624
Collection, purification and dis			214	3,902	3,44Z	3,040	114	392	40	122	2,120	024
Total	34,425	1,161	1.643	6,298	11,986	4,781	447	1,936	210	373	4,803	787
Male	17,080	638	1,429	2,396	6,545	4,781 941	333	1,545	162	251	4,803	164
Female	17,345	522	214	2,390 3,902	5,442	3,840	114	392	48	122	2,077	624
MAJOR GROUP 5: CONSTR		522	214	3,902	J,44Z	3,040	114	392	40	122	2,120	024
Total	380,581	28,593	5,645	57,826	108,345	7,571	3,002	48,403	1,542	2,987	108,645	8,021
Male	318,433	23,478	4,683	43,665	93,928	7,284	2,697	40,403	1,138	2,5907	89,610	6,465
Female	62,148	23,478 5,115	4,003 962	43,005	93,920 14,417	287	305	5,509	404	2,390	19,035	1,556
CONSTRUCTION	02,140	5,115	302	14,101	14,417	201	505	5,509	404	557	19,000	1,550
Total	380,581	28,593	5,645	57,826	108,345	7,571	3,002	48,403	1,542	2,987	108,645	8,021
Male	318,433	23,478	4,683	43,665	93,928	7,284	2,697	42,894	1,138	2,590	89,610	6,465
Female	62,148	5,115	962	14,161	14,417	287	305	5,509	404	397	19,035	1,556
Site preparation												
Total	25,121	4,203	12	1,423	3,834	432	42	4,487	305	-	10,221	163
Male	21,222	2,383	12	1,120	3,713	401	42	3,931	237	-	9,222	163
Female	3,899	1,819	-	303	121	32	-	556	68	-	1,000	-
Building complete construction	n or parts the	reof; civil	engineei	ring								
Total	293,953	15,856	5,384	47,266	91,517	5,844	2,387	36,195	1,180	2,322	78,887	7,116
Male	243,508	13,701	4,432	34,420	78,287	5,626	2,200	32,070	844	1,977	64,391	5,559
Female	50,444	2,155	952	12,846	13,229	218	187	4,125	336	345	14,495	1,556
Building installation												
Total	20,647	1,417	31	2,856	4,074	1,115	157	2,938	23	147	7,665	224
Male	17,138	1,154	21	2,487	3,062	1,078	62	2,698	23	147	6,183	224
Female	3,509	263	10	369	1,012	37	96	240	-	-	1,483	-
Building completion	-											
Total	37,759	7,001	219	6,281	8,640	143	416	3,665	34	518	10,324	519
Male	34,756	6,122	219	5,639	8,586	143	394	3,396	34	466	9,239	519
Female	3,003	879	-	642	54	-	23	269	-	52	1,084	-

Major Industrial Divisions	 					Re	gion			· · · · ·		
and Sex	Tatal	T :	A.£	Arrest	0	0	Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SINNPR	Gambela	Hararı	Ababa	Dawa
Renting of construction or dem			operato	I	000	07					4 5 4 5	
Total	3,101	118	-	-	280	37	-	1,119	-	-	1,547	-
Male .	1,808	118	-	-	280	37	-	799	-	-	574	-
Female	1,293	-						320			973	-
MAJOR GROUP 6 : WHOLES GOODS	SALE AND R		KADE; R		- MOTOR	VEHICLES	S, MOTORCYC	LES AND	PERSON	AL AND I	HOUSEHO	LD
Total	1,099,758	72 440	10 560	183,331	361,317	44,906	5 7 <i>1</i> 7	132,204	7 262	14,576	239,267	26 1 15
Male	580.722	35,622	6,744	86.599	189,849	44,900 17,262	2,566	76,778	1,982	5,651	146.720	
Female	519,036	36,817	5,818	96,732	171,468	27,643	2,500	55,426	5,281	3,031 8,925	92,548	-,
WHOLESALE AND RETAIL T												15,190
Total	66,193	2,743	164	5,340	13,038	947	256 ID FERSONAL	6,870	147	681	34,506	1,501
Male	58,036	2,743	164	5,340 4,868	11,996	947 947	230	6,209	147	437	29,290	1,349
Female	8,156	2,413	104	4,000	1,041	947	215 40	661	147	437 244	29,290	1,349
Sale of motor vehicles	0,150	550	-	472	1,041	-	40	001	-	244	5,215	152
Total	2,972	67		350						41	2,363	152
Male	2,972	- 07	-	350 350	-	-	-	-	-	41	2,363	152
Female	2,144	- 67	-	350	-	-	-	-	-	41	609	- 152
Maintenance and repair of mo			- a and ar	- Dasina	-	-	-	-	-	-	009	152
Total	39,335	1,935	y anu yn 11	3,046	7,603	886	217	4,224	124	340	20,040	908
Male	39,335 36,555	1,935	11	3,046 2,748	6,860	886	197	4,224 3,780	124	340 259	20,040	908 908
Female	2,779	1,955		2,740	742	000	20	3,780 444	124	259 81	1,193	900
Sale of motor vehicle parts an		-	-	290	142	-	20	444	-	01	1,195	-
Total	9,179	s 90		724	307		_	245		164	7,506	143
Male	9,179 6,446	90 90	-	647	143	-	-	245 245	-	41	7,508 5,137	143
Female	2,733	90	-	77	143	-	-	240	-	123	2,369	143
Sale, maintenance and repair		- ond role	- ated par			-	-	-	-	123	2,309	-
Total	8,496	311 311	aieu pan 136	1,125	2,754	61		1,039	23	52	2,776	218
Male	8,496 7,704	311	136	1,125	2,754	61	-	1,039	23	52 52	2,776	210 218
Female	7,704 792	- 311	130	1,029 96	2,734	-	-	1,039	23	52	2,081	210
Retail sale of automotive fuel	192	-	-	90	-	-	-	-	-	-	090	-
Total	6,211	340	18	94	2,374		38	1,361		85	1,821	80
Male	6,211 5,186	340 77	18	94 94	2,374 2,239	-	30 18	1,301	-	65 44	1,621	80 80
	5,166 1,025	264	10	94	2,239	-	20	217	-		349	60
Female WHOLESALE AGENCIES, IM			- \RTEP9	- םוסדפות					-	41	349	-
Total	158,705	13,110	2,830	22,587	ботока, в 50,823	4,446	125 AND TENDER	11,366	356	1,596	45,243	6,221
Male			2,830		•	•	55					
	99,546 50 158	8,099 5 011	,	13,426	35,889	2,102	55 70	5,717	111 245	1,023	27,815	3,138
Female	59,158	5,011	658	9,161	14,934	2,345	70	5,649	245	573	17,428	3,083

						Re	gion					
Major Industrial Divisions							Benishangul			[] [Addis	Dire
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Wholesale agencies, importers	and exporte				nder	1				1		
Total	33,988	3,435	128	2,065	11,909	1,159	38	1,324	-	216	10,945	2,769
Male	26,559	2,657	64	1,988	9,951	735	38	1,265	-	216	8,462	1,181
Female	7,430	778	64	76	1,957	424	-	59	-	-	2,483	1,587
Wholesale of agricultural raw r	naterials, live	e animals,	food, be	everages a	and tobacco)						
Total	68,248	3,607	1,138	10,845	24,817	3,218	86	7,065	218	657	14,410	2,186
Male	34,372	1,553	958	4,177	15,270	1,324	17	2,761	61	308	6,433	1,511
Female	33,876	2,054	180	6,669	9,548	1,895	70	4,304	157	349	7,977	674
Wholesale of household goods	5											
Total	46,993	5,736	1,491	8,424	10,488	70	-	2,184	115	684	16,823	979
Male	31,748	3,640	1,077	6,229	7,751	43	-	1,051	27	460	11,174	297
Female	15,246	2,096	414	2,195	2,738	26	-	1,133	88	224	5,650	682
Wholesale of non-agricultural	intermediate	products,	waste a	and scrap								
Total	6,458	181	73	1,254	2,396	-	-	418	23	39	1,787	288
Male	4,585	181	73	1,032	1,939	-	-	264	23	39	885	149
Female	1,873	-	-	222	457	-	-	154	-	-	902	139
Wholesale of machinery, equp		oplies										
Total	3,017	151	-	-	1,213	-	-	376	-	-	1,278	-
Male	2,283	68	-	-	978	-	-	376	-	-	862	-
Female	734	83	-	-	235	-	-	-	-	-	416	-
Other wholesale												
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
RETAIL TRADE, OF MOTOR												
Total	874,860	56,587		155,404	297,457	39,512		113,967		12,299	159,518	
Male	423,139	25,111	4,408	68,305	141,964	14,213	2,296	64,852	1,724	4,191	89,614	6,461
Female	451,721	31,476	5,160	87,099	155,493	25,299	3,071	49,116	5,035	8,107	69,904	11,961
Non-specialized retail trade in												
Total	165,629	12,654	2,465	25,421	52,209	4,585	737	16,652	377	1,709	46,933	1,888
Male	92,610	6,549	1,511	14,226	29,261	2,658	409	9,385	255	823	26,761	771
Female	73,019	6,105	954	11,195	22,949	1,927	328	7,267	122	886	20,172	1,117
Retail sale of food, beverages	and tobacco	in specia	lized sto	res								
Total	199,200	13,325	1,721	42,360	66,522	4,569	1,267	30,427	233	1,561	34,712	2,503
Male	85,993	5,280	838	16,883	29,255	1,708	529	12,558	171	802	17,335	633
Female	113,207	8,044	883	25,477	37,267	2,861	738	17,869	61	759	17,377	1,870

						Re	gion					
Major Industrial Divisions							Benishangul				Addis	Dire
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Other retail trade of new good								-				
Total	114,985	7,048	329	22,591	27,928	3,097	676	10,072	4,647	1,661	34,785	2,150
Male	72,121	4,848	21	14,790	19,690	1,325	487	8,186	521	1,086	19,911	1,257
Female	42,864	2,200	308	7,802	8,238	1,772	189	1,887	4,126	575	14,874	893
Retail sale of second-hand go	ods in stores				,	·					,	
Total	10,698	61	41	1,739	2,358	87	20	2,642	-	414	2,954	382
Male	7,537	61	23	1,202	1,562	87	20	1,645	-	336	2,379	222
Female	3,161	-	18	538	796	-	-	997	-	78	574	160
Retail trade not in stores												
Total	329,984	21,426	4,489	58,454	130,778	20,760	2,667	43,772	1,292	6,333	29,715	10,298
Male	131,614	7,093	1,844	18,795	51,425	5,412	851	25,229	639	827	16,485	3,016
Female	198,370	14,333	2,645	39,659	79,353	15,348	1,816	18,543	653	5,506	13,230	7,283
Repair of personal and housel	nold goods											
Total	54,363	2,073	523	4,838	17,662	6,414	-	10,401	212	621	10,419	1,201
Male	33,263	1,280	171	2,409	10,772	3,023	-	7,849	138	317	6,743	562
Female	21,100	793	352	2,428	6,890	3,391	-	2,553	74	303	3,676	639
MAJOR GROUP 7: HOTELS	AND RESTA	URANTS	5									
Total	408,076	27,482	4,427	79,505	135,683	6,781	5,439	67,492	3,526	1,251	73,300	3,190
Male	105,545	3,595	601	13,527	34,593	1,838	463	22,063	382	494	27,016	973
Female	302,531	23,887	3,826	65,977	101,091	4,943	4,977	45,429	3,144	757	46,284	2,217
HOTELS AND RESTAURANT	S											
Total	408,076	27,482	4,427	79,505	135,683	6,781	5,439	67,492	3,526	1,251	73,300	3,190
Male	105,545	3,595	601	13,527	34,593	1,838	463	22,063	382	494	27,016	973
Female	302,531	23,887	3,826	65,977	101,091	4,943	4,977	45,429	3,144	757	46,284	2,217
Hotels; camping sites and othe	er provision o	of short-sta	ay accor	nmodation								
Total	87,904	3,091	173	10,820	28,678	910	594	11,640	236	644	29,903	1,215
Male	35,601	738	9	5,097	10,204	355	108	7,081	74	365	10,877	693
Female	52,303	2,353	164	5,722	18,474	555	485	4,559	162	279	19,026	522
Restaurants, bars and												
canteens												
Total	320,172	24,391	4,254	68,685	107,006	5,871	4,845	55,852	3,290	607	43,397	1,974
Male	69,944	2,857	592	8,430	24,389	1,483	354	14,983	308	129	16,139	280
Female	250,229	21,534	3,662	60,255	82,616	4,388	4,491	40,870	2,982	478	27,258	1,695
MAJOR GROUP 8:TRANSPO												
Total	242,410	14,522	1,195	38,524	72,106	7,175	1,466	21,452	701	1,744	77,222	6,303
Male	226,779	13,080	1,195	36,074	70,452	6,804	1,349	20,774	667	1,744	68,699	5,940
Female	15,631	1,442	-	2,450	1,654	370	118	678	35	-	8,523	363

Major Industrial Divisions				· · · · ·		Re	gion	1	1	· · · · · ·		
and Sex	Tatal	Tierrey	A far	Ameliana	Oramia	Comolio	Benishangul		Cambala	Llorori	Addis	Dire
LAND TRANSPORT; TRANSP		Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
							0.50	10.047		4 0 0 0		=
Total	199,335	10,720	723	31,916	63,041	5,046	953	16,847	461	1,663	62,333	5,631
Male	193,052	10,720	723	31,558	62,443	4,976	953	16,491	461	1,663	57,650	5,415
Female	6,282	-	-	358	598	70	-	356	-	-	4,684	216
Transport via railways												
Total	2,156	-	-	-	633	1,214	-	-	-	-	-	309
Male	2,156	-	-	-	633	1,214	-	-	-	-	-	309
Female	-	-	-	-	-	-	-	-	-	-	-	-
Other land transport												
Total	197,179	10,720	723	31,916	62,409	3,832	953	16,847	461	1,663	62,333	5,322
Male	190,897	10,720	723	31,558	61,810	3,762	953	16,491	461	1,663	57,650	5,107
Female	6,282	-	-	358	598	70	-	356	-	-	4,684	216
WATER TRANSPORT												
Total	3,599	729	281	198	546	1,737	-	59	49	-	-	-
Male	3,166	486	281	198	513	1,579	-	59	49	-	-	-
Female	434	243	-	-	33	158	-	-	-	-	-	-
Sea and coastal water transpor	rt											
Total	2,649	486	155	76	86	1,737	-	59	49	-	-	-
Male	2,215	243	155	76	53	1,579	-	59	49	-	-	-
Female	434	243	-	-	33	158	-	-	-	-	-	-
Inland water transport												
Total	951	243	126	122	460	-	-	-	-	-	-	-
Male	951	243	126	122	460	-	-	-	-	-	-	-
Female AIR TRANSPORT	-	-	-	-	-	-	-	-	-	-	-	-
Total	5,251	396	12	403	452	_	_	_	10	_	3,898	80
Male	4,655	396	12	403	452	_	_		10	_	3,301	80
Female	4,000 597	- 590	12	403	452	-	-		-	_	597	
Scheduled air transport	557	-	-	-	-	-	-	-	-	-	591	_
Total	4,150	-	-	122	40	-	-	-	10	-	3,898	80
Male	3,554	-	-	122	40	-	-	-	10	-	3,301	80
Female	597	-	-	-	-	-	-	-	-	-	597	-
Non-scheduled air transport	001										001	
Total	1,101	396	12	281	413	_	_	_	_	_	_	-
Male	1,101	396	12	281	413	_	_	_	-	-	_	-
Female	1,101	590	12	201	415	-	-	-	-	-	-	-

Major Industrial Divisions						Re	gion		-			
and Sex					. .	.	Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
SUPPORTING AND AUXILIA											. =	
Total	14,726	896	115	2,746	3,518	117	77	2,179	57	-	4,783	237
Male	12,463	896	115	2,451	2,741	80	77	2,129	23	-	3,787	162
Female	2,263	-	-	. 294	777	37	-	50	35	-	996	75
Supporting and auxiliary trans	-			-								
Total	14,726	896	115	2,746	3,518	117	77	2,179	57	-	4,783	237
Male	12,463	896	115	2,451	2,741	80	77	2,129	23	-	3,787	162
Female	2,263	-	-	294	777	37	-	50	35	-	996	75
POST AND TELECOMMUNIC												
Total	19,498	1,781	64	3,261	4,548	275	436	2,367	124	81	6,208	354
Male	13,444	583	64	1,464	4,302	169	318	2,096	124	81	3,961	282
Female	6,055	1,198	-	1,797	246	106	118	271	-	-	2,247	73
Post and courier activities												
Total	4,437	-	-	824	824	43	-	648	80	-	1,875	142
Male	3,213	-	-	727	824	43	-	444	80	-	952	142
Female	1,224	-	-	97	-	-	-	204	-	-	923	-
Telecommunications												
Total	15,061	1,781	64	2,437	3,724	231	436	1,719	44	81	4,332	213
Male	10,231	583	64	737	3,478	126	318	1,651	44	81	3,009	140
Female	4,830	1,198	-	1,700	246	106	118	67	-	-	1,323	73
MAJOR GROUP 9: FINANCIA	AL INTERMI	DIATION										
Total	91,143	9,545	471	10,610	32,909	1,645	943	10,059	493	288	22,508	1,674
Male	51,635	5,465	458	6,897	18,307	1,370	526	6,286	348	87	10,581	1,312
Female	39,508	4,080	13	3,713	14,602	275	417	3,773	145	201	11,927	362
FINANCIAL INTERMEDIATIO	N, EXCEPT	INSURAN	ICE AND) PENSIO	N FUNDING	G						
Total	59,150	5,895	184	7,551	18,542	696	159	5,838	225	163	18,663	1,237
Male	32,842	4,156	171	5,191	9,206	653	159	3,225	201	41	8,813	1,026
Female	26,308	1,738	13	2,360	9,336	42	-	2,613	23	122	9,850	211
Monetary intermediation	·				,							
Total	37,389	2,502	171	4,463	10,946	425	73	2,665	137	163	14,746	1,098
Male	20,825	1,885	171	2,750	5,681	382	73	2,086	114	41	6,754	888
Female	16,564	616	-	1,713	5,265	42	-	579	23	122	7,993	211
Other financial intermediation	-,			,	,_ .				_0		,	
Total	21,761	3,393	13	3,088	7,595	271	86	3,173	88	-	3,916	139
Male	12,017	2,271	-	2,441	3,524	271	86	1,139	88	-	2,059	139
Female	9,744	1,122	13	647	4,071		-	2,034	50	_	1,857	

Major Industrial Divisions			r			Re	gion					r
and Sex					. .		Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
INSURANCE AND PENSION						URITY						
Total	1,995	364	10	46	878	-	51	-	-	-	574	71
Male	920	84	10	-	774	-	51	-	-	-	-	-
Female	1,075	281	-	46	104	-	-	-	-	-	574	71
Insurance and pension funding	•			•								
Total	1,995	364	10	46	878	-	51	-	-	-	574	71
Male	920	84	10	-	774	-	51	-	-	-	-	-
Female	1,075	281	-	46	104	-	-	-	-	-	574	71
ACTIVITIES AUXILIARY TO F			DIATION									
Total	29,998	3,286	277	3,014	13,489	949	732	4,220	268	126	3,270	366
Male	17,874	1,225	277	1,706	8,327	717	315	3,060	146	46	1,768	286
Female	12,125	2,061	-	1,308	5,163	232	417	1,160	121	79	1,503	80
Activities auxiliary to financial	intermediatio	n, except	insurand	ce and per	nsion fundir	ig						
Total	29,387	3,286	277	3,014	13,189	949	732	4,220	268	126	2,960	366
Male	17,762	1,225	277	1,706	8,215	717	315	3,060	146	46	1,768	286
Female	11,625	2,061	-	1,308	4,974	232	417	1,160	121	79	1,192	80
Activities auxiliary to insurance	e and pensio	n funding										
Total	611	-	-	-	300	-	-	-	-	-	311	-
Male	112	-	-	-	112	-	-	-	-	-	-	-
Female	499	-	-	-	189	-	-	-	-	-	311	-
MAJOR GROUP 10: REAL E	STATE, REN		ID BUSI	NESS AC	TIVITIES							
Total	116,708	3,982	962	15,486	27,678	1,089	849	15,898	393	572	48,175	1,623
Male	76,784	1,972	709	9,988	16,434	894	740	10,595	292	355	33,703	1,101
Female	39,924	2,011	253	5,499	11,243	195	109	5,303	101	217	14,472	522
REAL ESTATE ACTIVITIES												
Total	7,547	396	-	770	570	-	-	454	-	-	5,204	152
Male	4,945	-	-	726	405	-	-	227	-	-	3,435	152
Female	2,602	396	-	44	166	-	-	227	-	-	1,769	-
Real estate activities with own	n or leased pr	operty										
Total	3,150	-	-	46	119	-	-	454	-	-	2,379	152
Male	1,751	-	-	46	119	-	-	227	-	-	1,207	152
Female	1,399	-	-	-	-	-	-	227	-	-	1,172	-
Real estate activities on a fee	or contract b	asis										
Total	4,397	396	-	724	452	-	-	-	-	-	2,826	-
Male	3,194	-	-	680	286	-	-	-	-	-	2,228	-
Female	1,203	396	-	44	166	-	-	-	-	-	597	-

Major Industrial Divisions						Re	gion					
Major Industrial Divisions and Sex							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
RENTING OF MACHINERY A	ND EQUIPN	IENT WIT	HOUT C	PERATO	R AND OF	PERSONA	AL AND HOUS	EHOLD G	GOODS			
Total	11,987	41	196	1,745	3,115	240	149	1,674	99	42	4,390	295
Male	9,701	41	196	966	2,700	240	149	1,508	99	-	3,506	295
Female	2,287	-	-	779	415	-	-	165	-	42	884	-
Renting of transport equipmen	nt											
Total	2,509	-	186	869	416	-	21	584	76	-	287	71
Male	2,307	-	186	767	315	-	21	584	76	-	287	71
Female	202	-	-	102	101	-	-	-	-	-	-	-
Renting of other machinery an	t equipment											
Total	3,464	41	-	435	632	98	-	479	23	-	1,755	-
Male	2,798	41	-	79	632	98	-	479	23	-	1,446	-
Female	666	-	-	356	-	-	-	-	-	-	309	-
Renting of personal and house	ehold goods	n.e.c.										
Total	6,014	-	10	441	2,068	142	128	611	-	42	2,347	224
Male	4,595	-	10	120	1,753	142	128	445	-	-	1,773	224
Female	1,419	-	-	321	315	-	-	165	-	42	574	-
COMPUTER AND RELATED	ACTIVITIES/	SERVICE	=/									
Total	29,918	1,702	137	2,574	4,692	389	172	4,738	106	95	15,017	297
Male	17,288	411	-	1,254	3,127	304	128	1,552	37	55	10,268	152
Female	12,630	1,291	137	1,319	1,565	85	44	3,186	69	41	4,749	144
Hardware consultancy service	•											
Total	1,415	396	-	-	-	-	54	58	-	-	908	-
Male	408	-	-	-	-	-	54	58	-	-	297	-
Female	1,006	396	-	-	-	-	-	-	-	-	611	-
Software consultancy service												
Total	7,020	275	-	1,012	2,494	-	27	1,563	69	55	1,454	72
Male	2,691	-	-	189	1,103	-	-	179	-	55	1,166	-
Female	4,328	275	-	823	1,391	-	27	1,384	69	-	287	72
Data processing/analysis/ serv												
Total	5,952	67	137	616	289	-	56	1,454	37	-	3,224	73
Male	3,985	-	-	575	169	-	56	787	37	-	2,363	-
Female	1,967	67	137	41	120	-	-	667	-	-	862	73
Data base activities												
Total	2,678	168	-	-	681	252	-	677	-	-	901	-
Male	2,100	-	-	-	681	252	-	266	-	-	901	-
Female	579	168	-	-	-	-	-	411	-	-	-	-

Major Industrial Divisions						Re	gion					
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Manitenance and repair of offi								-				
Total	5,989	114	-	374	1,131	-	18	357	-	-	3,914	80
Male	5,334	114	-	374	1,131	-	18	-	-	-	3,616	80
Female	655	-	-	-	-	-	-	357	-	-	298	-
Other computer related activiti	ies											
Total	6,864	682	-	572	97	137	17	630	-	41	4,616	72
Male	2,770	297	-	117	44	52	-	263	-	-	1,925	72
Female RESEARCH AND DEVELOPM	4,094 MENT	385	-	455	53	85	17	367	-	41	2,691	-
Total	11,598	263	84	1,557	4,823	43	120	1,322	35	-	3,276	75
Male	8,624	263	84	1,557	2,726	-	120	1,084	35	-	2,682	75
Female	2,973		-	-	2,097	43	-	238	-	-	595	-
Research and experimental de engineering(NSE)		on natural	science	s and	_,							
Total	8,403	263	-	1,496	3,226	43	120	829	12	-	2,415	-
Male	6,529	263	-	1,496	1,766	-	120	765	12	-	2,107	-
Female	1,874	-	-	-	1,460	43	-	64	-	-	307	-
Research and experimental de	evelopment o	on social s	ciences	and huma	nities (SSH)						
Total	3,195	-	84	62	1,597	-	-	494	23	-	862	75
Male	2,096	-	84	62	959	-	-	319	23	-	574	75
Female	1,100	-	-	-	638	-	-	175	-	-	287	-
OTHER BUSINESS ACTIVITI	ES											
Total	55,658	1,581	545	8,840	14,477	417	408	7,710	153	435	20,288	805
Male	36,225	1,257	430	5,484	7,477	351	343	6,224	121	301	13,812	427
Female	19,433	324	115	3,356	7,000	67	65	1,486	32	134	6,476	377
Legal accounting, book-keepir and public opinion polling; bus	ng and auditir siness and ma	ng activitie anagemer	es; tax consu	onsultancy Itancy	; market re	search						
Total	30,057	924	309	3,607	8,673	235	205	5,015	63	209	10,678	139
Male	17,882	790	193	2,345	4,132	235	140	3,638	63	169	6,038	139
Female	12,175	134	115	1,263	4,541	-	65	1,377	-	41	4,640	-
Architectural, engineering and	l other technic	cal activiti	es									
Total	3,182	105	172	414	1,160	29	-	622	32	-	574	75
Male	2,273	-	172	318	845	29	-	622	-	-	287	-
Female	909	105	-	96	315	-	-	-	32	-	287	75
Advertising												
Total	3,162	-	-	921	1,281	-	115	195	35	41	574	-
Male	1,804	-	-	636	495	-	115	195	35	41	287	-
Female	1,358	_	_	285	786						287	

Major Industrial Divisions						Re	gion					
and Sex							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Business activities n.e.c.												
Total	19,257	552	64	3,897	3,363	154	88	1,879	23	185	8,461	591
Male	14,267	467	64	2,185	2,004	87	88	1,769	23	91	7,199	289
Female	4,990	85	-	1,712	1,358	67	-	110	-	94	1,262	303
MAJOR GROUP 11: PUBLIC												
Total	324,536	20,544	4,269	64,545	95,175	16,919	5,321	49,938	4,192	4,468	55,106	4,059
Male	222,301	13,782	2,383	47,150	66,388	14,977	3,727	34,769	2,658	2,817	30,759	2,890
Female	102,234	6,762	1,885	17,395	28,787	1,942	1,594	15,169	1,534	1,650	24,348	1,169
PUBLIC ADMINISTRATION A	ND DEFENC	COMF	PULSOR	Y SOCIAL	. SECURIT							
Total	324,536	20,544	4,269	64,545	95,175	16,919	5,321	49,938	4,192	4,468	55,106	4,059
Male	222,301	13,782	2,383	47,150	66,388	14,977	3,727	34,769	2,658	2,817	30,759	2,890
Female	102,234	6,762	1,885	17,395	28,787	1,942	1,594	15,169	1,534	1,650	24,348	1,169
Administration of the state and			•	•	mmunity							
Total	204,487	8,299	2,541	39,408	66,425	9,948	3,822	33,145	2,635	2,021	34,445	1,798
Male	128,867	4,809	1,374	27,744	41,879	8,396	2,474	21,946	1,565	1,313	16,167	1,200
Female	75,620	3,491	1,166	11,663	24,546	1,552	1,348	11,199	1,070	709	18,278	598
Provision of services to the co	mmunity as a	a whole										
Total	117,736	11,981	1,728	25,137	28,643	6,921	1,478	16,526	1,558	2,324	19,179	2,261
Male	92,542	8,711	1,009	19,406	24,402	6,531	1,233	12,772	1,094	1,422	14,273	1,690
Female	25,194	3,271	719	5,731	4,241	390	245	3,754	464	901	4,906	571
Compulsory social security ac	tivities and pe	ension se	curity									
Total	2,313	263	-	-	107	50	21	266	-	123	1,483	-
Male	892	263	-	-	107	50	21	50	-	82	319	-
Female	1,420	-	-	-	-	-	-	216	-	41	1,164	-
MAJOR GROUP 12: EDUCA	TION											
Total	346,352	23,384	2,284	64,073	103,827	8,815	4,314	62,262	1,678	4,658	66,991	4,066
Male	201,521	11,138	1,771	35,247	66,183	6,778	2,887	35,911	1,172	2,380	35,147	2,908
Female	144,831	12,246	513	28,825	37,644	2,038	1,427	26,352	506	2,278	31,844	1,158
EDUCATION												
Total	346,352	23,384	2,284	64,073	103,827	8,815	4,314	62,262	1,678	4,658	66,991	4,066
Male	201,521	11,138	1,771	35,247	66,183	6,778	2,887	35,911	1,172	2,380	35,147	2,908
Female	144,831	12,246	513	28,825	37,644	2,038	1,427	26,352	506	2,278	31,844	1,158
Primary education	•	·						•				
Total	165,910	11,483	1,807	30,244	49,295	5,995	1,958	28,289	874	2,481	32,241	1,241
Male	82,359	4,997	1,373	11,950	27,960	4,933	924	14,360	652	1,094	13,344	772
Female	83,550	6,487	434	18,295	21,335	1,062	1,034	13,929	222	1,387	18,898	470
	,-00	-,		,	,	.,	.,501			.,	,	

Major Industrial Divisions						Re	gion	n	1			
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Secondary education	TULAI	пугау	Alai	Annala	Offinia	Somalie	-Guilluz	SININER	Gampela	Tiatatt	Ababa	Dawa
Total	88,771	5,042	51	17,030	26,672	2,290	1,739	16,983	642	627	16,842	851
Male	66,155	2,779	29	13,146	20,606	1,538	1,574	12,275	474	342	12,541	851
Female	22,616	2,263	22	3,884	6,066	752	165	4,708	168	285	4,301	001
Technical and vocational, drivin			22	5,004	0,000	152	105	4,700	100	200	+,501	
Total	23,081	3,822	204	5,475	3,361	_	83	4,203	23	233	5,309	368
Male	15,786	2,134	204	4,929	2,139	_	41	2,587	23	233	3,270	226
Female	7,295	1,688	- 204	546	1,222	_	42	1,616	20	- 200	2,039	142
Higher education/college and u			_	540	1,222	_	72	1,010	_	-	2,000	172
Total	68,591	3,036	221	11,323	24,499	531	534	12,787	140	1,316	12,599	1,606
Male	37,221	1,228	164	5,222	15,478	307	348	6,689	23	710	5,992	1,000
Female	31,370	1,808	57	6,100	9,021	224	186	6,098	116	606	6,606	546
MAJOR GROUP 13: HEALTH				0,100	3,021	224	100	0,030	110	000	0,000	540
Total	161,675	11,191	3,086	24,170	45,905	6,024	2,934	23,984	894	2,528	38,753	2,207
Male	80,565	6,024	1,938	11,581	24,876	3,779	1,498	12,418	669	1,016	15,719	1,048
Female	81,110	0,024 5,167	1,148	12,589	24,870	2,245	1,436	11,566	225	1,512	23,034	1,159
HEALTH AND SOCIAL WORK	,	5,107	1,140	12,569	21,020	2,245	1,430	11,500	225	1,312	23,034	1,158
Total	、 161,675	11,191	3,086	24,170	45,905	6,024	2,934	23,984	894	2,528	38,753	2,207
Male	80,565	6,024	1,938	11,581	24,876	3,779	1,498	12,418	669	1,016	15,719	1,048
Female	81,110	5,167	1,148	12,589	21,028	2,245	1,436	11,566	225	1,512	23,034	1,159
Human health activities	01,110	0,107	1,110	12,000	21,020	2,210	1,100	11,000	220	1,012	20,001	1,100
Total	136,703	10,212	2,951	20,325	40.762	4,882	2,492	20,044	737	2,443	29,932	1,923
Male	66,109	5,621	1,803	9,005	21,956	2,689	1,216	9,557	513	972	11,869	909
Female	70,594	4,591	1,148	11,320	18,807	2,000	1,277	10,487	225	1,471	18,063	1,014
Veterinary activities	70,001	1,001	1,110	11,020	10,007	2,102	1,211	10,107	220	1,171	10,000	1,011
Total	8,301	222	135	2,770	1,832	1,061	209	1,263	157	-	584	69
Male	6,712	111	135	2,277	1,218	1,061	135	1,263	157	-	287	69
Female	1,589	111	-	493	614	-	74	-,200	-	-	297	
Social work activities	1,000			100	011		, ,				201	
Total	16,671	757	-	1,075	3,311	81	232	2,678	-	85	8,237	214
Male	7,744	292	-	299	1,703	29	147	1,598	-	44	3,563	69
Female	8,926	465	-	776	1,608	53	86	1,080	-	41	4,674	145
MAJOR GROUP 14: OTHER								1,000		11	7,074	1-10
Total	459,078	24,267	2,454	76,847	136,213	13,199	4,750	88,303	1,078	4,388	100,719	6,860
Male	276,244	14,842	1,816	50,316	86,549	6,853	2,971	45,222	730	2,939	59,007	4,999
Female	182,834	9,424	638	26,531	49,664	6,347	1,779	43,081	348	1,449	41,712	1,861
	102,034	3,424	030	20,001	49,004	0,547	1,779	40,001	540	1,449	41,712	1,00

Table 5.10 Cont'd

Major Industrial Divisions						Re	gion	n				
and Sex	Tatal	T .			o	0	Benishangul				Addis	Dire
SEWAGE AND REFUSE DISF	Total	Tigray	Afar	Amhara		Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
						044	40	0.040		101	0 700	454
Total	11,807	996	121	3,542	1,532	211	43	2,246	-	184	2,782	151
Male	5,321	307	-	1,175	1,127	-	23	1,360	-	46	1,283	
Female	6,486 Conitation o	688 ed Circilo	121	2,366	405	211	20	886	-	138	1,499	151
Sewage and Refuse Disposal,					4 500	044	10	0.040		404	0 700	4 - 4
Total	11,807	996	121	3,542	1,532	211	43	2,246	-	184	2,782	151
Male	5,321	307	-	1,175	1,127	-	23	1,360	-	46	1,283	
Female	6,486	688	121	2,366	405	211	20	886	-	138	1,499	151
ACTIVITIES OF MEMBERSHI												
Total	53,294	3,852	318	10,015	14,121	1,073	360	5,666	252	339	16,457	841
Male	41,758	3,566	307	7,778	11,492	811	334	4,935	211	293	11,335	695
Female	11,536	286	10	2,237	2,629	262	26	731	41	46	5,122	146
Other Activities of business an		•	•									
Total	13,187	1,102	67	2,220	1,520	382	51	664	50	86	6,967	75
Male	8,512	816	67	1,112	1,286	339	51	272	33	41	4,495	-
Female	4,674	286	-	1,108	235	43	-	392	17	46	2,472	75
Activities of trade unions												
Total	369	-	-	369	-	-	-	-	-	-	-	-
Male	369	-	-	369	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	
Activities of other membership	organization	IS										
Total	39,738	2,750	251	7,425	12,600	690	309	5,002	202	252	9,490	766
Male	32,876	2,750	240	6,296	10,207	472	283	4,663	178	252	6,840	695
Female	6,862	-	10	1,129	2,394	219	26	339	25	-	2,650	71
RECREATIONAL, CULTURAL	_ AND SPOR	RTING AC	TIVITIE	5								
Total	37,725	2,461	134	2,608	6,898	-	275	4,851	102	328	19,550	518
Male	25,993	1,843	134	2,020	5,669	-	213	3,041	21	275	12,344	433
Female	11,732	618	-	588	1,229	-	62	1,810	80	53	7,206	85
Motion picture, radio, television			ent activ		.,0		02	1,010		00	.,_00	00
Total	18,669	1,663	115	1,143	2,120	-	88	2,640	42	144	10.642	73
Male	12,599	1,201	115	736	1,972	-	88	1,959	21	91	6,344	73
Female	6,070	462	-	407	148	-	-	681	21	53	4,298	
News agency activities	0,070	-102	_	101	1-10	_	_	001	21	00	7,200	
Total	1,530	324	10	36	392	-	23	386	-	-	287	72
Male	1,079	324	10	36	- 592	_	23	327	_	_	287	72
Female	451	524	10	- 50	392	-	- 23	59	-	-	207	12
remaie	451	-	-	-	392	-	-	59	-	-	-	

Table 5.10 Cont'd

and Sex Total Tigray Atar Amhara Oromia Somalie -Gumuz SNNPR Gambela Harari Ababa Dawa Library, archives, museums and other cultural activities 4,901 189 - 227 857 - 62 500 25 - 2,904 13 Male 2,628 100 - 119 817 - - - - 1,453 13 Female 2,273 89 - 108 3,529 - 103 1,326 35 184 5,717 23 Male 9,688 2,187 9 1,203 3,529 - 571 35 1,447 4,4260 15 Female 2,937 67 - 73 650 - 571 35 1,447 4,353 61,930 5,34 Male 203,172 9,126 1,375 39,42 68,273 1,671 3,654 4488 2,245	Major Industrial Divisions						Re	gion					
India India India Oronaita Somaita Gameia Formational Procession Somaita Formational Procession Adda beside Data Total 4,901 189 - 227 857 - 62 500 25 - 2,904 13 Male 2,628 100 - 119 817 - - - 1,453 13 Female 2,628 100 - 119 817 - - - 1,453 13 Sporting and other recreational activities - 73 650 - 103 7,55 - 184 4,260 15 Female 2,937 67 - 73 650 - - 571 35 - 1,457 8 OTHER SERVICE ACTIVITIES - 1,326 26,81 1,366 1,375 3,462 1,417 1,403 3,437 10 79 4,734 64 Fema													Dire
Total 4,901 189 - 227 857 - 62 500 25 - 2,904 13 Male 2,628 100 - 119 817 - - - 1,453 13 Sporting and other recreational activities - 12,625 285 9 1,203 3,529 - 103 1,326 35 184 5,717 23 Male 9,688 218 9 1,129 2,879 - 103 755 - 184 4,260 15 Female 2,937 67 - 73 650 - - 571 35 61,930 5,34 Male 2,93172 9,126 1,375 39,342 68,261 6,042 2,401 35,886 498 2,325 34,045 3,87 Female 153,080 7,832 507 2,134 45,015 5,873 1,611 90 6,763 55 <td< td=""><td></td><td></td><td>0,</td><td></td><td>Amhara</td><td>Oromia</td><td>Somalie</td><td>-Gumuz</td><td>SNNPR</td><td>Gambela</td><td>Harari</td><td>Ababa</td><td>Dawa</td></td<>			0,		Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Male 2,628 100 - 119 817 - - - - - 1,453 13 Female 2,273 89 - 108 39 - 62 500 25 - 1,453 13 Sporting and other recreational activities 12,625 2,85 9 1,203 3,529 - 103 1,326 35 184 5,717 7.8 Male 9,688 218 9 1,203 3,529 - 103 7,55 - 1,447 8 OTHER SERVICE ACTIVITES 2,937 67 - 73 650 - 5,541 7,24 5,376 6,193 3,68 4,002 7,5,541 7,24 5,347 6,642 2,401 35,886 498 2,325 3,40,45 3,87 Repairing of old cloths - 71,340 45,401 5,876 1,673 5,34 4,637 1,61 190 6,763 55 44 <td>•</td> <td></td> <td></td> <td>ies</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	•			ies									
Female 2,273 89 - 108 39 - 62 500 25 - 1,451 Sporting and other recreational activities 1 12,625 285 9 1,203 3,529 - 103 1,326 35 184 4,200 15 Female 2,937 67 - 73 660 - - 571 35 - 1,457 80 OTHER SERVICE ACTIVITIES 356,252 16,958 1,816 60,683 113,662 1,911 4,072 7,541 724 3,537 61,930 5,34 Male 203,172 9,126 1,375 39,342 68,261 6,042 2,401 35,686 498 2,325 34,045 3,87 Female 163,060 6,513 308 7,028 22,269 541 196 10,199 66 543 17,118 1,020 Male 28,949 2,241 21 4,637 1,265 380<				-			-	62	500	25	-		139
Sporting and other recreational activities Total 12,625 285 9 1,203 3,529 - 103 1,326 35 184 5,717 23 Male 9,688 218 9 1,129 2,879 - 103 1,326 35 184 4,260 15 Female 2,937 67 - 73 650 - 571 35 - 1,457 8 OTHER SERVICE ACTIVITIES 366,252 16,958 1,881 60,683 113,662 11,915 4,072 75,541 724 3,537 61,930 5,34 Male 203,172 9,126 1,375 39,42 68,261 6,042 2,401 38,664 498 2,425 3,4045 3,470 66 543 1,7118 1,222 7,885 1,477 Repairing of old cloths - 104 66,857 4,272 287 2,391 9,614 161 90 6,763 55				-			-		-	-	-		139
Total 12,625 285 9 1,203 3,529 - 103 1,326 35 184 5,717 23 Male 9,688 218 9 1,129 2,879 - 103 755 - 184 4,260 15 Female 2,937 67 - 73 650 - - 571 35 - 1,457 8 OTHER SERVICE ACTIVITIES 0356,252 16,958 1,881 60,683 113,662 11,915 4,072 75,541 724 3,537 61,930 5,34 Male 203,172 9,126 1,375 39,342 68,261 6,042 2,401 35,864 482 2,225 34,045 3,87 1,671 39,654 226 1,212 27,885 1,47 Repairing of old clots - Total 65,806 6,513 308 7,028 22,699 541 196 10,733 55 464 1,324 3,437			89	-	108	39	-	62	500	25	-	1,451	-
Male 9,688 218 9 1,129 2,879 - 103 755 - 184 4,260 15 Female 2,937 67 - 73 650 - - 571 35 - 1.84 4,260 15 OTHER SERVICE ACTIVITES Total 356,252 16,958 1,881 60,683 113,662 11,915 4,072 75,541 724 3,537 61,930 5,344 Male 203,172 9,126 1,375 507 21,340 45,401 5,873 1,611 39,654 226 1,212 27,885 1,47 Repairing of old cloths Total 65,806 6,513 308 7,028 22,269 341 196 10,199 66 543 17,118 1,02 Male 28,949 2,241 21 4,637 12,655 380 106 3,437 10 79 4,734 44 Female 116,223 3,560 200 18,949 3,787 5,712 1,581 32,891 170 749<	Sporting and other recreationa	al activities											
Female 2,937 67 73 650 - - 571 35 - 1,457 8 OTHER SERVICE ACTIVITIES Total 236,252 16,958 1,881 60,683 113,662 11,915 4,072 75,541 724 3,537 61,930 5,34 Male 203,172 9,126 1,375 39,342 68,261 6,042 2,401 35,886 498 2,325 34,045 3,67 Repailing of old clotts Total 65,806 6,513 308 7,028 22,269 541 196 10,199 66 543 17,118 1,02 Male 28,949 2,241 21 4,637 12,655 380 106 3,437 10 79 4,734 64 12,384 37 Other service activities Total 290,446 10,445 1,573 53,654 91,393 11,374 3,876 65,341 658 2,94 44,812 4,32 Female </td <td></td> <td></td> <td></td> <td>9</td> <td>,</td> <td></td> <td>-</td> <td></td> <td></td> <td>35</td> <td>184</td> <td></td> <td>235</td>				9	,		-			35	184		235
OTHER SERVICE ACTIVITIES Total 356,252 16,958 1,881 60,683 113,662 11,915 4,072 75,541 724 3,537 61,930 5,34 Male 203,172 9,126 1,375 39,342 66,261 6,042 2,401 35,866 488 2,325 34,045 3,87 Repairing of old cloths Total 65,806 6,513 308 7,028 22,269 541 196 10,199 66 543 17,118 1,02 Male 28,949 2,241 21 4,637 12,655 380 106 3,437 10 79 4,734 64 Female 36,857 4,272 287 2,391 9,614 161 90 6,763 55 464 12,384 37 Total 290,446 10,445 1,573 53,654 91,393 11,374 3,876 65,341 658 2,994 4,812 4,322 Male 174,222	Male		218	9	1,129	2,879	-	103	755	-	184	4,260	150
Total 356,252 16,958 1,881 60,683 113,662 11,915 4,072 75,541 724 3,337 61,303 5,34 Male 203,172 9,126 1,375 39,342 68,261 6,042 2,401 35,886 498 2,325 34,045 3,87 Repairing of old cloths Total 65,806 6,513 308 7,028 22,269 541 196 10,199 66 543 17,118 1,02 Male 28,949 2,241 2,1 4,637 12,655 380 106 3,437 10 79 4,734 64 Female 36,857 4,272 287 2,391 9,614 161 90 6,763 55 46 12,384 37 Other service activities Total 290,446 10,455 15,506 5,602 2,294 32,450 488 2,246 29,311 3,22 Female 174,222 6,865 1,375 5,767 </td <td></td> <td></td> <td>67</td> <td>-</td> <td>73</td> <td>650</td> <td>-</td> <td>-</td> <td>571</td> <td>35</td> <td>-</td> <td>1,457</td> <td>85</td>			67	-	73	650	-	-	571	35	-	1,457	85
Male 203,172 9,126 1,375 39,342 68,261 6,042 2,401 35,886 498 2,325 34,045 3,87 Female 153,080 7,832 507 21,340 45,401 5,873 1,671 39,654 226 1,212 27,885 1,47 Repairing of old cloths 7014 65,806 6,513 308 7,028 22,269 541 196 10,199 66 543 17,118 1,02 Male 28,949 2,241 21 4,637 12,655 380 106 3,437 10 79 4,734 64 Female 36,857 4,272 287 53,654 91,393 11,374 3,876 65,341 658 2,994 44,812 4,32 Male 174,222 6,854 1,573 5,365 5,612 2,294 3,245 648 2,494 2,462 2,911 3,22 Female 161,223 3,560 220 1	OTHER SERVICE ACTIVITIES	S											
Female 153,080 7,832 507 21,340 45,401 5,873 1,671 39,654 226 1,212 27,885 1,47 Repairing of old cloths -	Total	356,252	16,958	1,881	60,683	113,662	11,915	4,072	75,541	724	3,537	61,930	5,349
Repairing of old cloths Total 65,806 6,513 308 7,028 22,269 541 196 10,199 66 543 17,118 1,02 Male 28,949 2,241 21 4,637 12,655 380 106 3,437 10 79 4,734 64 Female 36,857 4,272 287 2,391 9,614 161 90 6,763 55 464 12,384 37 Other service activities Total 290,446 10,445 1,573 53,654 91,393 11,374 3,876 65,341 658 2,994 44,812 4,32 Male 116,223 3,560 220 18,949 35,787 5,712 1,581 32,891 170 749 15,501 1,10 MJOR GROUP 15: PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 1,663 107,665 5,300 Male 249,981 966 66 2,607 5,057 100	Male	203,172	9,126	1,375	39,342	68,261	6,042	2,401	35,886	498	2,325	34,045	3,871
Total 65,806 6,513 308 7,028 22,269 541 196 10,199 66 543 17,118 1,02 Male 28,949 2,241 21 4,637 12,655 380 106 3,437 10 79 4,734 64 Chter service activities 36,857 4,272 287 2,391 9,614 161 90 6,763 55 40 12,384 37 Chter service activities 174,222 6,885 1,573 53,654 91,393 11,374 3,876 65,341 658 2,994 44,812 4,32 Male 174,222 6,885 1,354 34,705 55,676 5,662 2,294 32,450 488 2,466 29,311 3,22 Female 116,223 3,560 220 18,949 35,787 5,712 1,581 32,891 170 79 4,526 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 64<		153,080	7,832	507	21,340	45,401	5,873	1,671	39,654	226	1,212	27,885	1,478
Male 28,949 2,241 21 4,637 12,655 380 106 3,437 10 79 4,734 64 Female 36,857 4,272 287 2,391 9,614 161 90 6,763 55 464 12,384 37 Other service activities 7 290,446 10,445 1,573 53,654 91,393 11,374 3,876 65,341 658 2,994 44,812 4,32 Male 174,222 6,885 1,354 34,705 55,606 5,662 2,294 32,450 488 2,246 29,311 3,227 Female 116,223 3,560 20 18,949 35,787 5,712 1,813 32,891 170 749 15,501 1,10 MAJOR GROUP 15: PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561	Repairing of old cloths												
Female 36,857 4,272 287 2,391 9,614 161 90 6,763 55 464 12,384 37 Other service activities 70tal 290,446 10,445 1,573 53,654 91,393 11,374 3,876 65,341 658 2,994 44,812 4,32 Male 174,222 6,885 1,354 34,705 55,606 5,662 2,294 32,450 488 2,246 29,911 3,22 Female 116,223 3,560 220 18,949 35,787 5,712 1,581 32,891 170 749 15,501 1,100 MAJOR GROUP 15: PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 240,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female	Total	65,806	6,513	308	7,028	22,269	541	196	10,199	66	543	17,118	1,024
Other service activities Total 290,446 10,445 1,573 53,654 91,393 11,374 3,876 65,341 658 2,994 44,812 4,322 Male 174,222 6,885 1,354 34,705 55,606 5,662 2,294 32,450 488 2,246 29,311 3,22 Female 116,223 3,560 220 18,949 35,787 5,712 1,581 32,911 170 749 15,501 1,10 MAJOR GROUP 15: PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86	Male	28,949	2,241	21	4,637	12,655	380	106	3,437	10	79	4,734	649
Total 290,446 10,445 1,573 53,654 91,393 11,374 3,876 65,341 658 2,994 44,812 4,322 Male 174,222 6,885 1,354 34,705 55,606 5,662 2,294 32,450 488 2,246 29,311 3,22 Female 116,223 3,560 220 18,949 35,787 5,712 1,581 32,891 170 749 15,501 1,10 MAJOR GROUP 15: PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,300 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,563 107,665 5,300 Male 24,981 966 66 2,607 5,057 100 23 1,561 - <t< td=""><td>Female</td><td>36,857</td><td>4,272</td><td>287</td><td>2,391</td><td>9,614</td><td>161</td><td>90</td><td>6,763</td><td>55</td><td>464</td><td>12,384</td><td>376</td></t<>	Female	36,857	4,272	287	2,391	9,614	161	90	6,763	55	464	12,384	376
Male 174,222 6,885 1,354 34,705 55,606 5,662 2,294 32,450 488 2,246 29,311 3,22 Female 116,223 3,560 220 18,949 35,787 5,712 1,581 32,891 170 749 15,501 1,10 MAJOR GROUP 15: PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 <td>Other service activities</td> <td></td>	Other service activities												
Female 116,223 3,560 220 18,949 35,787 5,712 1,581 32,891 170 749 15,501 1,10 MAJOR GROUP 15: PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS 7 100 23 1,561 - 135 14,026 44 Female 203,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female	Total	290,446	10,445	1,573	53,654	91,393	11,374	3,876	65,341	658	2,994	44,812	4,325
MAJOR GROUP 15: PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS V V V V V 135 14,026 44 Female 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,663 107,665 5,30 </td <td>Male</td> <td>174,222</td> <td>6,885</td> <td>1,354</td> <td>34,705</td> <td>55,606</td> <td>5,662</td> <td>2,294</td> <td>32,450</td> <td>488</td> <td>2,246</td> <td>29,311</td> <td>3,222</td>	Male	174,222	6,885	1,354	34,705	55,606	5,662	2,294	32,450	488	2,246	29,311	3,222
Total230,3628,9101,67242,29245,4631,82661814,8481041,663107,6655,30Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS71,67242,29245,4631,82661814,8481041,663107,6655,30Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86Private households with employed persons77,9431,60639,68540,4061,72659513,2871041,52893,6394,86Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,287104 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>5,712</td> <td>1,581</td> <td>32,891</td> <td>170</td> <td>749</td> <td>15,501</td> <td>1,103</td>							5,712	1,581	32,891	170	749	15,501	1,103
Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 Private households with employed persons Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561	MAJOR GROUP 15: PRIVAT	E HOUSEHO	OLDS WIT	H EMPI	OYED PE	RSONS							
Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONSTotal230,3628,9101,67242,29245,4631,82661814,8481041,663107,6655,30Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86Private households with employed persons7,9431,60639,68540,4061,72659513,2871041,663107,6655,30Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND MGOS?Total37,4252,1023467,5324,3833,091724,09646042914,36954Male25,4931,	Total	230,362	8,910	1,672	42,292	45,463	1,826	618	14,848	104	1,663	107,665	5,302
PRIVATE HOUSEHOLDS WITH EMPLOYED PERSONS Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 Private households with employed persons Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595	Male	24,981	966	66	2,607	5,057	100	23	1,561	-	135	14,026	440
Total230,3628,9101,67242,29245,4631,82661814,8481041,663107,6655,30Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86Private households with employed persons742,29245,4631,82661814,8481041,663107,6655,30Male24,981966662,6075,057100231,561-13514,02644Female205,3817,9431,60639,68540,4061,726231,561-13514,02644Female205,3817,9431,60639,68540,4061,72659513,2871041,52893,6394,86MAJOR GROUP 16:EXTRA-TERRITORIALVGANIXATIONNE NEUVV1,52893,6394,86V1,72659513,2871041,52893,6394,86Male25,4931,5522,1023467,5324,3833,091724,09646042914,36954Male25,4931,5523263,7313,9151,844512,79534331310,31530	Female	205,381	7,943	1,606	39,685	40,406	1,726	595	13,287	104	1,528	93,639	4,861
Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 Private households with employed persons 7 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND MGOS! Total 37,425 2,102 346 7,532 4,383 3,091 72 4,096 460 429 14,369 54 <td>PRIVATE HOUSEHOLDS WIT</td> <td>TH EMPLOY</td> <td>ED PERS</td> <td>ONS</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	PRIVATE HOUSEHOLDS WIT	TH EMPLOY	ED PERS	ONS									
Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 Private households with employed persons Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND NGOS? Total 37,425 2,102 346 7,532 4,383 3,091 72 4,096 460 429 14,369 54 Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30	Total	230,362	8,910	1,672	42,292	45,463	1,826	618	14,848	104	1,663	107,665	5,302
Private households with employed persons Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND NGOS? Total 37,425 2,102 346 7,532 4,383 3,091 72 4,096 460 429 14,369 54 Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30	Male	24,981	966	66	2,607	5,057	100	23	1,561	-	135	14,026	440
Total 230,362 8,910 1,672 42,292 45,463 1,826 618 14,848 104 1,663 107,665 5,30 Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND NGOS? Total 37,425 2,102 346 7,532 4,383 3,091 72 4,096 460 429 14,369 54 Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30	Female	205,381	7,943	1,606	39,685	40,406	1,726	595	13,287	104	1,528	93,639	4,861
Male 24,981 966 66 2,607 5,057 100 23 1,561 - 135 14,026 44 Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND NGOS? Total 37,425 2,102 346 7,532 4,383 3,091 72 4,096 460 429 14,369 54 Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30	Private households with emplo	yed persons	6										
Female 205,381 7,943 1,606 39,685 40,406 1,726 595 13,287 104 1,528 93,639 4,86 MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND NGOS? Total 37,425 2,102 346 7,532 4,383 3,091 72 4,096 460 429 14,369 54 Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30	Total	230,362	8,910	1,672	42,292	45,463	1,826	618	14,848	104	1,663	107,665	5,302
MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND NGOS? Total 37,425 2,102 346 7,532 4,383 3,091 72 4,096 460 429 14,369 54 Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30								23	1,561	-		14,026	440
MAJOR GROUP 16:EXTRA-TERRITORIAL ORGANIZATIONS AND NGOS? Total 37,425 2,102 346 7,532 4,383 3,091 72 4,096 460 429 14,369 54 Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30	Female	205,381	7,943	1,606	39,685	40,406	1,726	595	13,287	104	1,528	93,639	4,861
Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30	MAJOR GROUP 16:EXTRA-1	ERRITORIA	L ORGA	NIZATIC	NS AND I								
Male 25,493 1,552 326 3,731 3,915 1,844 51 2,795 343 313 10,315 30	Total	37,425	2,102	346	7,532	4,383	3,091	72	4,096	460	429	14,369	543
			•			3,915							309
	Female	11,931	550		3,801	468	1,247		1,302	117		4,054	234

Table 5.10 Cont'd

Major Industrial Divisions						Re	egion					
and Sex	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul -Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
EXTRA-TERRITORIAL ORGA	NIZATIONS											
Total	37,425	2,102	346	7,532	4,383	3,091	72	4,096	460	429	14,369	543
Male	25,493	1,552	326	3,731	3,915	1,844	51	2,795	343	313	10,315	309
Female	11,931	550	21	3,801	468	1,247	21	1,302	117	116	4,054	234
Extra-territorial organizations												
Total	37,425	2,102	346	7,532	4,383	3,091	72	4,096	460	429	14,369	543
Male	25,493	1,552	326	3,731	3,915	1,844	51	2,795	343	313	10,315	309
Female	11,931	550	21	3,801	468	1,247	21	1,302	117	116	4,054	234
Not stated												
Total	2,188	341	-	926	553	-	-	299	-	-	-	69
Male	1,296	341	-	429	457	-	-	-	-	-	-	69
Female	892	-	-	497	95	-	-	299	-	-	-	-

	buntry-1 ot						Status	in Employme	ent						
					Paid Er	nployees	C tatut								
Age Group Sex	Total Employed Population	Total Paid Employees	Employee- Gov't	Employee- Government Parasitatal	Employee- Private organization	Employee- NGO (including International organization)	Employee Domestic	Other Employees	Self Employ ed	Unpaid Family Worker	Employer	Apprentice	Member of Co-operatives	Other	Not Stated
All Ages									2,000,0						
Total	5,139,831	2,544,615	993,158	98,868	995,660	86,746	323,515	46,668		433,630	41,650	6,857	39,861	71,968	1,195
Male	2,928,953	1,537,512	636,591	67,274	656,355	55,816	85,736	35,739		176,819	34,103	5,046	25,582	41,225	455
Female 10-14	2,210,879	1,007,104	356,567	31,594	339,304	30,930	237,779	10,929	891,844	256,811	7,547	1,811	14,279	30,743	740
Total	144,811	27,621	-	-	4,275	-	23,346	-	16,725	94,727	-	558	-	5,180	-
Male	71,614	4,717	-	-	1,914	-	2,803	-	11,818	52,643	-	558	-	1,877	-
Female 15-19	73,197	22,904	-	-	2,361	-	20,543	-	4,907	42,084	-	-	-	3,303	-
Total	429,517	207,088	13,871	3,309	90,536	1,743	96,409	1,221			-	1,713	2,549	8,085	55
Male	192,665	68,766	5,511	2,766	49,211	619	9,439	1,221	56,324	60,604	-	957	1,238	4,721	55
Female 20-24	236,852	138,322	8,361	543	41,325	1,123	86,970	-	40,265	52,835	-	756	1,311	3,364	-
Total	797,478	472,677	133,337	15,208	234,743	13,631	69,237	6,521	228,371	70,458	3,395	2,629	7,076	12,466	405
Male	404,296	230,277	61,847	8,142	136,947	5,283	13,743		125,281	31,253	2,500	1,749	5,154	8,081	-
Female 25-29	393,182	242,400	71,490	7,066	97,796	8,349	55,495	2,205	103,091	39,205	894	880	1,922	4,385	405
Total	1,009,623	566,288	230,384	16,574	240,989	16,211	52,030	10,101	351,984	54,722	6,942	1,757	9,859	17,395	676
Male	579,802	342,486	143,680	10,898	153,510	8,618	18,224		196,937	17,239	5,621	1,582	7,212	8,384	341
Female 30-34	429,821	223,802	86,703	5,676	87,479	7,592	33,805	2,546	155,047	37,483	1,321	175	2,647	9,011	335
Total	719,297	363,130	146,435	14,492	147,957	17,707	28,335	8,204		34,839	8,892	-	7,336	10,278	59
Male	439,566	242,682	92,155	10,817	108,285	12,628	13,091		172,766	4,995	7,643	-	4,888	6,534	59
Female 35-39	279,731	120,448	54,280	3,675	39,672	5,080	15,244		121,998	29,844	1,249	-	2,448	3,744	-
Total	656,750	325,176	161,814	13,290	111,066	14,817	18,286		291,537	24,181	6,358	200	4,416	4,881	-
Male	363,332	205,216	99,310	9,525	76,304	8,924	7,307		144,969	2,963	4,965	200	1,543	3,475	-
Female 40-44	293,418	119,960	62,504	3,764	34,762	5,893	10,979		146,568	21,218	1,393	-	2,873	1,405	-
Total	409,957	190,036	110,394	8,798	49,729	7,317	10,152		197,492	11,190	7,033	-	1,641	2,565	-
Male	269,401	142,147	85,816	4,454	36,817	6,349	5,641		117,579	1,365	5,560	-	1,201	1,549	-
Female	140,556	47,888	24,578	4,344	12,912	968	4,511	575	79,913	9,825	1,473	-	440	1,016	-

Table 5.11 Currently Employed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Status In Employment Country-Total: 2011

Table 5.11 Cont'd

							Status	in Employme	nt						
					Paid Er	nployees									
Age Group Sex	Total Employed Population	Total Paid Employees	Employee- Gov't	Employee- Government Parasitatal	Employee- Private organization	Employee-NGO (including International organization)	Employee Domestic	Other Employees	Self Employ ed	Unpaid Family Worker	Employer	Apprentice	Member of Co-operatives	Other	Not Stated
45-49															
Total	326,320	151,249	85,137	9,462	43,456	4,626	6,541	2,027	153,875	10,450	2,898	-	3,647	4,201	-
Male	196,881	112,317	62,963	7,124	33,477	3,419	3,948	1,386	75,729	1,659	2,548	-	2,052	2,576	-
Female 50-54	129,439	38,932	22,174	2,338	9,979	1,207	2,594	640	78,146	8,791	350	-	1,595	1,626	-
Total	232,556	107,891	59,373	7,903	24,973	4,341	7,961	3,341	110,717	8,589	1,064	-	1,101	3,194	-
Male	138,821	76,172	41,319	5,723	17,397	4,271	4,197	3,265	59,166	665	577	-	475	1,766	-
Female	93,735	31,719	18,054	2,179	7,576	70	3,764	75	51,551	7,924	488	-	626	1,427	-
55-59															
Total	152,901	63,833	33,246	4,966	17,151	3,358	3,700	1,412	77,331	6,057	3,068	-	955	1,656	-
Male	92,826	50,605	27,111	4,250	13,742	2,709	1,713	1,079	36,561	1,291	2,731	-	613	1,026	-
Female 60-64	60,074	13,229	6,135	716	3,409	648	1,987	333	40,770	4,766	336	-	342	631	-
Total	108,220	32,155	10,536	2,072	14,794	1,052	2,140	1,561	70,772	1,988	967	-	1,033	1,304	-
Male	72,874	28,719	9,496	1,405	13,473	1,052	1,733	1,561	41,286	287	967	-	1,033	581	-
Female	35,346	3,436	1,040	667	1,322	-	408	-	29,486	1,701	-	-	-	724	-
65+															
Total	152,402	37,470	8,631	2,795	15,991	1,944	5,377	2,733	109,898	2,990	1,032	-	247	764	-
Male	106,875	33,407	7,383	2,169	15,280	1,944	3,898	2,733	69,795	1,854	990	-	173	655	-
Female	45,527	4,063	1,248	626	712	-	1,479	-	40,103	1,135	42	-	75	109	-

							Statu	is in Employ	ment						
					Paid Emplo	yees									
Age Group and Sex	Total Employed Population	Total Paid Employees	Employee- Gov't	Employee- Gov't Parasitatal	Employee- Private organization	Employee- NGO (including Internation al Org.)	Employee Domestic	Other Employee	Self Employed	Unpaid Family Worker	Employer	Apprentice	Member of Co-operatives	Other	Not Stated
Country Total															
Total	5,139,831	2,544,615	993,158	98,868	995,660	86,746	323,515		2,000,055	433,630		6,857	39,861	71,968	1,195
Male	2,928,953	1,537,512	636,591	67,274	656,355	55,816	85,736	35,739	1,108,211	176,819		5,046	25,582	41,225	455
Female	2,210,879	1,007,104	356,567	31,594	339,304	30,930	237,779	10,929	891,844	256,811	7,547	1,811	14,279	30,743	740
Tigray Region															
Total	321,569	168,875	69,430	5,862	62,449	3,192	24,986	2,955	124,261	17,654	1	580	4,668	2,536	341
Male	173,501	99,581	41,727	3,479	36,385	2,562	12,955	2,474	57,440	8,634		580	3,287	1,522	341
Female	148,068	69,293	27,704	2,383	26,064	629	12,032	482	66,821	9,020	539	-	1,381	1,014	-
Affar Region															
Total	71,536	41,121	16,514	5,146	16,869	562	2,020	10	28,747	626		-	94	927	-
Male	45,023	27,270	10,672	3,956	11,897	521	213	10	16,894	232		-	94	524	-
Female	26,512	13,851	5,841	1,190	4,972	41	1,807	-	11,854	395	10	-	-	403	-
Amhara Region															
Total	921,246	377,588	174,790	10,473	121,862	10,212	54,299	5,952	411,784	99,526		1,145	6,744	17,126	464
Male	505,459	217,148	112,544	5,421	78,597	5,086	10,932	4,569	224,382	44,830		969	4,848	6,768	59
Female	415,787	160,440	62,246	5,052	43,265	5,127	43,368	1,383	187,402	54,696	415	176	1,896	10,358	405
Oromia Region															
Total	1,609,473	731,002	295,214	24,933	312,631	21,893	62,980	13,350	681,437	153,206		2,608	11,364	12,772	90
Male	930,772	470,542	196,729	18,037	213,889	13,765	18,491	9,631	363,943	62,373		1,243	8,426	10,018	55
Female Somali Region	678,701	260,459	98,485	6,896	98,742	8,128	44,489	3,720	317,494	90,833	2,822	1,364	2,938	2,755	35
Total	167,435	59,105	36,466	5,806	8,663	4,288	2,717	1,166	77,775	27,410	1,916	38	585	606	-
Male	98,864	44,322	28,793	3,918	7,041	2,851	731	987	42,903	9,229		-	585	304	-
Female	68,571	14,784	7,673	1,888	1,622	1,436	1,986	179	34,872	18,181		38	-	302	-
Benishangul Gu		,			• -										
Total	53,221	22,441	15,069	831	4,364	461	1,061	654	24,550	5,222	303	193	117	396	-
Male	30,493	15,254	10,136	698	3,284	378	198	560	12,299	2,206		80	96	286	-
Female	22,729	7,187	4,934	134	1,080	83	863	93	12,251	3,016		112	22	110	-

Table 5.12 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Status In Employment: 2011

Table 5.12 Cont'd

							Statu	is in Employ	ment						
					Paid Emplo	oyees									
Age Group and Sex	Total Employed Population	Total Paid Employees	Employee- Gov't	Employee- Gov't Parasitatal	Employee- Private organization	Employee- NGO (including Internation al Org.)		Other Employee	Self Employed	Unpaid Family Worker	Employer	Apprentice	Member of Co-operatives	Other	Not Stated
SNNP Region															
Total	680,560	295,972	158,106	10,314	86,510	11,571	24,769	4,701	282,994	85,170	2,659	837	4,790	7,839	299
Male	387,610	187,250	100,934	8,588	59,819	7,461	7,343	3,104	160,452	31,305	1,550	837	3,123	3,093	-
Female	292,950	108,722	57,172	1,726	26,691	4,110	17,426	1,597	122,542	53,865	1,109	-	1,667	4,745	299
Gambella Region															
Total	33,342	12,811	8,808	131	2,730	643	200	300	13,329	6,939	67	23	37	138	-
Male	16,628	8,930	6,183	108	1,856	404	79	300	5,912	1,671	42	23	-	49	-
Female	16,715	3,882	2,625	23	874	239	121	-	7,417	5,267	25	-	37	88	-
Harari Region															
Total	48,450	25,399	11,836	1,001	8,284	877	2,365	1,037	17,624	3,487	834	46	740	321	-
Male	25,806	14,963	6,796	809	5,385	527	496	951	7,522	1,731	781	-	488	321	-
Female Addis Ababa City Administration	22,644	10,436	5,040	192	2,899	350	1,869	86	10,102	1,755	53	46	252	-	-
Total	1,148,974	769,581	196,406	32,092	351,204	31,983	142,501	15,396	305,238	27,801	7,323	1,239	9,824	27,968	_
Male	667,973	427,036	190,400	20,883	223,895	21,811	33,684	12,155	200,955	11,662		1,239	3,901	17,283	
Female	481,001	342,546	81,798	11,210	127,309	10,172	108,816	3,240	104,283	16,138		1,237	5,923	10,685	-
Dire Dawa Administration	401,001	542,540	01,770	11,210	127,307	10,172	100,010	3,240	104,203	10,130	1,420	-	5,725	10,005	-
Total	84,025	40,721	10,518	2,279	20,093	1,066	5,617	1,147	32,316	6,591	2,009	149	899	1,340	-
Male	46,825	25,217	7,469	1,378	14,308	450	614	997	15,509	2,944	1,286	75	736	1,058	-
Female	37,200	15,504	3,049	901	5,785	616	5,003	150	16,807	3,646		75	163	282	-

Table 5.13 C						ed l'en rears		Employment							
					Paid Em	plovees									
Major Towns and Sex	Total Employed Population	Total Paid Employees	Employee- Government	Employee- Government Parasitatal	Employee- Private organization	Employee- NGO (including International organization)	Employee Domestic	Other Employee	Self Employed	Unpaid Family Worker	Employer	Apprentice	Member of Co- Operatives	Other	Not Stated
COUNTRY TOTA	L														
Total	1,965,425	1,221,824	360,182	46,705	547,545	45,135	200,006	22,251	593,965	80,677	13,638	2,172	18,841	34,124	185
Male	1,134,199	690,196	216,375	30,220	350,296	30,108	45,422	17,776	365,277	34,802	10,492	1,751	10,103	21,463	114
Female	831,226	531,628	143,807	16,485	197,249	15,027	154,584	4,475	228,688	45,874	3,146	421	8,737	12,661	71
MEKELE TOWN															
Total	87,546	55,231	18,066	2,775	23,412	942	9,793	243	26,660	2,976	1,096	84	892	606	-
Male	49,393	32,555	10,744	1,375	14,904	684	4,695	153	13,604	1,393	725	84	731	302	-
Female	38,153	22,677	7,322	1,401	8,508	258	5,097	90	13,056	1,583	371	-	162	304	-
ASAYITA TOWN															
Total	6,472	3,156	1,498	360	876	60	352	10	3,064	211	20	-	21	-	-
Male	3,810	2,029	938	318	709	19	35	10	1,711	38	10	-	21	-	-
Female	2,663	1,127	560	42	167	41	317	-	1,353	173	10	-	-	-	-
GONDER TOWN															
Total	82,743	32,272	11,253	572	15,391	646	3,789	622	29,417		123	-	635	1,635	
Male	45,705	17,707	6,433	174	9,883	474	367	377	17,864	8,366	123	-	556	1,090	
Female	37,038	14,565	4,820	398	5,508	172	3,422	245	11,552	10,297	-	-	79	545	-
DESSIE TOWN															
Total	46,017	25,776	10,658	723	8,272	933	4,401	789	17,469	1,671	127	106	477	355	
Male	25,986	14,343	6,912	639	5,170	697	396	529	10,585	518	83	51	193	214	
Female	20,032	11,433	3,746	84	3,102	236	4,005	260	6,885	1,154	44	54	284	141	36
BAHIR DAR TOW	'N														
Total	89,993	57,983	19,099	1,905	26,381	1,094	8,826	679	25,798	4,490	73	198	530	862	
Male	46,674	29,670	11,244	1,389	15,276	684	499	578	13,792	2,108	73	76	308	587	59
Female BISHOFTU TOWN	43,319	28,313	7,855	516	11,105	410	8,326	101	12,006	2,382	-	122	221	275	-
Total	40,673	25,710	7,543	582	13,825	881	2,468	410	12,308	899	251	79	1,016	375	35
Male	22,604	14,843	4,539	305	8,328	349	946	375	6,397	430	124	40	528	241	
Female	18,069	10,867	3,004	278	5,497	532	1,522	35	5,911	469	127	39	487	133	35
ADAMA TOWN															
Total	101,178	57,300	14,857	1,901	33,300	1,189	5,516	536	39,100	3,269	395	-	980	135	-
Male	60,623	32,881	9,796	1,246	19,451	957	895	536	25,541	1,049	395	-	623	135	-
Female	40,555	24,419	5,061	655	13,849	233	4,621	-	13,559	2,220	-	-	357	-	-
JIMMA TOWN															
Total	55,260	38,264	17,810	679	13,130	1,696	4,405	545	14,664	1,421	432	47	378	-	55
Male	33,297	23,063	10,903	483	9,386	1,083	755	452	8,905	613	432	-	230	-	55
Female	21,963	15,201	6,907	195	3,744	613	3,650	93	5,758	808	-	47	148	-	-

Table 5.13 Currently Employed Population of Major Towns Aged Ten Years and Above by Sex and Status in Employment: 2011

							Status in E	mployment							
					Paid Em	oloyees									
Major Towns and Sex	Total Employed Population	Total Paid Employees	Employee- Government	Employee- Government Parasitatal	Employee- Private organization	Employee- NGO (including International organization)	Employee Domestic	Other Employee	Self Employed	Unpaid Family Worker	Employer	Apprentice	Member of Co- Operatives	Other	Not Stated
SHASHEMENE T	OWN				. 2		•								
Total	54,907	17,521	4,908	497	7,790	414	3,698	213	33,249	2,765	280	-	861	232	-
Male	32,752	10,623	3,439	497	5,405	269	800	213	19,785	1,530	49	-	766	-	-
Female	22,155	6,898	1,470	-	2,385	145	2,898	-	13,464	1,235	232	-	95	232	-
JIJIGA TOWN															
Total	29,089	15,134	7,371	193	5,646	595	1,279	50	12,345	1,105	375	38	92	-	-
Male	20,272	11,155	5,589	119	4,542	384	471	50	8,210	526	289	-	92	-	-
Female	8,817	3,979	1,783	74	1,104	211	808	-	4,135	579	87	38	-	-	-
ASOSA TOWN															
Total	13,319	8,808	5,383	283	2,412	144	428	158	3,647	614	174	21	56	-	-
Male	7,830	5,617	3,356	192	1,803	104	46	116	1,955	62	142	21	34	-	-
Female HAWASSA TOWN	5,489	3,191	2,027	91	610	40	382	42	1,692	552	31	-	22	-	-
Total	64,568	40,868	16,981	732	16,327	2,098	4,437	294	17,335	4,401	84	144	1,440	296	-
Male	37,212		9,723	306		1,249	-	163	11,111	1,690	42		895	232	
Female	27,356	17,770	7,257	427	5,311	849	3,795	131	6,224	2,711	42	-	545	64	-
GAMBELLA TOW	N														
Total	12,209	8,099	5,995	131	1,202	518	132	121	3,732	313	42	23	-	-	-
Male	7,437	5,398	3,887	108	835	367	79	121	1,831	142	42	23	-	-	-
Female	4,772	2,700	2,107	23	367	151	52	-	1,901	171	-	-	-	-	-
HARAR TOWN															
Total	48,450	25,399	11,836	1,001	8,284	877	2,365	1,037	17,624	3,487	834	46	740	321	-
Male	25,806	14,963	6,796	809	5,385	527	496	951	7,522	1,731	781	-	488	321	-
Female	22,644	10,436	5,040	192	2,899	350	1,869	86	10,102	1,755	53	46	252	-	-
ADDIS ABABA CI	TY														
Total	1,148,974	769,581	196,406	32,092	351,204	31,983	142,501	15,396	305,238	27,801	7,323	1,239	9,824	27,968	-
Male	667,973	427,036	114,608	20,883	223,895	21,811	33,684	12,155	200,955		5,897	1,239		17,283	-
Female	481,001	342,546	81,798	11,210	127,309	10,172	108,816	3,240	104,283	16,138	1,426	-	5,923	10,685	-
DIRE DAWA TOW	/N														
Total	84,025	40,721	10,518	2,279	20,093	1,066	5,617	1,147	32,316	6,591	2,009	149	899	1,340	-
Male	46,825	25,217	7,469	1,378	14,308	450	614	997	15,509	2,944	1,286	75	736	1,058	-
Female	37,200	15,504	3,049	901	5,785	616	5,003	150	16,807	3,646	723	75	163	282	-

				N		upational G	roupe				
				I I			loups		Γ	[
Major Industrial Division and Sex	Total Employed	Legislator Senior Officials &	Destancional	Technical and Associate	Olaska	Service, Shop & Market sales	Skilled agricultural and	Crafts and related	Plant ,Machine Operators &		Not
Total Employed Population	Persons	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
	5 400 004		000 044	000.000	000 004	4 075 000	004.047	050 074	000 000	4 400 004	4 000
Total Male	5,139,831	153,155				1,275,292		958,971	268,268	1,132,631	1,209
	2,928,953	123,323 29,832	226,454		119,703	576,401		576,176	246,870	581,582	470
Female Agriculture, Hunting, Forestry &	2,210,879	29,832	109,557	111,781	170,288	698,891	134,548	382,795	21,398	551,049	740
	•		0.400	5.070	0.040	4 004	0.40,000	0.470	0.040	400.040	
Total Male	486,402	4,414	2,138	5,279	6,319	4,821	349,329 233,128	3,170	2,919	108,013	-
Female	311,909	3,486 928	2,006 132	4,477 802	3,296	2,365		2,919 251	2,919	57,313 50,700	-
Mining & Quarrying	174,493	928	132	802	3,023	2,455	116,202	201	-	50,700	-
Total	18,178	1,805	949	62	287	50	277	7,207	1 077	5,555	
Male	15,790	1,805		62	- 201	59 -	277	6,442	1,977 1,977	5,003	-
Female	2,388	353	370	- 02	- 287	- 59	- 211	765	1,977	552	-
Manufacturing	2,500	555	572	-	207		-	705	-	552	-
Total	675,481	7,446	10,946	9,364	25,508	34,903	6.030	434,934	62,560	83,780	_
Male	370,099	6,270	6,981	7,122	13,248	11,367		227,660	47,108	47,192	
Female	305,382	1,177	3,965	2,242	12,260	23,536		207,274	15,452	36,588	-
Electricity, Gas and Water Sup		1,177	0,000	2,272	12,200	20,000	2,000	201,214	10,402	00,000	
Total	59,456	3,464	3,194	4,830	7,524	3,565	-	11,870	6,696	18,314	_
Male	38,832	2,652		4,306	2,326	1,158	-	11,084	6,287	8,352	-
Female	20,624	812		524	5,198	2,407	-	785	410	9,962	-
Construction	20,021	012	020	021	0,100	2,101			110	0,002	
Total	380,581	5,752	12,945	10,010	10,799	2,282	-	221,965	12,982	103,846	-
Male	318,433	5,426	10,023	7,998	4,187	1,309		200,459	12,528	76,504	-
Female	62,148	326	2,923	2,011	6,612	973	-		455	27,342	-
Whole Sale and Retail Trade	,		,	,	,					,	
Total	1,099,758	4,393	4,611	12,704	12,562	803,371	25,627	103,427	17,348	115,715	-
Male	580,722	3,609	2,235	10,089	5,239	390,345	16,573	72,073	15,920	64,638	-
Female	519,036	784	2,375	2,615	7,324	413,025	9,053	31,354	1,428	51,077	-
Hotels and Restaurants				-			•	,	-	•	
Total	408,076	5,821	1,942	1,480	13,786	244,988	340	104,189	1,184	34,346	-
Male	105,545	3,830	495	747	5,592	70,970	42	10,069	1,184	12,616	-
Female	302,531	1,992	1,447	733	8,193	174,018	298	94,120	-	21,730	-

Table 5.14Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Occupation Groups and Major
Industrial Divisions-Country Total: 2011

Table 5.14 Cont'd

Persons Managers Professional Professionals Clerks Workers Fishery trades Assemblers Transport, Storage and Communications Total 242,410 3,871 3,595 7,413 13,946 9,579 2,255 5,320 132,905 Male 226,779 3,189 2,250 7,124 7,299 8,010 2,255 4,827 131,498 Female 15,631 682 1,345 289 6,647 1,568 - 493 1,407 Financial Intermediation Total 91,143 9,498 16,792 8,570 40,829 - - 237 1,814	63,526 60,327 3,199 13,402 6,280	Not Stated - -
Transport, Storage and Communications Total 242,410 3,871 3,595 7,413 13,946 9,579 2,255 5,320 132,905 Male 226,779 3,189 2,250 7,124 7,299 8,010 2,255 4,827 131,498 Female 15,631 682 1,345 289 6,647 1,568 - 493 1,407 Financial Intermediation Total 91,143 9,498 16,792 8,570 40,829 - - 237 1,814	60,327 3,199 13,402 6,280	-
Male226,7793,1892,2507,1247,2998,0102,2554,827131,498Female15,6316821,3452896,6471,568-4931,407Financial IntermediationTotal91,1439,49816,7928,57040,8292371,814	60,327 3,199 13,402 6,280	- -
Female 15,631 682 1,345 289 6,647 1,568 - 493 1,407 Financial Intermediation Total 91,143 9,498 16,792 8,570 40,829 - - 237 1,814	3,199 13,402 6,280	-
Financial Intermediation 91,143 9,498 16,792 8,570 40,829 - - 237 1,814	13,402 6,280	-
Total 91,143 9,498 16,792 8,570 40,829 237 1,814	6,280	
	6,280	
		-
Male 51,635 8,619 9,721 4,963 20,002 237 1,814		-
Female 39,508 878 7,072 3,608 20,828	7,122	-
Real Estate, Renting and Business Activities		
Total 116,708 8,980 17,690 16,795 29,681 10,764 33 15,019 5,408	12,339	-
Male 76,784 7,439 12,028 12,788 12,502 9,328 33 10,061 5,026	7,580	-
Female 39,924 1,540 5,662 4,007 17,179 1,436 - 4,958 382 PUBLIC ADMINISTRATION AND DEFENCE	4,759	-
Total 324,536 71,693 37,973 58,159 59,757 47,981 464 6,632 9,062	32,815	-
Male 222,301 56,378 28,550 44,773 22,454 39,924 55 3,103 8,908	18,156	-
Female 102,234 15,315 9,423 13,386 37,304 8,057 409 3,528 153	14,659	-
Education	1,000	
Total 346,352 8,101 161,466 102,197 21,685 7,051 178 4,810 3,220	37,645	-
Male 201,522 6,447 110,303 54,664 4,440 956 178 2,266 3,220	19,047	-
Female 144,831 1,654 51,163 47,532 17,245 6,095 - 2,543 -	18,598	-
Health and Social Work	-,	
Total 161,675 6,366 40,180 62,778 18,061 7,643 189 1,240 3,051	22,168	-
Male 80,565 5,172 21,625 35,509 4,806 1,963 189 875 3,051	7,374	-
Female 81,110 1,194 18,555 27,269 13,254 5,679 - 364 -	14,794	-
Other Community, Social and Personal Service Activities		
Total 459,078 5,474 17,519 25,515 23,105 73,679 8,211 30,743 3,864	270,969	-
Male 276,244 4,323 14,626 20,730 13,106 35,776 2,514 22,461 2,153	160,557	-
Female 182,834 1,151 2,893 4,785 9,999 37,903 5,697 8,282 1,711	110,412	-

Table 5.14 Cont'd

				Ν	lajor Occu	upational G	roups				
Major Industrial Division		Legislator		Technical		Service, Shop &	Skilled	Crafts	Plant		
and Sex	Total	Senior		and		Market	agricultural	and	,Machine		
and Sex	Employed	Officials &		Associate		sales	and	related	Operators &	Elementary	Not
	Persons	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
Private Households with Emplo	yed Persons	6									
Total	230,362	451	-	297	561	22,020	630	7,042	84	199,277	-
Male	24,981	-	-	-	-	1,791	630	636	84	21,840	-
Female	205,381	451	-	297	561	20,229	-	6,406	-	177,437	-
Extra-Territorial Organizations a	and NGOs										
Total	37,425	5,626	4,072	4,834	5,467	2,527	76	710	3,194	10,919	-
Male	25,493	5,032	2,367	3,154	1,183	1,138	76	547	3,194	8,802	-
Female	11,931	594	1,705	1,680	4,283	1,389	-	163	-	2,117	-
Not Stated											
Total	2,188	-	-	-	92	60	369	457	-	-	1,209
Male	1,296	-	-	-	-	-	369	457	-	-	470
Female	892	-	-	-	92	60	-	-	-	-	740

				N		upational G	roupe				
				I I			loups		Γ	[
Major Industrial Division and Sex	Total Employed	Legislator Senior Officials &	Destancional	Technical and Associate	Olaska	Service, Shop & Market sales	Skilled agricultural and	Crafts and related	Plant ,Machine Operators &		Not
Total Employed Population	Persons	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
	5 400 004		000 044	000.000	000 004	4 075 000	004.047	050 074	000 000	4 400 004	4 000
Total Male	5,139,831	153,155				1,275,292		958,971	268,268	1,132,631	1,209
	2,928,953	123,323 29,832	226,454		119,703	576,401		576,176	246,870	581,582	470
Female Agriculture, Hunting, Forestry &	2,210,879	29,832	109,557	111,781	170,288	698,891	134,548	382,795	21,398	551,049	740
	•		0.400	5.070	0.040	4 004	0.40,000	0.470	0.040	400.040	
Total Male	486,402	4,414	2,138	5,279	6,319	4,821	349,329 233,128	3,170	2,919	108,013	-
Female	311,909	3,486 928	2,006 132	4,477 802	3,296	2,365		2,919 251	2,919	57,313 50,700	-
Mining & Quarrying	174,493	928	132	802	3,023	2,455	116,202	201	-	50,700	-
Total	18,178	1,805	949	62	287	50	277	7,207	1 077	5,555	
Male	15,790	1,805		62	- 201	59 -	277	6,442	1,977 1,977	5,003	-
Female	2,388	353	370	- 02	- 287	- 59	- 211	765	1,977	552	-
Manufacturing	2,500	555	572	-	207		-	705	-	552	-
Total	675,481	7,446	10,946	9,364	25,508	34,903	6.030	434,934	62,560	83,780	_
Male	370,099	6,270	6,981	7,122	13,248	11,367		227,660	47,108	47,192	
Female	305,382	1,177	3,965	2,242	12,260	23,536		207,274	15,452	36,588	-
Electricity, Gas and Water Sup		1,177	0,000	2,272	12,200	20,000	2,000	201,214	10,402	00,000	
Total	59,456	3,464	3,194	4,830	7,524	3,565	-	11,870	6,696	18,314	_
Male	38,832	2,652		4,306	2,326	1,158	-	11,084	6,287	8,352	-
Female	20,624	812		524	5,198	2,407	-	785	410	9,962	-
Construction	20,021	012	020	021	0,100	2,101			110	0,002	
Total	380,581	5,752	12,945	10,010	10,799	2,282	-	221,965	12,982	103,846	-
Male	318,433	5,426	10,023	7,998	4,187	1,309		200,459	12,528	76,504	-
Female	62,148	326	2,923	2,011	6,612	973	-		455	27,342	-
Whole Sale and Retail Trade	,		,	,	,					,	
Total	1,099,758	4,393	4,611	12,704	12,562	803,371	25,627	103,427	17,348	115,715	-
Male	580,722	3,609	2,235	10,089	5,239	390,345	16,573	72,073	15,920	64,638	-
Female	519,036	784	2,375	2,615	7,324	413,025	9,053	31,354	1,428	51,077	-
Hotels and Restaurants				-			•	,	-	•	
Total	408,076	5,821	1,942	1,480	13,786	244,988	340	104,189	1,184	34,346	-
Male	105,545	3,830	495	747	5,592	70,970	42	10,069	1,184	12,616	-
Female	302,531	1,992	1,447	733	8,193	174,018	298	94,120	-	21,730	-

Table 5.14Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Occupation Groups and Major
Industrial Divisions-Country Total: 2011

Table 5.14 Cont'd

Persons Managers Professional Professionals Clerks Workers Fishery trades Assemblers Transport, Storage and Communications Total 242,410 3,871 3,595 7,413 13,946 9,579 2,255 5,320 132,905 Male 226,779 3,189 2,250 7,124 7,299 8,010 2,255 4,827 131,498 Female 15,631 682 1,345 289 6,647 1,568 - 493 1,407 Financial Intermediation Total 91,143 9,498 16,792 8,570 40,829 - - 237 1,814	63,526 60,327 3,199 13,402 6,280	Not Stated - -
Transport, Storage and Communications Total 242,410 3,871 3,595 7,413 13,946 9,579 2,255 5,320 132,905 Male 226,779 3,189 2,250 7,124 7,299 8,010 2,255 4,827 131,498 Female 15,631 682 1,345 289 6,647 1,568 - 493 1,407 Financial Intermediation Total 91,143 9,498 16,792 8,570 40,829 - - 237 1,814	60,327 3,199 13,402 6,280	-
Male226,7793,1892,2507,1247,2998,0102,2554,827131,498Female15,6316821,3452896,6471,568-4931,407Financial IntermediationTotal91,1439,49816,7928,57040,8292371,814	60,327 3,199 13,402 6,280	- -
Female 15,631 682 1,345 289 6,647 1,568 - 493 1,407 Financial Intermediation Total 91,143 9,498 16,792 8,570 40,829 - - 237 1,814	3,199 13,402 6,280	-
Financial Intermediation 91,143 9,498 16,792 8,570 40,829 - - 237 1,814	13,402 6,280	-
Total 91,143 9,498 16,792 8,570 40,829 237 1,814	6,280	
	6,280	
		-
Male 51,635 8,619 9,721 4,963 20,002 237 1,814		-
Female 39,508 878 7,072 3,608 20,828	7,122	-
Real Estate, Renting and Business Activities		
Total 116,708 8,980 17,690 16,795 29,681 10,764 33 15,019 5,408	12,339	-
Male 76,784 7,439 12,028 12,788 12,502 9,328 33 10,061 5,026	7,580	-
Female 39,924 1,540 5,662 4,007 17,179 1,436 - 4,958 382 PUBLIC ADMINISTRATION AND DEFENCE	4,759	-
Total 324,536 71,693 37,973 58,159 59,757 47,981 464 6,632 9,062	32,815	-
Male 222,301 56,378 28,550 44,773 22,454 39,924 55 3,103 8,908	18,156	-
Female 102,234 15,315 9,423 13,386 37,304 8,057 409 3,528 153	14,659	-
Education	1,000	
Total 346,352 8,101 161,466 102,197 21,685 7,051 178 4,810 3,220	37,645	-
Male 201,522 6,447 110,303 54,664 4,440 956 178 2,266 3,220	19,047	-
Female 144,831 1,654 51,163 47,532 17,245 6,095 - 2,543 -	18,598	-
Health and Social Work	-,	
Total 161,675 6,366 40,180 62,778 18,061 7,643 189 1,240 3,051	22,168	-
Male 80,565 5,172 21,625 35,509 4,806 1,963 189 875 3,051	7,374	-
Female 81,110 1,194 18,555 27,269 13,254 5,679 - 364 -	14,794	-
Other Community, Social and Personal Service Activities		
Total 459,078 5,474 17,519 25,515 23,105 73,679 8,211 30,743 3,864	270,969	-
Male 276,244 4,323 14,626 20,730 13,106 35,776 2,514 22,461 2,153	160,557	-
Female 182,834 1,151 2,893 4,785 9,999 37,903 5,697 8,282 1,711	110,412	-

Table 5.14 Cont'd

				Ν	lajor Occu	upational G	roups				
Major Industrial Division		Legislator		Technical		Service, Shop &	Skilled	Crafts	Plant		
and Sex	Total	Senior		and		Market	agricultural	and	,Machine		
and Sex	Employed	Officials &		Associate		sales	and	related	Operators &	Elementary	Not
	Persons	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
Private Households with Emplo	yed Persons	6									
Total	230,362	451	-	297	561	22,020	630	7,042	84	199,277	-
Male	24,981	-	-	-	-	1,791	630	636	84	21,840	-
Female	205,381	451	-	297	561	20,229	-	6,406	-	177,437	-
Extra-Territorial Organizations a	and NGOs										
Total	37,425	5,626	4,072	4,834	5,467	2,527	76	710	3,194	10,919	-
Male	25,493	5,032	2,367	3,154	1,183	1,138	76	547	3,194	8,802	-
Female	11,931	594	1,705	1,680	4,283	1,389	-	163	-	2,117	-
Not Stated											
Total	2,188	-	-	-	92	60	369	457	-	-	1,209
Male	1,296	-	-	-	-	-	369	457	-	-	470
Female	892	-	-	-	92	60	-	-	-	-	740

			IRY IOTA	. 2011		Status Ir	n Employme	nt						
				Daid Emr	lovoos		гспрюуте							
				Paid Emp	Employee-NGO									
Major Industrial	Total		Employee-	Employee-	(including				Unpaid					
Divisions and Sex	Employed	Employee-	Government	Private	International	Employee	Other	Self	Family			Member of		Not
	Population	Government	Parasitatal	organization	organization)	Domestic		Employed	Worker	Employer	Apprentice	Co-Operatives	Other	Stated
Total Employed Populat	ion											•		
Total	5,139,831	993,158	98,868	995,660	86,746	323,515	46,668	2,000,055	433,630	41,650	6,857	39,861	71,968	1,195
Male	2,928,953	636,591	67,274	656,355	55,816	85,736	35,739	1,108,211	176,819	34,103	5,046	25,582	41,225	455
Female	2,210,879	356,567	31,594	339,304	30,930	237,779	10,929	891,844	256,811	7,547	1,811	14,279	30,743	740
Agriculture, Hunting For	estry & Fishir	ng												
Total	486,402	17,787	7,943	52,383	682	10,353	909		140,711	3,934	-	1,574	3,670	-
Male	311,909	12,395	4,755	26,495	423	8,370	909	179,770	71,822	3,539	-	1,459	1,972	-
Female	174,493	5,392	3,188	25,888	260	1,983	-	66,686	68,888	395	-	116	1,698	-
Mining & Quarrying														
Total	18,178	3,084	40	4,702	93	-	75	7,349	319	333	-	2,019	164	-
Male	15,790	2,102	40	4,049	62	-	75	6,974	260	333	-	1,731	164	-
Female	2,388	981	-	654	31	-	-	375	59	-	-	288	-	-
Manufacturing														
Total	675,481	33,229	21,884	202,927	3,796	14,951	3,972	313,317	55,863	9,201	421	7,904	8,017	-
Male	370,099	24,332	12,019	139,047	2,102	7,741	2,324	146,274	17,454	7,961	421	5,239	5,185	-
Female	305,382	8,897	9,865	63,880	1,693	7,210	1,648	167,043	38,409	1,240	-	2,665	2,832	-
Electricity, Gas and Wa														
Total	59,456	26,437	11,751	5,154	533	-	46	2,438	10,756	774	-	63	1,504	-
Male	38,832	20,924	9,825	4,062	366	-	-	1,678	1,140	774	-	63	-	-
Female	20,624	5,512	1,926	1,092	167	-	46	760	9,616	-	-	-	1,504	-
Construction														
Total	380,581	29,697	17,713	188,405	5,876	18,461	11,461	81,876	2,435	3,350	711	12,775	7,820	-
Male	318,433	22,764	14,676	149,153	4,756	17,219	9,471	78,038	2,435	3,350	711	8,940	6,919	-
Female	62,148	6,933	3,038	39,252	1,121	1,242	1,990	3,838	-	-	-	3,835	901	-
Wholesale and Retail T														
Total	1,099,758	8,242	3,931	134,418	2,670	8,661	1,425	812,151		9,420	2,196	4,965	8,708	55
Male	580,722	5,503	1,866	90,913	2,338	4,566	1,138	409,151	46,931	8,603	2,196	2,095	5,366	55
Female	519,036	2,739	2,065	43,505	332	4,095	287	403,000	55,984	816	-	2,870	3,342	-
Hotels and Restaurants														
Total	408,076	6,757	388	120,599	78	13,192	665	208,526	45,770	8,166	60	1,994	1,879	-
Male	105,545	2,934	348	49,162	39	3,044	368	32,202	13,116	4,272	60	-	-	-
Female	302,531	3,823	40	71,437	39	10,148	297	176,325	32,655	3,894	-	1,994	1,879	-
Transport, Storage and														
Total	242,410	33,424	13,886	93,069	1,976	11,780	2,909	75,017	3,376	2,918	436	1,002	2,618	-
Male	226,779	26,133	10,660	89,834	1,688	11,205	2,909	74,373	3,376	2,846	436	700	2,618	-
Female	15,631	7,291	3,225	3,234	287	574	-	644	-	72	-	302	-	-

Table 5.15 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Industrial Divisions and Status in Employment - COUNTRY TOTAL: 2011

Table 5.15 Cont'd							_							
		1				Status Ir	n Employme	nt		1	1			1
				Paid Emp										
Major Industrial Divisions and Sex	Total Employed Population	Employee- Government	Employee- Government Parasitatal	Employee- Private organization	Employee-NGO (including International organization)	Employee Domestic	Other Employee	Self Employed	Unpaid Family Worker	Employer	Apprentice	Member of Co-Operatives	Other	Not Stated
Financial Intermediation	on													
Total	91,143	61,269	7,916	18,795	1,941	-	462	612	10	-	-	-	139	-
Male	51,635	33,798	5,199	10,697	789	-	462	612	10	-	-	-	69	-
Female	39,508	27,471	2,717	8,098	1,152	-	-	-	-	-	-	-	69	-
Real Estate, Renting a	and Business A	Activities												
Total	116,708	44,661	3,085	25,990	3,197	1,788	311	33,292	1,876	565	1,027	342	574	-
Male	76,784	26,493	1,241	16,176	1,649	650	23	28,128	787	447	275	342	574	-
Female	39,924	18,169	1,844	9,814	1,548	1,138	287	5,165	1,089	118	752	-	-	-
PUBLIC ADMINISTRA														
Total	324,536	322,668	649	646	254	-	-	-	-	-	-	319	-	-
Male	222,301	221,080	649	-	254	-	-	-	-	-	-	319	-	-
Female	102,234	101,588	-	646	-	-	-	-	-	-	-	-	-	-
Education														
Total	346,352	271,409	3,676	57,043	6,635	536	670	3,761	362	415	337	1,384	123	-
Male	201,522	167,869	2,338		2,107	488	-	2,379	-	98	76	86	-	-
Female	144,831	103,540	1,338	30,963	4,528	48	670	1,381	362	318	261	1,299	123	-
Health and Social Wor														
Total	161,675	106,786	1,865	22,904	18,759	353	824	6,253	1,185	690	1,140	-	916	-
Male	80,565	53,207	1,208		9,330	66	367	4,590	533	617	343	-	335	-
Female	81,110	53,579	656		9,429	287	457	1,663	652	73	798	-	580	-
Other Community, So														
Total	459,078	26,684	3,504	64,005	8,641	21,549	22,713	206,697	62,144	1,426	528	5,451	35,734	-
Male	276,244	16,165	2,010		7,724	10,457	17,467	142,680	16,987	804	528	4,540	18,022	-
Female	182,834	10,519	1,494		918	11,092	5,246	64,017	45,157	622	-		17,713	-
Private Households wi														
Total	230,362	-	-	1,068	-	221,508	-	1,880	5,906	-	-	-	-	-
Male	24,981	-	-	400	-	21,620	-		1,967	-	-	-	-	-
Female	205,381	-	-	667	-	199,888	-		3,939	-	-	-	-	-
Extra-Territorial Organ		IGOs							-,					
Total	37,425	909	637	3,552	31,615	384	225	-	-	-	-	-	101	-
Male	25,493	869	440		22,190	310	225	-	-	-	-	-	-	-
Female	11,931	40	198		9,425	75	-	-	-	-	-	-	101	-
Not Stated		10		-,	.,.20									
Total	2,188	92	-		-	-	-	429	-	457	-	69	-	1,140
Male	1,296	-	-	-	-	-	-	369	-	457	-	69	-	400
Female	892	92	-	-	-	-	-	60	-	-	-	-	-	740

				SNIKT IOTAL.		Status In F	mployment							
Major					Employee NCO		проупен		1					1
Occupational	Total	Employee	Employee-		Employee-NGO (including				Unpaid					
Groups	Employed	Linbioyee	Government	Employee-Private	International	Employee	Other	Self	Family			Member of		Not
Croups	Population	Government	Parasitatal	organization	organization)	Domestic	Employee		Worker	Employer	Apprentice	Co-Operatives	Other	Stated
Total Employed Per		Covoninion	1 di doltatal	organization	organizationy	Domostio	Employee	Employed	montor	Employor	rippiontioo	oo operatives	Outlot	otatou
Total	5,139,831	993,158	98,868	995,660	86,746	323,515	46 668	2,000,055	433 630	41,650	6,857	39,861	71 968	1,195
Male	2,928,953	636,591	67,274	656,355	55,816	85,736	35,739	1,108,211		34,103	5,046	25,582		455
Female	2,210,879	356,567	31,594	339,304	30,930	237,779	10,929	891,844		7,547	1,811	14,279	30,743	740
Legislator Senior Of			51,574	557,504	50,750	231,117	10,727	071,044	230,011	1,541	1,011	14,277	30,743	740
Total	153,155	110,714	2,892	15,375	10,130	498	75	5,791	2,313	4,056	-	395	914	
Male	123,323	89,099	1,971	13,183	8,232	490	75	4,053	1,932	3,485		395	851	
Female	29,832	21,615	921	2,193	1,898	451		1,738	381	570	-	- 575	64	-
Professional	27,032	21,015	721	2,175	1,070	451		1,730	301	570	-	-	04	-
Total	336,011	234,289	9,322	54,653	15,299	344	3,984	13,387	635	512	863	-	2,724	
Male	226,454	163,552	5,887	30,673	7,667	236	3,904	10,955	318	472	426	-	2,724	-
Female	220,454	70,736	3,434	23,980	7,632	108	3,900	2,431	318	39	420	-	2,307	-
Technical and Asso			3,434	23,700	7,032	100	04	2,431	310	39	437	-	307	-
Total	330,286	219,277	6,032	44,203	18,208	607	10,378	24,010	785	592	481	1,623	4,089	
Male	218,506	140,220	4,822	28,363	10,200	310	8,408	24,010	238	592 592		1,023	4,089 3,051	-
										592	- 481	444		-
Female Clerks	111,781	79,058	1,210	15,839	6,971	297	1,970	3,925	548	-	461	444	1,038	-
	200 001	1/0 7/5	12 100	72.007	0.000	2 / 2 /	2 50/	10 010	F 44/	2 212	000	1 707	2 02 4	
Total	289,991	160,745	13,190	72,886	8,983	2,624	2,596	15,819	5,446	2,212	828	1,727	2,934	-
Male	119,703	64,243	6,756	25,288	3,022	1,114	688	13,186	2,528	1,467	223	1,093	94	-
Female	170,288	96,502	6,434	47,598	5,961	1,510	1,907	2,633	2,919	745	605	634	2,841	-
Service, Shop & Ma					= / 0=							(
Total	1,275,292	62,927	2,242	155,691	5,637	33,836	2,288	859,014		11,257	1,218	6,100	9,031	55
Male	576,401	44,325	1,330	71,634	2,043	6,922	1,002	384,033	51,711	6,504	930	1,359	4,553	55
Female	698,891	18,602	912	84,057	3,595	26,914	1,286	474,980	74,285	4,753	289	4,741	4,478	-

Table 5.16 Currently Employed Population of Urban Areas Aged Ten Years and Above by Major Occupational Groups, Sex and Status in Employment - COUNTRY TOTAL: 2011

Table 5.16 Cont'd

						Status In E	mployment							
Major Occupational Groups	Total Employed Population	Employee - Government	Employee- Government Parasitatal	Employee-Private organization	Employee-NGO (including International organization)	Employee Domestic		Self Employed	Unpaid Family Worker	Employer	Apprentice	Member of Co-Operatives	Other	Not Stated
Skilled agricultural a														
Total	394,017	4,474	3,524	20,916	76	5,171	19	245,213	104,479	5,617	-	1,116	3,411	-
Male	259,469	2,915	1,479	9,763	76	4,046	19	182,314	50,015	5,222	-	1,116	2,505	-
Female	134,548	1,559	2,045	11,154	-	1,125	-	62,899	54,465	395	-	-	906	-
Crafts and related t	rades													
Total	958,971	38,296	17,683	225,948	4,580	34,958	10,218	514,736	68,472	11,726	2,073	18,525	11,757	-
Male	576,176	28,536	13,467	185,406	3,778	21,711	7,639	261,484	21,092	10,999	2,073	11,966	8,026	-
Female	382,795	9,760	4,216	40,542	802	13,248	2,579	253,251	47,380	727	-	6,559	3,731	-
Plant ,Machine Ope	erators & Ass	emblers												
Total	268,268	40,341	10,996	129,600	6,756	12,111	2,645	51,202	3,614	4,277	1,172	1,734	3,820	-
Male	246,870	38,315	10,308	116,391	6,652	11,426	2,645	48,247	3,202	4,277	1,172	990	3,246	-
Female	21,398	2,026	689	13,209	104	685	-	2,955	412	-	-	743	574	-
Elementary Occupa	ition													
Total	1,132,631	122,095	32,988	276,387	17,077	233,365	14,464	270,883	121,889	1,402	222	8,572	33,287	-
Male	581,582	65,386	21,255	175,655	13,109	39,925	11,362	183,853	45,784	1,084	222	7,415	16,533	-
Female	551,049	56,709	11,733	100,732	3,968	193,440	3,103	87,031	76,104	318	-	1,157	16,754	-
Not Stated														
Total	1,209	-	-	-	-	-	-	-	-	-	-	69	-	1,140
Male	470	-	-	-	-	-	-	-	-	-	-	69	-	400
Female	740	-	-	-	-	-	-	-	-	-	-	-	-	740

						of Hours of	Worked Pe	er Week				[Mean
Age Group and Sex												Not	No. of Hours
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Stated	Worked
All Ages		-	-		-						-		
Total	5,139,831	211,368	11,295	256,863	345,250	469,427	571,179	1,218,306	752,139	771,399	532,105	501	45.0
Male	2,928,953	122,134	4,609	97,551	148,573	230,621	312,072	723,732	464,264	496,515	328,749	134	47.2
Female	2,210,879	89,235	6,687	159,312	196,677	238,806	259,107	494,574	287,875	274,884	203,356	368	41.9
10-14													
Total	144,811	757	1,319	36,344	25,502	22,515	12,670	11,377	9,553	13,125	11,649	-	31.9
Male	71,614	757	904	17,072	11,132	13,010	8,350	6,718	3,986	6,715	2,969	-	30.0
Female	73,197	-	415	19,272	14,369	9,504	4,320	4,659	5,568	6,410	8,680	-	33.8
15-19													
Total	429,517	9,588	1,391	43,985	38,870	45,603	36,388	70,842	49,955	72,692	60,159	44	44.9
Male	192,665	4,840	679	17,910	16,580	25,935	15,099	36,321	22,227	32,533	20,495	44	43.4
Female	236,852	4,748	711	26,074	22,290	19,668	21,289	34,520	27,729	40,159	39,664	-	46.2
20-24													
Total	797,478	24,097	1,410	30,878	51,396	68,792	74,700	193,965	126,327	130,442	95,379	93	47.4
Male	404,296	10,612	342	9,715	21,821	34,396	35,640	97,957	68,204	75,035	50,530	44	49.4
Female	393,182	13,485	1,069	21,162	29,575	34,395	39,060	96,008	58,122	55,408	44,849	48	45.3
25-29													
Total	1,009,623	38,268	1,480	41,461	59,094	78,627	110,662	268,169	153,335	154,835	103,692	-	45.9
Male	579,802	19,130	44	17,339	24,809	35,392	57,537	156,215	99,012	100,345	69,979	-	48.8
Female	429,821	19,139	1,436	24,122	34,285	43,235	53,125	111,954	54,323	54,490	33,713	-	42.1
30-34													
Total	719,297	32,295	620	24,663	40,584	62,398	78,529	177,783	117,994	111,962	72,420	48	45.9
Male	439,566	19,647	213	6,660	16,828	29,817	44,140	113,216	77,173	78,955	52,917	-	49.0
Female	279,731	12,647	407	18,003	23,756	32,582	34,389	64,568	40,822	33,007	19,503	48	41.1
35-39													
Total	656,750	26,048	2,038	23,246	32,466	59,880	83,948	171,759	97,879	98,988	60,474	23	45.3
Male	363,332	13,504	449	6,963	11,596	26,476	38,455	100,915	60,820	63,833	40,297	23	48.6
Female	293,418	12,545	1,589	16,283	20,870	33,403	45,493	70,844	37,059	35,155	20,176	-	41.2
40-44													
Total	409,957	18,959	194	10,471	24,142	33,244	54,652	107,874	64,415	55,734	40,274	-	45.5
Male	269,401	14,476	62	3,914	9,681	16,565	36,678	72,449	44,014	42,166	29,396	-	47.7
Female	140,556	4,482	132	6,557	14,461	16,679	17,974	35,425	20,401	13,568	10,877	-	41.4
45-49													
Total	326,320	17,222	260	12,551	17,528	29,240	36,815	85,567	49,225	47,587	30,032	294	44.8
Male	196,881	11,283	-	3,375	8,793	14,235	22,546	51,106	33,366	32,846	19,309	23	46.9
Female	129,439	5,938	260	9,176	8,736	15,006	14,269	34,461	15,859	14,741	10,722	271	41.5

Table 5.17 Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Age Group and Number of Hours Worked Per Week-COUNTRY TOTAL: 2011

					Number of	of Hours of	Worked P	er Week					Mean
Age Group and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Not Stated	No. of Hours Worke
50-54													
Total	232,556	14,744	741	8,566	15,491	21,846	32,501	58,141	32,929	26,768	20,829	-	42
Male	138,821	8,639	424	3,118	6,297	10,815	20,971	35,082	20,440	19,770	13,264	-	45
Female	93,735	6,105	317	5,448	9,194	11,031	11,530	23,059	12,489	6,998	7,565	-	39
55-59													
Total	152,901	8,453	411	7,515	11,958	20,450	20,329	32,117	20,444	19,394	11,829	-	41
Male	92,826	3,354	369	2,825	7,280	8,213	12,687	23,496	13,903	13,492	7,205	-	44
Female	60,074	5,099	41	4,690	4,678	12,237	7,642	8,621	6,540	5,902	4,623	-	37
60-64													
Total	108,220	5,993	88	6,053	13,120	11,321	12,639	18,812	12,306	17,066	10,821	-	42
Male	72,874	3,541	-	1,715	6,515	6,362	8,793	14,879	9,509	12,421	9,140	-	46
Female	35,346	2,453	88	4,338	6,605	4,960	3,847	3,933	2,797	4,645	1,681	-	33
65+													
Total	152,402	14,944	1,344	11,130	15,101	15,510	17,345	21,899	17,776	22,806	14,548	-	39
Male	106,875	12,350	1,123	6,943	7,241	9,404	11,175	15,379	11,610	18,403	13,247	-	42
Female	45,527	2,593	222	4,187	7,860	6,106	6,170	6,520	6,166	4,403	1,301	-	34

Table 5.17 Cont'd

Per Week	. 2011				Number	of Hours of	Worked Pe	r Week					Mean
Pogion and					TRUTIDE								No. Of
Region and Sex												Not	Hours
	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Stated	Worked
Country Total													
Total	5,139,831	211,368	11,295	256,863	345,250	469,427	571,179	1,218,306	752,139	771,399	532,105	501	45.0
Male	2,928,953	122,134	4,609	97,551	148,573	230,621	312,072	723,732	464,264	496,515	328,749	134	47.2
Female	2,210,879	89,235	6,687	159,312	196,677	238,806	259,107	494,574	287,875	274,884	203,356	368	41.9
Tigray Region													
Total	321,569	12,200	-	6,629	18,013	28,618	26,880	83,335	61,083	49,180	35,360	271	47.6
Male	173,501	8,030	-	2,819	5,683	12,933	14,561	45,899	33,602	30,084	19,891	-	49.1
Female	148,068	4,171	-	3,810	12,330	15,685	12,320	37,436	27,481	19,097	15,469	271	45.9
Affar Region													
Total	71,536	2,578	-	2,336	4,445	5,437	10,286	9,461	16,792	12,164	8,036	-	47.8
Male	45,023	1,827	-	316	2,123	3,875	6,463	6,532	11,667	7,211	5,010	-	48.5
Female	26,512	751	-	2,019	2,322	1,563	3,823	2,929	5,125	4,953	3,026	-	46.4
Amhara Region													
Total	921,246	46,474	3,953	55,618	82,905	102,274	124,339	179,936	120,246	118,001	87,312	185	41.9
Male	505,459	23,458	1,563	24,803	33,367	46,626	67,049	105,032	76,629	75,685	51,158	89	44.4
Female	415,787	23,017	2,390	30,815	49,539	55,648	57,290	74,904	43,617	42,316	36,155	97	38.8
Oromia Region													
Total	1,609,473	57,532	2,460	77,532	116,026	172,411	196,348	349,463	231,761	247,880	158,059	-	44.5
Male	930,772	35,798	1,023	31,066	48,967	81,636	105,905	220,304	144,332	156,035	105,706	-	46.9
Female	678,701	21,734	1,437	46,467	67,058	90,775	90,443	129,159	87,429	91,845	52,353	-	41.1
Somali Region													
Total	167,435	7,138	311	10,639	16,775	18,309	26,331	22,577	21,289	23,661	20,404	-	43.0
Male	98,864	5,486	-	3,401	8,031	11,155	15,259	13,229	15,259	14,979	12,065	-	44.7
Female	68,571	1,652	311	7,238	8,743	7,154	11,072	9,348	6,030	8,683	8,339	-	40.5
Benishangul Gumuz	Region												
Total	53,221	5,346	206	3,871	6,250	7,625	10,865	7,512	4,896	4,181	2,469	-	34.2
Male	30,493	3,000	93	1,412	2,739	3,794	6,460	4,894	3,670	2,779	1,653	-	37.3
Female	22,729	2,346	113	2,459	3,511	3,831	4,406	2,618	1,226	1,403	817	-	30.2
SNNP Region													
Total	680,560	29,839	2,785	62,506	54,452	65,270	70,602	154,273	90,085	85,231	65,517	-	41.9
Male	387,610	15,159	1,309	21,443	25,120	30,670	40,962	95,126	55,706	59,644	42,469	-	45.7
Female	292,950	14,680	1,476	41,063	29,332	34,600	29,640	59,146	34,379	25,587	23,048	-	36.9
Gambella Region													
Total	33,342	1,725	88	2,618	3,316	4,304	3,374	7,412	3,902	2,801	3,757	45	40.2
Male	16,628	593	-	571	1,004	2,296	1,440	4,720	2,555	1,143	2,261	45	44.6
Female	16,715	1,132	88	2,047	2,311	2,008	1,934	2,693	1,348	1,658	1,496	-	35.8

Table 5.18 Currently Employed Population of Urban Areas Aged Ten Years and Above by Region, Sex and Number of Hours Worked Per Week: 2011

Table 5.18 Cont'd

					Number o	f Hours of	Worked Pe	er Week					Mean
Region and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Not Stated	No. Of Hours Worked
Harari Region													
Total	48,450	2,435	-	2,089	3,090	4,201	4,140	10,308	8,129	8,633	5,425	-	46.5
Male	25,806	1,108	-	784	1,372	1,962	2,184	5,138	4,704	4,936	3,619	-	49.5
Female	22,644	1,327	-	1,305	1,718	2,239	1,956	5,171	3,425	3,698	1,806	-	43.0
Addis Ababa City Adm	ninistration												
Total	1,148,974	41,049	1,256	27,693	34,697	54,891	89,795	376,481	180,256	205,170	137,686	-	49.8
Male	667,973	25,168	620	8,524	17,647	32,456	47,456	212,050	108,345	135,578	80,128	-	51.0
Female	481,001	15,881	636	19,169	17,050	22,435	42,338	164,431	71,911	69,592	57,558	-	48.3
Dire Dawa Administra	tion												
Total	84,025	5,051	235	5,332	5,281	6,088	8,218	17,547	13,698	14,496	8,078	-	44.6
Male	46,825	2,508	-	2,410	2,519	3,219	4,333	10,809	7,794	8,443	4,789	-	46.3
Female	37,200	2,543	235	2,922	2,762	2,869	3,885	6,738	5,904	6,053	3,289	-	42.5

Number			JI WEEK			of Hours of	Worked F	Per Week					Mean
													No. of
Major Industrial division												Not	Hours
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Stated	Worked
Total Employed Population													
Total	5,139,831	211,368	11,295	256,863	345,250	469,427	571,179	1,218,306	752,139	771,399	532,105	501	45.0
Male	2,928,953	122,134	4,609	97,551	148,573	230,621	312,072	723,732	464,264	496,515	328,749	134	47.2
Female	2,210,879	89,235	6,687	159,312	196,677	238,806	259,107	494,574	287,875	274,884	203,356	368	41.9
Agriculture, Hunting Forestry	/ & Fishing												
Total	486,402	34,663	51	45,007	61,688	66,656	61,338	81,547	65,337	49,533	20,581	-	35.7
Male	311,909	24,091	-	23,531	33,895	41,086	42,858	52,033	41,794	36,164	16,457	-	37.3
Female	174,493	10,572	51	21,476	27,792	25,570	18,480	29,515	23,543	13,369	4,124	-	32.8
Mining & Quarrying													
Total	18,178	183	38	460	1,045	1,174	1,419	5,896	3,716	3,401	846	-	47.1
Male	15,790	142	38	460	1,045	1,174	510	5,072	3,716	2,787	846	-	47.5
Female	2,388	41	-	-	· -	<i>.</i> –	909	825	-	614	-	-	44.2
Manufacturing													
Total	675,481	34,367	1,796	28,578	41,652	75,463	61,195	176,142	108,522	105,183	42,583	-	43.3
Male	370,099	17,668	1,017	6,607	12,045	28,309	28,875	104,693	70,861	72,409	27,614	-	47.5
Female	305,382	16,698	779	21,971	29,607	47,154	32,320	71,449	37,661	32,774	14,970	-	38.3
Electricity, Gas and Water S		,		,	,	,	,	,	,	,			
Total	59,456	1,836	-	7,028	4,040	3,814	8,408	19,958	6,057	4,824	3,492	-	39.3
Male	38,832	1,660	-	604	1,069	1,953	5,338	16,791	4,576	3,771	3,071	-	45.8
Female	20,624	176	-	6,424	2,972	1,861	3,070	3,167	1,481	1,053	420	-	27.1
Construction				,	,	,	,	,	,	,			
Total	380.581	16,496	-	11,113	21,761	35.328	36,699	131,780	70,895	44,989	11,519	-	42.9
Male	318,433	14,105	-	9,663	18,499	31,338	31,689	104,471	59,610	38,243	10,813	-	42.9
Female	62,148	2,391	-	1,450	3,261	3,990	5,010	27,309	11,285	6,746	706	-	42.8
Wholesale and Retail Trade	- , -	,		,	-, -	-,	-,	,	,	-, -			-
Total	1,099,758	47,477	2,869	53,909	81,706	116,084	95,179	176,201	158,290	227,791	139,913	338	46.7
Male	580,722	22,602	2,100	19,027	28,716	45,491	40,684	96,819	91,176	143,306	90,735	67	50.9
Female	519,036	24,874	770	34,882	52,991	70,593	54,495	79,383	67,114	84,485	49,178	271	42.0
Hotels and Restaurants		,		,	,	,	,	,	,	,			
Total	408,076	14,345	222	14,206	25,760	33,447	38,048	55,174	55,144	79,318	92,412	-	52.7
Male	105,545	2,708	-	842	3,554	5,464	6,067	14,625	13,864	20,062	38,359	-	62.3
Female	302,531	11,638	222	13,365	22,206	27,983	31,981	40,549	41,280	59,256	54,053	-	49.4
Transport, Storage and Com				-,	,==••	,	- ,	-,•	,0	, ••	,		
Total	242,410	10.773	62	4,154	3.866	13,805	12,240	47,900	34,767	59,204	55,594	44	56.0
Male	226,779	10,555	62	3,867	3,830	12,862	10,119	41,175	31,583	57,960	54,722	44	56.6
Female	15,631	218	-	287	37	943	2,121	6,725	3,184	1,244	871	-	46.3
	. 0,001	2.0		201	01	0.0	_,	0,120	2,101	.,	0.1		10.0

Table 5.19Currently Employed Population of Urban Areas Aged Ten Years and Above by Major Industrial Divisions, Sex and
Number of Hours Worked Per Week - COUNTRY TOTAL: 2011

Table 5.19 Cont'd

				١	Number of	Hours of	Worked P	er Week					Mean
Major Industrial division												Not	No. of Hours
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Stated	Worked
Financial Intermediation	1												
Total	91,143	2,814	-	492	956	3,147	21,039	43,459	13,138	4,145	1,953	-	43.1
Male	51,635	1,893	-	443	751	991	10,632	23,835	8,205	3,244	1,641	-	44.2
Female	39,508	921	-	49	205	2,156	10,407	19,624	4,933	901	312	-	41.6
Real Estate, Renting and Busine	ess Activities												
Total	116,708	4,171	391	3,605	3,088	4,570	13,768	43,700	16,853	16,192	10,370	-	46.8
Male	76,784	2,494	-	2,616	2,385	2,762	8,237	23,963	10,863	14,517	8,946	-	49.4
Female	39,924	1,677	391	988	703	1,807	5,531	19,737	5,990	1,675	1,424	-	41.8
PUBLIC ADMINISTRATION AN	D DEFENCE												
Total	324,536	9,250	-	1,283	3,365	11,489	72,972	128,377	53,217	25,026	19,557	-	45.8
Male	222,301	5,234	-	1,021	698	7,413	49,184	78,246	42,573	20,607	17,326	-	47.9
Female	102,234	4,016	-	262	2,668	4,076	23,788	50,131	10,644	4,419	2,230	-	41.3
Education													
Total	346,352	6,188	825	2,348	33,707	41,843	63,979	134,328	38,326	14,134	10,651	23	39.4
Male	201,522	1,741	365	797	19,528	22,364	37,557	77,077	22,079	11,435	8,558	23	41.0
Female	144,831	4,447	460	1,551	14,180	19,480	26,422	57,251	16,247	2,699	2,093	-	37.3
Health and Social Work													
Total	161,675	7,724	-	2,063	4,616	5,324	22,624	55,713	30,853	19,583	13,175	-	47.0
Male	80,565	3,623	-	887	687	2,580	12,110	23,440	15,364	13,771	8,103	-	49.6
Female	81,110	4,101	-	1,176	3,929	2,744	10,514	32,273	15,489	5,812	5,072	-	44.5
Other Community, Social and Po	ersonal Servic	e Activitie	S										
Total	459,078	15,756	4,676	71,572	45,237	42,620	41,061	76,400	58,231	63,182	40,343	-	39.0
Male	276,244	10,824	869	24,563	20,961	24,427	24,855	48,267	42,581	50,093	28,805	-	44.2
Female	182,834	4,932	3,807	47,008	24,277	18,193	16,206	28,133	15,651	13,089	11,538	-	31.2
Private Households with Employ													
Total	230,362	2,193	199	10,362	10,803	11,568	16,125	28,082	34,705	50,722	65,506	97	57.9
Male	24,981	618	158	2,489	272	646	345	3,515	2,461	5,144	9,333	-	60.8
Female	205,381	1,575	41	7,873	10,531	10,923	15,780	24,567	32,244	45,578	56,173	97	57.5
Extra-Territorial Organizations a	ind NGOs												
Total	37,425	1,992	166	681	1,958	2,665	5,084	13,556	3,537	4,173	3,611	-	44.7
Male	25,493	1,775	-	134	640	1,392	3,011	9,618	2,500	3,003	3,421	-	47.8
Female	11,931	217	166	548	1,319	1,273	2,073	3,937	1,037	1,170	191	-	38.2
Not Stated													
Total	2,188	1,140	-	-	-	429	-	69	549	-	-	-	20.8
Male	1,296	400	-	-	-	369	-	69	457	-	-	-	29.9
Female	892	740	-	-	-	60	-	-	92	-	-	-	7.5

		TOIRCUI					f worked	ner week					
Major Occupational							Workeu						Mean
Groups and Sex	Total												No. of Hours
	Employed											Not	Worked
	Population	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Stated	wonco
Total Employed Persons													
Total	5,139,831	211,368	11,295					1,218,306				501	45.0
Male	2,928,953	122,134				230,621		723,732	464,264	496,515	328,749	134	47.2
Female	2,210,879	89,235	6,687	159,312	196,677	238,806	259,107	494,574	287,875	274,884	203,356	368	42.0
Legislator Senior Officials &	Managers												
Total	153,155	4,212	-	630	2,318	7,308	29,554	65,397	28,657	9,271	5,807	-	44.5
Male	123,323	2,525	-	554	2,053	5,258	23,506	53,795	23,043	7,541	5,049	-	45.1
Female	29,832	1,687	-	76	265	2,049	6,048	11,603	5,615	1,730	759	-	42.1
Professional													
Total	336,011	8,210	55	1,662	18,134	26,784	57,869	144,786	49,396	20,920	8,195	-	41.7
Male	226,454	3,564	55	1,064	12,852	16,219	40,389	90,627	37,659	18,317	5,707	-	42.8
Female	109,557	4,646	-	598	5,282	10,565	17,480	54,159	11,737	2,603	2,488	-	39.4
Technical and Associate Pro	ofessionals												
Total	330,286	12,905	770	5,923	22,867	22,932	54,078	129,760	42,857	21,398	16,797	-	41.8
Male	218,506	7,451	310	4,487	11,370	15,181	35,759	83,102	28,364	18,484	13,997	-	43.5
Female	111,781	5,454	460	1,436	11,496	7,751	18,319	46,658	14,493	2,914	2,800	-	38.6
Clerks													
Total	289,991	8,619	166	5,337	4,503	9,361	60,049	120,607	40,327	25,244	15,777	-	44.9
Male	119,703	4,458	-	2,143	1,739	2,859	24,043	42,102	17,360	14,026	10,972	-	47.2
Female	170,288	4,162	166	3,194	2,764	6,501	36,006	78,506	22,967	11,218	4,804	-	43.2
Service, Shop & Market sale	es Workers												
Total	1,275,292	45,711	2,331	48,952	87,201	120,341	110,790	180,866	187,061	280,511	211,189	338	49.9
Male	576,401	18,216	1,053	11,848	26,879	43,856	40,339	81,437	93,030	149,709	109,966	67	53.8
Female	698,891	27,495	1,278	37,103	60,323	76,484	70,451	99,429	94,031	130,802	101,223	271	46.7
Skilled agricultural and Fishe	ery												
Total	394,017	30,170	356	44,748	49,833	55,533	45,652	61,561	49,092	38,984	18,087	-	34.7
Male	259,469	21,025	217	23,818	28,908	35,908	34,318	41,177	30,901	29,240	13,955	-	36.1
Female	134,548	9,145	139	20,930	20,925	19,625	11,334	20,384	18,191	9,744	4,132	-	31.8
Crafts and related trades													
Total	958,971	53,486	2,819	52,385	75,954	119,199	97,998	227,083	146,553	132,147	51,347	-	40.9
Male	576,176	31,601	1,728	18,874	30,493	53,019	51,586	161,628	104,158	91,500	31,591	-	44.0
Female	382,795	21,885	1,091	33,511	45,461	66,180	46,412	65,455	42,395	40,648	19,756	-	36.3

Table 5.20Currently Employed Population of Urban Areas Aged Ten Years and Above by Sex, Major Occupational Groups and
Number of Hours Worked Per Week - COUNTRY TOTAL: 2011

Table 5.20 Cont'd

					Number c	of Hours c	f worked p	oer week					Mean
Major Occupational Groups and Sex	Total Employed Population	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Not Stated	No. of Hours Worked
Plant ,Machine Operators &	Assemblers												
Total	268,268	13,939	-	2,808	3,705	11,258	14,103	58,197	46,969	67,008	50,281	-	54.9
Male	246,870	13,498	-	2,073	2,730	8,731	12,967	48,029	44,316	65,499	49,028	-	55.9
Female	21,398	440	-	735	975	2,528	1,136	10,169	2,653	1,509	1,253	-	43.3
Elementary Occupation													
Total	1,132,631	32,976	4,798	94,418	80,735	96,711	101,085	229,979	161,226	175,916	154,624	164	46.0
Male	581,582	19,395	1,245	32,689	31,549	49,590	49,164	121,767	85,433	102,201	88,483	67	48.8
Female	551,049	13,581	3,552	61,730	49,186	47,122	51,921	108,212	75,793	73,715	66,141	97	43.2
Not Stated													
Total	1,209	1,140	-	-	-	-	-	69	-	-	-	-	2.7
Male	470	400	-	-	-	-	-	69	-	-	-	-	6.9
Female	740	740	-	-	-	-	-	-	-	-	-	-	-

Hours worked Per week - COUNTRY TOTAL: 2011 Number of Hours of Worked Per Week Mean Number of Hours of Worked Per Week No. of													
								er vveek					No. of
Status in Employment												Not	Hours
And sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	stated	Worked
Total Employed Population													
Total	5,139,831	211,368	11,295	256,863	345,250	469,427	571,179	1,218,306	752,139	771,399	532,105	501	45.0
Male	2,928,953	122,134	4,609	97,551	148,573	230,621	312,072	723,732	464,264	496,515	328,749	134	47.2
Female	2,210,879	89,235	6,687	159,312	196,677	238,806	259,107	494,574	287,875	274,884	203,356	368	41.9
Employee-Government													
Total	993,158	27,714	515	5,866	42,067	60,790	189,765	399,402	151,280	67,942	47,794	23	43.7
Male	636,591	13,547	55	3,258	23,121	35,285	118,611	244,618	103,217	54,349	40,509	23	45.6
Female	356,567	14,167	460	2,608	18,946	25,506	71,155	154,784	48,063	13,594	7,285	-	40.3
Employee-Government Parasit	tatal												
Total	98,868	3,144	-	420	1,533	5,419	9,236	43,774	16,338	11,673	7,331	-	47.5
Male	67,274	2,674	-	23	950	3,079	5,450	26,103	11,711	10,365	6,918	-	49.7
Female	31,594	471	-	398	582	2,339	3,786	17,671	4,626	1,309	413	-	42.8
Employee-Private organization	1												
Total	995,660	36,611	410	16,087	23,849	55,025	67,310	303,034	178,878	193,048	121,363	44	51.0
Male	656,355	27,340	112	9,602	15,977	35,274	43,032	179,893	116,505	136,599	91,977	44	52.0
Female	339,304	9,271	299	6,484	7,872	19,752	24,278	123,141	62,372	56,450	29,386	-	49.3
Employee-NGO (including Inte	rnational orga	nization)											
Total	86,746	4,330	166	872	2,897	4,935	11,215	34,451	13,095	7,825	6,959	-	44.9
Male	55,816	3,524	-	324	1,215	2,176	6,970	20,384	9,242	5,628	6,354	-	47.3
Female	30,930	806	166	548	1,682	2,759	4,246	14,067	3,853	2,198	605	-	40.7
Employee Domestic													
Total	323,515	5,038	391	8,489	16,582	20,272	23,552	48,788	47,521	67,779	85,006	97	56.7
Male	85,736	2,664	310	661	3,554	7,230	3,989	17,261	10,159	18,691	21,216	-	56.4
Female	237,779	2,374	81	7,827	13,029	13,043	19,563	31,527	37,362	49,088	63,790	97	56.8
Other Employee													
Total	46,668	1,426	-	2,409	2,914	5,559	5,755	11,007	6,699	6,040	4,860	-	44.3
Male	35,739	1,426	-	1,711	2,318	4,071	2,988	7,567	5,278	5,596	4,783	-	46.6
Female	10,929	-	-	698	595	1,487	2,767	3,440	1,421	443	77	-	36.8
Self Employed													
Total	2,000,055	112,545	5,831	110,252	163,973	237,270		301,645	281,974	350,484	224,454	338	43.9
Male	1,108,211	60,139	3,228	45,316	68,435	104,849	105,703	183,382	175,923	222,925	138,244	67	46.8
Female	891,844	52,405	2,603	64,936	95,538	132,422	105,587	118,263	106,051	127,559	86,210	271	40.3

Table 5.21 Currently Employed Population Urban Areas Aged Ten Years and Above by Sex, Status in Employment and Number of Hours Worked Per Week - COUNTRY TOTAL: 2011

Table 5.21 Cont'd

					Number o	Hours of	Worked Pe	er Week					Mean
Status in Employment And sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Not stated	No. of Hours Worked
Unpaid Family Worker					-								
Total	433,630	12,286	1,876	95,127	77,027	64,198	41,285	45,820	33,293	40,192	22,526	-	31.1
Male	176,819	6,019	904	29,422	25,263	28,757	18,235	20,639	14,974	22,038	10,569	-	34.5
Female	256,811	6,268	972	65,706	51,765	35,441	23,049	25,181	18,318	18,154	11,957	-	28.8
Employer													
Total	41,650	1,837	-	441	1,177	4,175	1,508	7,146	7,674	11,561	6,129	-	52.6
Male	34,103	981	-	441	1,106	3,414	734	6,761	6,708	9,278	4,681	-	52.8
Female	7,547	857	-	-	72	761	774	386	966	2,284	1,448	-	51.6
Apprentice													
Total	6,857	-	195	658	813	765	565	2,800	485	469	106	-	37.4
Male	5,046	-	-	658	759	450	406	2,166	200	347	60	-	36.1
Female	1,811	-	195	-	54	315	160	634	285	122	46	-	41.0
Member of Co-Operatives													
Total	39,861	1,413	-	2,021	1,545	2,857	4,514	9,539	8,172	6,428	3,373	-	46.0
Male	25,582	997	-	1,553	236	2,384	2,724	6,470	5,195	4,320	1,702	-	45.4
Female	14,279	415	-	468	1,309	473	1,790	3,069	2,977	2,107	1,671	-	47.1
Others													
Total	71,968	3,882	1,911	14,167	10,872	8,161	5,184	10,899	6,732	7,957	2,204	-	30.8
Male	41,225	2,422	-	4,526	5,639	3,652	3,230	8,488	5,151	6,379	1,736	-	37.6
Female	30,743	1,460	1,911	9,641	5,233	4,509	1,953	2,411	1,580	1,577	468	-	21.7
Not Stated													
Total	1,195	1,140	-	55	-	-	-	-	-	-	-	-	0.4
Male	455	400	-	55	-	-	-	-	-	-	-	-	1.0
Female	740	740	-	-	-	-	-	-	-	-	-	-	-

COUNTRY TOTAL Country	Table 5.22	Paid	Employ	ees of l	Jrban	ropula	ation A	gea I	en re	ears ar			Regio	n, sex	and	і уре с	or Payl	nent:	2011			
Pregent and Yang Table Finallyses-Covernment Engloyes-Covernment Eng					•						Paid en	nployees										
COUNTRY TOTAL Country			Total		Emplo	yee-Gover	rnment							(incluc	ling Interr	national	Empl	oyee Dor	nestic	Oth	ner Emplo	oyee
Incash 2.544.615 ISJF.21 10.100 93.83 67.07 81.967 81.988 67.07 81.999 96.907 68.717 27.143 55.90 23.55 73.78 73.787 10.408 73.787 10.781 73.787 10.781 73.787 12.777 77.07 5.82 3.477 2.788 5.808 3.787 12.787 12.78		Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female
In kind 2.463.814 1.498.236 965.578 978.432 627.12 35.751 94.726 63.774 96.776 437.743 85.807 55.00 32.080 293.447 78.942 21.446 4.149 34.607 1.74 7.74 7.75 7.713 85.807 722 72 72 72 7.83 30.702 1.743 7.85 25.70 7.72 7.7 7.5 <	COUNTRY TOTAL	-																				
In kind 23.78 12.003 10.844 3.895 1.745 2.150 797 66. 131 12.393 6.131 6.230 722 72 7.5 7.80 2.383 3.03 1.74 1.74 2.74 Both 55.315 25.303 30.012 10.613 7.745 7.10 6.76 7.104 7.80 7.312 2.342 2.312 2.342 2.342 2.342 2.342 2.342 2.342 2.342 2.342 2.342 2.342 2.342 3.142 2.342 3.142 2.342 3.142 3.142 3.14 7.4 7	Total	2,544,615	1,537,512	1,007,104	993,158	636,591	356,567	98,868	67,274	31,594	995,660	656,355	339,304	86,746	55,816	30,930	323,515	85,736	237,779	46,668	35,739	10,929
Beh 55,315 25,303 30.012 10.43 7.726 2.76 7.76 7.6 7.66 7.6 7.66	In cash	2,463,814	1,498,236	965,578	978,632	627,121	351,511	94,726	63,787	30,939	965,917	638,774	327,143	85,893	55,003	30,890	293,447	78,984	214,463	45,199	34,567	10,632
No stated 1.699 1.070 6.29 - - 76 7.6 7.0 7.00 <t< td=""><td>In kind</td><td>23,787</td><td>12,903</td><td>10,884</td><td>3,895</td><td>1,745</td><td>2,150</td><td>797</td><td>666</td><td>131</td><td>12,393</td><td>6,813</td><td>5,580</td><td>722</td><td>722</td><td>-</td><td>5,807</td><td>2,783</td><td>3,023</td><td>174</td><td>174</td><td>-</td></t<>	In kind	23,787	12,903	10,884	3,895	1,745	2,150	797	666	131	12,393	6,813	5,580	722	722	-	5,807	2,783	3,023	174	174	-
TIGRAY REGION Total 162,805 97,50 65,299 64,39 41,727 26,812 5,862 3,479 2,383 58,468 34,749 23,70 31,92 2,562 62,90 12,102 12,122 2,181 2,562 642 24,167 12,925 12,122 2,182 2,664 42,107 12,937 31,92 2,562 629 24,167 12,925 12,122 2,182 2,121 2,182 2,182 12,122 2,182 2,121 2,183 12,121 2,183 12,121 2,183 13,17 14 14,121 2,770 13,851 16,514 10,672 5,841 4,946 3,759 11,971 16,869 11,971 4,972 5,62 5,21 4,41 2,020 2,13 1,807 10	Both	55,315	25,303	30,012	10,631	7,726	2,905	3,270	2,745	525	16,302	10,038	6,264	132	92	40	23,950	3,969	19,981	1,031	734	297
Total 168,875 99,581 69,293 69,293 61,293 61,293 51,202 23,29 52,404 32,50 62,50 64,79 11,202 23,35 64,649 31,402 25,52 649 21,407 12,505 12,022 23,85 64,649 31,407 22,203 130 2.5.5 649 24,107 12,505 14,20 12,305 14,207 14,203 14,207 14,203 <td>Not stated</td> <td>1,699</td> <td>1,070</td> <td>629</td> <td>-</td> <td>-</td> <td>-</td> <td>76</td> <td>76</td> <td>-</td> <td>1,048</td> <td>730</td> <td>318</td> <td>-</td> <td>-</td> <td>-</td> <td>312</td> <td>-</td> <td>312</td> <td>264</td> <td>264</td> <td>-</td>	Not stated	1,699	1,070	629	-	-	-	76	76	-	1,048	730	318	-	-	-	312	-	312	264	264	-
In cash 162,806 97,507 65,299 68,599 41,72 26,872 3,479 2,383 58,468 3,479 2,387 3,192 2,562 6,292 1,617 1,293 2,518 2,043 4,727 2,783 3,708 2,792 3,785 2,793 3,785 2,793 3,785 2,793 3,785 2,793 3,895 1,933 1,933 1,936 - - 3,785 2,793 3,86 2,785 3,785 1,933 1,936 1,933 1,936 1,933 1,93 1,936 - - - - 1,933 1,986 1,01 1,010 - </td <td>TIGRAY REGION</td> <td></td>	TIGRAY REGION																					
In kind 3.257 912 2.345 672 . 672 . . 2.043 738 1.306 . . 3.68 1.74 1.74 1.74 Both 2.548 898 1.650 160 . 100 . . 1.737 898 1.030 .	Total	168,875	99,581	69,293	69,430	41,727	27,704	5,862	3,479	2,383	62,449	36,385	26,064	3,192	2,562	629	24,986	12,955	12,032	2,955	2,474	482
Both 2,54 8,99 1,65 1,60 . . . 1,937 8,98 1,03 . </td <td>In cash</td> <td>162,806</td> <td>97,507</td> <td>65,299</td> <td>68,599</td> <td>41,727</td> <td>26,872</td> <td>5,862</td> <td>3,479</td> <td>2,383</td> <td>58,468</td> <td>34,749</td> <td>23,720</td> <td>3,192</td> <td>2,562</td> <td>629</td> <td>24,167</td> <td>12,955</td> <td>11,212</td> <td>2,518</td> <td>2,036</td> <td>482</td>	In cash	162,806	97,507	65,299	68,599	41,727	26,872	5,862	3,479	2,383	58,468	34,749	23,720	3,192	2,562	629	24,167	12,955	11,212	2,518	2,036	482
Not stated 264 276 76	In kind	3,257	912	2,345	672	-	672	-	-	-	2,043	738	1,306	-	-	-	368	-	368	174	174	-
AFFAR REGION Total 41,121 27.20 13.81 16.14 10.62 5.841 4,194 3,756 11.90 16.669 11.971 4,962 4.89 4.94 4.90 2.6 11.91 4.962 4.99	Both	2,548	898	1,650	160	-	160	-	-	-	1,937	898	1,038	-	-	-	452	-	452	-	-	-
Total 41,12 27,270 13,851 16,514 10,72 5,841 5,146 3,950 1,190 16,673 11,111 4,962 499 441 2,020 213 1,807 10		264	264	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	264	264	-
In cash 40,655 26,814 13,841 16,174 10,072 5,81 4,949 3,759 17,16 17,6 17,73 73 - - - - - - - 77 77 73 - - - - - - - - 77 77 - - - - - - - - - 77 77 - <th< td=""><td>AFFAR REGION</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	AFFAR REGION																					
In kind 249 243 240 <	Total	41,121	27,270	13,851	16,514	10,672	5,841	5,146	3,956	1,190	16,869	11,897	4,972	562	521	41	2,020		1,807	10	10	-
Both 141 131 10 121 121 121 120 10 10 1				13,841	16,514	10,672	5,841	4,949	3,759	1,190			4,962			41	2,020	213	1,807	10	10	-
Not stated AMMARA REGION 76 7				-	-	-	-	-	-	-				73	73	-	-	-	-	-	-	-
AMHARA REGON Solution				10	-	-	-			-	20	10	10	-	-	-	-	-	-	-	-	-
In cash 370,858 213,098 157,760 174,606 112,260 62,264 9,799 5,199 4,600 119,268 76,577 42,741 10,212 5,086 5,127 5,106 9,442 41,645 5,866 4,848 1,383 In kind 3,083 2,366 717 101 101 101 - - 1,683 1,218 465 - - 1,299 1,046 252 - - - - 1,815 443 1,451 86 86 - <td< td=""><td>AMHARA</td><td>76</td><td>76</td><td>-</td><td>-</td><td>-</td><td>-</td><td>76</td><td>76</td><td>-</td><td>-</td><td>-</td><td>-</td><td></td><td></td><td>-</td><td>-</td><td>-</td><td>-</td><td></td><td>-</td><td>-</td></td<>	AMHARA	76	76	-	-	-	-	76	76	-	-	-	-			-	-	-	-		-	-
In kind 3.083 2.366 717 101 101 - - - 1.683 1.218 465 - - 1.299 1.046 252 - - - Both 3.647 1.684 1.963 83 83 674 222 452 910 851 59 - - 1.895 443 1.451 86 86 - - - 1.895 443 1.451 86 86 - - - - - 1.895 443 1.451 86 86 - <td>Total</td> <td>377,588</td> <td>217,148</td> <td>160,440</td> <td>174,790</td> <td>112,544</td> <td>62,246</td> <td>10,473</td> <td>5,421</td> <td>5,052</td> <td>121,862</td> <td>78,597</td> <td>43,265</td> <td>10,212</td> <td>5,086</td> <td>5,127</td> <td>54,299</td> <td>10,932</td> <td>43,368</td> <td>5,952</td> <td>4,569</td> <td>1,383</td>	Total	377,588	217,148	160,440	174,790	112,544	62,246	10,473	5,421	5,052	121,862	78,597	43,265	10,212	5,086	5,127	54,299	10,932	43,368	5,952	4,569	1,383
Both Not stated QROMIA REGION 3,647 1,684 1,963 83 83 6.74 222 452 910 851 59 - 1,895 443 1,451 86 86 - Not stated QROMIA REGION .	In cash	370,858	213,098	157,760	174,606	112,360	62,246	9,799	5,199	4,600	119,268	76,527	42,741	10,212	5,086	5,127	51,106	9,442	41,664	5,866	4,484	1,383
Not stated OROMIA REGION 731,002 470,542 260,459 295,214 196,729 98,485 24,933 18,037 6.896 312,631 21,389 98,742 21,893 13,765 8,128 62,980 18,491 44,489 13,300 96,31 37,200 In cash 705,610 456,698 260,912 288,668 191,922 96,33 23,445 16,590 6,856 30,870 260,760 21,893 13,765 8,128 62,980 18,491 44,489 13,300 9,613 37,200 In cash 705,610 456,698 260,912 288,668 191,922 9,633 23,445 16,59 6,855 3,060 2,616 21,693 13,455 8,088 57,74 16,386 41,348 13,350 9,613 3,720 Both 15,945 10,199 5,746 4,749 1,848 1,846 3,471 1,992 100 60 40 4,44 4,44 4,44 4,44 4,43 4,45 4,4	In kind	3,083	2,366	717	101	101	-	-	-	-	1,683	1,218	465	-	-	-	1,299	1,046	252	-	-	-
OROMIA REGION Solution	Both	3,647	1,684	1,963	83	83	-	674	222	452	910	851	59	-	-	-	1,895	443	1,451	86	86	-
In cash 705,610 454,698 250,912 288,668 191,932 96,737 23,445 16,589 6,856 300,870 206,706 94,164 21,542 13,455 8,088 57,734 16,386 41,348 13,350 9,631 3,720 In kind 8,692 5,202 3,489 1,842 979 863 84 44 40 5,855 3,269 2,586 250 2.50 - 659 659 -	OROMIA	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
In kind 8,692 5,202 3,489 1,842 979 863 84 44 40 5,855 3,269 2,586 250 250 - 659 659 -	Total	731,002	470,542	260,459	295,214	196,729	98,485	24,933	18,037	6,896	312,631	213,889	98,742	21,893	13,765	8,128	62,980	18,491	44,489	13,350	9,631	3,720
Both 15,945 10,199 5,746 4,703 3,818 885 1,404 1,404 - 5,464 3,471 1,992 100 60 40 4,274 1,446 2,829 - - - Not stated SOMALL REGION 755 443 312 - - - 443 443 - - - 312 - 312 - 312 - 312 - 312 - 312 -	In cash	705,610	454,698	250,912	288,668	191,932	96,737	23,445	16,589	6,856	300,870	206,706	94,164	21,542	13,455	8,088	57,734	16,386	41,348	13,350	9,631	3,720
Not stated SOMALI REGION 755 443 312 443 443 312 312 312 <td>In kind</td> <td>8,692</td> <td>5,202</td> <td>3,489</td> <td>1,842</td> <td>979</td> <td>863</td> <td>84</td> <td>44</td> <td>40</td> <td>5,855</td> <td>3,269</td> <td>2,586</td> <td>250</td> <td>250</td> <td>-</td> <td>659</td> <td>659</td> <td>-</td> <td>-</td> <td>-</td> <td>-</td>	In kind	8,692	5,202	3,489	1,842	979	863	84	44	40	5,855	3,269	2,586	250	250	-	659	659	-	-	-	-
SOMALI REGION SOMALI NEGON Soma Negoni	Both	15,945	10,199	5,746	4,703	3,818	885	1,404	1,404	-	5,464	3,471	1,992	100	60	40	4,274	1,446	2,829	-	-	-
In cash 58,402 43,701 14,701 36,229 28,557 7,673 5,502 3,615 1,888 8,624 7,041 1,583 4,256 2,819 1,436 2,625 682 1,943 1,166 987 179 In kind 584 540 43 237 237 304 304 - - - - 43 - 43 - - - - - 43 - <t< td=""><td>SOMALI</td><td>755</td><td>443</td><td>312</td><td>-</td><td>-</td><td>-</td><td>-</td><td>-</td><td>-</td><td>443</td><td>443</td><td>-</td><td>-</td><td>-</td><td>-</td><td>312</td><td>-</td><td>312</td><td>-</td><td>-</td><td>-</td></t<>	SOMALI	755	443	312	-	-	-	-	-	-	443	443	-	-	-	-	312	-	312	-	-	-
In kind 584 540 43 237 237 - 304 304 - - - - 43 - 43 - - - - 43 - 43 - - - - 43 - 43 - - - - 43 - - - - - 43 - - - - - - - 43 - - - - - 40 32 32 - 43 - - - - - - 40 - 40 32 32 - 43 - - - - - - - 43 - - - - - 40 32 32 32 - 43 - 43 - - - - <th< td=""><td>Total</td><td>59,105</td><td>44,322</td><td>14,784</td><td>36,466</td><td>28,793</td><td>7,673</td><td>5,806</td><td>3,918</td><td>1,888</td><td>8,663</td><td>7,041</td><td>1,622</td><td>4,288</td><td>2,851</td><td>1,436</td><td>2,717</td><td>731</td><td>1,986</td><td>1,166</td><td>987</td><td>179</td></th<>	Total	59,105	44,322	14,784	36,466	28,793	7,673	5,806	3,918	1,888	8,663	7,041	1,622	4,288	2,851	1,436	2,717	731	1,986	1,166	987	179
Both 120 81 40 40 - 40 32 32 - 49 49	In cash	58,402	43,701	14,701	36,229	28,557	7,673	5,502	3,615	1,888	8,624	7,041	1,583	4,256	2,819	1,436	2,625	682	1,943	1,166	987	179
	In kind	584	540	43	237	237	-	304	304	-	-	-	-	-	-	-	43	-	43	-	-	-
Not stated	Both	120	81	40	-	-	-	-	-	-	40	-	40	32	32	-	49	49	-	-	-	-
	Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

Table 5.22 Paid Employees of Urban Population Aged Ten Years and Above by Region, Sex and Type of Payment: 2011

										Paid er	nployees										
Region and Type of Payment		Total		Emplo	yee-Gover	nment		yee-Gove Parasitata			ployee-Priv organizatio		(includ	nployee-N ling Intern rganizatio	ational	Empl	oyee Dor	nestic	Oth	er Emplo	iyee
	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female	Total	Male	Female
BENISHANGUL GU	MUZ REGIC	N																			
Total	22,441	15,254	7,187	15,069	10,136	4,934	831	698	134	4,364	3,284	1,080	461	378	83	1,061	198	863	654	560	93
In cash	22,080	15,084	6,996	14,979	10,136	4,843	831	698	134	4,246	3,266	980	461	378	83	909	46	863	654	560	93
In kind	252	152	100	-	-	-	-	-	-	100	-	100	-	-	-	152	152	-	-	-	-
Both	109	18	91	91	-	91		-	-	18	18	-	-	-	-	-	-	-	-		-
Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
SNNP REGION																					
Total	295,972	187,250	108,722	158,106	100,934	57,172	10,314	8,588	1,726	86,510	59,819	26,691	11,571	7,461	4,110	24,769	7,343	17,426	4,701	3,104	1,597
In cash	289,150	184,815	104,336	156,752	99,580	57,172	10,314	8,588	1,726	83,536	59,090	24,446	11,571	7,461	4,110	22,277	6,991	15,285	4,701	3,104	1,597
In kind	2,291	730	1,561	-	-	-	-	-	-	1,111	378	732	-	-	-	1,180	352	828	-	-	-
Both	4,213	1,705	2,508	1,354	1,354	-	-	-	-	1,546	351	1,196	-	-	-	1,312	-	1,312	-	-	-
Not stated	318	-	318	-	-	-	-	-	-	318	-	318	-	-	-	-	-	-	-	-	-
GAMBELLA REGIO	Ν																				
Total	12,811	8,930	3,882	8,808	6,183	2,625	131	108	23	2,730	1,856	874	643	404	239	200	79	121	300	300	-
In cash	12,531	8,713	3,818	8,763	6,138	2,625	100	77	23	2,555	1,715	840	643	404	239	172	79	92	300	300	-
In kind	179	145	34	35	35	-	31	31	-	114	80	34	-	-	-	-	-	-	-	-	-
Both	100	72	29	10	10	-	-	-	-	61	61	-	-	-	-	29	-	29	-	-	-
Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
HARARI REGION																					
Total	25,399	14,963	10,436	11,836	6,796	5,040	1,001	809	192	8,284	5,385	2,899	877	527	350	2,365	496	1,869	1,037	951	86
In cash	24,705	14,670	10,034	11,415	6,544	4,871	960	768	192	8,201	5,385	2,816	877	527	350	2,215	496	1,719	1,037	951	86
In kind	122	41	81	81	41	41	-	-	-	-	-	-	-	-	-	41	-	41	-	-	-
Both	573	252	321	340	211	129	41	41	-	83	-	83	-	-	-	109	-	109	-	-	-
Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-
ADDIS ABABA CITY	Y																				
Total	769,581	427,036	342,546	196,406	114,608	81,798	32,092	20,883	11,210	351,204	223,895	127,309	31,983	21,811	10,172	142,501	33,684	108,816	15,396	12,155	3,240
In cash	738,019	415,005	323,015	191,884	112,156	79,727	31,169	19,959	11,210	344,152	218,890	125,261	31,584	21,412	10,172	124,781	31,079	93,702	14,450	11,507	2,943
In kind	4,850	2,497	2,353	927	352	574	287	287	-	1,172	884	287	399	399	-	2,065	574	1,491	-	-	-
Both	26,425	9,247	17,178	3,595	2,099	1,496	636	636	-	5,594	3,833	1,761	-	-	-	15,655	2,031	13,624	945	648	297
Not stated	287	287	-	-	-	-	-	-	-	287	287	-	-	-	-	-	-	-	-	-	-
DIRE DAWA TOWN																					
Total	40,721	25,217	15,504	10,518	7,469	3,049	2,279	1,378	901	20,093	14,308	5,785	1,066	450	616	5,617	614	5,003	1,147	997	150
In cash	38,998	24,132	14,866	10,223	7,319	2,904	1,795	1,057	738	19,325	13,694	5,631	1,066	450	616	5,441	614	4,827	1,147	997	150
In kind	229	69	160	-	-	-	91	-	91	139	69	69	-	-	-	-	-	-	-	-	-
Both	1,493	1,015	478	295	151	145	393	321	73	629	544	85	-	-	-	176	-	176		-	-
Not stated	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-

Table 5.22 Cont'd

j	<u> </u>	10tal. 2011	D	aid Employee	s]
			r		s Employee-		
Period of					NGO		
Payment	Total		Employee-	Employee-	(including		
and Sex	paid	Employee-	Government	Private	International	Employee	Other
	Employees	Government	Parasitatal	organization	organization)	Domestic	Employees
Total			L	0	<u> </u>		
Total	2,544,615	993,158	98,868	995,660	86,746	323,515	46,668
Male	1,537,512	636,591	67,274	656,355	55,816	85,736	35,739
Female	1,007,104	356,567	31,594	339,304	30,930	237,779	10,929
Hourly							
Total	24,082	3,105	1,213	12,808	386	5,029	1,541
Male	18,213	2,144	685	9,262	386	4,247	1,489
Female	5,869	961	528	3,546	-	782	51
Daily							
Total	126,225	2,569	1,633	95,941	277	19,722	6,083
Male	100,525	1,525	738	78,518	-	14,682	5,063
Female	25,699	1,044	895	17,423	277	5,040	1,020
Weekly							
Total	106,716	1,649	3,129	85,297	421	11,988	4,232
Male	87,146	1,251	2,907	69,118	254	9,671	3,945
Female	19,569	398	222	16,179	166	2,317	287
Half Month							
Total	87,174	5,163	7,251	66,456	3,985	3,098	1,222
Male	61,553	3,729	3,097	48,809	3,606	1,276	1,035
Female	25,622	1,433	4,154	17,648	378	1,821	187
Monthly							
Total	2,151,919	979,363	84,373	707,938	81,186	270,525	28,533
Male	1,233,109	626,697	59,030	428,145	51,077	48,364	19,795
Female	918,810	352,667	25,343	279,793	30,108	222,161	8,738
Yearly							
Total	9,297	498	224	2,307	-	6,225	44
Male	7,499	498	224	2,181	-	4,553	44
Female	1,798	-	-	126	-	1,672	-
Other/specify							
Total	36,880	811	970	23,242	492	6,616	4,749
Male	27,772	747	517	18,970	492	2,942	4,103
Female	9,108	64	452	4,271	-	3,674	646
Not Stated							
Total	2,323	-	76	1,671	-	312	264
Male	1,694	-	76	1,354	-	-	264
Female	629	-	-	318	-	312	-

Table 5.23 Paid Employees of Urban Population Aged Ten Years and Above by Sex and Period of Payment - Country Total: 2011

					of Total Pa				
Period of Payment and Sex	Total	<50	50-99	100-199	200-399	400-699	700-999	1000+	Not Stated
Total									
Total	2,544,615	16,746	51,204	144,560	374,906	535,600	395,370	1,010,204	16,025
Male	1,537,512	2,625	11,687	36,681	138,215	316,242	268,832	754,454	8,775
Female Hourly	1,007,104	14,121	39,517	107,879	236,691	219,358	126,537	255,750	7,250
Total Male Female	24,082 18,213 5,869	39 - 39	1,548 720 828	2,257 1,202 1,056	2,836 2,517 319	6,041 3,648 2,392	4,719 4,082 637	6,642 6,045 597	-
Daily Total Male Female	126,225 100,525 25,699	2,659 788 1,871	2,901 2,352 549	6,284 4,752 1,532	18,991 11,136 7,856	39,579 29,486 10,093	29,112 26,073 3,039	26,626 25,865 760	73 73
Weekly Total Male Female	106,716 87,146 19,569	3,058 1,070 1,988	667 37 630	5,934 3,074 2,860	17,081 10,879 6,201	32,945 27,260 5,685	21,750 20,176 1,574	24,659 24,100 559	622 550 72
Half Month Total Male Female	87,174 61,553 25,622	493 41 452	2,382 527 1,854	4,014 301 3,713	13,681 4,577 9,104	28,989 21,439 7,550	18,049 15,795 2,254	19,246 18,552 693	320 320
Monthly Total Male Female Yearly	2,151,919 1,233,109 918,810	8,181 446 7,735	39,297 5,888 33,409	24,826	314,939 104,090 210,849	226,853	194,535	923,406 670,815 252,591	12,205 5,656 6,548
Total Male Female Other	9,297 7,499 1,798	704 - 704	2,116 1,744 373	1,142 973 169	2,047 1,647 400	946 794 152	-	2,342 2,342 -	-
Total Male Female Not Stated	36,880 27,772 9,108	1,611 280 1,331	2,293 420 1,874	3,485 1,554 1,931	5,331 3,369 1,962	7,440 6,762 678	8,953 8,171 782	7,284 6,734 550	482 482 -
Total Male Female	2,323 1,694 629	-	-	-	-	-	-	-	2,323 1,694 629

Table 5.24 Paid Employees of Urban Population Aged Ten Years and Over by Sex, Period Of Payment and Amount of Total Payment - Country Total: 2011

COUNTRY TOTAL: 2011 Amount of Total Payment per month													
			Ar	nount of T	otal Paym	ent per mo	onth						
Major Industrial Divisions and Sex	Total	<50	50-99	100-199	200-399	400-699	700-999	1000+	Not Stated	Mean Amount of Total Payment			
Total Employed													
Total	2,544,615	16,746	51,204	144,560	374,906	535,600	395,370	1,010,204	16,025	1,063.2			
Male	1,537,512	2,625	11,687	36,681	138,215	316,242	268,832	754,454	8,775	1,280.4			
Female Agriculture, Hur	-	y & Fishi	ng				126,537	255,750	7,250	731.1			
Total	90,057	2,177	6,173	5,497	18,366	28,936	9,298	18,372	1,238	746.9			
Male	53,347	821	3,532	3,181	9,348	14,452	7,485	14,199	329	897.3			
Female Mining & Quarry	36,710	1,356	2,642	2,316	9,018	14,484	1,813	4,173	908	524.3			
Total	7,994		101		695	1 200	1,511	4,308		1,350.6			
Male	6,328	-	101	-	685 393	1,388 1,026	1,224	4,308	-	1,350.8			
Female	1,666	-	-	_	292	362	287	725	-	962.5			
Manufacturing	1,000				202	002	201	120		002.0			
Total	280,758	1,837	4,464	8,904	41,433	92,310	58,346	69,470	3,995	874.4			
Male	187,564	618	1,176	3,904	18,101	58,537	45,782	56,134	3,312	996.7			
Female	93,193	1,218	3,288	5,000	23,332	33,773	12,564	13,335	683	630.8			
Electricity, Gas	and Water S	Supply											
Total	43,921	333	523	52	2,963	4,115	11,273	24,639	23	1,172.2			
Male	35,178	287	65	-	1,554	3,248	9,123	20,877	23	1,238.1			
Female	8,743	46	458	52	1,410	866	2,150	3,761	-	907.4			
Construction													
Total	271,613	671	1,511	5,924	28,019	79,085	57,961	96,874	1,568	1,073.7			
Male .	218,038	118	1,243	4,110	16,541	58,667	49,382	86,675	1,305	1,146.4			
Female	53,575 Detail Trade	554	269	1,814	11,479	20,418	8,580	10,200	263	778.2			
Wholesale and			4 074	0 450	00.000	50 740	05 007	00.007	4 075	700.0			
Total	159,348	463	1,374	6,452	38,002	52,719	25,327	33,937	1,075	783.0			
Male	106,325	65	306	2,950 3,502	20,584	33,076	20,165	28,520	658	892.9			
Female Hotels and Res	53,024	398	1,068	3,502	17,418	19,642	5,162	5,417	417	562.3			
Total	141,680	470	4,153	23,845	59,627	34,246	9,902	8,285	1,151	404.8			
Male	55,895		1,658	7,400	19,787	16,251	5,636	4,876	287	489.0			
Female	85,785		2,495	16,445	39,840	17,995	4,266	3,410	864	349.7			
Transport, Stora				,	00,010	,	.,_00	0,0		0.011			
Total	157,042	-	-	2,340	9,394	36,893	36,026	71,539	851	1,204.3			
Male	142,430	-	-	2,340	8,507	34,134	32,684	63,915	851	1,193.7			
Female	14,612	-	-	-	886	2,759	3,342	7,624	-	1,306.9			
Financial Interm	nediation												
Total	90,382	-	-	298	1,611	11,853	12,904	63,290	426	1,663.9			
Male	50,944	-	-	298	152	3,640	5,851	40,916	88	1,988.2			
Female	39,438	-	-	-	1,459	8,213	7,053	22,374	339	1,242.1			
Real Estate, Re		usiness A											
Total	79,031	-	347	774	5,419	13,077	13,606	45,196	611	1,507.5			
Male	46,231	-	-	603	1,202	6,419	7,230	30,459	318	1,775.9			
Female	32,800	-	347	171	4,217	6,658	6,376	14,738	293	1,128.4			

Table 5.25Paid Employees of Urban Population Aged Ten Years and Above by Major
Industrial Divisions, Sex and Amount of Payment per month –
COUNTRY TOTAL: 2011

1 able 5.25	Cont a									
			Ar	nount of T	otal Paym	ent per mo	onth			
Major Industrial Divisions									Not	Mean Amount of Total
and Sex	Total	<50	50-99	100-199	200-399	400-699	700-999	1000+	Stated	Payment
PUBLIC ADMI	NISTRATION	N AND D	EFENCE							
Total	324,217	-	340	1,668	14,191	39,779	52,107	215,676	456	1,519.1
Male	221,983	-	-	790	6,169	19,392	33,773	161,762	96	1,654.8
Female	102,234	-	340	878	8,022	20,387	18,335	53,914	359	1,223.4
Education										
Total	339,970	48	277	1,810	23,933	46,291	50,735	215,985	891	1,378.2
Male	198,883	-	277	826	8,464	19,271	22,470	147,363	211	1,590.0
Female	141,087	48	-	984	15,469	27,020	28,265	68,622	679	1,078.6
Health and So	cial Work									
Total	151,491	-	263	1,229	9,752	26,105	25,921	87,498	723	1,341.8
Male	74,147	-	263	704	1,864	8,725	8,531	53,388	672	1,666.7
Female	77,345	-	-	525	7,889	17,380	17,390	34,109	52	1,033.0
Other Commur	hity, Social a	nd Perso	nal Serv	ice Activiti	es					
Total	147,097	3,832	3,621	10,785	32,510	43,163	20,911	31,119	1,156	765.2
Male	92,683	672	1,930	5,546	16,995	27,701	15,178	24,397	264	909.2
Female	54,414	3,160	1,691	5,239	15,515	15,462	5,733	6,722	892	516.4
Private House	holds with En	nployed	Persons							
Total	222,575	6,915	27,982	74,342	86,222	20,514	3,113	1,987	1,501	218.9
Male	22,020	43	1,137	3,732	7,853	7,898	454	902	-	389.5
Female	200,555	6,871	26,845	70,610	78,369	12,616	2,659	1,085	1,501	200.0
Extra-Territoria	I Organizatio	ons and I	NGOs							
Total	37,323	-	75	639	2,779	5,126	6,383	21,961	361	1,877.2
Male	25,493	-	-	297	701	3,804	3,865	16,466	361	2,135.1
Female	11,830	-	75	342	2,077	1,322	2,518	5,495	-	1,329.2
Not Stated										
Total	92	-	-	-	-	-	46	46	-	950.0
Male	-	-	-	-	-	-	-	-	-	-
Female	92	-	-	-	-	-	46	46	-	950.0

Table 5.25 Cont'd

k	by Region	, Sex, ar	nd Reason	For Not W		011		
				Reason fo				
Region and Sex		Sick	Holiday/	Training/	Off	No work	Other	Not
	Total	Injury	vacation	Education	Season	temporarily	/Specify	Stated
Country Total								
Total	220,306	57,308	33,900	7,756	24,323	42,813	54,205	-
Male	126,218	29,679	19,423	6,400	17,007	30,669	23,040	-
Female	94,088	27,630	14,477	1,356	7,316	12,144	31,165	-
Tigray Region								
Total	12,612	2,094	1,854	877	4,681	2,300	805	-
Male	8,030	1,101	861	627	3,612	1,527	301	-
Female	4,582	993	993	249	1,069	773	505	-
Affar Region								
Total	2,436	561	426	9	73	777	589	-
Male	1,820	201	405	-	73	691	450	-
Female	616	360	21	9	-	86	139	-
Amhara Region								
Total	50,565	15,198	7,641	1,119	5,070	12,235	9,302	-
Male	25,942	7,459	3,615	842	2,920	7,392	3,715	-
Female	24,623	7,739	4,026	277	2,151	4,843	5,587	-
Oromia Region								
Total	60,002	14,720	9,309	2,759	8,342	10,940	13,932	-
Male	36,279	6,969	5,749	2,210	6,697	8,816	5,838	-
Female	23,723	7,751	3,559	549	1,644	2,124	8,095	-
Somali Region	-, -	, -	-,		, -	,	-,	
Total	7,748	1,400	1,438	-	790	724	3,397	-
Male	5,627	856	1,140	-	790	360	2,481	-
Female	2,122	544	298	-	-	364	916	-
Benishangul Gumu		• • •	200				0.0	
Total	5,429	1,000	961	24	1,692	1,067	684	-
Male	3,031	543	567		846	878	197	-
Female	2,397	457	394	24	846	189	487	-
SNNP Region	2,007	107	001	21	010	100	107	
Total	29,297	9,995	3,104	1,158	1,922	5,640	7,478	_
Male	15,733	5,759	1,451	985	981	4,236	2,322	_
Female	13,563	4,236	1,653	173	942	1,404	5,156	-
Gambella Region	13,303	4,230	1,000	175	342	1,404	5,150	-
Total	1,844	485	118	32	492	548	170	
Male	649	405	17	32	184	267	34	-
			101	52				-
Female	1,195	370	101	-	308	281	136	-
Harari Region	0 404	FF 4	400	00	4.4	0.14	4 407	
Total	2,481	554	422	96	41	241	1,127	-
Male	1,154	322	146	96	41	201	349	-
Female	1,327	233	276	-	-	41	778	-
Addis Ababa City A								
Total	42,607	9,954	7,910	1,608	862	7,387	14,887	-
Male	25,445	5,748	5,157	1,608	574	5,770	6,587	-
Female	17,162	4,205	2,752	-	287	1,617	8,300	-
Dire Dawa Administ								
Total	5,286	1,348	718	75	359	952	1,835	-
Male	2,508	607	314	-	289	531	767	-
Female	2,779	741	404	75	69	421	1,068	-

Table 5.26Total Employed Population Not at Work Aged Ten Years and Above
by Region, Sex, and Reason For Not Working: 2011

				I
Whether A	vailable and Ready	to Work Additional H	lours	
				Percent Available and Ready to
Total Employed	Seeking or Available to Work	Not Seeking or Available to Work	Not	work Additional
F EISUIIS	Auditional Hours	Auditional Hours	Slaleu	Hours
5,139,831	2,392,409	2,745,604	1.818	53.4
				52.6
2,210,879	1,006,537	1,203,601	740	54.4
144,811	28,534	116,277	-	80.3
71,614	16,619	54,995	-	76.8
73,197	11,915	61,282	-	83.7
400 547	470 477	055 074	000	50.0
				59.6
			300	56.3 62.3
			-	
				48.9 46.8
				40.8 51.2
				49.0
				48.1 50.2
429,021	210,000	215,054	555	50.2
719.297	353.610	365.629	59	50.8
				50.9
279,731	137,681	142,050	-	50.8
656,750	320,418	336,331	-	51.2
363,332	174,226	189,105	-	52.0
293,418	146,192	147,226	-	50.2
,			-	52.4
			-	53.8
			-	49.9
			-	54.7
			-	54.6
			-	54.9
			-	55.8
			-	54.9
			-	57.3
152,901	61,219	91,682	-	60.0
			-	59.8
			-	60.3
			-	62.1
			-	60.7
35,346	12,327	23,018	-	65.1
152.402	48,737	103.665	-	68.0
106,875	37,623	69,252	-	64.8
45,527	11,114	34,413	-	75.6
	Total Employed Persons 5,139,831 2,928,953 2,210,879 144,811 71,614 73,197 429,517 192,665 236,852 797,478 404,296 393,182 1,009,623 579,802 429,821 719,297 439,566 279,731 656,750 363,332 293,418 409,957 269,401 140,556 326,320 196,881 129,439 232,556 138,821 93,735 152,901 92,826 60,074 108,220 72,874 35,346	Total Employed PersonsSeeking or Available to Work Additional Hours5,139,8312,392,4092,928,9531,385,8712,210,8791,006,537144,81128,53471,61416,61973,19711,915429,517173,177192,66583,783236,85289,394797,478406,472404,296214,818393,182191,6541,009,623513,915579,802300,322429,821213,593719,297353,610439,566215,928279,731137,681656,750320,418363,332174,226293,418146,192409,957194,945269,401124,579140,55670,366326,320147,707196,88189,300129,43958,407232,556102,712138,82162,67493,73540,038152,90161,21992,82637,36260,07423,857108,22040,96572,87428,63835,34612,327152,40248,737106,87537,623	Total Employed Persons Seeking or Available to Work Additional Hours Not Seeking or Available to Work Additional Hours 5,139,831 2,392,409 2,745,604 2,928,953 1,385,871 1,542,003 2,210,879 1,006,537 1,203,601 144,811 28,534 116,277 71,614 16,619 54,995 73,197 11,915 61,282 429,517 173,177 255,974 192,665 83,783 108,516 236,852 89,394 147,458 797,478 406,472 390,290 404,296 214,818 189,167 393,182 191,654 201,123 1,009,623 513,915 495,033 579,802 300,322 279,139 429,821 213,593 215,894 719,297 353,610 365,629 439,566 215,928 223,579 279,731 137,681 142,050 656,750 320,418 336,331 363,332 174,226	Employed PersonsAvailable to Work Additional HoursAvailable to Work Additional HoursNot stated5,139,8312,392,4092,745,6041,8182,928,9531,385,8711,542,0031,0782,210,8791,006,5371,203,601740144,81128,534116,277-71,61416,61954,995-73,19711,91561,282-429,517173,177255,974366192,66583,783108,516366236,85289,394147,458-797,478406,472390,290717404,296214,818189,167312393,182191,654201,1234051,009,623513,915495,033676579,802300,322279,139341429,821213,593215,894335719,297353,610365,629593332174,226189,105-293,418146,192147,266-409,957194,945215,012-409,957194,945215,012-232,556102,712129,844-138,82162,67476,147-93,73540,03853,698-152,90161,21991,682-92,82637,36255,464-60,07423,85736,217-108,22040,96567,255-72,87428,63844,237-

Table 5.27Currently Employed Population of Urban Areas Aged Ten Years and Above
by Age Group, Sex, Whether Available and Ready to Work Additional Hours
and Percent Available and Ready to Work Additional Hours- Country Total: 2011

Hours	and Number	of Hours	s worke	ed Per we	ек by Ag	e Group	and Sex -	Country	Total: 20	11			-
					Number of	of Hours of	Worked Pe	er Week					Mean
Age Group and Sex												Not	Number of Hours Worked
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Stated	Per Week
All Ages	Total	0	10	712	10 21	22 00	01 00	10 10	40 07	0074	701	Olalea	
Total	2,392,409	92,926	4,302	123,246	178,682	255,730	281,506	563,879	356,542	333,927	201,398	271	44.1
Male	1,385,871	59,169	1,258	50,088	83,565	125,413	155,507	346,697	221,046	217,844	125,286		45.5
Female	1,006,537	33,757	3,045	73,158	95,118	130,318	125,999	217,182	135,496	116,084	76.112	271	41.8
10-14	.,,	00,101	0,010	. 0, . 00	00,110	,	0,000	,	,	,			
Total	28,534	32	-	7,172	4,831	4,257	847	2,510	2,827	4,804	1,253	-	29.6
Male	16,619	32	-	4,474	2,728	1,855	689	1,752	1,316	3,419	355	-	32.7
Female	11,915	-	-	2,699	2,104	2,402	158	757	1,512	1,385	898	-	22.3
15-19													
Total	173,177	3,985	718	14,569	17,747	20,001	15,667	32,922	20,029	29,500	18,039	-	41.4
Male	83,783	1,847	369	4,438	7,504	10,537	6,724	18,113	9,828	15,664	8,757	-	42.5
Female	89,394	2,138	349	10,131	10,243	9,463	8,943	14,809	10,200	13,836	9,282	-	40.2
20-24													
Total	406,472	10,868	1,017	18,959	31,207	41,020	43,956	95,492	64,704	62,711	36,537	-	45.9
Male	214,818	5,710	-	7,233	14,568	20,710	23,390	50,970	35,534	36,215	20,488	-	47.6
Female	191,654	5,158	1,017	11,726	16,639	20,310	20,566	44,522	29,170	26,496	16,049	-	43.6
25-29													
Total	513,915	16,299	851	25,929	33,318	48,752	61,954	132,012	80,706	73,044	41,050	-	45.1
Male	300,322	12,096	-	11,577	14,350	20,488	31,577	82,464	52,213	48,739	26,818	-	46.7
Female	213,593	4,203	851	14,352	18,968	28,264	30,377	49,547	28,493	24,306	14,233	-	42.3
30-34													
Total	353,610	15,681	232	16,000	23,962	38,253	40,851	88,206	54,457	47,249	28,719	-	45.3
Male	215,928	9,672	112	4,254	12,770	18,880	23,999	58,198	35,916	33,525	18,602	-	46.5
Female	137,681	6,009	120	11,746	11,192	19,373	16,852	30,009	18,541	13,723	10,116	-	43.2
35-39													
Total	320,418	11,664	538	12,345	21,167	38,666	44,641	77,298	46,165	43,157	24,777	-	44.4
Male	174,226	6,528	407	5,137	7,321	18,069	21,570	43,717	27,243	28,654	15,578	-	45.7
Female	146,192	5,136	130	7,208	13,845	20,597	23,071	33,580	18,922	14,503	9,199	-	42.2
40-44	4040		105				00 EC /	17.00.1	00 -	o (7 70	40.000		
Total	194,945	11,673	132	5,850	14,314	19,411	26,564	47,964	30,571	21,773	16,692	-	44.8
Male	124,579	9,140	-	2,323	6,034	8,567	17,268	33,630	19,180	16,132	12,305	-	46.9
Female	70,366	2,533	132	3,527	8,280	10,844	9,296	14,333	11,392	5,641	4,387	-	40.3

Table 5.28	Currently Employed Population of Urban Areas Aged Ten Years and Above Who Sought or Available to Work Additional	
Ηοι	urs and Number of Hours Worked Per Week by Age Group and Sex - Country Total: 2011	

					Number of	of Hours of	Worked Pe	r Week					Mean
Age Group and Sex	Total	0	1-3	4-12	13-21	22.20	31-39	40.49	40.57	EQ 74	75 .	N.S	Number of Hours Worked Per Week
45.40	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	IN.3	Fei Week
45-49													
Total	147,707	6,994	-	6,671	10,139	15,071	14,888	37,263	22,919	19,974	13,517	271	44.9
Male	89,300	4,445	-	2,350	4,703	9,180	8,864	23,387	15,914	13,262	7,195	-	45.0
Female	58,407	2,549	-	4,321	5,435	5,892	6,024	13,876	7,004	6,712	6,322	271	44.7
50-54													
Total	102,712	6,492	686	4,307	6,337	11,034	14,219	25,138	15,761	10,730	8,008	-	40.8
Male	62,674	3,680	369	1,883	2,919	6,600	9,559	14,574	11,154	6,959	4,976	-	42.2
Female	40,038	2,812	317	2,424	3,417	4,434	4,660	10,564	4,607	3,771	3,032	-	39.0
55-59													
Total	61,219	2,627	41	4,330	4,562	8,992	8,003	12,650	8,599	6,652	4,763	-	41.3
Male	37,362	1,338	-	2,164	3,624	3,270	4,140	9,312	6,179	4,369	2,967	-	41.9
Female	23,857	1,289	41	2,165	938	5,722	3,863	3,338	2,420	2,283	1,796	-	40.4
60-64													
Total	40,965	1,637	88	3,170	5,785	5,612	5,542	6,264	3,947	5,257	3,662	-	39.4
Male	28,638	1,248	-	1,097	3,431	3,473	4,645	5,516	2,560	3,724	2,942	-	42.3
Female	12,327	389	88	2,073	2,354	2,138	897	748	1,387	1,533	720	-	31.3
65+					•					•			
Total	48,737	4,973	-	3,943	5,313	4,663	4,373	6,160	5,856	9,076	4,380	-	40.3
Male	37,623	3,433	-	3,159	3,611	3,784	3,081	5,062	4,009	7,182	4,302	-	40.6
Female	11,114	1,540	-	784	1,702	879	1,292	1,098	1,847	1,894	78	-	39.6

Table 5.28 Cont'd

Hours	and Number	of Hours	s worke	ed Per we	ек by Ag	e Group	and Sex -	Country	Total: 20	11			-
					Number of	of Hours of	Worked Pe	er Week					Mean
Age Group and Sex												Not	Number of Hours Worked
and Sex	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	Stated	Per Week
All Ages	Total	0	10	712	10 21	22 00	01 00	10 10	40 07	0074	701	Olaica	
Total	2,392,409	92,926	4,302	123,246	178,682	255,730	281,506	563,879	356,542	333,927	201,398	271	44.1
Male	1,385,871	59,169	1,258	50,088	83,565	125,413	155,507	346,697	221,046	217,844	125,286		45.5
Female	1,006,537	33,757	3,045	73,158	95,118	130,318	125,999	217,182	135,496	116,084	76.112	271	41.8
10-14	.,,	00,101	0,010	. 0, . 00	00,110	,	0,000	,	,	,			
Total	28,534	32	-	7,172	4,831	4,257	847	2,510	2,827	4,804	1,253	-	29.6
Male	16,619	32	-	4,474	2,728	1,855	689	1,752	1,316	3,419	355	-	32.7
Female	11,915	-	-	2,699	2,104	2,402	158	757	1,512	1,385	898	-	22.3
15-19													
Total	173,177	3,985	718	14,569	17,747	20,001	15,667	32,922	20,029	29,500	18,039	-	41.4
Male	83,783	1,847	369	4,438	7,504	10,537	6,724	18,113	9,828	15,664	8,757	-	42.5
Female	89,394	2,138	349	10,131	10,243	9,463	8,943	14,809	10,200	13,836	9,282	-	40.2
20-24													
Total	406,472	10,868	1,017	18,959	31,207	41,020	43,956	95,492	64,704	62,711	36,537	-	45.9
Male	214,818	5,710	-	7,233	14,568	20,710	23,390	50,970	35,534	36,215	20,488	-	47.6
Female	191,654	5,158	1,017	11,726	16,639	20,310	20,566	44,522	29,170	26,496	16,049	-	43.6
25-29													
Total	513,915	16,299	851	25,929	33,318	48,752	61,954	132,012	80,706	73,044	41,050	-	45.1
Male	300,322	12,096	-	11,577	14,350	20,488	31,577	82,464	52,213	48,739	26,818	-	46.7
Female	213,593	4,203	851	14,352	18,968	28,264	30,377	49,547	28,493	24,306	14,233	-	42.3
30-34													
Total	353,610	15,681	232	16,000	23,962	38,253	40,851	88,206	54,457	47,249	28,719	-	45.3
Male	215,928	9,672	112	4,254	12,770	18,880	23,999	58,198	35,916	33,525	18,602	-	46.5
Female	137,681	6,009	120	11,746	11,192	19,373	16,852	30,009	18,541	13,723	10,116	-	43.2
35-39													
Total	320,418	11,664	538	12,345	21,167	38,666	44,641	77,298	46,165	43,157	24,777	-	44.4
Male	174,226	6,528	407	5,137	7,321	18,069	21,570	43,717	27,243	28,654	15,578	-	45.7
Female	146,192	5,136	130	7,208	13,845	20,597	23,071	33,580	18,922	14,503	9,199	-	42.2
40-44			105				00 EC /	17.00.1	00 -	o (7 70	40.000		
Total	194,945	11,673	132	5,850	14,314	19,411	26,564	47,964	30,571	21,773	16,692	-	44.8
Male	124,579	9,140	-	2,323	6,034	8,567	17,268	33,630	19,180	16,132	12,305	-	46.9
Female	70,366	2,533	132	3,527	8,280	10,844	9,296	14,333	11,392	5,641	4,387	-	40.3

Table 5.28	Currently Employed Population of Urban Areas Aged Ten Years and Above Who Sought or Available to Work Additional	
Ηοι	urs and Number of Hours Worked Per Week by Age Group and Sex - Country Total: 2011	

					Number of	of Hours of	Worked Pe	r Week					Mean
Age Group and Sex	Total	0	1-3	4-12	13-21	22.20	31-39	40.49	40.57	EQ 74	75 .	N.S	Number of Hours Worked Per Week
45.40	Total	0	1-3	4-12	13-21	22-30	31-39	40-48	49-57	58-74	75+	IN.3	Fei Week
45-49													
Total	147,707	6,994	-	6,671	10,139	15,071	14,888	37,263	22,919	19,974	13,517	271	44.9
Male	89,300	4,445	-	2,350	4,703	9,180	8,864	23,387	15,914	13,262	7,195	-	45.0
Female	58,407	2,549	-	4,321	5,435	5,892	6,024	13,876	7,004	6,712	6,322	271	44.7
50-54													
Total	102,712	6,492	686	4,307	6,337	11,034	14,219	25,138	15,761	10,730	8,008	-	40.8
Male	62,674	3,680	369	1,883	2,919	6,600	9,559	14,574	11,154	6,959	4,976	-	42.2
Female	40,038	2,812	317	2,424	3,417	4,434	4,660	10,564	4,607	3,771	3,032	-	39.0
55-59													
Total	61,219	2,627	41	4,330	4,562	8,992	8,003	12,650	8,599	6,652	4,763	-	41.3
Male	37,362	1,338	-	2,164	3,624	3,270	4,140	9,312	6,179	4,369	2,967	-	41.9
Female	23,857	1,289	41	2,165	938	5,722	3,863	3,338	2,420	2,283	1,796	-	40.4
60-64													
Total	40,965	1,637	88	3,170	5,785	5,612	5,542	6,264	3,947	5,257	3,662	-	39.4
Male	28,638	1,248	-	1,097	3,431	3,473	4,645	5,516	2,560	3,724	2,942	-	42.3
Female	12,327	389	88	2,073	2,354	2,138	897	748	1,387	1,533	720	-	31.3
65+					•					•			
Total	48,737	4,973	-	3,943	5,313	4,663	4,373	6,160	5,856	9,076	4,380	-	40.3
Male	37,623	3,433	-	3,159	3,611	3,784	3,081	5,062	4,009	7,182	4,302	-	40.6
Female	11,114	1,540	-	784	1,702	879	1,292	1,098	1,847	1,894	78	-	39.6

Table 5.28 Cont'd

		y Region and Type of	Work Sought or Av	ailable for	
		Турс ог	Work Obugint of 7.0	Other job with	
Region and Sex		At the	Other job in	more hours to	
5		present	addition to the	replace the	Not
	Total	main job	present	present job	Stated
Country Total		· · ·			
total	2,394,227	787,576	1,071,379	533,387	1,886
male	1,386,950	489,498	615,948	280,358	1,146
female	1,007,277	298,078	455,430	253,029	740
Tigray Region		,	,	,	
total	146,493	48,968	69,805	27,379	341
male	77,837	29,189	34,882	13,424	341
female	68,656	19,778	34,923	13,955	-
Affar Region	,	,	,	,	
total	32,247	7,941	20,395	3,911	-
male	19,907	4,760	12,642	2,505	-
female	12,340	3,181	7,753	1,406	-
Amhara Region					
total	408,820	163,962	163,738	80,655	464
male	233,735	96,644	96,320	40,712	59
female	175,085	67,318	67,418	39,943	405
Oromia Region					
total	859,709	274,137	370,101	214,690	781
male	490,124	168,460	205,921	114,997	746
female	369,585	105,677	164,180	99,692	35
Somali Region	,	,	,	,	
total	65,391	23,823	22,559	19,009	-
male	39,577	16,445	14,343	8,789	-
female	25,814	7,377	8,216	10,220	-
Benishangul gumuz Re					
total	24,767	7,317	13,014	4,436	-
male	14,093	4,460	7,250	2,383	-
female	10,674	2,858	5,764	2,053	-
SNNP Region					
total	288,485	93,710	151,271	43,204	299
male	167,114	59,111	86,873	21,130	-
female	121,371	34,600	64,398	22,074	299
Gambella Region					
total	18,336	6,539	9,737	2,060	-
male	8,938	3,702	4,091	1,144	-
female	9,399	2,836	5,646	916	-
Harari Region					
total	23,206	7,261	10,995	4,950	-
male	12,552	4,521	5,283	2,748	-
female	10,654	2,740	5,712	2,202	-
Addis Ababa City					
Administration					
total	487,976	141,978	220,947	125,051	-
male	300,252	94,107	138,035	68,111	-
female	187,724	47,871	82,913	56,940	-
Dire Dawa					
Administration	00				
total	38,796	11,940	18,816	8,040	-
male	22,821	8,097	10,309	4,414	-
female	15,976	3,842	8,507	3,626	-

Table 5.29Currently Employed Population of Urban Areas Aged Ten Years and
Above who were willing and Available to Work Additional Hours and
Type of Work Sought by Region and Sex: 2011

of Informal- Country Total: 2011 Employment Sector Percentage											
Age Group	Γ	Employmen	t Sector	N 1 - 4	Percentage						
and Sex	Total	Formal	Informal	Not Identified	Share of Informal						
All Ages	iUlai	i unidi	monnal	IUEIIIIIEU	monna						
Total	4,586,774	2,867,400	1,673,473	45,901	36.5						
Male	2,683,100	1,903,672	751,922	27,506	28.0						
Female	1,903,674	963,728	921,551	18,395	48.4						
10-14	1,303,074	303,720	321,001	10,000	-0.4						
Total	110,101	21,779	87,779	542	79.7						
Male	61,732	15,473	45,943	315	74.4						
Female	48,369	6,306	41,835	227	86.5						
15-19	10,000	0,000	11,000		00.0						
Total	310,666	149,567	157,846	3,252	50.8						
Male	168,297	91,292	75,361	1,644	44.8						
Female	142,369	58,275	82,485	1,609	57.9						
20-24	,	,	,	.,							
Total	704,208	469,685	223,695	10,829	31.8						
Male	375,763	261,238	109,979	4,546	29.3						
Female	328,445	208,447	113,716	6,283	34.6						
25-29	-										
Total	921,855	638,224	273,378	10,253	29.7						
Male	541,388	407,957	126,857	6,575	23.4						
Female	380,467	230,267	146,522	3,678	38.5						
30-34											
Total	667,379	435,911	224,064	7,404	33.6						
Male	409,219	300,150	103,439	5,631	25.3						
Female	258,160	135,761	120,626	1,773	46.7						
35-39											
Total	612,133	389,551	217,054	5,528	35.5						
Male	339,923	249,633	86,999	3,291	25.6						
Female	272,209	139,918	130,055	2,236	47.8						
40-44 Tatal	270.002	055 400	400.004	4 007	22.2						
Total	379,003 247,567	255,422	122,284 50,783	1,297	32.3						
Male Female	131,436	195,596 59,827	50,785 71,501	1,188 109	20.5 54.4						
45-49	131,430	59,027	71,501	109	54.4						
Total	305,474	187,690	114,854	2,931	37.6						
Male	183,656	137,330	44,837	1,488	24.4						
Female	121,818	50,359	70,016	1,442	57.5						
50-54	121,010	00,000	70,010	1,112	01.0						
Total	211,327	133,915	76,846	565	36.4						
Male	124,353	94,389	29,523	441	23.7						
Female	86,974	39,526	47,323	125	54.4						
55-59	,-		,								
Total	138,243	81,733	55,922	589	40.5						
Male	82,770	62,081	20,291	397	24.5						
Female	55,474	19,651	35,631	192	64.2						
60-64											
Total	93,764	48,471	44,775	519	47.8						
Male	59,407	40,208	18,967	232	31.9						
Female	34,357	8,263	25,807	287	75.1						
65+											
Total	132,621	55,452	74,977	2,192	56.5						
Male	89,025	48,324	38,944	1,757	43.7						
Female	43,595	7,128	36,033	434	82.7						

Table 5.30Currently Employed Population of Urban Areas Aged Ten Years and
Above by Age Group, Sex, Employment Sector and Percentage Share
of Informal- Country Total: 2011

F	Percentage S	hare of Info	rmal: 2011		
		Employme	ent Sector		Percentage
Region and				Not	Share of
Sex	Total	Formal	Informal	Identified	Informal
Country Total					
Total	4,586,774	2,867,400	1,673,473	45,901	36.5
Male	2,683,100	1,903,672	751,922	27,506	28.0
Female	1,903,674	963,728	921,551	18,395	48.4
Tigray Region					
Total	284,675	193,103	85,863	5,708	30.2
Male	151,743	115,308	33,296	3,139	21.9
Female	132,932	77,795	52,567	2,570	39.5
Affar Region					
Total	54,216	31,740	22,278	198	41.1
Male	34,550	23,138	11,214	198	32.5
Female	19,667	8,602	11,065	-	56.3
Amhara Region					
Total	829,103	489,016	331,673	8,413	40.0
Male	463,825	318,967	140,647	4,211	30.3
Female	365,278	170,049	191,026	4,203	52.3
Oromia Region					
Total	1,459,397	815,326	634,800	9,271	43.5
Male	852,204	563,743	282,676	5,785	33.2
Female	607,194	251,583	352,124	3,487	58.0
Somali Region					
Total	146,315	67,339	78,300	676	53.5
Male	85,964	50,364	34,963	638	40.7
Female	60,350	16,975	43,338	38	71.8
Benishangul - Gumuz Region					
Total	43,919	23,974	19,305	640	44.0
Male	24,946	16,129	8,302	516	33.3
Female	18,973	7,845	11,003	124	58.0
SNNP Region					
Total	619,621	344,821	271,416	3,385	43.8
Male	356,793	232,886	122,554	1,353	34.3
Female	262,829	111,935	148,862	2,032	56.6
Gambella Region					
Total	29,197	14,108	14,920	170	51.1
Male	13,648	9,223	4,255	170	31.2
Female	15,549	4,884	10,665	-	68.6
Harari Region					
Total	45,681	28,205	17,026	449	37.3
Male	24,960	17,888	6,812	259	27.3
Female	20,721	10,317	10,214	190	49.3
Addis Ababa City Administration					
Total	996,632	815,415	165,484	15,734	16.6
Male	628,648	525,693	92,671	10,285	14.7
Female	367,984	289,722	72,813	5,449	19.8
Dire Dawa					
Administration					
Total	78,018	44,353	32,408	1,257	41.5
Male	45,820	30,333	14,534	953	31.7
Female	32,198	14,020	17,874	304	55.5

Table 5.31Currently Employed Population of Urban Areas Aged Ten Years
and above by Region, Sex, Employment Sector and
Percentage Share of Informal: 2011

Informal: 2011											
		Employmen	t Sector		Percentage						
Major Towns				Not	Share of						
and Sex	Total	Formal	Informal	Identified	Informal						
TOTAL MAJOR TOWNS											
Total	1,735,489	1,296,298	411,772	27,419	23.7						
Male	1,067,793	834,181	214,999	18,613	20.1						
Female	667,695	462,117	196,773	8,806	29.5						
MEKELE TOWN											
Total	76,565	60,211	14,862	1,491	19.4						
Male	44,113	36,203	6,628	1,282	15.0						
Female	32,451	24,009	8,234	209	25.4						
ASAYITA TOWN											
Total	5,500	3,411	2,066	22	37.6						
Male	3,221	2,398	801	22	24.9						
Female	2,279	1,013	1,265	-	55.5						
GONDER TOWN	, -	, -	, -								
Total	74,273	36,565	36,438	1,270	49.1						
Male	40,985	23,659	16,349	978	39.9						
Female	33,287	12,906	20,089	293	60.4						
DESSIE TOWN	, -	,	-,								
Total	41,098	28,716	12,034	348	29.3						
Male	25,408	18,627	6,622	159	26.1						
Female	15,690	10,089	5,412	189	34.5						
BAHIR DAR TOWN		,	-,								
Total	79,886	53,761	23,970	2,154	30.0						
Male	44,962	33,128	10,957	878	24.4						
Female	34,924	20,634	13,013	1,276	37.3						
BISHOFTU TOWN	- ,-	-,	-,	, -							
Total	35,216	24,935	9,559	722	27.1						
Male	19,821	14,733	4,502	586	22.7						
Female	15,395	10,201	5,057	136	32.8						
ADAMA TOWN	,		,								
Total	94,161	65,683	26,331	2,147	28.0						
Male	58,846	41,280	15,916	1,650	27.0						
Female	35,315	24,402	10,415	498	29.5						
JIMMA TOWN	,	,	,								
Total	49,925	38,065	11,806	55	23.6						
Male	31,754	24,833	6,922	-	21.8						
Female	18,171	13,232	4,884	55	26.9						
SHASHEMENE	,	,	.,								
TOWN											
Total	47,409	18,851	28,294	264	59.7						
Male	28,968	14,296	14,507	166	50.1						
Female	18,441	4,556	13,788	97	74.8						
JIJIGA TOWN											
Total	27,499	17,278	9,545	676	34.7						
Male	19,539	13,062	5,838	638	29.9						
Female	7,960	4,216	3,707	38	46.6						
ASOSA TOWN	•	•	·								
Total	12,413	8,804	3,299	310	26.6						
Male	7,484	5,719	1,528	237	20.4						
Female	4,929	3,085	1,771	73	35.9						
	,	-,	, -	÷							

Table 5.32Currently Employed Populations of Major Towns Aged Ten Years
and above by Sex, Employment Sector and Percentage share of
Informal: 2011

Table 5.32 Cont'd

			Percentage		
Major Towns and Sex	Tatal	Formel	lus forma al	Not	Share of Informal
	Total	Formal	Informal	Identified	moma
HAWASSA TOWN					
Total	59,392	43,738	15,304	350	25.8
Male	36,138	26,752	9,035	350	25.0
Female	23,255	16,986	6,269	-	27.0
GAMBELLA TOWN					
Total	11,822	8,308	3,345	170	28.3
Male	7,125	5,577	1,378	170	19.3
Female	4,697	2,730	1,967	-	41.9
HARAR TOWN	,	,			
Total	45,681	28,205	17,026	449	37.3
Male	24,960	17,888	6,812	259	27.3
Female	20,721	10,317	10,214	190	49.3
ADDIS ABABA CITY A	OMINISTRATION	J			
Total	996,632	815,415	165,484	15,734	16.6
Male	628,648	525,693	92,671	10,285	14.7
Female	367,984	289,722	72,813	5,449	19.8
DIRE DAWA TOWN					
Total	78,018	44,353	32,408	1,257	41.5
Male	45,820	30,333	14,534	953	31.7
Female	32,198	14,020	17,874	304	55.5

Note: The above figure in this table does not include those who were engaged in subsistence farming and in private households.

Group and F	Cegiona. 201		10/ 11								
			Worki	ng in the Infor	mal Se		lajor Occupa	tional Grou	qu	1	
Region And Sex		Legislator		Technical		Service, Shop &	Skilled	Crafts	Plant		
- 3	Total	Senior		and		Market	agricultural	and	,Machine		
	Employed	Officials &		Associate		sales	and	related	Operators &		
	Persons	Managers	Professional	Professionals	Clerks	Workers	Fishery	trades	Assemblers	Occupation	Stated
COUNTRY TOTAL											
Total	1,673,473	1,624	4,738			594,383	134,381	528,354	16,177		-
Male	751,922	1,551	3,893	15,529		187,002		239,698	14,398		-
Female	921,551	73	845	2,532	4,187	407,381	62,269	288,656	1,780	153,829	-
TIGRAY REGION											
Total	85,863	-	460	589	250	38,434		26,370	397		-
Male	33,296	-	460	325	250	10,044		10,696	356	•	-
Female	52,567	-	-	264	-	28,390	3,789	15,673	41	4,410	-
AFFAR REGION											
Total	22,278	40	-	-	135	9,203	3,886	5,913	10	3,091	-
Male	11,214	40	-	-	115	3,653	3,703	2,113	-	1,589	-
Female	11,065	-	-	-	20	5,550	183	3,800	10	1,501	-
AMHARA REGION											
Total	331,673	554	641	2,113	654	108,221	25,258	115,057	1,405	77,771	-
Male	140,647	554	641	2,113	412	26,396	18,032	46,359	1,405	44,735	-
Female	191,026	-	-	-	242	81,825	7,226	68,697	-	33,036	-
OROMIA REGION											
Total	634,800	-	804	6,037	2,651	231,856	50,780	215,590	6,023	121,060	-
Male	282,676	-	506	5,324	1,107	76,439	21,680	103,070	5,093	69,458	-
Female	352,124	-	298	713	1,544	155,417	29,100	112,521	930	51,602	-
SOMALI REGION											
Total	78,300	-	-	375	462	35,232	9,927	7,853	536	23,917	-
Male	34,963	-	-	375	462	9,498	8,539	4,571	536	10,983	-
Female	43,338	-	-	-	-	25,734	1,388	3,282	-	12,934	-
BENISHANGUL											
GUMUZ REGION											
Total	19,305	-	-	361	-	4,966	2,537	5,851	40	5,548	-
Male	8,302	-	-	161	-	981	1,847	2,195	40	3,078	-
Female	11,003	-	-	201	-	3,985	690	3,657	-	2,470	-
SNNP REGION											
Total	271,416	-	651	1,171	2,102	93,573	23,499	79,970	2,242	68,208	-
Male	122,554	-	120	1,120	1,610	35,009	11,083	30,635	2,063	40,913	-
Female	148,862	-	531	50	492	58,564	12,417	49,335	178	27,295	-

Table 5.33: Distribution of Urban Population Aged Ten years and above Working in the Informal Sector by Major Occupational Group and Regions: 2011

Table 5.33 Cont d											
				Maj	or Occup	pational Gr	oup				
Region And Sex	Total Employed Persons	Legislator Senior Officials &	Professional	Technical and Associate Professionals	Clerks	Service, Shop & Market sales Workers	Skilled agricultur al and	Crafts and related trades	Plant ,Machine Operators & Assemblers	Elementary Occupation	Not
	Feisons	Managers	Professional	FIDIESSIDITAIS	CIEIKS	WOIKEIS	Fishery	liaues	Assemblers	Occupation	Stated
GAMBELLA REGION Total	14,920	_	17	25	23	2,786	7,374	3,483	23	1,189	_
Male	4,255		-	25	23	726	1,879	652	23	927	_
Female	10,665	-	17	- 20	- 20	2,060	5,495	2,831	- 20	262	-
HARARI REGION	. 0,000					_,	0,100	_,			
Total	17,026	-	44	125	273	8,808	177	3,636	229	3,734	-
Male	6,812	-	44	84	183	1,815	131	1,639	229	2,687	-
Female	10,214	-	-	41	90	6,993	46	1,997	-	1,047	-
ADDIS ABABA CITY ADMINISTRATION											
Total	165,484	958	2,123	6,799	3,971	48,060	2,197	56,818	4,818	39,741	-
Male	92,671	958	2,123	5,627	2,391	18,600	916	33,884	4,198	23,973	-
Female DIRE DAWA ADMINISTRATION	72,813	-	-	1,172	1,580	29,460	1,281	22,934	620	15,767	-
Total	32,408	73	-	467	430	13,246	1,417	7,812	454	8,508	-
Male	14,534	-	-	377	211	3,842	762	3,883	454	5,004	-
Female	17,874	73	-	91	219	9,404	654	3,929	-	3,504	-

Table 5.33 Cont'd

Informal Sector by Region, Sex and Major Branches of Industry: 2011 Major Branches of Industry Manufacturing, Manufacturing,												
				ninuustry	Agriculture							
		Construction,	Whole Sale	Other	Hunting,							
Region and		Mining &	and Retail	Service	Forestry &	Not						
Sex	Total	Quarrying	Trade	Sectors	Fishing	Stated						
Country Total												
Total	1,673,473	432,267	557,127	532,753	150,897	429						
Male	751,922	231,812	212,230	226,782	80,729	369						
Female	921,551	200,455	344,897	305,972	70,168	60						
Tigray Region	021,001	200,100	011,001	000,012	10,100	00						
Total	85,863	24,270	30,556	24,521	6,515	-						
Male	33,296	12,373	9,786	7,683	3,454	-						
Female	52,567	11,897	20,771	16,838	3,061	-						
Affar Region	02,007	11,007	20,111	10,000	0,001							
Total	22,278	4,509	9,619	4,451	3,700	_						
Male	11,214	2,148	4,153	1,385	3,528	_						
Female	11,065	2,360	5,466	3,066	172	_						
Amhara Region	11,005	2,300	5,400	3,000	172	-						
Total	331,673	84,141	102,779	107,046	37,337	369						
Male	140,647	43,100	31,805	43,569	21,804	369						
Female	191,026	41,041	70,974	63,478	15,534	303						
Oromia Region	191,020	41,041	10,914	05,470	15,554	-						
Total	634,800	177,378	215,586	186,942	54,834	60						
Male	282,676	98,521	215,586 84,319	76,331	23,505	00						
Female	352,124	78,857		110,611	23,505 31,329	-						
	552,124	70,007	131,267	110,011	31,329	60						
Somali Region	79 200	9 00F	22 112	22.074	15 010							
Total Male	78,300 34,963	8,005	32,112 9,608	22,971 10,385	15,213 10,342	-						
Female	43,338	4,628		12,586	4,871	-						
Benishangul Gumuz Reg		3,377	22,504	12,500	4,071	-						
Total	19,305	3,464	2,934	8,197	4,710							
Male	8,302	1,635	2,934 870	2,854	2,943	-						
Female	11,003	1,829	2,064	5,343	1,767	-						
Snnp Region	11,003	1,029	2,004	5,545	1,707	-						
Total	271,416	66,195	84,818	99,653	20,750							
Male	122,554	30,271	39,724	99,055 40,763	20,750	-						
Female	148,862	35,924	45,093	40,703 58,890	8,955	-						
Gambella Region	140,002	55,924	45,095	56,690	6,955	-						
-	14 020	1 022	E GEA	2 556	2 777							
Total Mala	14,920	1,932	5,654	3,556	3,777	-						
Male	4,255	892	970	746	1,647	-						
Female Harari Bagian	10,665	1,040	4,685	2,810	2,130	-						
Harari Region Total	17.026	2 000	0.002	2 057	177							
	17,026	3,800	9,092	3,957	177	-						
Male	6,812	1,871	2,146	2,664	131	-						
Female	10,214 stration	1,929	6,947	1,293	46	-						
Addis Ababa City Admini		E4 007	10 607	60.000	0.040							
Total Malo	165,484	51,227	48,627	63,382	2,248	-						
Male	92,671	31,908	24,165	35,940	658 1 501	-						
Female	72,813	19,319	24,462	27,442	1,591	-						
Dire Dawa Administration		7 0 4 0	15 0 40	0.077	4 005							
Total	32,408	7,346	15,349	8,077	1,635	-						
Male	14,534	4,465	4,685	4,462	922	-						
Female	17,874	2,882	10,664	3,615	713	-						

Table 5.34: Distribution of Urban Population Aged Ten Years and Above Working in the Informal Sector by Region, Sex and Major Branches of industry: 2011

	Mean N	Number of Hours	of Work by Majo	r Branches of	Industry
		Manufacturing,	Whole Sale	Other	Agriculture,
	Total	Construction,	and Retail	Service	Hunting,
Region and Sex	TOLAI	Mining &	Trade	Sectors	Forestry &
		Quarrying	Trade	00013	Fishing
COUNTRY TOTAL					
Total	38.5	37.3	40.3	38.8	33.7
Male	42.0	40.5	44.5	43.5	36.0
Female	35.5	33.5	37.7	35.4	31.0
TIGRAY REGION					
Total	44.6	43.5	45.6	46.2	38.0
Male	50.8	48.3	57.1	49.2	45.8
Female	40.6	38.6	40.2	44.8	29.2
AFFAR REGION					
Total	48.3	43.8	47.1	56.1	47.4
Male	49.4	52.2	44.1	66.5	47.2
Female	47.2	36.3	49.4	51.4	51.8
AMHARA REGION					
Total	35.3	32.3	34.7	38.8	33.9
Male	39.0	35.0	41.3	41.3	39.2
Female	32.6	29.4	31.8	37.1	26.4
OROMIA REGION					
Total	40.2	39.2	41.5	42.2	31.7
Male	42.9	41.3	44.3	46.6	32.4
Female	38.1	36.6	39.7	39.1	31.2
SOMALI REGION					
Total	39.3	39.3	46.4	30.7	37.1
Male	40.7	40.4	46.0	40.2	36.3
Female	38.1	37.8	46.6	22.8	38.7
BENISHANGUL GUMUZ F	REGION				
Total	27.3	28.5	24.9	30.0	23.3
Male	28.1	31.1	29.9	27.9	26.0
Female	26.8	26.1	22.7	31.2	18.8
SNNP REGION					
Total	34.4	35.3	37.5	31.9	31.3
Male	38.6	39.1	40.7	38.6	30.3
Female	31.0	32.1	34.7	27.3	32.6
GAMBELLA REGION					
Total	38.3	38.7	27.9	42.9	49.5
Male	46.5	40.1	44.1	37.8	55.4
Female	35.1	37.5	24.5	44.3	45.0
HARARI REGION					
Total	43.3	49.7	42.9	37.6	49.4
Male	44.9	51.6	42.0	42.4	48.4
Female	42.2	47.9	43.2	27.9	52.0
ADDIS ABABA CITY ADM	INISTRATION				
Total	40.8	37.8	44.7	40.4	40.5
Male	46.2	43.4	50.8	45.6	43.7
Female	34.0	28.4	38.7	33.5	39.2
DIRE DAWA ADMINISTRA	ATION				
Total	36.6	33.1	39.9	35.3	26.7
Male	39.3	36.8	42.7	40.3	29.7
Female	34.3	27.3	38.7	29.2	22.7

Table 5.35: Distribution of Urban Population Aged Ten years and above Working in the
Informal Sector by Regions, Sex, Major Economic Sector and Mean Number
Of Hours Worked Per week: 2011

		,		Highest Grade Completed											
								J			ond Seconda	ary			
									TEVT	,					
						Secondary			10+1 &		Diploma	Degree		Diploma	
Region and	All	All		Non	Grades	not	Secondary		10+2	10+2 not	not	not	Certificate	and above	Not
Sex	Persons	Illiterate	Pre-school	Formal	1-8	completed	Completed	Preparatory	Completed	Completed	completed	completed	Completed	Completed	Stated
COUNTRY TOTAL															
Total	1,673,473				746,868	87,487	139,652	7,378	7,820	2,953	5,015	1,381	2,278	23,287	310
Male		169,978			397,579	52,082	77,408	4,001	3,945	1,091	2,114	914	1,561	15,666	310
Female	921,551	439,273	1,354	13,167	349,289	35,405	62,244	3,377	3,875	1,863	2,901	466	717	7,621	-
TIGRAY REGION															
Total	85,863	38,706	557	1,300	32,935	3,385	6,721	565	341	-	361	-	-	991	-
Male	33,296	9,958	286	862	16,743	1,668	2,024	305	341	-	361	-	-	747	-
Female	52,567	28,748	271	438	16,192	1,716	4,697	260	-	-	-	-	-	245	-
AFFAR REGION															
Total	22,278	8,753	-	.,	10,715	466	510	9	-	-	10	-	-	569	-
Male	11,214	2,946	-	1,001	6,640	186	321	9	-	-	-	-	-	78	-
Female	11,065	5,807	-	212	4,075	280	189	-	-	-	10	-	-	491	-
AMHARA REGION															
Total	331,673	146,700			117,850	11,923	26,069	1,097	1,426	726	1,423	551	369	1,464	-
Male	140,647	36,463	352	15,294	64,755	6,086	13,637	1,019	628	557	691	175	319	674	-
Female	191,026	110,237	-	6,430	53,096	5,837	12,432	78	798	169	732	376	50	790	-
OROMIA REGION															
Total	634,800	202,752	1,413	6,148	313,451	36,546	55,485	2,832	3,690	1,424	1,902	432	691	8,033	-
Male	282,676	61,361	380	2,518	159,163	21,264	28,330	1,087	1,921	358	445	432	480	4,937	-
Female	352,124	141,392	1,032	3,630	154,288	15,282	27,155	1,746	1,770	1,065	1,457	-	211	3,096	-
SOMALI REGION															
Total	78,300	48,084	100	965	23,608	2,805	2,000	307	-	-	-	-	-	431	-
Male	34,963	17,166	50	426	13,495	1,843	1,665	-	-	-	-	-	-	318	-
Female	43,338	30,918	50	539	10,113	962	335	307	-	-	-	-	-	113	-
BENISHANGUL															
GUMUZ REGION															
Total	19,305	9,897	-	670	6,886	758	675	134	-	-	73	-	60	150	-
Male	8,302	3,151	-	560	3,697	358	435	31	-	-	20	-	-	50	-
Female	11,003	6,746	-	110	3,190	400	241	103	-	-	53	-	60	100	-
SNNP REGION															
Total	271,416	86,791	-		145,520	14,105	17,922	933	774	435	290	-	721	1,893	-
Male	122,554	18,689	-	1,821	81,303	8,366	9,822	386	245	-	73	-	475	1,373	-
Female	148,862	68,102	-	210	64,217	5,739	8,100	547	529	435	217	-	245	520	-

Table 5.36: Distribution of Urban Population Aged Ten years and Above Working in the Informal Sector by Literacy Status and Educational Level, Sex and Regions: 2011

Table 5.36 Cont'd

				Highest Grade Completed											
										Bey	ond Seconda	ary			
									TEVT						
Region and	A II	A 11				Secondary			10+1 &		Diploma	Degree		Diploma	Niet
Sex	All Persons	All Illiterate		Non	Grades	not	Secondary	D	10+2	10+2 not	not	not	Certificate	and above	
		IIIIterate	Pre-school	Formal	1-8	completed	Completed	Preparatory	Completed	Completed	completed	completed	Completed	Completed	Stateu
GAMBELLA REGIO	N														
Total	14,920	7,887	-	102	4,916	1,399	436	-	63	31	51	-	-	34	-
Male	4,255	1,532	-	56	1,954	465	161	-	31	31	26	-	-	-	-
Female	10,665	6,355	-	46	2,962	935	274	-	32	-	26	-	-	34	-
HARARI REGION															
Total	17,026	5,166	-	205	8,941	1,204	977	48	-	42	77	-	81	284	-
Male	6,812	1,110	-	81	4,045	564	690	-	-	-	38	-	-	284	-
Female	10,214	4,056	-	124	4,896	641	287	48	-	42	39	-	81	-	-
ADDIS ABABA CITY	ADMINIST	RATION													
Total	165,484	41,806	-	2,892	68,089	13,605	25,994	1,451	1,296	-	606	308	287	8,840	310
Male	92,671	13,652	-	1,553	38,383	10,361	18,950	1,164	707	-	319	308	287	6,678	310
Female	72,813	28,154	-	1,340	29,706	3,244	7,044	287	589	-	287	-	-	2,162	-
DIRE DAWA ADMIN	ISTRATION	١													
Total	32,408	12,708	-	87	13,955	1,290	2,862	-	230	296	222	91	69	598	-
Male	14,534	3,951	-	-	7,401	921	1,373	-	73	145	141	-	-	529	-
Female	17,874	8,757	-	87	6,554	369	1,489	-	158	151	80	91	69	69	-

	y Regions	,				of Wor	kers					
					5.20							
Region and Sex												Not
	Total	<10	11-19	20-29	30-39	40-49	50-59	60-69	70-79	80-89	90+	Stated
COUNTRY TOTAL												
Total	1,673,473	1,628,366						604	109		4,346	1,189
Male	751,922	718,400					1,330	26	-		2,743	1,140
Female	921,551	909,965	5,117	1,132	76	37	713	578	109	-	1,603	49
TIGRAY REGION												
Total	85,863	82,829	2,648	93	93	93	-	-	-	-	-	108
Male	33,296	30,658	2,252	93	93	93	-	-	-	-	-	108
Female	52,567	52,172	396	-	-	-	-	-	-	-	-	-
AFFAR REGION												
Total	22,278	22,079	180	9	-	-	-	-	-	-	-	-
Male	11,214	11,033	180	-	-	-	-	-	-	-	-	-
Female	11,065	11,046	-	9	-	-	-	-	-	-	-	-
AMHARA												
REGION												
Total	331,673	325,109	3,048	1,258	373	102	591	-	109	-	375	409
Male	140,647	135,373	2,344	1,036	297	102	591	-	-	-	273	409
Female	191,026	189,736	704	222	76	-	-	-	109	-	101	-
OROMIA REGION												
Total	634,800	619,218	8,477	2,068	120	105	601	578	-	94	923	97
Male	282,676	273,190	5,489	1,478	120	68	281	-	-	94	341	49
Female	352,124	346,028	2,988	590	-	37	320	578	-	-	582	49
SOMALI REGION	,	,	,									
Total	78,300	76,971	532	43	206	-	-	-	-	-	49	-
Male	34,963	33,794	532	43	206	-	-	-	-	-	49	-
Female	43,338	43,177		-		-	-	-	-	-	-	-
BENISHANGUL GL												
Total	19,305	18,818	282	58	23	-	62	-	-	62	-	-
Male	8,302	7,815	282	58	23	-	62	-	-	62	-	-
Female	11,003	11,003	- 202		- 20	-	- 02	-	-		-	-
SNNP REGION	11,000	11,000										
Total	271,416	267,256	2,853	286	63	229	323	-	-	-	_	-
Male	122,554	119,303	2,513	286	63	229		-	-	-	_	-
Female	148,862	147,953	340	200		- 223	323	_	_	_	_	_
GAMBELLA REGIC		177,300	540	-	-	-	525	-	-	-	-	-
Total	14,920	14,804	67	-	23	_	-	26	-	-	-	-
Male	4,255	4,185	21	-	23	-	-	26			-	-
Female	4,235	10,619	46	-	23	-	-	20	-	-	-	-
HARARI REGION	10,005	10,019	40	-	-	-	-	-	-	-	-	-
	17.006	16 674	250			EE					10	
Total	17,026	16,674	250	-	-	55 55	-	-	-	-	48	-
Male	6,812	6,460	250	-	-	55	-	-	-	-	48	-
	10,214	10,214	-	-	-	-	-	-	-	-	-	-
ADDIS ABABA CIT			4 000	4 005	505	000	000				0.054	
Total	165,484	153,548		1,625	595	929	322	-	-		2,951	574
Male	92,671	83,197		1,314	595	929	322	-	-	-	2,032	574
Female	72,813	70,351	574	311	-	-	-	-	-	-	919	-
DIRE DAWA ADMII												
Total	32,408	31,058	592	155	320	-	144	-	-	-	-	-
Male	14,534	13,392	523	155	320	-	75	-	-	-	-	-
Female	17,874	17,666	69	-	-	-	69	-	-	-	-	-

Table 5.37: Distribution of Urban Population Aged Ten Years and Above Working in the InformalSector by Regions, Sex and Size of Workers: 2011

	J Flaces.						Туре	of Worki	ng Places						
Region and Sex		Business		At	On	Gulit or Open	Farm Area/		Quarrying/	Any Where	Construction	Where	At lakes/ River/		Not
COA	Total	house	Office	home	Street	Market	Field	Factory	mining	as Found	Area	available	Pond	Others	
COUNTRY TOTAL	_														
Total	1,673,473	169,346	16,861	508,301	131,037	292,612	97,926	3,388	4,551	219,370	56,764	70,475	32,407	70,436	-
Male	751,922	88,460	10,127	117,167	69,429	89,385	59,207	2,589	3,982	182,027	49,263	51,017	7,006	22,263	-
Female	921,551	80,886	6,734	391,134	61,608	203,227	38,719	799	569	37,343	7,501	19,457	25,402	48,173	-
TIGRAY REGION															
Total	85,863	9,726	238	41,866	5,677	16,044	2,399	264	-	7,234	1,461	912	-	41	-
Male	33,296	5,487	134	9,592	3,077	4,652	2,399	264	-	5,855	1,461	376	-	-	-
Female	52,567	4,239	105	32,275	2,600	11,392	-	-	-	1,379	-	536	-	41	-
AFFAR REGION															
Total	22,278	2,726	10	8,010	2,935	1,352	642	19	10	2,070	818	278	-	3,409	-
Male	11,214	1,698	-	1,408	1,314	146	497	9	10	2,060	779	55	-	3,236	-
Female	11,065	1,029	10	6,602	1,621	1,205	145	9	-	10	38	223	-	172	-
AMHARA REGION															
Total	331,673	27,549	4,691	,	21,185	66,098	20,969	905	593	43,435	10,819	13,811	4,435	8,843	-
Male	140,647	12,095	2,689	22,329	13,367	15,655	14,673	905	95	34,937	8,278	10,796	937	3,891	-
Female	191,026	15,454	2,002	86,012	7,818	50,442	6,295	-	499	8,498	2,541	3,015	3,497	4,952	-
OROMIA REGION															
Total	634,800	69,725	5,697	200,846		116,814	26,693	1,773	2,547	75,444	22,378		21,700		-
Male	282,676	39,620	3,865	49,878	21,787	38,486	13,281	983	2,547	63,097	19,307	19,862	4,363	5,599	-
Female	352,124	30,105	1,833	150,967	25,783	78,328	13,412	790	-	12,347	3,071	8,518	17,336	9,634	-
SOMALI REGION															
Total	78,300	11,806	61	5,508	6,299	13,141	21,155	-	237	12,404	1,095	2,651	1,555	2,388	-
Male	34,963	5,432	61	814	2,716	2,825	12,164	-	237	7,416	882	2,015	-	401	-
Female	43,338	6,374	-	4,694	3,583	10,317	8,991	-	-	4,988	213	636	1,555	1,987	-
BENISHANGUL G		-	004	F 000	140	0.005	0.007			0.400	540	004	400	4 054	
Total	19,305	1,199	331	5,683	440	2,365	3,927	-	-	2,429	516	331	433	1,651	-
Male	8,302	269	48	823	304	582	2,378	-	-	2,275	416	231	135	841	-
Female SNNP REGION	11,003	929	282	4,860	136	1,783	1,549	-	-	154	100	100	298	810	-
	074 440	04 004	0 400	74.000	47 407	E4 007	45 000	~~~	C45	07.057	0.074	44 040	2 200	22 750	
Total Male	271,416	21,004 9.562	3,422 1.551	74,296 14,117	17,107	54,697	15,629	66 66	615 615	37,657	8,274	11,619		23,750	-
	122,554	- ,	,		11,288	20,549	10,946	00	015	31,680 5,977	7,420 854	9,269	1,398	4,095	-
Female	148,862	11,442	1,872	60,179	5,819	34,149	4,683	-	-	5,977	854	2,350	1,882	19,656	-

Table 5.38: Distribution of Urban Population Aged Ten years and Above Working in the Informal Sector by Regions, Sex and Type of Working Places: 2011

Table 5.38 Cont'd

		Type of Working Places													
Region and Sex	Total	Business house	Office	At home	On Street	Gulit or Open Market	Farm Area/ Field	Factory	Quarrying/ mining	Any Where As Found	Construction Area	Where customer available	At lakes/ River/ Pond	Others	Not stated
GAMBELLA REGIO	N														
Total	14,920	241	61	3,548	567	1,317	3,610	-	-	1,372	151	124	878	3,050	-
Male	4,255	143	27	290	250	318	1,611	-	-	1,116	81	101	88	231	-
Female	10,665	99	34	3,258	317	999	1,999	-	-	256	70	23	791	2,819	-
HARARI REGION															
Total	17,026	2,264	242	4,019	3,097	2,962	136	41	262	2,173	609	496	127	599	-
Male	6,812	840	190	1,122	668	645	90	41	192	1,752	570	402	84	216	-
Female	10,214	1,423	52	2,897	2,430	2,317	46	-	70	422	39	94	42	383	-
ADDIS ABABA CITY	Y ADMINIST	FRATION													
Total	165,484	19,460	1,657	49,238	20,568	13,707	729	322	287	30,468	8,721	10,564	-	9,763	-
Male	92,671	11,511	1,346	16,000	12,993	3,869	-	322	287	28,084	8,146	7,038	-	3,073	-
Female	72,813	7,949	311	33,237	7,575	9,838	729	-	-	2,385	574	3,526	-	6,689	-
DIRE DAWA ADMIN	ISTRATIO	N													
Total	32,408	3,646	450	6,945	5,592	4,116	2,037	-	-	4,683	1,923	1,308	-	1,708	-
Male	14,534	1,803	216	794	1,667	1,657	1,168	-	-	3,755	1,923	871	-	679	-
Female	17,874	1,842	234	6,151	3,925	2,459	869	-	-	928	-	436	-	1,029	-

Region Type of Difficulties Benishangul-Addis Dire Total Tigray Afar Amhara Oromia Somalie Gumuz SNNPR Gambela Harari Ababa Dawa **First Difficulties** Total 1,673,473 85,863 22,278 331,673 634,800 78,300 19,305 271,416 14,920 17,026 165,484 32,408 I do not have Difficulties 521,095 36,180 6,569 115,343 169,939 27,158 8,384 107,854 4,366 3,537 35,884 5,881 Lack of Premises 163,766 7,422 1,973 38,032 54,534 5,646 1,530 20,929 367 2,544 28,421 2,367 3.299 Lack of Working Capital 361.961 25.102 5.234 72.883 134.112 8.865 3.378 69.957 2.085 33.186 3.861 Limited Market 235,950 9,244 2,907 56,459 98,348 8,917 2,508 27,574 722 1,273 22,135 5.863 No Training/Inadequate Skill/ 14,801 2,009 232 2.124 4,116 611 60 762 10 136 4,408 334 **Difficulty in Obtaining License** 10.171 68 355 503 4,364 592 70 846 46 222 2,399 708 Lack of credit 1,364 13,179 1,260 966 285 3.955 1,255 311 _ 3,545 237 -Access to raw material 98,206 1.213 1.417 8.323 58.198 2,970 313 10.240 584 1.982 7.307 5.658 Lack of Working Material 13.221 604 664 1.149 7.686 619 54 1.063 23 178 1.035 146 Lack of Enough Information 2,976 -121 739 440 52 62 60 21 1,481 --Pressure from Government Regulation 9.191 243 161 1.392 2,560 1,086 65 1,721 194 91 1,231 447 Inadequate salary 38,775 478 3,781 714 1,219 10 6,667 18,847 600 34 5,880 545 Inconvenient work place 58.392 292 387 417 1.219 9.012 26.571 2.607 5.579 1.169 9.785 1.353 8,786 Others 130,085 882 449 17,057 17.446 1.679 19.686 6.080 1,881 5,008 51,130 Not Stated 1,705 1,705 Second Difficulties Total 1,673,473 85,863 22,278 331,673 634,800 78,300 19,305 271,416 14,920 17.026 165,484 32,408 I do not have Difficulties 927,540 53,965 12,643 195,192 323,524 57,330 14,332 166,421 10,070 8,358 69,203 16,504 Lack of Premises 87,710 3,882 2,232 20,419 37,654 1,449 529 6.316 1,077 12,736 1,301 115 Lack of Working Capital 173,028 8,583 2,083 1,891 41,369 65,444 3,746 1,236 25,060 546 21,648 1,423 Limited Market 191,655 7,909 35,717 33,566 1,036 4,317 1,898 83,210 3,374 1.258 204 19,167 No Training/Inadequate Skill/ 15.335 1.585 10 3.425 5.476 1.716 140 762 41 2.108 72 -**Difficulty in Obtaining License** 12.105 1.582 4,282 315 1.042 158 3.623 158 -868 76 -Lack of credit 37,852 1,144 596 3.055 14,441 2,374 178 11.031 89 463 4,178 304 Access to raw material 63,668 3,008 700 8,026 30,306 1,573 513 4,493 217 1,331 10,837 2,666 Lack of Working Material 14,024 286 375 1,839 5,532 576 54 2,256 55 2,477 469 105 Lack of Enough Information 262 53 6.261 286 1,500 2,964 155 902 139 ---Pressure from Government Regulation 6.643 396 240 780 343 22 1.183 425 922 678 1.654 -Pressure from Government 425 678 Regulation 6,643 396 240 780 1,654 343 22 1,183 922 -Inadequate salary 13.065 53 10 1.942 6.362 22 1.764 10 84 2.303 400 113 Inconvenient work place 63,453 4,240 386 7,052 25,536 4,004 251 9,459 283 1,009 9,514 1,719 Others 59,430 527 236 8,070 28,415 1,127 696 7,909 3,281 1,046 5,865 2,258 Not Stated 1.705 1.705 _ .

Table 5.39: Distribution of Urban Population Aged Ten Years and Above Working in the Informal Sector by Regions, Order and
Type of Difficulties Faced: 2011

Table 5.39 Cont'd

						R	egion					
Type of Difficulties							Benishangul				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	-Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Third Difficulties												
Total	1,673,473	85,863	22,278	331,673	634,800	78,300	19,305	271,416	14,920	17,026	165,484	32,408
I do not have Difficulties	1,322,586	73,521	17,073	270,797	493,956	68,542	18,045	225,668	12,072	13,471	103,297	26,143
Lack of Premises	31,240	1,060	514	7,673	12,144	654	26	2,459	21	331	5,779	579
Lack of Working Capital	42,783	1,590	523	9,222	19,664	1,157	156	3,003	111	693	5,432	1,232
Limited Market	63,755	1,576	1,634	12,027	27,747	462	337	7,296	50	588	11,276	763
No Training/Inadequate Skill/	11,944	915	435	1,742	4,970	687	-	-	-	81	3,046	69
Difficulty in Obtaining License	10,008	925	107	1,313	3,969	887	-	549	25	-	2,155	78
Lack of credit	28,782	1,374	202	4,549	12,784	784	119	4,768	97	287	3,747	69
Access to raw material	39,569	1,757	294	6,597	12,534	908	223	5,494	47	437	10,264	1,013
Lack of Working Material	12,066	994	307	3,513	4,541	792	129	1,008	63	99	620	-
Lack of Enough Information Pressure from Government	4,982	-	64	375	1,203	315	-	81	-	81	2,792	69
Regulation Pressure from Government	8,229	561	84	1,117	1,531	259	-	1,782	12	44	2,595	244
Regulation	8,229	561	84	1,117	1,531	259	-	1,782	12	44	2,595	244
Inadequate salary	5,990	-	263	1,249	2,387	52	-	-	23	44	1,746	226
Inconvenient work place	52,855	1,272	757	5,767	20,322	2,542	116	10,776	200	697	9,534	871
Others	36,980	320	21	4,025	17,048	262	153	8,532	2,197	172	3,200	1,050
Not Stated	1,705	-	-	1,705	-	-	-	-	-	-	-	-

CHAPTER VI

UNEMPLOYEMENT IN URBAN AREAS

6.1. Introduction

Unemployment is one of the indicators that have been used to measure the extent of unutilized human resource in the economy at a given area or region in a certain period of time. It indicates the quantity and quality of the human labour a nation or a region would have, and should be used to promote its economic development, but not actually used. The unutilized human resource as a result of unemployment has its own social and economic implication. Therefore, the availability of reliable and timely statistical data on unemployment is vital to formulate and design new strategies as well as to monitor the existing employment policies.

A wide range of information regarding unemployment and the characteristics of unemployed population were collected in the 2011 Urban Employment and Unemployment Survey. This information is useful to estimate the number of unemployed population and rate of unemployment at various reporting levels.

This chapter presents the size of unemployed population and rate of unemployment at national, regional and major urban levels. In addition, the characteristics of unemployed population such as age, sex, education, training status, marital status, previous work experience, duration of unemployment, type of jobs looking for, problems in establishing own business or enterprise, actions taken to seek work ... etc., are presented in detail. These information were collected from persons aged ten years and above by using the two known measurement approaches, namely current and usual approaches. But, the results presented in this chapter refer only the current activity status approach.

During the data collection period, respondents in the sampled household were asked detailed labour force questions using structured questionnaires. These includes:-

- a. whether they were seeking for employment, if not reason for not seeking.
- b. whether available for work during the coming one month, if not reason for not being available.
- c. type of job looking for /i.e. self employment or paid employment or other/
- d. type of problems faced to establish own business.
- e. previous work experience and duration of unemployment.

Based on the responses obtained from the respondents aged ten years and above, the measurement of unemployment was done using various methods.

6.2 Measurement of Unemployment

The measurement of unemployment is based on the following three criteria that must be satisfied simultaneously: "without work", "currently available for work" 'and "seeking work" (ILO, 1983). The standard definition of unemployment is based on the "seeking work" criterion that can be interpreted as activities or efforts that non-working persons performed to look for jobs during a specified reference period or prior to it (i.e., paid or self employment). The specific steps include registration at a public or private exchange; applying to employers; checking at worksites, farms, factory gates, market or other assembly places; placing or answering news paper advertisements; seeking assistance of friends or relatives; visit vacancy boards; looking for land, building, machinery or equipment to establish own enterprise, and arranging financial resources; applying for work permits and licenses...etc.

However, the standard definition of unemployment with its emphasis on seeking work criterion might be restrictive and might not fully capture the prevailing employment situations in many developing countries including Ethiopia, where the labour market is largely unorganized or of limited scope, where labour absorption is inadequate or where the labour force is largely self-employed. Hence, the International standards introduced provisions, which allows for the relaxation of the seeking work criterion in certain situations. The provisions are two types, namely, partial relaxation and complete relaxation. Following the recommendations of the International standard and considering the prevailing national situations, the employment and unemployment and related labour force surveys used the above provisions to measure the different forms of unemployment. The definition of the two provisions is described as follows.

Under partial relaxation, the definition of unemployment includes discouraged job seekers, future start and layoffs in addition to persons satisfying the standard definition. Discouraged job seekers are persons who want a job but did not take any active step to search for work because they believe that they cannot find one. Future starts are those persons without work who have made arrangements to take up paid employment or to undertake self-employment activity at a date subsequent to the reference period. Layoff is a person whose contract of employment or activity has been suspended by the employer for a specified or unspecified period at the end of which the person concerned has a recognized right or recognized expectation to cover employment with the employer (ILO, 1990).

Under the completely relaxed definition, unemployment includes persons without work and those who are available for work, including those who were or were not seeking work. That is, the seeking work criterion is completely relaxed and unemployment is based on the "without work" and "availability" criterion only. The availability in this situation is tested by asking the willingness to take up work for wage or salary in locally prevailing terms, or readiness to undertake self-employment activity; given the necessary resources and facilities. It should be noted that fulltime students are considered as available if and only if they are ready to withdraw from their studies in order to accept a job.

This survey collected unemployment data in the standard, partially relaxed and completely relaxed options of measurements. After thorough evaluation and assessment of the results obtained using the three alternative and complementary measures; the rates obtained using the completely relaxed definition was found most plausible and hence selected for reporting. In this survey, those persons aged ten years and above who had no work or did not work at least four hours or did not have job to return to, were asked to respond whether they were available or willing to work if job was found during the coming one month. Those who responded "Yes" answer to this question were further tested whether they were ready to take a job under prevailing conditions. By doing so, those persons who had no job but available to work are considered as unemployed at the time of the survey.

6.3. Size and Rate of Urban Unemployment

The proportion of unemployed population aged ten years and above is described using unemployment rate. Unemployment rate is computed as percentage of the unemployed population over the total number of economically active population. The rate indicates that the number of persons who are ready to participate or engaged in the production of goods and services. A higher rate of unemployment implies the loss of individual contributions to the growth of Growth Domestic Product (GDP) as well as the national economy, and hence a waste of human resource.

The distribution of currently unemployed population and unemployment rate of urban areas of the country and regions aged ten years and above by age group, sex is presented in Table 6.1 and 6.1.1 - 6.1.11, respectively. The survey result reveals that the unemployment rate at national urban level was 18.0 percent. The corresponding rates for males and females were 11.4 percent and 25.3 percent, respectively (Table 6.1). Unemployment rate in the regions illustrates that the highest rate was recorded in Addis Ababa 25.1 percent and Dire Dawa administration 22.9 percent, while the lowest rate as compared to the highest ones was found in Benishangul-Gumuz (9.3 percent) and Gambella (7.7 percent) regions. But in the other regions of the country, the rate was found in the range 10 - 20 percent (Table 6.1.1 - 6.1.11).

Table 6.2 and 6.2.1 - 6.2.15 shows the distribution of currently unemployed population of total major towns and sixteen selected major towns classified by age group and sex. Among the major towns, the highest unemployment rate above 20.0 percent was recorded in Addis Ababa, Dire Dawa, Jijiga, Adama, Bahir Dar and Mekele towns. The lowest unemployment rate as compared to the highest ones was found in Asosa town (9.7 percent).

With regard to literacy status and educational attainment of the currently unemployed population of each region is shown in Table 6.3 and 6.4, while the training status and types of training of the unemployed illustrated in Table 6.5. The distribution of unemployed persons disaggregated by marital status, age group and sex is presented in Table 6.6. Table 6.7 shows similar information for each region classified by sex.

Those persons aged ten years and above who were looking for work during the three months prior to date of interview were asked to report what steps they have taken to seek work and type of work they look for during this period. Accordingly, Table 6.8 presents the distribution of unemployed persons who were looking for work during the three months prior to the survey date by sex and step taken to seek work for urban areas of each region. Table 6.9 also provides data on type of job they were looking for by sex and each region.

The size of unemployed persons of urban areas of the country categorized by age group sex, previous work experience and duration of unemployment is presented in Table 6.10. The corresponding figure by sex and region is presented in Table 6.11. The distribution of unemployed persons according to the type of problems they faced to start own business by region and sex is shown in Table 6.12.

Age Group	Eco	nomically Act	tive	Ur	nemployed		Unemployment Rate				
	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	6,265,367	3,304,538	2,960,829	1,125,536	375,585	749,950	18.0	11.4	25.3		
10-14	163,037	79,007	84,029	18,226	7,394	10,832	11.2	9.4	12.9		
15-19	587,006	250,346	336,660	157,489	57,681	99,808	26.8	23.0	29.6		
20-24	1,106,287	510,306	595,982	308,809	106,010	202,800	27.9	20.8	34.0		
25-29	1,236,396	647,935	588,461	226,773	68,133	158,640	18.3	10.5	27.0		
30-34	845,564	481,734	363,830	126,267	42,168	84,099	14.9	8.8	23.1		
35-39	751,307	390,114	361,194	94,558	26,782	67,776	12.6	6.9	18.8		
40-44	467,174	286,581	180,594	57,218	17,180	40,038	12.2	6.0	22.2		
45-49	370,591	207,346	163,246	44,271	10,464	33,807	11.9	5.0	20.7		
50-54	267,551	149,792	117,760	34,995	10,971	24,024	13.1	7.3	20.4		
55-59	172,102	99,621	72,481	19,201	6,794	12,407	11.2	6.8	17.1		
60-64	124,728	80,140	44,589	16,508	7,265	9,243	13.2	9.1	20.7		
65+	173,622	121,618	52,004	21,220	14,743	6,477	12.2	12.1	12.5		

Table 6.1Currently Unemployed Population of Urban Areas Aged Ten Years and Above
by Age Group, Sex and Unemployment Rate, COUNTRY TOTAL: 2011

 Table 6.1.1
 Currently Unemployed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, TIGRAY REGION: 2011

Age Group	Eco	nomically Ac	tive	Ur	employed		Une	employme Rate	ent
, igo oroup	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	396,327	194,559	201,769	74,758	21,057	53,701	18.9	10.8	26.6
10-14	6,503	3,224	3,279	2,909	1,269	1,640	44.7	39.4	50.0
15-19	36,835	14,757	22,078	13,080	4,035	9,045	35.5	27.3	41.0
20-24	75,919	29,227	46,692	24,267	7,266	17,001	32.0	24.9	36.4
25-29	73,384	35,064	38,321	13,299	3,426	9,873	18.1	9.8	25.8
30-34	53,572	28,249	25,323	6,911	1,353	5,558	12.9	4.8	21.9
35-39	47,657	23,600	24,057	5,361	1,444	3,917	11.2	6.1	16.3
40-44	30,865	17,112	13,753	2,845	849	1,996	9.2	5.0	14.5
45-49	20,548	12,177	8,371	1,348	61	1,287	6.6	0.5	15.4
50-54	16,898	8,785	8,112	2,253	490	1,763	13.3	5.6	21.7
55-59	13,169	8,180	4,990	682	-	682	5.2	-	13.7
60-64	10,270	6,544	3,726	1,049	366	683	10.2	5.6	18.3
65+	10,708	7,640	3,068	752	497	255	7.0	6.5	8.3

Table 6.1.2	2	Currently Unemployed Pop	ulation of Urban Areas Ag	ed Ten Years and Above
		by Age Group, Sex and Une	employment Rate, AFFAR	REGION: 2011
				فيترج معربها مراجع والل

	Fee	onomically Ac	tivo	Line	mplove	d	Un	Unemploymen Rate Total Male 17.4 8.9 68.5 66.8 28.9 12.2 37.7 29.0 19.7 10.8 9.7 3.2 8.1 1.7 12.0 5.8 22.1 8.6 16.7 13.7		
Age Group		,,,						1		
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	86,602	49,402	37,200	15,066	4,378	10,688	17.4	8.9	28.7	
10-14	337	173	164	231	115	115	68.5	66.8	70.3	
15-19	3,553	1,770	1,782	1,025	217	809	28.9	12.2	45.4	
20-24	10,882	4,118	6,764	4,105	1,194	2,911	37.7	29.0	43.0	
25-29	17,210	8,719	8,491	3,392	942	2,450	19.7	10.8	28.9	
30-34	14,366	9,104	5,262	1,397	293	1,103	9.7	3.2	21.0	
35-39	11,436	7,793	3,643	930	133	797	8.1	1.7	21.9	
40-44	8,820	4,237	4,582	1,059	247	811	12.0	5.8	17.7	
45-49	6,531	4,153	2,378	1,441	355	1,086	22.1	8.6	45.7	
50-54	4,537	1,949	2,588	757	267	490	16.7	13.7	18.9	
55-59	3,024	2,682	342	-	-	-	-	-	-	
60-64	3,515	2,842	674	699	584	115	19.9	20.5	17.1	
65+	2,390	1,860	530	31	31	-	1.3	1.6	-	

Age Group	Ecor	nomically Act	tive	Un	employed		Ur	Unemployme Rate 19.3 12.6 12.0 11.7 30.0 26.4 31.8 23.6 21.5 13.7 14.9 7.6 14.8 9.8 10.1 4.2		
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	1,141,582	578,586	562,996	220,336	73,128	147,208	19.3	12.6	26.1	
10-14	35,807	17,636	18,171	4,297	2,066	2,231	12.0	11.7	12.3	
15-19	121,303	53,615	67,688	36,333	14,151	22,181	30.0	26.4	32.8	
20-24	198,760	84,432	114,328	63,143	19,912	43,230	31.8	23.6	37.8	
25-29	220,876	114,748	106,128	47,422	15,751	31,671	21.5	13.7	29.8	
30-34	128,384	69,794	58,590	19,129	5,274	13,855	14.9	7.6	23.6	
35-39	124,390	58,194	66,197	18,434	5,679	12,755	14.8	9.8	19.3	
40-44	89,893	53,895	35,998	9,062	2,276	6,785	10.1	4.2	18.8	
45-49	68,647	36,647	32,000	6,238	1,375	4,862	9.1	3.8	15.2	
50-54	47,293	28,177	19,116	4,833	1,194	3,639	10.2	4.2	19.0	
55-59	35,138	17,886	17,252	3,898	1,424	2,474	11.1	8.0	14.3	
60-64	28,210	15,060	13,150	2,174	941	1,233	7.7	6.2	9.4	
65+	42,880	28,503	14,377	5,375	3,084	2,292	12.5	10.8	15.9	

Table 6.1.3Currently Unemployed Population of Urban Areas Aged Ten Years and Above
by Age Group, Sex and Unemployment Rate, AMHARA REGION: 2011

Table 6.1.4Currently Unemployed Population of Urban Areas Aged Ten Years and Above
by Age Group, Sex and Unemployment Rate, OROMIA REGION: 2011

Age Group	Ecc	nomically Ac	tive	Ur	nemployed		Une	employm Rate	2	
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	1,874,241	1,014,905	859,336	264,769	84,133	180,636	14.1	8.3	21.0	
10-14	54,693	3 26,387	28,306	3,015	629	2,386	5.5	2.4	8.4	
15-19	181,588	8 80,668	100,921	43,070	15,952	27,118	23.7	19.8	26.9	
20-24	356,623	3 171,418	185,205	87,181	26,166	61,015	24.4	15.3	32.9	
25-29	363,653	3 202,892	160,761	48,246	13,020	35,226	13.3	6.4	21.9	
30-34	261,593	152,695	108,898	34,358	11,707	22,651	13.1	7.7	20.8	
35-39	220,456	5 117,656	102,800	16,427	4,225	12,202	7.5	3.6	11.9	
40-44	129,991	84,279	45,712	7,762	2,366	5,396	6.0	2.8	11.8	
45-49	108,242	63,481	44,761	7,024	2,226	4,798	6.5	3.5	10.7	
50-54	64,510	33,496	31,013	6,190	1,739	4,451	9.6	5.2	14.4	
55-59	51,539	9 29,387	22,152	4,002	1,414	2,588	7.8	4.8	11.7	
60-64	31,710	20,448	11,263	3,233	1,260	1,974	10.2	6.2	17.5	
65+	49,642	32,098	17,544	4,261	3,431	830	8.6	10.7	4.7	

Table 6.1.5Currently Unemployed Population of Urban Areas Aged Ten Years and Above
by Age Group, Sex and Unemployment Rate, SOMALI REGION: 2011

Age Group	Eco	nomically Ac	tive	Un	employed		Un	17.212.213.83.933.128.126.218.119.816.114.18.012.011.017.513.9			
	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	202,113	112,599	89,514	34,678	13,735	20,943	17.2	12.2	23.4		
10-14	8,432	3,642	4,790	1,164	141	1,023	13.8	3.9	21.4		
15-19	15,207	6,477	8,730	5,034	1,822	3,212	33.1	28.1	36.8		
20-24	25,702	11,507	14,195	6,744	2,080	4,664	26.2	18.1	32.9		
25-29	32,593	16,690	15,902	6,469	2,682	3,787	19.8	16.1	23.8		
30-34	31,335	18,072	13,263	4,408	1,443	2,965	14.1	8.0	22.4		
35-39	26,114	13,864	12,250	3,143	1,518	1,625	12.0	11.0	13.3		
40-44	20,803	13,627	7,177	3,647	1,888	1,758	17.5	13.9	24.5		
45-49	12,570	7,761	4,810	1,850	472	1,378	14.7	6.1	28.6		
50-54	11,885	7,289	4,596	1,238	857	382	10.4	11.8	8.3		
55-59	4,865	3,225	1,640	379	229	150	7.8	7.1	9.1		
60-64	3,191	2,593	599	50	50	-	1.6	1.9	-		
65+	9,415	7,852	1,563	553	553	-	5.9	7.0	-		

Age Group	Ecc	pnomically Act	tive	Ur	nemployed		Un	9.3 4.2 10.9 13.2 22.9 12.2 15.5 12.0 8.3 0.5 7.2 5.2 4.4 2.2 2.2 -		
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	58,691	31,818	26,873	5,470	1,326	4,144	9.3	4.2	15.4	
10-14	1,556	763	793	170	100	70	10.9	13.2	8.8	
15-19	5,098	1,954	3,144	1,169	239	930	22.9	12.2	29.6	
20-24	9,698	4,105	5,593	1,502	492	1,010	15.5	12.0	18.1	
25-29	11,562	6,005	5,557	964	27	937	8.3	0.5	16.9	
30-34	8,121	4,332	3,789	581	225	356	7.2	5.2	9.4	
35-39	8,694	5,112	3,583	387	113	274	4.4	2.2	7.7	
40-44	3,863	2,472	1,390	86	-	86	2.2	-	6.2	
45-49	3,314	1,878	1,436	441	108	333	13.3	5.8	23.2	
50-54	2,996	1,959	1,037	27	-	27	0.9	-	2.6	
55-59	1,215	1,003	213	21	-	21	1.7	-	9.8	
60-64	975	733	241	100	-	100	10.3	-	41.6	
65+	1,599	1,502	97	21	21	-	1.3	1.4	-	

Table 6.1.6Currently Unemployed Population of Urban Areas Aged Ten Years and Above by
Age Group, Sex and Unemployment Rate, BENISHANGUL GUMUZ REGION: 2011

 Table 6.1.7
 Currently Unemployed Population of Urban Areas Aged Ten Years and Above by Age Group, Sex and Unemployment Rate, SNNP REGION: 2011

Age Group	Eco	nomically Acti	ve	U	nemployed		Une	11.8 6.7 7.3 7.8 21.7 16.9 19.7 12.5 11.6 5.2 7.9 3.1 7.4 2.3			
	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	771,271	415,270	356,001	90,711	27,660	63,050	11.8	6.7	17.7		
10-14	35,971	21,476	14,495	2,614	1,678	937	7.3	7.8	6.5		
15-19	88,008	40,831	47,177	19,128	6,895	12,233	21.7	16.9	25.9		
20-24	125,087	59,015	66,072	24,587	7,405	17,183	19.7	12.5	26.0		
25-29	144,883	74,630	70,253	16,798	3,894	12,904	11.6	5.2	18.4		
30-34	110,284	62,478	47,806	8,705	1,948	6,757	7.9	3.1	14.1		
35-39	96,584	51,651	44,933	7,167	1,184	5,984	7.4	2.3	13.3		
40-44	58,315	37,028	21,288	2,884	908	1,976	4.9	2.5	9.3		
45-49	41,484	24,193	17,292	3,903	701	3,202	9.4	2.9	18.5		
50-54	28,037	17,040	10,997	1,573	918	655	5.6	5.4	6.0		
55-59	16,813	9,750	7,063	1,331	877	454	7.9	9.0	6.4		
60-64	11,285	7,760	3,525	1,061	297	765	9.4	3.8	21.7		
65+	14,520	9,420	5,100	958	958	-	6.6	10.2	-		

Table 6.1.8	Currently Unemployed Population of Urban Areas Aged Ten Years and Above
	by Age Group, Sex and Unemployment Rate, GAMBELLA REGION: 2011

Age Group	Ecc	onomically Ac	tive	Ur	nemployed	I	Un	7.7 7.6 17.0 26.8 10.8 13.3 12.7 12.6 8.5 11.4 4.3 1.1 1.9 1.3 4.1 5.1 4.4 8.0 5.8 7.7		
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	36,127	17,995	18,132	2,785	1,368	1,417	7.7	7.6	7.8	
10-14	926	228	697	157	61	96	17.0	26.8	13.7	
15-19	3,239	1,238	2,001	349	165	184	10.8	13.3	9.2	
20-24	6,485	3,022	3,463	820	380	440	12.7	12.6	12.7	
25-29	9,208	3,585	5,624	784	408	376	8.5	11.4	6.7	
30-34	5,269	3,084	2,185	226	34	192	4.3	1.1	8.8	
35-39	4,985	3,503	1,482	94	46	48	1.9	1.3	3.2	
40-44	2,335	1,368	968	95	70	26	4.1	5.1	2.6	
45-49	1,831	699	1,132	80	56	25	4.4	8.0	2.2	
50-54	635	474	161	37	37	-	5.8	7.7	-	
55-59	354	237	117	31	-	31	8.7	-	26.3	
60-64	619	319	301	56	56	-	9.1	17.7	-	
65+	239	239	-	55	55	-	23.1	23.1	-	

Age Group	Eco	onomically Ac	tive	U	nemploye	d	Unemployme Rate Total Male 13.6 8.1 5.2 9.4 18.5 18.2 21.7 12.1 13.8 7.5 12.9 6.2 8.0 1.9 10.8 3.8 14.2 10.1 7.0 -		ent
	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	56,074	28,073	28,001	7,624	2,267	5,357	13.6	8.1	19.1
10-14	1,345	747	599	70	70	-	5.2	9.4	-
15-19	4,029	1,906	2,123	745	347	398	18.5	18.2	18.7
20-24	7,724	3,132	4,592	1,673	378	1,295	21.7	12.1	28.2
25-29	11,021	5,540	5,481	1,521	417	1,104	13.8	7.5	20.1
30-34	7,761	3,665	4,096	997	226	771	12.9	6.2	18.8
35-39	7,451	4,238	3,214	595	78	517	8.0	1.9	16.1
40-44	3,929	2,124	1,805	423	80	343	10.8	3.8	19.0
45-49	4,016	2,296	1,719	568	231	337	14.2	10.1	19.6
50-54	3,115	1,311	1,804	217	-	217	7.0	-	12.0
55-59	2,620	1,281	1,340	292	85	207	11.1	6.6	15.5
60-64	1,697	934	763	354	274	80	20.9	29.3	10.5
65+	1,366	899	467	167	81	86	12.2	9.0	18.5

 Table 6.1.9 Currently Unemployed Population of Urban Areas Aged Ten Years and Above by

 Age Group Sex and Unemployment Rate, HARARI REGION: 2011

Table 6.1.10Currently Unemployed Population of Urban Areas Aged Ten Years and Above
by Age Group Sex and Unemployment Rate, ADDIS ABABA CITY
ADMINISTRATION: 2011

Age Group	Econ	omically Ac	tive	L	Inemployed	k	Un	employme Rate	ent
-	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	1,533,291	807,307	725,984	384,317	139,334	244,982	25.1	17.3	33.7
10-14	14,839	3,561	11,278	3,524	1,189	2,335	23.7	33.4	20.7
15-19	118,673	43,870	74,803	34,626	13,491	21,134	29.2	30.8	28.3
20-24	269,660	132,803	136,857	88,017	39,177	48,839	32.6	29.5	35.7
25-29	331,299	168,407	162,892	83,267	26,336	56,931	25.1	15.6	35.0
30-34	213,032	123,953	89,079	47,111	19,058	28,054	22.1	15.4	31.5
35-39	189,427	97,431	91,996	39,242	11,533	27,709	20.7	11.8	30.1
40-44	109,171	65,493	43,678	27,618	7,491	20,128	25.3	11.4	46.1
45-49	95,017	49,242	45,774	20,348	4,500	15,848	21.4	9.1	34.6
50-54	82,390	46,684	35,705	16,663	5,155	11,508	20.2	11.0	32.2
55-59	39,780	23,643	16,137	7,754	2,400	5,353	19.5	10.2	33.2
60-64	31,375	21,992	9,383	7,415	3,123	4,292	23.6	14.2	45.7
65+	38,628	30,228	8,400	8,732	5,881	2,851	22.6	19.5	33.9

Table 6.1.11Currently Unemployed Population of Urban Areas Aged Ten Years and Above
by Age Group Sex and Unemployment Rate, DIRE DAWA ADMINISTRATION:
2011

Age Group				Ur	nemploye	ed	Unemployment Rate			
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	109,048	54,024	55,024	25,024	7,200	17,824	22.9	13.3	32.4	
10-14	2,628	3 1,171	1,457	75	75	-	2.8	6.4	-	
15-19	9,474	3,260	6,214	2,931	368	2,563	30.9	11.3	41.2	
20-24	19,747	7,528	12,219	6,770	1,559	5,211	34.3	20.7	42.6	
25-29	20,707	11,655	9,052	4,610	1,231	3,379	22.3	10.6	37.3	
30-34	11,847	6,307	5,540	2,443	608	1,836	20.6	9.6	33.1	
35-39	14,111	7,074	7,038	2,778	829	1,949	19.7	11.7	27.7	
40-44	9,189	4,946	4,243	1,736	1,005	732	18.9	20.3	17.2	
45-49	8,390	4,818	3,572	1,030	379	651	12.3	7.9	18.2	
50-54	5,256	5 2,626	2,630	1,208	315	893	23.0	12.0	34.0	
55-59	3,584	2,348	1,236	813	366	447	22.7	15.6	36.2	
60-64	1,881	915	966	315	315	-	16.8	34.4	-	
65+	2,234	1,376	858	314	152	162	14.1	11.0	18.9	

Age Group	Eco	nomically Ac	tive	L	Inemploye	d	Unemployment Rate			
o 1	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	2,545,533	1,334,633	1,210,901	580,108	200,434	379,674	22.8	15.0	31.4	
10-14	37,423	13,706	23,717	5,636	2,167	3,469	15.1	15.8	14.6	
15-19	213,828	79,806	134,022	58,646	20,716	37,930	27.4	26.0	28.3	
20-24	469,749	219,905	249,844	148,133	56,911	91,222	31.5	25.9	36.5	
25-29	542,138	280,437	261,701	125,345	39,881	85,464	23.1	14.2	32.7	
30-34	346,704	200,861	145,843	68,467	25,198	43,268	19.7	12.5	29.7	
35-39	303,139	160,405	142,734	53,321	15,009	38,312	17.6	9.4	26.8	
40-44	182,525	109,124	73,401	37,186	10,337	26,849	20.4	9.5	36.6	
45-49	153,658	85,080	68,579	27,105	6,992	20,113	17.6	8.2	29.3	
50-54	121,528	68,042	53,485	22,937	6,957	15,980	18.9	10.2	29.9	
55-59	65,782	38,302	27,480	11,371	3,392	7,980	17.3	8.9	29.0	
60-64	48,794	33,469	15,324	9,910	4,943	4,967	20.3	14.8	32.4	
65+	60,266	45,496	14,771	12,051	7,930	4,120	20.0	17.4	27.9	

Table 6.2Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, MAJOR TOWN TOTAL: 2011

Table 6.2.1Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, MEKELE TOWN: 2011

Age Group	Eco	nomically Ac	tive	U	nemploye	d	Unemployment Rate			
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	109,946	56,478	53,468	22,400	7,085	15,315	20.4	12.5	28.6	
10-14	898	708	190	100	100	-	11.1	14.1	- 20.0	
15-19	10,851	4,045	6,806	2,692	895	1,797	24.8	22.1	26.4	
20-24	21,522	8,970	12,552	7,451	2,892	4,560	34.6	32.2	36.3	
25-29	22,805	11,516	11,289	5,087	1,701	3,386	22.3	14.8	30.0	
30-34	15,293	8,283	7,010	2,416	462	1,954	15.8	5.6	27.9	
35-39	13,334	7,366	5,969	1,619	291	1,328	12.1	4.0	22.3	
40-44	8,814	5,935	2,879	590	304	286	6.7	5.1	9.9	
45-49	5,900	3,595	2,304	361	61	300	6.1	1.7	13.0	
50-54	4,350	1,980	2,370	1,271	90	1,181	29.2	4.5	49.8	
55-59	2,250	1,337	914	183	-	183	8.1	-	20.1	
60-64	1,978	1,311	667	206	123	83	10.4	9.4	12.5	
65+	1,951	1,433	518	422	167	255	21.6	11.7	49.2	

Table 6.2.2Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, ASAYITA TOWN: 2011

Age	Ecor	nomically Ac	tive	U	nemploye	d	Unemployment Rate			
Group	Total	Male	Female	Total	Male	Female	Total	Male	Female	
	7 000	4 0 0 7	0.070	040	400	040		4.0	40.7	
All Ages	7,283	4,007	3,276	810	198	613	11.1	4.9	18.7	
10-14	49	-	49	-	-	-	-	-	-	
15-19	504	165	339	91	19	72	18.1	11.7	21.2	
20-24	946	434	513	203	60	143	21.5	13.8	27.9	
25-29	1,650	827	824	241	55	187	14.6	6.6	22.6	
30-34	1,090	691	400	65	-	65	6.0	-	16.3	
35-39	991	649	342	80	12	68	8.0	1.8	19.8	
40-44	644	331	312	32	-	32	4.9	-	10.1	
45-49	469	242	227	32	11	21	6.8	4.5	9.2	
50-54	292	169	124	36	10	26	12.2	6.0	20.7	
55-59	243	160	82	-	-	-	-	-	-	
60-64	166	124	42	-	-	-	-	-	-	
65+	238	216	22	31	31	-	12.9	14.2	-	

Age Group	Eco	nomically Ac	tive	U	Inemploye	d	Unemployment Rate		
	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	101,170	52,572	48,598	18,426	6,867	11,559	18.2	13.1	23.8
10-14	7,554	3,493	4,061	153	153	-	2.0	4.4	-
15-19	12,608	5,704	6,905	2,651	1,005	1,646	21.0	17.6	23.8
20-24	19,019	8,972	10,047	5,293	1,361	3,931	27.8	15.2	39.1
25-29	18,308	10,077	8,231	4,445	1,694	2,752	24.3	16.8	33.4
30-34	10,276	5,520	4,757	1,697	799	898	16.5	14.5	18.9
35-39	8,823	4,388	4,435	1,308	262	1,045	14.8	6.0	23.6
40-44	6,769	3,796	2,973	1,048	458	590	15.5	12.1	19.9
45-49	6,733	3,878	2,855	595	350	245	8.8	9.0	8.6
50-54	3,554	2,289	1,265	507	384	123	14.3	16.8	9.7
55-59	2,225	770	1,454	172	-	172	7.7	-	11.8
60-64	2,144	1,457	686	157	157	-	7.3	10.8	-
65+	3,156	2,227	929	401	244	157	12.7	11.0	16.9

Table 6.2.3 Currently Unemployed Populations of Major Towns Aged Ten Years and Above by Sex, Age Group and Unemployment Rate, GONDER TOWN: 2011

Table 6.2.4Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, DESSIE TOWN: 2011

Age Group	Eco	nomically Ac	tive	U	Inemploye	d	Unemployment Rate			
· ·	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	56,380	29,297	27,083	10,363	3,311	7,051	18.4	11.3	26.0	
10-14	373	131	242	31	31	-	8.3	23.6	-	
15-19	4,940	1,925	3,015	1,261	239	1,021	25.5	12.4	33.9	
20-24	9,965	4,749	5,216	3,449	1,205	2,244	34.6	25.4	43.0	
25-29	11,126	5,314	5,812	2,450	978	1,472	22.0	18.4	25.3	
30-34	6,816	3,228	3,588	1,115	247	868	16.4	7.6	24.2	
35-39	6,794	3,934	2,859	803	213	590	11.8	5.4	20.6	
40-44	4,643	2,844	1,800	541	107	434	11.6	3.8	24.1	
45-49	3,399	2,149	1,250	96	-	96	2.8	-	7.7	
50-54	3,008	1,900	1,108	239	149	90	7.9	7.8	8.1	
55-59	1,993	999	994	102	-	102	5.1	-	10.3	
60-64	1,628	1,110	518	142	142	-	8.7	12.8	-	
65+	1,695	1,014	681	134	-	134	7.9	-	19.6	

Table 6.2.5Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, BAHIR DAR TOWN: 2011

Age Group	Econ	nomically Ac	tive	U	nemploye	d	Unemployment Rate			
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	116,828	55,550	61,278	26,835	8,876	17,959	23.0	16.0	29.3	
10-14	3,984	1,152	2,832	769	366	403	19.3	31.8	14.2	
15-19	13,085	4,303	8,782	3,347	1,136	2,211	25.6	26.4	25.2	
20-24	28,082	10,934	17,148	10,220	3,464	6,756	36.4	31.7	39.4	
25-29	25,836	12,885	12,951	5,687	2,078	3,609	22.0	16.1	27.9	
30-34	15,742	8,713	7,029	2,702	626	2,077	17.2	7.2	29.5	
35-39	10,200	5,793	4,407	1,130	308	822	11.1	5.3	18.7	
40-44	6,565	4,132	2,433	665	141	524	10.1	3.4	21.5	
45-49	3,986	2,625	1,362	479	59	419	12.0	2.3	30.8	
50-54	3,178	1,236	1,942	387	-	387	12.2	-	19.9	
55-59	2,379	1,038	1,342	638	92	547	26.8	8.9	40.7	
60-64	1,583	1,055	528	365	365	-	23.0	34.6	-	
65+	2,207	1,683	523	446	242	204	20.2	14.4	39.0	

Age Group	Eco	nomically Ac	tive	Unemployed			Unemployment Rate		
· ·	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	50,588	25,583	25,005	9,915	2,978	6,936	19.6	11.6	27.7
10-14	303	209	94	37	37	-	12.2	17.6	-
15-19	3,688	1,328	2,360	1,075	497	579	29.2	37.4	24.5
20-24	9,790	4,351	5,439	2,349	721	1,628	24.0	16.6	29.9
25-29	9,524	4,513	5,011	1,960	521	1,439	20.6	11.5	28.7
30-34	7,710	4,230	3,481	1,257	429	828	16.3	10.1	23.8
35-39	5,541	2,978	2,563	879	75	804	15.9	2.5	31.4
40-44	3,444	1,678	1,766	651	123	528	18.9	7.3	29.9
45-49	3,587	1,880	1,707	492	121	371	13.7	6.4	21.7
50-54	2,933	1,750	1,183	513	126	387	17.5	7.2	32.7
55-59	1,490	822	669	251	83	168	16.8	10.1	25.1
60-64	1,057	860	197	159	120	39	15.1	13.9	20.0
65+	1,521	984	536	292	127	165	19.2	12.9	30.7

Table 6.2.6Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, BISHOFTU TOWN: 2011

Table 6.2.7Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, ADAMA TOWN: 2011

Age Group	Eco	nomically Ac	tive	U	nemploye	d	Unemployment Rate			
° .	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	128,275	68,596	59,679	27,097	7,972	19,125	21.1	11.6	32.0	
10-14	611	270	341	70	-	70	11.4	-	20.4	
15-19	10,749	3,540	7,210	2,341	650	1,691	21.8	18.4	23.5	
20-24	29,509	12,474	17,034	9,588	2,393	7,195	32.5	19.2	42.2	
25-29	27,417	15,437	11,980	5,253	1,629	3,624	19.2	10.6	30.3	
30-34	17,604	11,332	6,272	3,012	929	2,084	17.1	8.2	33.2	
35-39	15,701	8,616	7,085	1,947	627	1,319	12.4	7.3	18.6	
40-44	8,841	5,058	3,783	1,399	-	1,399	15.8	-	37.0	
45-49	5,873	4,277	1,596	823	557	266	14.0	13.0	16.7	
50-54	4,356	2,562	1,795	1,147	360	787	26.3	14.0	43.9	
55-59	3,651	2,304	1,347	622	217	404	17.0	9.4	30.0	
60-64	1,291	826	464	406	213	194	31.5	25.8	41.7	
65+	2,673	1,900	773	488	397	91	18.3	20.9	11.8	

Table 6.2.8Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, JIMMA TOWN: 2011

Age Group	Eco	nomically Ac	tive	U	nemploye	d	Unemployment Rate			
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
	05 00 4	00.045	~~ ~~~	40.074						
All Ages	65,634	36,345	29,289	10,374	3,048	7,326	15.8	8.4	25.0	
10-14	1,319	350	969	360	107	253	27.3	30.7	26.1	
15-19	6,537	2,395	4,143	1,683	484	1,198	25.7	20.2	28.9	
20-24	13,586	6,842	6,744	3,414	1,111	2,304	25.1	16.2	34.2	
25-29	15,533	9,235	6,298	2,338	599	1,739	15.1	6.5	27.6	
30-34	7,775	4,990	2,785	613	246	367	7.9	4.9	13.2	
35-39	6,985	4,041	2,944	323	55	268	4.6	1.4	9.1	
40-44	4,441	2,419	2,022	525	43	482	11.8	1.8	23.8	
45-49	3,002	1,770	1,232	576	165	410	19.2	9.3	33.3	
50-54	2,547	1,541	1,006	186	40	146	7.3	2.6	14.5	
55-59	1,680	1,199	481	-	-	-	-	-	-	
60-64	968	703	265	160	-	160	16.5	-	60.2	
65+	1,260	860	400	197	197	-	15.6	22.9	-	

Age Group	Eco	nomically Ac	tive	U	Inemploye	d	Unemployment Rate			
- ge ereip	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	65,297	35,887	29,410	10,390	3,135	7,255	15.9	8.7	24.7	
10-14	934	731	202	39	39	-	4.2	5.4	-	
15-19	8,225	3,224	5,001	2,416	648	1,768	29.4	20.1	35.3	
20-24	13,326	6,837	6,490	3,120	767	2,353	23.4	11.2	36.3	
25-29	13,396	8,204	5,192	2,015	871	1,144	15.0	10.6	22.0	
30-34	9,101	5,424	3,677	1,147	293	854	12.6	5.4	23.2	
35-39	6,548	3,295	3,253	685	143	541	10.5	4.3	16.6	
40-44	3,618	2,020	1,598	438	87	351	12.1	4.3	21.9	
45-49	3,856	2,181	1,675	237	127	111	6.2	5.8	6.6	
50-54	1,963	1,117	845	165	85	80	8.4	7.6	9.4	
55-59	1,569	969	600	39	-	39	2.5	-	6.5	
60-64	1,217	887	330	30	30	-	2.4	3.3	-	
65+	1,543	997	546	59	44	15	3.8	4.4	2.7	

Table 6.2.9Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, SHASHEMENE TOWN: 2011

Table 6.2.10Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, JIJIGA TOWN: 2011

Age Group	Ecor	nomically Ac	tive	U	nemploye	d	Unemployment Rate			
° .	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	38,304	23,655	14.649	9,216	3.384	5,832	24.1	14.3	39.8	
10-14	492	221	271	89		89	18.1	-	32.8	
15-19	1,993	860	1,133	1,003	350	652	50.3	40.7	57.6	
20-24	5,467	2,372	3,095	1,733	359	1,374	31.7	15.2	44.4	
25-29	9,028	5,171	3,857	2,104	631	1,473	23.3	12.2	38.2	
30-34	6,448	4,574	1,874	1,715	624	1,091	26.6	13.6	58.2	
35-39	4,667	3,072	1,595	909	397	512	19.5	12.9	32.1	
40-44	4,159	3,214	945	552	344	208	13.3	10.7	22.0	
45-49	2,889	2,011	878	402	189	213	13.9	9.4	24.3	
50-54	1,381	845	537	314	244	70	22.7	28.8	13.1	
55-59	723	471	252	237	87	150	32.7	18.4	59.5	
60-64	528	391	137	50	50	-	9.5	12.8	-	
65+	530	456	75	108	108	-	20.3	23.7	-	

Table 6.2.11Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, ASOSA TOWN: 2011

Age Group	Economically Active			Unemployed			Unemployment Rate		
	Total	Male	Female	Total	Male	Female	Total	Male	Female
All Ages	14,757	8,154	6,603	1,438	324	1,115	9.7	4.0	16.9
10-14	119	61	58	-	-	-	-	-	-
15-19	1,048	279	769	210	57	153	20.1	20.5	19.9
20-24	2,855	1,149	1,706	450	109	341	15.8	9.5	20.0
25-29	2,993	1,563	1,430	248	27	221	8.3	1.7	15.5
30-34	2,504	1,462	1,042	243	63	180	9.7	4.3	17.3
35-39	2,135	1,424	712	100	-	100	4.7	-	14.0
40-44	1,007	678	329	26	-	26	2.6	-	7.9
45-49	969	674	295	92	47	46	9.5	6.9	15.5
50-54	438	291	146	27	-	27	6.2	-	18.5
55-59	268	193	75	21	-	21	7.8	-	27.8
60-64	216	175	40	-	-	-	-	-	-
65+	207	207	-	21	21	-	10.1	10.1	-

Age Group	Eco	nomically Ac	tive	U	Inemploye	d	Une	mploymen Rate	t
	Total	Male	Female	Total	Male	Female	Total	Male	Female
	70 500	40 700	27 052	14.017	2 5 2 4	10 100	17.0	0.6	27.7
All Ages	78,586	40,733	37,852	14,017	3,521	10,496	17.8	8.6	
10-14	1,820	859	961	285	-	285	15.7	-	29.7
15-19	6,887	2,792	4,095	1,403	453	950	20.4	16.2	23.2
20-24	15,941	7,080	8,861	3,705	1,037	2,668	23.2	14.6	30.1
25-29	18,143	8,303	9,840	3,611	830	2,781	19.9	10.0	28.3
30-34	11,503	7,046	4,457	1,829	557	1,273	15.9	7.9	28.6
35-39	8,173	4,369	3,804	831	139	692	10.2	3.2	18.2
40-44	6,119	3,794	2,325	882	120	762	14.4	3.2	32.8
45-49	4,813	2,943	1,871	892	139	753	18.5	4.7	40.2
50-54	2,410	1,520	890	59	-	59	2.5	-	6.6
55-59	1,090	620	470	217	63	155	19.9	10.1	32.9
60-64	776	540	236	119	-	119	15.3	-	50.3
65+	910	868	42	184	184	-	20.2	21.2	-

Table 6.2.12 Currently Unemployed Populations of Major Towns Aged Ten Years and Above by Sex, Age Group and Unemployment Rate, HAWASSA TOWN: 2011

Table 6.2.13Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, GAMBELLA TOWN: 2011

Age Group	Eco	Economically Active			Economically Active Unemployed				d	Une	mploymen Rate	t
· ·	Total	Male	Female	Total	Male	Female	Total	Male	Female			
All Ages	14,073	8,372	5,701	1,864	935	929	13.2	11.2	16.3			
-	,	,			935			11.2				
10-14	155	42	113	35	-	35	22.2	-	30.6			
15-19	536	210	326	172	77	96	32.1	36.5	29.3			
20-24	2,608	1,277	1,332	698	319	379	26.8	25.0	28.5			
25-29	3,351	1,792	1,560	505	284	221	15.1	15.8	14.2			
30-34	2,202	1,444	758	103	34	70	4.7	2.3	9.2			
35-39	2,259	1,740	519	94	46	48	4.1	2.6	9.2			
40-44	1,173	663	511	59	33	26	5.0	5.0	5.0			
45-49	760	499	261	80	56	25	10.6	11.2	9.4			
50-54	358	222	136	-	-	-	-	-	-			
55-59	235	149	86	31	-	31	13.1	-	36.0			
60-64	289	189	101	32	32	-	11.0	16.8	-			
65+	147	147	-	55	55	-	37.8	37.8	-			

Table 6.2.14Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, HARAR TOWN: 2011

Age Group	Ecor	nomically Ac	tive	U	Unemployed Rate				t
	Total	Male	Female	Total	Male	Female	Total	Male	Female
AU A	50.074	00.070	00.004	7.004	0.007	F 0.57	40.0	0.4	40.4
All Ages	56,074	28,073	28,001	7,624	2,267	5,357	13.6	8.1	19.1
10-14	1,345	747	599	70	70	-	5.2	9.4	-
15-19	4,029	1,906	2,123	745	347	398	18.5	18.2	18.7
20-24	7,724	3,132	4,592	1,673	378	1,295	21.7	12.1	28.2
25-29	11,021	5,540	5,481	1,521	417	1,104	13.8	7.5	20.1
30-34	7,761	3,665	4,096	997	226	771	12.9	6.2	18.8
35-39	7,451	4,238	3,214	595	78	517	8.0	1.9	16.1
40-44	3,929	2,124	1,805	423	80	343	10.8	3.8	19.0
45-49	4,016	2,296	1,719	568	231	337	14.2	10.1	19.6
50-54	3,115	1,311	1,804	217	-	217	7.0	-	12.0
55-59	2,620	1,281	1,340	292	85	207	11.1	6.6	15.5
60-64	1,697	934	763	354	274	80	20.9	29.3	10.5
65+	1,366	899	467	167	81	86	12.2	9.0	18.5

		NISTRALI	ON: 2011							
Age Group	Ecor	nomically Ac	tive	U	nemploye	d	Unemployment Rate			
	Total	Male	Female	Total	Male	Female	Total	Male	Female	
All Ages	1,533,291	807,307	725,984	384,317	139,334	244,982	25.1	17.3	33.7	
10-14	14,839	3,561	11,278	3,524	1,189	2,335	23.7	33.4	20.7	
15-19	118,673	43,870	74,803	34,626	13,491	21,134	29.2	30.8	28.3	
20-24	269,660	132,803	136,857	88,017	39,177	48,839	32.6	29.5	35.7	
25-29	331,299	168,407	162,892	83,267	26,336	56,931	25.1	15.6	35.0	
30-34	213,032	123,953	89,079	47,111	19,058	28,054	22.1	15.4	31.5	
35-39	189,427	97,431	91,996	39,242	11,533	27,709	20.7	11.8	30.1	
40-44	109,171	65,493	43,678	27,618	7,491	20,128	25.3	11.4	46.1	
45-49	95,017	49,242	45,774	20,348	4,500	15,848	21.4	9.1	34.6	
50-54	82,390	46,684	35,705	16,663	5,155	11,508	20.2	11.0	32.2	
55-59	39,780	23,643	16,137	7,754	2,400	5,353	19.5	10.2	33.2	
60-64	31,375	21,992	9,383	7,415	3,123	4,292	23.6	14.2	45.7	
65+	38,628	30,228	8,400	8,732	5,881	2,851	22.6	19.5	33.9	

Table 6.2.15Currently Unemployed Populations of Major Towns Aged Ten Years and
Above by Sex, Age Group and Unemployment Rate, ADDIS ABABA CITY
ADMINISTRATION: 2011

Table 6.2.16	Currently Unemployed Populations of Major Towns Aged Ten Years and
	Above by Sex, Age Group and Unemployment Rate, DIRE DAWA TOWN:2011

Age Group	Ecor	nomically Ac	tive	U	Inemploye	d	Une	11.3 20.7 10.6 9.6 11.7 20.3 7.9 12.0	ent		
. .	Total	Male	Female	Total	Male	Female	Total	Male	Female		
All Ages	109,048	54,024	55,024	25,024	7,200	17,824	22.9	13.3	32.4		
10-14	2,628	1,171	1,457	75	75	-	2.8	6.4	-		
15-19	9,474	3,260	6,214	2,931	368	2,563	30.9	11.3	41.2		
20-24	19,747	7,528	12,219	6,770	1,559	5,211	34.3	20.7	42.6		
25-29	20,707	11,655	9,052	4,610	1,231	3,379	22.3	10.6	37.3		
30-34	11,847	6,307	5,540	2,443	608	1,836	20.6	9.6	33.1		
35-39	14,111	7,074	7,038	2,778	829	1,949	19.7	11.7	27.7		
40-44	9,189	4,946	4,243	1,736	1,005	732	18.9	20.3	17.2		
45-49	8,390	4,818	3,572	1,030	379	651	12.3	7.9	18.2		
50-54	5,256	2,626	2,630	1,208	315	893	23.0	12.0	34.0		
55-59	3,584	2,348	1,236	813	366	447	22.7	15.6	36.2		
60-64	1,881	915	966	315	315	-	16.8	34.4	-		
65+	2,234	1,376	858	314	152	162	14.1	11.0	18.9		

			All Literate											
Age Group and Sex	All Persons	All Illiterate	Non Formal	Pre- school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate (TVET, 10+1,10+2 and in Old Cur.) Completed	Diploma Completed	Degree and above Completed			
COUNTRY TOTAL										·				
Total														
Total	1,125,536	213,791	10,875	698	362,994	109,199	246,021	41,985	26,323	91,727	21,924			
Male	375,585	32,907	3,568	99	117,846	41,070	97,478	19,128	9,476	39,772	14,242			
Female	749,950	180,883	7,306	598	245,148	68,128	148,543	22,858	16,848	51,955	7,683			
5-9														
Total	-	-	-	-	-	-	-	-	-	-	-			
Male	-	-	-	-	-	-	-	-	-	-	-			
Female 10-14	-	-	-	-	-	-	-	-	-	-	-			
Total	18,226	2,634	-	-	14,977	615	-	-	-	-	-			
Male	7,394	884	-	-	6,182	328	-	-	-	-	-			
Female 15-19	10,832	1,750	-	-	8,795	287	-	-	-	-	-			
Total	157,489	17,048	349	142	57,717	17,372	42,738	6,996	5,398	9,728	-			
Male	57,681	4,444	-	-	20,893	8,766	14,977	2,395	2,252	3,952	-			
Female 20-24	99,808	12,604	349	142	36,824	8,606	27,760	4,601	3,146	5,776	-			
Total	308,809	25,041	69	361	86,936	27,557	83,334	22,057	10,980	39,646	12,827			
Male	106,010	2,527	-	-	23,682	9,889	32,374	11,398	4,233	15,065	6,842			
Female 25-29	202,800	22,515	69	361	63,255	17,669	50,959	10,659	6,746	24,581	5,985			
Total	226,773	42,564	765	-	69,229	23,596	50,907	8,296	6,486	19,126	5,805			
Male	68,133	3,389	51	-	20,740	6,289	19,324	3,598	2,131	8,171	4,439			
Female 30-34	158,640	39,174	715	-	48,489	17,307	31,583	4,698	4,355	10,954	1,365			
Total	126,267	25,607	137	-	45,151	16,060	27,499	2,500	1,099	7,122	1,092			
Male	42,168	4,362	137	-	15,263	6,324	10,816	738	220	3,256	1,052			
Female 35-39	84,099	21,245	-	-	29,889	9,736	16,682	1,762	879	3,866	40			
Total	94,558	18,299	3,135	10	34,527	11,426	18,729	1,134	673	6,334	291			
Male	26,782	1,644	1,816	-	8,879	3,592	7,311	571	271	2,408	291			
Female	67,776	16,655	1,320	10	25,648	7,834	11,419	563	402	3,926	-			

Table 6.3 Currently Unemployed Population of Urban Areas Aged Ten Years and Over by Age Group, Sex, Literacy Status and Educational Attainment, Country Total: 2011

Table	6.3	Cont'd
-------	-----	--------

							All Litera	ate			
Age Group and Sex	All Persons	All Illiterate	Non Formal	Pre-school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate (TVET, 10+1,10+2 and in Old Cur.) Completed	Diploma Completed	Degree and above Completed
40-44			1 onna		10	completed	Completed	rioparatory	Completed	Completed	Completed
Total	57,218	18,286	2,805	-	17,701	4,430	10,498	-	381	2,537	579
Male	17,180	3,399	724	-	5,773	1,729	4,298	-	-	970	287
Female	40,038	14,887	2,081	-	11,929	2,701	6,201	-	381	1,567	292
45-49	,		, -		, -	, -	, -			, -	
Total	44,271	18,298	555	-	13,146	2,684	5,985	287	713	1,957	646
Male	10,464	1,059	-	-	3,200	517	3,369	-	76	1,598	646
Female	33,807	17,238	555	-	9,947	2,167	2,617	287	637	360	-
50-54											
Total	34,995	17,717	1,494	175	9,281	1,789	2,616	715	-	1,127	80
Male	10,971	3,472	136	90	3,210	912	1,868	428	-	775	80
Female 55-59	24,024	14,245	1,359	85	6,071	877	748	287	-	352	-
Total	19,201	10,313	713	-	5,359	1,082	738	-	302	473	221
Male	6,794	2,464	142	-	2,377	954	451	-		186	221
Female	12,407	7,849	571	-	2,982	128	287	-	302	287	-
60-64											
Total	16,508	8,599	104	-	3,081	1,580	810	-	226	2,012	96
Male	7,265	1,725	104	-	2,088	778	523	-	226	1,725	96
Female	9,243	6,873	-	-	993	802	287	-	-	287	-
65+											
Total	21,220	9,385	746	10	5,888	1,007	2,167	-	66	1,665	287
Male	14,743	3,538	459	10	5,559	993	2,167	-	66	1,665	287
Female	6,477	5,847	287	-	329	14	-	-	-	-	-

	ducationa	i Attainn	nent: 20	11							
							All Li	terate			
Region and Sex	All Persons	All Illiterate	Non Formal	Pre- school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate(TVET, 10+1,10+2 and in Old Cur.) Completed	Diploma Completed	Degree and above Completed
COUNTRY TOTAL											
Total	1,125,536	213,791	10,875	698	362,994	109,199	246,021	41,985	26,323	91,727	21,924
Male	375,585	32,907	3,568	99	117,846	41,070	97,478	19,128	9,476	39,772	14,242
Female	749,950	180,883	7,306	598	245,148	68,128	148,543	22,858	16,848	51,955	7,683
TIGRAY REGION											
Total	74,758	12,795	90	90	27,326	7,128	17,096	4,958	815	3,085	1,375
Male	21,057	437	90	90	8,605	2,313	5,085	1,584	715	1,332	806
Female	53,701	12,358	-	-	18,721	4,815	12,011	3,374	100	1,753	570
AFFAR REGION											
Total	15,066	5,690	206	20	5,435	755	1,122	446	133	1,249	10
Male	4,378	703	136	10	1,766	261	242	446	64	750	-
Female	10,688	4,987	70	10	3,669	493	880	-	69	499	10
AMHARA REGION											
Total	220,336	50,311	3,593	-	70,183	15,444	50,019	5,837	5,859	14,038	5,051
Male	73,128	10,083	1,579	-	24,055	4,902	19,630	2,114	1,773	5,128	3,863
Female	147,208	40,228	2,014	-	46,128	10,542	30,389	3,723	4,086	8,911	1,188
OROMIA REGION											
Total	264,769	49,978	1,916	-	93,412	23,762	52,221	7,111	6,644	26,740	2,986
Male	84,133	5,774	555	-	29,364	9,773	18,983	3,628	3,535	11,080	1,441
Female	180,636	44,204	1,361	-	64,047	13,989	33,238	3,483	3,109	15,660	1,545
SOMALI REGION											
Total	34,678	17,168	1,633	142	6,630	2,817	5,082	488	44	310	365
Male	13,735	3,607	1,133	-	3,215	1,748	2,951	488	-	229	365
Female BENISHANGUL GUMUZ REGION	20,943	13,561	501	142	3,414	1,069	2,131	-	44	81	-
Total	5,470	1,025	-	-	2,533	236	1,271	45	141	203	17
Male	1,326	234	-	-	668	-	381	-	-	27	17
Female	4,144	791	-	-	1,865	236	890	45	141	176	-
SNNP REGION											
Total	90,711	18,028	365	-	33,597	9,598	13,865	3,735	2,062	8,881	581
Male	27,660	3,675	-	-	9,415	4,539	3,939	1,699	455	3,674	265
Female	63,050	14,353	365	-	24,182	5,059	9,926	2,036	1,607	5,207	316

 Table 6.4
 Currently Unemployed Population of Urban Areas Aged Ten Years and Over by Region, Sex, Literacy Status and Educational Attainment: 2011

							All Li	terate			
Region and Sex	All Persons	All Illiterate	Non Formal	Pre-school	Grades 1-8	Secondary not completed	Secondary Completed	Preparatory	Certificate(TVET, 10+1,10+2 and in Old Cur.) Completed	Diploma Completed	Degree and above Completed
GAMBELLA REGION											
Total	2,785	573	23	-	1,093	366	453	-	66	178	32
Male	1,368	185	-	-	486	234	365	-	-	65	32
Female	1,417	388	23	-	607	132	88	-	66	113	-
HARARI REGION											
Total	7,624	1,183	41	-	3,019	846	1,556	240	309	388	41
Male	2,267	162	-	-	880	453	473	80	85	134	-
Female ADDIS ABABA CITY ADMINISTRATION	5,357	1,021	41	-	2,139	393	1,083	159	225	254	41
Total	384,317	49,496	2,794	361	112,275	46,739	98,846	18,612	9,656	34,594	10,944
Male	139,334	7,252	-	-	37,707	16,222	43,105	9,019	2,706	16,248	7,074
Female DIRE DAWA ADMINISTRATION	244,982	42,243	2,794	361	74,568	30,517	55,741	9,592	6,950	18,346	3,870
Total	25,024	7,544	213	85	7,491	1,508	4,489	516	595	2,060	522
Male	7,200	795	75	-	1,684	623	2,324	69	144	1,105	380
Female	17,824	6,748	139	85	5,807	885	2,165	446	451	955	143

·						R	egion					
Training status and							Benishangul-				Addis	Dire
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
All Persons		<u> </u>						-				
Total	1,125,536	74,758	15,066	220,336	264,769	34,678	5,470	90,711	2,785	7,624	384,317	25,024
Male	375,585	21,057	4,378	73,128	84,133	13,735	1,326	27,660	1,368	2,267	139,334	7,200
Female	749,950	53,701	10,688	147,208	180,636	20,943	4,144	63,050	1,417	5,357	244,982	17,824
Not Trained												
Total	871,054	62,364	13,332	174,542	212,009	32,723	4,713	73,702	2,485	5,974	270,453	18,756
Male	263,573	14,820	3,243	54,033	61,410	12,132	1,017	21,379	1,215	1,350	88,697	4,278
Female	607,481	47,545	10,089	120,510	150,600	20,590	3,696	52,323	1,270	4,623	181,756	14,479
Trained												
Total	254,194	12,394	1,734	45,793	52,760	1,955	757	17,008	299	1,650	113,576	6,267
Male	112,012	6,238	1,135	19,095	22,723	1,602	309	6,281	152	917	50,638	2,922
Female	142,182	6,156	599	26,698	30,036	353	448	10,727	147	733	62,939	3,345
Not Stated												
Total	287	-	-	-	-	-	-	-	-	-	287	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	287	-	-	-	-	-	-	-	-	-	287	-
Types of Training	9											
Agriculture												
Total	2,874	406	240	671	-	142	-	507	-	-	908	-
Male	2,338	243	240	298	-	142	-	507	-	-	908	-
Female	536	163	-	373	-	-	-	-	-	-	-	-
Water and Soil Cor												
Total	222	-	172	50	-	-	-	-	-	-	-	-
Male	50	-	-	50	-	-	-	-	-	-	-	-
Female	172	-	172	-	-	-	-	-	-	-	-	-
forestry Science												
Total	287	-	-	-	-	-	-	-	-	-	287	-
Male	287	-	-	-	-	-	-	-	-	-	287	-
Female	-	-		-	-	-	-	-	-	-	-	-
General medicine,		gy Patho	logy, Inte	rnal medici	-	ology, Surge	ery, Anesthesiolo	gу				
Total	1,116	-	-	-	475	-	-	-	32	-	609	-
Male	709	-	-	-	355	-	-	-	32	-	322	-
Female	407	-	-	-	120	-	-	-	-	-	287	-
Medical Laboratory												
Total	1,736	-	-	540	500	-	-	84	-	-	612	-
Male	579	-	-	369	210	-	-	-	-	-	-	-
Female	1,157	-	-	170	291	-	-	84	-	-	612	-

Table 6.5 Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Region, Sex, Training Status and Type of Training: COUNTRY TOTAL 2011

Training status and						R	egion					
Types of Training							Benishangul-				Addis	Dire
	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
Pharmacy/Pharmac	y Techniciar	1 IIII										
Total	3,703	-	-	603	958	-	-	535	-	92	1,436	78
Male	2,025	-	-	306	835	-	-	178	-	53	574	78
Female	1,678	-	-	298	122	-	-	357	-	39	862	
Sanitary Science												
Total	310	-	-	-	-	-	-	-	-	-	310	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	310	-	-	-	-	-	-	-	-	-	310	
Clinical Science/Nur	rsing											
Total	14,210	204	-	1,810	6,585	38	-	3,009	-	80	2,045	438
Male	4,257	100	-	792	2,512	-	-	813	-	40	-	
Female	9,953	105	-	1,018	4,072	38	-	2,196	-	40	2,045	43
Public health Nurse,	, Health offic	er/ Sanita	rian/Post	Health Offi	cer							
Total	556	-	-	-	490	-	-	66	-	-	-	
Male	66	-	-	-	-	-	-	66	-	-	-	
Female	490	-	-	-	490	-	-	-	-	-	-	
Personal business r	nanagement											
Total	3,566	-	-	703	948	50	27	133	32	-	1,450	224
Male	1,380	-	-	138	694	50	27	-	32	-	287	15
Female	2,186	-	-	565	254	-	-	133	-	-	1,163	72
Economics												
Total	3,492	153	-	1,026	176	-	-	269	-	-	1,867	
Male	2,749	100	-	805	-	-	-	269	-	-	1,575	
Female	743	53	-	222	176	-	-	-	-	-	292	
Book Keeping/Acco	unting											
Total	16,409	654	19	2,053	4,268	38	37	1,667	33	41	7,384	214
Male	5,566	392	-	594	1,246	38	17	451	33	-	2,654	142
Female	10,843	263	19	1,458	3,022	-	20	1,217	-	41	4,731	73
Secretarial Science	-,			,	-,-			,			, -	
Total	9,437	776	59	1,852	2,440	43	67	479	-	40	3,611	69
Male	61	-	-	-	-	-	-	61	-	-	-,	
Female	9,376	776	59	1,852	2,440	43	67	419	-	40	3,611	69
Banking and Financ			20	.,	_, •		0.				-,	
Total	2,298	187	-	392	-	-	-	152	-	-	1,567	
Male	1,103	-	-	222	-	-	-	-	-	-	881	
Female	1,194	187	-	170	-	-	-	152	-	-	686	

raining status and						R	egion					
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
aw				L								
Total	5,841	89	-	1,263	1,052	-	-	595	-	41	2,731	6
Male	3,413	89	-	976	112	-	-	365	-	-	1,802	6
Female	2,428	-	-	287	941	-	-	230	-	41	929	
Language and Liter		ian and Fo	oreign)		• • •							
Total	1,179	-	-	168	47	-	-	389	-	-	574	
Male	765	-	-	-	47	-	-	144	-	-	574	
Female	413	-	-	168		-	-	245	-	-	-	
Theatrical art	110			100				210				
Total	312	-	-	-	312	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	312	-	-	-	312	-	-	-	-	-	-	
Educational Pedago	ogy, Teaching	g curriculu	m instruc	tion Educat	tional admi	nistration, A	Adults education					
Total	5,759	271	19	1,941	1,127	44	-	347	-	126	1,813	7
Male	2,378	271	19	1,104	517	-	-	63	-	40	293	7
Female	3,381	-	-	837	610	44	-	285	-	86	1,520	
Geography	·											
Total	532	-	-	141	-	-	-	311	-	-	-	6
Male	221	-	-	141	-	-	-	-	-	-	-	8
Female	311	-	-	-	-	-	-	311	-	-	-	
History												
Total	326	-	-	326	-	-	-	-	-	-	-	
Male	326	-	-	326	-	-	-	-	-	-	-	
Female		-	-		-	-	-	-	-	-	-	
Political science												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	_	-	-		-	_	_	_	_	_		
Female	_	-	-		-	_	_	_	_	_		
Sociology and Anth	voology											
Total	894	_	_	_	597	_	_	_	_	_	297	
Male	894	_	_	_	597	_	_	_	_	_	297	
Female				_		_	_	_		-	231	
Psychology	_	_	_	_	_	_	_	_	_	_	_	
Total	732	-	-	76	-	-	-	-	-	-	656	
Male	732	-	-	76	-	-	-	-	-	-	656	
Female	-	-	-	-	-	-	-	-	-	-	-	
Civil Engineering/Bu	uilding											
Total	873	-	-	-	55	179	-	-	-	-	640	
Male	585	-	-	-	55	179	-	-	-	-	352	
Female	287	-	-	-	-	-	-	-	-	-	287	

Training status and						R	egion					
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Mechanical Engine	ering										•	
Total	1,680	-	-	421	638	-	-	-	-	-	621	
Male	577	-	-	72	186	-	-	-	-	-	319	
Female	1,103	-	-	349	452	-	-	-	-	-	302	
Electrical Engineerii	ng											
Total	1,671	-	-	378	120	-	-	-	-	-	1,172	
Male	1,394	-	-	101	120	-	-	-	-	-	1,172	
Female	277	-	-	277	-	-	-	-	-	-	-	
Industrial Engineeri	ng											
Total	676	-	-	154	230	-	-	-	-	-	292	
Male	154	-	-	154	-	-	-	-	-	-	-	
Female	522	-	-	-	230	-	-	-	-	-	292	
Municipality Engine	ering/Archite	cture										
Total	103	-	-	-	103	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	103	-	-	-	103	-	-	-	-	-	-	
Other types of Engi		Land, Bio-	diversity	, Hydraulic	Engineerin	g Irrigation	Engineering					
Total	1,847	-	64	369	450	-	31	357	-	-	574	
Male	1,528	-	64	369	450	-	-	357	-	-	287	
Female	319	-	-	-	-	-	31	-	-	-	287	
Automotive technology												
Total	1,425	-	136	392	297	-	-	-	-	-	600	
Male	1,425	-	136	392	297	-	-	-	-	-	600	
Female	, -	-	-	-	-	-	-	-	-	-	-	
Construction techno	ology											
Total	1,987	422	-	457	706	-	92	-	-	-	310	
Male	881	360	-	159	51	-	-	-	-	-	310	
Female	1,106	61	-	298	655	-	92	-	-	-	-	
Drafting technology	,	-					-					
Total	2,031	-	-	294	232	-	-	-	-	-	1,436	6
Male	304	-	-	72	232	-	-	-	-	-	-	•
Female	1,727	-	-	222		-	-	-	-	-	1,436	6
Manufacturing techno											.,	Ŭ
Total	384	-	-	97	-	-	-	-	-	-	287	
Male	-	-	-	-	-	-	-	-	-	-		
Female	384	-	-	97	-	-	-	-	-	-	287	
Metal technology	001			01							201	
Total	41	-	-	-	-	-	-	-	-	41	-	
Male	41	-	-	-	-	-	-	-	-	41	-	
Female		_	_	_	_	_	_	_	_	וד	_	

Fraining status and						R	egion					
ypes of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Veterinary Science,	Animal husb							•	•	•		
Total	51	-	9	-	-	-	-	42	-	-	-	
Male	51	-	9	-	-	-	-	42	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Communication												
Total	421	90	-	-	44	-	-	-	-	-	287	
Male	44	-	-	-	44	-	-	-	-	-	-	
Female	378	90	-	-	-	-	-	-	-	-	287	
Power, Energy and c	others like In	strumenta		ctrical Powe	er Mechani	cal and Ma	nufacturing Powe	er				
Total	1,062	-	115	204	98	-	-	66	-	-	579	
Male	558	-	-	107	98	-	-	66	-	-	287	
Female	504	-	115	97	-	-	-	-	-	-	292	
Biology												
Total	954	-	-	646	-	-	-	-	-	-	307	
Male	646	-	-	646	-	-	-	-	-	-	-	
Female	307	-	-	-	-	-	-	-	-	-	307	
Chemistry												
Total	162	-	-	-	-	-	-	82	-	-	-	1
Male	162	-	-	-	-	-	-	82	-	-	-	8
Female	-	-	-	-	-	-	-	-	-	-	-	
Geology												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Physics												
Total	231	-	10	72	148	-	-	-	-	-	-	
Male	108	-	10	-	98	-	-	-	-	-	-	
Female	122	-	-	72	50	-	-	-	-	-	-	
Mathematics												
Total	675	100	-	-	288	-	-	-	-	-	287	
Male	388	100	-	-	288	-	-	-	-	-	-	
Female	287	-	-	-	-	-	-	-	-	-	287	
Statistics												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Demography												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	

raining status and						R	egion					
ypes of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Computer Science/Ir	formation S	cience										
Total	9,341	271	68	817	1,764	-	58	322	-	-	5,895	14
Male	1,375	-	-	-	138	-	-	-	-	-	1,162	7
Female	7,966	271	68	817	1,627	-	58	322	-	-	4,733	6
Health and Physical												
Total	2,116	-	-	-	497	49	48	120	-	-	1,260	14
Male	1,590	-	-	-	497	49	-	-	-	-	973	-
Female	527	-	-	-	-	-	48	120	-	-	287	
Meteorological Scien												
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Chemical Engineerin	g											
Total	-	-	-	-	-	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	-	-	-	-	-	-	-	-	-	-	-	
Textile Engineering												
Total	335	-	-	44	-	-	-	-	-	-	291	
Male	291	-	-	-	-	-	-	-	-	-	291	
Female	44	-	-	44	-	-	-	-	-	-		
Purchasing supply												
Total	2,158	-	-	296	960	-	-	318	-	-	584	
Male	758	-	-		143	-	-	318	-	-	297	
Female	1,400	-	-	296	817	-	-	-	-	-	287	
Software Engineering					• • •							
Total	515	138	-	-	49	-	-	-	-	41	287	
Male	349	61	-	-	-	-	_	_	-	-	287	
Female	166	77	-	-	49	-	-	-	-	41		
Home economics	100				10							
Total	3,594	123	23	776	610	-	-	227	-	-	1,694	14
Male	-	-	-	-	-	-	-		-	-		
Female	3,594	123	23	776	610	-	-	227	-	-	1,694	14
Leather handicraft te			_0		010						.,	
Total	1,629	737	-	-	-	-	-	-	-	-	892	
Male	518	216	-	-	-	-	-	-	-	-	302	
Female	1,111	522	-	-	-	-	-	-	-	-	589	
Decorative work trair		522									000	
Total	12,295	1,466	-	1,432	1,756	-	-	904	-	41	6,399	2
Male	1,737	251	-	50	509	-	-		-	-	927	2
Female	10,558	1,215	-	1,382	1,247			904	_	41	5,472	2

Table 6.5 cont'd												
Training status and						R	egion	1	1	1 1		
Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Video photographic	training											
Total	936	-	-	-	574	-	-	-	-	-	292	69
Male	161	-	-	-	91	-	-	-	-	-	-	69
Female	775	-	-	-	483	-	-	-	-	-	292	-
Pottery												
Total	287	-	-	-	-	-	-	-	-	-	287	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	287	-	-	-	-	-	-	-	-	-	287	-
Weavery												
Total	318	-	-	-	-	-	-	-	-	-	318	-
Male	318	-	-	-	-	-	-	-	-	-	318	-
Female	-	-	-	-	-	-	-	-	-	-	-	-
Wood work/carpentr	-y											
Total	4,492	123	-	1,372	180	-	-	128	-	53	2,567	69
Male	4,092	123	-	1,372	136	-	-		-	53	2,279	-
Female	400	-	-	-	44	-	-	-	-	-	287	69
Electrical work											-	
Total	9,263	733	64	2,020	1,844	-	-	282	-	87	3,635	597
Male	6,625	409	64	855	1,213	-	-	139	-	87	3,332	526
Female	2,638	324	-	1,165	631	-	-	144	-	-	303	71
Metal welding and b				,								
Total	3,097	-	-	1,332	812	-	-	-	-	44	909	-
Male	2,452	-	-	737	762	-	-	-	-	44	909	-
Female	645	-	-	595	50	-	-	-	-	-	-	-
Auto Mechanics incl		cycle, Airc	raft main									
Total	4,419	221	172	453	1,075	-	-	200	-	53	2,245	-
Male	3,770	221	172	231	648	-	-	200	-	53	2,245	-
Female	648		-	222	427	-	-		-	-	_,	-
Mason												
Total	4,339	263	-	224	704	-	-	210	-	-	2,939	-
Male	3,960	263	-	54	704	-	-		-	-	2,939	-
Female	380		-	170	-	-	-	210	-	-	_,	-
Typing												
Total	4,356	249	-	464	1,091	-	-	1,028	-	173	1,196	155
Male	442		-	-	55	-	-	59	-	41	287	-
Female	3,914	249	-	464	1,035	-	-	969	-	132	909	155
Tailoring	5,511				.,							
Total	11,806	395	-	1,935	2,123	-	-	58	34	38	6,691	531
Male	2,017	-	-	46	948	-	-	-	-	38	899	85
Female	9,789	395	-	1,889	1,175	-	-	58	34	-	5,792	446
	5,105	000	_	1,003	1,175	_	_	00	54	_	0,102	0

Training status and						R	egion	n				
Types of Training	Tatal	Tierret	۸ 4	A 100 k	0	Com alla	Benishangul-		Comboli	Llerer	Addis	Dire
Driver	Total	Tigray	Afar	Amhara	Oromia	Somalie	Gumuz	SNNPR	Gambela	Harari	Ababa	Dawa
	40,400	4 070	044	0 700	4 007	405	04	707		400	4 407	
Total	13,482	1,676	314	3,782	1,267	465	21	727	-	120	4,487	62
Male	12,524	1,676	305	3,689	1,017	465	21	727	-	120	3,882	62
Female	958	-	9	94	250	-	-	-	-	-	605	
Development Agent		400						040				
Total	364	123	-	-	-	-	-	210	32	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	364	123	-	-	-	-	-	210	32	-	-	
Traditional Birth Atte												
Total	88	-	-	-	88	-	-	-	-	-	-	
Male	-	-	-	-	-	-	-	-	-	-	-	
Female	88	-	-	-	88	-	-	-	-	-	-	
Computer Application	ons and Usa	ge										
Total	11,072	498	109	3,727	2,428	235	19	461	35	192	2,513	8
Male	1,139	-	-	472	278	43	-	59	-	-	287	
Female	9,933	498	109	3,255	2,150	192	19	402	35	192	2,226	8
Lower Scale training	g of Agricultu	ire (from A	garfa, Ar	daeta and I	Baco)							
Total	928	316	-	-	-	-	-	178	47	-	318	(
Male	565	-	-	-	-	-	-	178	-	-	318	(
Female	362	316	-	-	-	-	-	-	47	-	-	
Hotels and Tourism												
Total	7,684	100	13	779	1,191	-	-	214	-	-	5,387	
Male	1,025	-	-	-	451	-	-		-	-	574	
Female	6,658	100	13	779	740	-	-	214	-	-	4,812	
Police	0,000	100	10		1.10						1,012	
Total	1,818	_	-	_	589	-	-	59	-	53	893	22
Male	1,818	_	_	-	589	_	-	59	-	53	893	22
Female	1,010	-	-	-		-	-	- 55			- 095	~ ~ ~
Military	_	-	-	-	-	-	_	-	_	-	-	
Total	2,090		_	656	450	270	170	62	25	92	287	7
Male		-		379	450 450	270	170	62	25 25	92 92	287	-
	1,813	-	-							92	201	
Female	277	-	-	277	-	-	-	-	-	-	-	
Cobblestone						~-						
Total	3,677	668	-	1,645	348	37	-	82	-	81	662	1
Male	1,768	668	-	277	43	37	-	-	-	81	662	
Female	1,909	-	-	1,368	305	-	-	82	-	-	-	1
Embroidery												
Total	2,212	-	-	408	184	37	112	227	-	-	1,244	
Male	647	-	-	-	-	-	-	-	-	-	647	
Female	1,565	-	-	408	184	37	112	227	-	-	597	

Table 6.5 cont'd

Table 6.5 cont'd												
Training status and						R	egion					
Training status and Types of Training	Total	Tigray	Afar	Amhara	Oromia	Somalie	Benishangul- Gumuz	SNNPR	Gambela	Harari	Addis Ababa	Dire Dawa
Plumbery												
Total	1,649	-	-	1,014	-	-	-	268	-	-	298	69
Male	1,058	-	-	423	-	-	-	268	-	-	298	69
Female	591	-	-	591	-	-	-	-	-	-	-	-
Other Senior and J	unior Profes	sions/Spe	cified									
Total	10,640	145	-	1,390	2,469	92	-	109	-	81	6,135	218
Male	7,074	61	-	554	1,770	92	-	-	-	41	4,412	144
Female	3,566	84	-	836	699	-	-	109	-	41	1,723	75
Other Skilled and T	echnical Pro	fessions/S	Specified									
Total	31,420	797	126	4,128	6,322	237	74	1,524	31	40	17,706	434
Male	13,382	634	115	1,688	3,237	237	74	622	31	40	6,490	213
Female	18,038	163	11	2,440	3,086	-	-	902	-	-	11,216	221
Not Stated for type	of training											
Total	-	-	-	-	-	-	-	-	-	-	-	-
Male	-	-	-	-	-	-	-	-	-	-	-	-
Female	-	-	-	-	-	-	-	-	-	-	-	-

	Above by	Age Grou	ip, Sex and			NTRY TOTAL	2011
Age Group				Marital Statu	JS		
and Sex	All Persons	Never Married	Married	Divorced	Widowed	Separated	Live Together
All Ages		Marrioa	Marrioa	Bivereda	maomoa	Copulatoa	regenier
Total	1,125,536	490,088	508,260	48,288	50,438	21,665	6,798
Male	375,585	252,061	106,350	7,538	4,781	3,588	1,268
Female	749,950	238,027	401,910	40,750	45,657	18,077	5,530
10-14	,	,	,			,	-,
Total	18,226	18,191	35	-	-	-	-
Male	7,394	7,394	-	-	-	-	-
Female	10,832	10,798	35	-	-	-	-
15-19							
Total	157,489	125,533	29,115	566	74	608	1,594
Male	57,681	56,663	713	305	-	-	-
Female	99,808	68,869	28,402	261	74	608	1,594
20-24							
Total	308,809	191,956	103,805	6,781	89	3,935	2,243
Male	106,010	97,887	6,970	740	25	-	389
Female	202,800	94,069	96,835	6,041	64	3,935	1,855
25-29							
Total	226,773	95,151	114,720	9,040	2,389	4,493	979
Male	68,133	52,631	13,566	836	-	770	330
Female	158,640	42,520	101,155	8,204	2,389	3,723	649
30-34	400.007	05 70 4	74 504	7 070	0.000	0.405	4 400
Total	126,267	35,794	74,524	7,673	3,293	3,485	1,498
Male	42,168	21,113	18,483	990	347	766	468
Female 35-39	84,099	14,681	56,040	6,683	2,946	2,719	1,030
Total	94,558	15,874	66,161	5,071	4,818	2,430	204
Male	26,782	10,794	14,669	853	4,010	428	204 38
Female	67,776	5,081	51,492	4,218	4,818	2,002	165
40-44	07,770	5,001	51,432	4,210	4,010	2,002	100
Total	57,218	4,109	38,767	6,570	5,847	1,882	43
Male	17,180	2,752	12,639	607	628	511	43
Female	40,038	1,357	26,128	5,963	5,219	1,371	-
45-49	-,	,	-, -	-,	-, -	, -	
Total	44,271	862	28,469	5,090	8,918	933	-
Male	10,464	575	8,185	1,100	547	58	-
Female	33,807	287	20,284	3,990	8,371	876	-
50-54							
Total	34,995	1,723	19,734	2,481	9,036	1,870	152
Male	10,971	1,358	8,179	-	1,100	334	-
Female	24,024	365	11,555	2,481	7,936	1,536	152
55-59							
Total	19,201	-	10,863	1,311	6,099	844	85
Male	6,794	-	5,037	813	298	646	-
Female	12,407	-	5,826	497	5,801	198	85
60-64							
Total	16,508	179	8,886	1,577	5,120	746	-
Male	7,265	179	6,296	475	315	-	-
Female	9,243	-	2,590	1,102	4,805	746	-
65+ Totol	04 000	740	40 404	0 4 0 0	4 750	400	
Total Male	21,220 14,743	716 716	13,181 11,613	2,129 819	4,756 1,521	438 75	-
Female	6,477	110	1,568	1,310	3,236	363	-
Female	0,477	-	1,000	1,310	3,230	303	-

Table 6.6Currently Unemployed Populations of Urban Areas Aged Ten Years and
Above by Age Group, Sex and Marital Status, COUNTRY TOTAL: 2011

by	Region, Se	x and mar					
Region and Sex				Marital Statu	IS		
		Never	Manulad		\\/;-!	O an anata d	Live
	All Persons	Married	Married	Divorced	Widowed	Separated	Together
COUNTRY TOTAL		400.000		40.000	=0.400	04.005	
Total	1,125,536	490,088	508,260	48,288	50,438	21,665	6,798
Male	375,585	252,061	106,350	7,538	4,781	3,588	1,268
Female	749,950	238,027	401,910	40,750	45,657	18,077	5,530
TIGRAY REGION							
Total	74,758	34,357	31,446	4,458	2,606	1,891	-
Male	21,057	15,340	5,103	573	-	41	-
Female	53,701	19,018	26,343	3,885	2,606	1,849	-
AFFAR REGION							
Total	15,066	3,432	8,905	926	1,178	539	86
Male	4,378	2,440	1,682	64	182	10	-
Female	10,688	992	7,223	862	996	529	86
AMHARA REGION							
Total	220,336	89,016	103,954	13,915	8,449	4,544	458
Male	73,128	49,019	20,442	2,415	430	748	74
Female	147,208	39,997	83,512	11,500	8,019	3,796	384
OROMIA REGION							
Total	264,769	105,775	133,795	6,960	11,078	4,916	2,244
Male	84,133	56,078	24,219	1,532	888	714	702
Female	180,636	49,697	109,576	5,428	10,190	4,203	1,542
SOMALI REGION	,	,	,	-,	,	-,	.,
Total	34,678	13,328	18,380	834	1,085	1,052	-
Male	13,735	5,838	7,237	386	37	237	-
Female	20,943	7,490	11,143	448	1,048	815	_
BENISHANGUL	20,010	1,100	11,110	110	1,010	010	
GUMUZ REGION							
Total	5,470	1,791	3,296	147	235	-	-
Male	1,326	840	468	17	-	-	-
Female	4,144	951	2,828	130	235	-	-
SNNP REGION	,		,				
Total	90,711	35,448	47,571	2,048	3,161	1,133	1,350
Male	27,660	19,039	7,084	503	584	115	334
Female	63,050	16,409	40,486	1,545	2,576	1,018	1,016
GAMBELLA	00,000	,	,	.,	_,010	.,	.,
REGION							
Total	2,785	985	1,603	35	108	53	-
Male	1,368	701	544	25	61	37	-
Female	1,417	284	1,059	10	47	17	-
HARARI REGION	-,		.,				
Total	7,624	2,559	4,171	356	428	111	_
Male	2,267	1,351	875	41			_
Female	5,357	1,208	3,296	315	428	111	_
ADDIS ABABA CITY	5,557	1,200	3,230	515	420		_
ADMINISTRATION							
Total	384,317	194,097	144,374	17,353	20,694	6,762	1,036
Male	139,334	97,601	36,160	1,751	2,208	1,614	-,000
Female	244,982	96,496	108,214	15,602	18,486	5,148	1,036
DIRE DAWA ADMINI		55,450	100,214	10,002	10,400	5,140	1,000
Total	25,024	9,299	10,765	1,256	1,416	664	1,623
Male	7,200	9,299 3,815	2,535	230	390	72	1,623
	17,824	5,485	2,535 8,230	1,026	1,026	592	
Female	17,024	0,400	0,230	1,020	1,020	592	1,465

Table 6.7Currently Unemployed Population of Urban Areas Aged Ten Years and Above
by Region, Sex and Marital Status: 2011

			ek WOIK. 201		eps taken to	Seek Work				
Region and Sex	Total	Searching vacancy advertising boards	Through news paper, Radio and TV	Seeking assistance of friends, relatives, etc	Checking at work sites	Trying to establish own enterprise	Direct application to employee	l have unemployment card	Others	Not Stated
COUNTRY TOTAL										
Total	712,681	224,097	35,579	147,247	93,776	160,579	24,260	10,511	16,490	142
Male	282,404	93,036	16,958	59,495	47,627	44,225	12,269	3,229	5,422	142
Female	430,278	131,061	18,621	87,752	46,149	116,354	11,991	7,282	11,068	-
TIGRAY REGION										
Total	46,241	17,032	938	5,048	7,546	13,521	467	1,071	617	-
Male	15,763	6,614	567	1,780	4,202	2,300	217	-	83	-
Female	30,478	10,418	371	3,268	3,343	11,222	250	1,071	534	-
AFFAR REGION										
Total	8,577	1,205	160	2,111	2,785	1,590	602	10	115	-
Male	3,697	983	134	540	1,017	672	235	-	115	-
Female	4,880	221	26	1,571	1,767	918	367	10	-	-
AMHARA REGION										
Total	146,595	60,045	2,757	15,413	14,169	45,910	2,696	2,514	3,090	-
Male	54,791	24,194	1,382	7,456	6,926	13,166	825	443	398	-
Female	91,803	35,850	1,375	7,957	7,243	32,744	1,871	2,070	2,692	-
OROMIA REGION										
Total	160,707	46,737	5,512	39,278	21,319	37,360	4,833	3,294	2,373	-
Male	63,082	19,258	3,017	17,290	10,600	8,612	1,667	1,735	901	-
Female	97,625	27,479	2,495	21,988	10,718	28,748	3,166	1,559	1,472	-
SOMALI REGION										
Total	17,774	2,499	244	6,162	4,236	3,987	304	158	43	142
Male	9,341	1,671	-	2,537	3,475	1,256	261	-	-	142
Female	8,433	828	244	3,625	761	2,731	43	158	43	-
BENISHANGUL GUM	UZ REGION									
Total	2,572	1,125	163	233	229	640	132	26	24	-
Male	919	238	97	181	188	189	-	26	-	-
Female	1,652	886	66	52	41	451	132	-	24	-
SNNP REGION										
Total	55,482	17,329	1,063	8,285	7,173	17,732	1,469	634	1,797	-
Male	20,854	7,173	560	2,327	3,751	5,701	716	-	626	-
Female	34,628	10,156	503	5,957	3,422	12,031	754	634	1,171	-
		-		-	•	-				

Table 6.8Currently Unemployed Populations of Urban Areas Aged Ten Years and Above Who Were Seeking for Work by Region,
Sex and Steps Taken to Seek Work: 2011

				Ste	eps taken to S	Seek Work				
Region	Total	Searching vacancy advertising boards	Through news paper, Radio and TV	Seeking assistance of friends, relatives, etc	Checking at work sites	Trying to establish own enterprise	Direct application to employee	I have unemployment card	Others	Not Stated
GAMBELLA REGION										
Total	1,593	850	59	129	167	236	122	31	-	-
Male	936	539	27	74	46	98	122	31	-	-
Female	656	311	32	55	121	138	-	-	-	-
HARARI REGION										
Total	4,705	602	181	725	1,534	1,272	226	-	164	-
Male	1,635	53	-	325	791	333	93	-	41	-
Female	3,070	549	181	401	744	940	133	-	123	-
ADDIS ABABA CITY A	ADMINISTR/	ATION								
Total	254,420	72,809	24,285	66,500	32,532	34,924	12,904	2,773	7,693	-
Male	105,629	30,681	11,027	25,449	15,509	11,240	7,766	993	2,964	-
Female DIRE DAWA ADMINISTRATION	148,791	42,128	13,258	41,051	17,023	23,684	5,138	1,780	4,729	-
Total	14,017	3,865	217	3,362	2,087	3,405	507	-	573	-
Male	5,756	1,631	148	1,536	1,122	658	368	-	293	-
Female	8,261	2,234	69	1,826	965	2,748	139	-	280	-

Table 6.8 cont'd

Abo	ve by Reg	ion, Sex a	nd Type of jol				
			Type of	job looking f	or		
			Self				
Region and Sex		Any	Employed/ To				
		Available	Establish Own	Private	Government		Not
	Total	work	Business	Employees	Employees	Others	Stated
COUNTRY TOTAL							
Total	1,125,536	600,435	316,622	44,873	150,847	12,572	186
Male	375,585	220,463	78,778	18,491	53,205	4,648	-
Female	749,950	379,972	237,844	26,382	97,642	7,924	186
TIGRAY REGION							
Total	74,758	41,978	22,809	2,047	7,485	439	-
Male	21,057	13,658	3,779	766	2,854	-	-
Female	53,701	28,320	19,030	1,281	4,630	439	-
AFFAR REGION							
Total	15,066	7,498	3,710	761	3,097	-	-
Male	4,378	1,698	910	147	1,624	-	-
Female	10,688	5,801	2,800	614	1,473	-	-
AMHARA REGION		,					
Total	220,336	98,514	75,459	6,772	38,855	736	-
Male	73,128	35,112	19,195	2,364	15,827	629	-
Female	147,208	63,402	56,264	4,407	23,027	107	-
OROMIA REGION	,200	00,102	00,201	1,107	20,021		
Total	264,769	135,590	72,803	11,484	41,445	3,260	186
Male	84,133	47,780	18,586	4,779	12,170	818	100
Female	180,636	87,811	54,218	6,705	29,275	2,442	186
SOMALI REGION	100,030	07,011	54,210	0,705	29,215	2,442	100
Total	34,678	17,471	11,844	394	4,898	70	
	13,735	8,475	2,516	394 193		-	-
Male					2,551		-
Female	20,943	8,996	9,329	201	2,348	70	-
			747	000	070	400	
Total	5,470	3,530	747	220	873	100	-
Male	1,326	860	83	116	267	-	-
Female	4,144	2,669	665	104	606	100	-
SNNP REGION							
Total	90,711	40,131	29,689	2,340	17,256	1,293	-
Male	27,660	13,486	6,783	1,497	4,767	1,127	-
Female	63,050	26,645	22,906	844	12,489	166	-
GAMBELLA REGION							
Total	2,785	1,398	590	53	744	-	-
Male	1,368	690	202	53	424	-	-
Female	1,417	708	389	-	320	-	-
HARARI REGION							
Total	7,624	3,379	2,639	310	1,215	81	-
Male	2,267	1,227	470	207	282	81	-
Female	5,357	2,152	2,169	102	933	-	-
ADDIS ABABA CITY ADMINISTRATION							
Total	384,317	239,420	87,505	19,246	32,164	5,982	-
Male	139,334	93,146	25,296	8,077	11,363	1,452	-
Female	244,982	146,274	62,209	11,169	20,800	4,530	-
DIRE DAWA ADMINIS		, -	, -	,	,	,	
Total	25,024	11,525	8,826	1,246	2,816	611	-
Male	7,200	4,331	961	292	1,075	541	-
Female	17,824	7,194	7,865	955	1,741	69	-
	17,02-	r,10 1	,,000	000	.,	00	

Table 6.9Currently Unemployed Population of Urban Areas Aged ten Years and
Above by Region, Sex and Type of job looking for: 2011

Unemployed Duration of Unemployment (in Months) Age Group and Sex Total Unemployed Population Unemployed With Work Experience Not Experience Instruct Stated 1 6 7 12 13-24 25-36 37-48 49-60 61-72 73-84 85-95 96+ Not stated All Ages 1 1.25,536 582,277 543,073 186 35,465 405,695 246,242 141,018 64,175 30,539 23,698 14,229 6,438 697 155,681 1,420 Male 375,585 216,446 159,139 - 1,125 30,609 34,75 24,342 20,052 18,089 11,053 49,22 610 120,338 1313 10-14 1 18,226 3,833,34 14,390 - 2,129 6,113 3,104 4,007 - - - 2,434 - Total 18,734 23,323 5,062 - 1,254 3,153 2,845 1,809 515 566				pioyment - C	ountry		011		~								
Age Group and Sex Total Unemployed Population With Work Experience Not Stated < I <				yed					Dura	ition of U	nemployr	nent (In I	vionths)				
and Unemployed Population Work Experience With Work Experience Not Stated < 1 6 7.12 13.24 25.36 37.48 49-60 61.72 73.48 85-95 96.4 Not stated All Ages 701al 1,125,536 582,277 543.073 186 35,465 405,895 246,242 141.018 64,175 30,539 23,698 14,269 64.38 697 155,681 1,420 Male 375,585 216,446 155,139 186 23,556 25,197 158,405 94,75 24,322 20,032 18,089 11,053 4,922 610 12,083 1,313 10-14 18,226 3,836 14,390 - 2,129 6,113 3,104 4,007 - 439 - - 1,114 - Total 18,226 3,632 5,662 - 2,243 3,224 4,99 - - 1,1,14 - 15.19 Total 157,681																	
Population Experience Experience Stated <1 1-6 7.12 13-24 25-36 37-48 49-60 61-72 73-84 85-95 96+ stated All Ages Total 1,125,535 582,277 543,073 186 35,685 246,222 141,018 64,175 30.539 23,698 14,269 6,438 697 155,681 1,420 Male 375,585 216,446 159,139 -1 17,509 87,837 42,432 20,032 18,069 11,053 4,922 610 120,838 1,313 10-14 18,226 3,836 14,390 - 2,129 6,113 3,104 4,007 -439 - - - 1,114 - 1519 Total 10,822 3,052 5,052 -1,254 3,153 885 783 - - - 1,114 - 16,475 - 1,114 - 1,114 - 1,114 - 1,114 - <td></td> <td></td> <td></td> <td></td> <td>Net</td> <td></td> <td>Net</td>					Net												Net
All Ages Total 1,125,536 582,277 543,073 186 35,465 405,895 246,242 141,018 64,175 30,539 23,698 14,269 6,438 697 155,681 1,420 Male 375,585 216,446 159,139 - 1,879 150,699 87,837 47,7543 21,744 10,503 4,922 610 120,838 1,313 10-14 18,226 3,836 14,390 - 2,129 6,113 3,104 4,007 - 439 - - - 1,1319 - Female 10,832 1,505 9,328 875 2,960 2,219 3,224 - 439 - - - 1,114 - 157 666 119 - 16,202 - 439 - - - 1,114 - - 1,124 3,243 3,241 2,213 14,067 7,951 1,780 388 238 137 - 6,4	and Sex		-			. 1	1.0	7 10	10.04	25.26	27 40	40.60	61 70	72.04	0E 0E	061	
Total 1,125,536 562,277 543,073 186 35,465 405,895 246,242 141,018 64,175 30,539 23,698 14,269 6,438 697 155,681 1,420 Male 375,585 216,446 159,139 - 11,879 150,699 87,837 47,543 21,744 10,507 5,609 3,216 1,516 87 34,843 107 Fernale 749,950 365,81 14,390 - 2,129 6,113 3,104 4,007 - 439 - - - 1,319 Total 18,226 3,836 14,390 - 2,129 3,213 885 7783 - - - - 1,114 - Total 157,489 60,589 96,900 - 9,855 60,373 41,227 7,800 1,809 515 566 119 - 16,920 - - - 6,475 - Total 157,489		Population	Experience	Experience	Stated	< 1	1-0	7-12	13-24	20-30	37-40	49-00	01-72	13-04	00-90	90+	stated
Male 375,585 216,446 159,139 - 11,879 150,699 87,837 47,543 21,744 10.507 5,609 3,216 1,516 87 34,843 107 Female 749,950 365,831 383,934 16 23,586 255,197 158,405 93,475 42,432 20,032 18,089 11,053 4,922 610 120,838 1,313 10-14 18,226 3,836 14,390 - 2,129 6,113 3,104 4,007 - 439 - - - 1,319 Female 10,832 1,505 9,328 - 875 2,960 2,219 3,224 - 439 - - 1,114 - 15.19 Total 157,489 60,589 96,900 - 9,855 60,373 41,237 18,264 7,830 1,809 515 566 119 - 16,920 - Total 308,089 126,854 181	-		500 077	5 40 0 7 0	100	05 405					~~ ~~~	~~ ~~~			~~~		
Female 749,950 365,831 383,934 186 23,586 255,197 158,405 93,475 42,432 20,032 18,089 11,053 4,922 610 120,838 1,313 10-14 Total 18,226 3,836 14,390 - 2,129 6,113 3,104 4,007 - 439 - - - 2,2434 - 1,319 - - - 2,2434 - 1,319 - 1,319 - - - 1,114 - 1,519 - - - 1,114 - 1,519 - - - 1,114 - 1,620 - 1,114 - 1,627 - - - 6,757 - 6,475 - 6,473 3,7160 27,150 1,313 6,500 1,421 277 429 119 - 16,420 - 6,475 - 6,475 - 6,475 - 1,740 8,860 4,760				,		,	,	- ,	,	,	,	,	,	-,		,	, -
10-14 Total 18,226 3,836 14,390 - 2,129 6,113 3,104 4,007 - 439 - - - - 2,234 - 1,319 - Female 10,832 1,505 9,328 - 875 2,960 2,219 3,224 - 439 - - - 1,114 - 15:19 Total 157,489 60,589 96,900 - 9,855 60,373 41,237 18,264 7,830 1,809 515 566 119 - 16,920 - Male 57,681 25,198 32,483 - 3,413 23,213 14,067 7,951 1,780 388 238 137 - - 6,475 - 20-24 - - - 6,443 37,160 27,150 10,313 6,050 1,421 277 429 119 - 10,445 - 20-24 - - - 6,473 31,407 55,431 31,410 55 24,04		•				,			,	,		-,		,	-		-
Total 18,226 3,836 14,390 - 2,129 6,113 3,104 4,007 - 439 - 1,319 - - - 1,319 - - 1,319 - - - 1,319 - 1,319 - 1,319 - 1,114 - 1,319 - 1,114 - 1,114 - 1,114 - 1,519 3,243 3,413 23,213 14,067 7,951 1,780 388 238 137 - 6,6475 - 0,445 - 0,445 - 0,445 - 0,445 - 0,445 - 0,445 - 0,445 - 0,445 - 0,445 - 0,445 - 0,445 - 0,445 0,445 0,445 0,445 0,445 0,445 0,445 0,445 0,445 0,445 0,414 <td></td> <td>749,950</td> <td>365,831</td> <td>383,934</td> <td>186</td> <td>23,586</td> <td>255,197</td> <td>158,405</td> <td>93,475</td> <td>42,432</td> <td>20,032</td> <td>18,089</td> <td>11,053</td> <td>4,922</td> <td>610</td> <td>120,838</td> <td>1,313</td>		749,950	365,831	383,934	186	23,586	255,197	158,405	93,475	42,432	20,032	18,089	11,053	4,922	610	120,838	1,313
Male 7,394 2,332 5,062 1,254 3,153 885 783 - - - - - - - - - 1,319 - Female 10,832 1,505 9,328 - 875 2,960 2,219 3,224 - 439 - - - 1,114 - 15-19 10,832 15,7489 60,589 96,900 - 9,855 60,373 41,237 18,264 7,830 1,809 515 566 119 - 16,920 - Male 57,681 25,198 32,483 - 3,413 23,213 14,067 7,951 1,780 388 238 137 - 6,475 - 20-24 - - 1117,393 - 6,443 37,160 27,150 10,313 6,050 1,421 277 429 119 - 10,445 - 20-24 - - 117,393 - 6,848 15,820 23,681 10,278 5,895 2,970 1,4																	
Female 15-19 10,832 1,505 9,328 - 875 2,960 2,219 3,224 - 439 - - - 1,114 - Total Male 57,681 25,198 32,483 - 3,413 23,213 14,087 7,951 1,780 388 238 137 - 6,475 - 6,475 - 6,475 - 6,475 - 6,475 - 6,475 - 6,475 - 6,475 - 6,475 - 6,475 - 6,475 - 6,475 - 6,481 15,520 80,840 38,202 17,947 8,980 4,790 1,757 1,873 55 31,140 558 Male 106,010 41,447 64,562 - 2,765 41,079 25,439 14,521 7,669 3,086 1,820 319 1,071 - 8,134 107 Female 202,800 85,407 117,393 - 5,943		•			-					-	439	-	-	-	-		-
15-19 Total 157,489 60,589 96,900 - 9,855 60,373 41,237 18,264 7,809 515 566 119 - 16,920 - Male 57,681 25,198 32,483 32,413 23,213 14,087 7,951 1,780 388 238 137 - 6,6475 - 6,6475 - 6,475 - 6,6475 - 10,445 - 10,445 - 10,445 - 10,445 - 10,445 - 10,445 - 10,445 - 8,400 11,502 2,756 41,079 25,439 14,521 7,669 3,086 1,820 319 1,071 - 8,134 107 Female 202,800 85,407 117,393 - 4,083 74,741 55,402 23,681 10,278 5,895 2,970 1,437 802 55 29,928 107 Male 66,133 39,031 29,101 - 1,024 28,764 18,391 6,171 4,260 2,125 1,347 92 <t< td=""><td></td><td>•</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>-</td><td>-</td><td>-</td><td>-</td><td>-</td><td>-</td><td></td><td>-</td></t<>		•								-	-	-	-	-	-		-
Total 157,489 60,589 96,900 - 9,855 60,373 41,237 18,264 7,830 1,809 515 566 119 - 16,920 - Male 57,681 25,198 32,483 - 3,413 23,213 14,087 7,951 1,780 388 238 137 - - 6,475 - - 6,475 - - 6,475 - - 6,475 - - 6,475 - - 6,476 - - 6,476 - - 6,475 - - 6,443 37,100 27,150 10,313 6,050 1,421 27,67 1,014 - 6,473 10,044 - 6,348 116,920 1,071 - 8,134 107 - 8,134 107 - 8,134 107 - 8,143 107 - 8,143 107 - 8,133 39,031 29,101 - 1,024 28,674 18,391 6,171 4,262 2,125 1,343 1,130 55 24,348 1		10,832	1,505	9,328	-	875	2,960	2,219	3,224	-	439	-	-	-	-	1,114	-
Male 57,681 25,198 32,483 - 3,413 23,213 14,087 7,951 1,780 388 238 137 - - 6,475 - Female 99,808 35,392 64,416 - 6,443 37,160 27,150 10,313 6,050 1,421 277 429 119 - 10,445 - 20-24 Total 308,809 126,854 181,955 - 6,848 115,820 80,840 38,202 17,947 8,980 4,790 1,757 1,873 55 31,140 558 Male 106,010 41,447 64,562 - 2,765 41,079 25,439 14,521 7,669 3,086 1,820 319 1,071 - 8,134 107 Female 202,800 85,407 117,393 - 4,083 74,741 55,402 23,681 10,278 5,895 2,970 1,432 52 23,005 451 25-29 Total 226,773 121,609 105,164 - 5,943 82,522																	
Female 99,808 35,392 64,416 - 6,443 37,160 27,150 10,313 6,050 1,421 277 429 119 - 10,445 - Total 308,809 126,854 181,955 - 6,848 115,820 80,840 38,202 17,947 8,980 4,790 1,757 1,873 55 31,140 558 Male 106,010 41,447 64,6562 - 2,765 41,079 25,439 14,521 7,669 3,086 1,820 319 1,071 5 81,14 107 Female 202,800 85,407 117,393 - 4,083 74,741 55,402 23,681 10,278 5,895 2,970 1,437 802 55 23,005 451 25-29 Total 226,773 121,609 105,164 - 5,943 82,522 48,342 30,711 3,420 2,417 1,437 52 29,928 107 30-34 </td <td></td> <td></td> <td></td> <td></td> <td>-</td> <td></td> <td></td> <td></td> <td>•</td> <td></td> <td></td> <td></td> <td></td> <td>119</td> <td>-</td> <td></td> <td>-</td>					-				•					119	-		-
20-24 Total 308,809 126,854 181,955 - 6,848 115,820 80,840 38,202 17,947 8,980 4,790 1,757 1,873 55 31,140 558 Male 106,010 41,447 64,562 - 2,765 41,079 25,439 14,521 7,669 3,086 1,820 319 1,071 - 8,134 107 Female 202,800 85,407 117,393 - 4,083 74,741 55,402 23,681 10,278 5,895 2,970 1,437 802 55 23,005 451 25-29 Total 226,773 121,609 105,164 - 5,943 82,522 48,342 30,741 13,492 4,941 6,740 2,479 1,482 55 29,928 107 Male 68,133 39,031 29,101 - 1,024 28,764 18,391 6,171 4,260 2,125 1,347 92 379 - 5,580 - S0-34 Total 126,267 75,350 50,917 <t< td=""><td></td><td>•</td><td></td><td></td><td>-</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td>-</td><td></td><td>-</td></t<>		•			-										-		-
Total 308,809 126,854 181,955 - 6,848 115,820 80,840 38,202 17,947 8,980 4,790 1,757 1,873 55 31,140 558 Male 106,010 41,447 64,562 - 2,765 41,079 25,439 14,521 7,669 3,086 1,820 319 1,071 - 8,134 107 Female 202,800 85,407 117,393 - 4,083 74,741 55,402 23,681 10,278 5,895 2,970 1,437 802 55 23,005 451 25-29 70tal 226,773 121,609 105,164 - 5,943 82,522 48,342 30,741 13,492 4,941 6,740 2,479 1,482 55 29,928 107 Male 68,133 39,031 29,101 - 1,024 28,764 18,391 6,171 4,260 2,125 1,347 92 379 - 5,580 - 5,303 2,388 1,103 55 24,348 107 30-34<		99,808	35,392	64,416	-	6,443	37,160	27,150	10,313	6,050	1,421	277	429	119	-	10,445	-
Male 106,010 41,447 64,562 - 2,765 41,079 25,439 14,521 7,669 3,086 1,820 319 1,071 - 8,134 107 Female 202,800 85,407 117,393 - 4,083 74,741 55,402 23,681 10,278 5,895 2,970 1,437 802 55 23,005 451 25-29 Total 226,773 121,609 105,164 - 5,943 82,522 48,342 30,741 13,492 4,941 6,740 2,479 1,482 55 29,928 107 Male 68,133 39,031 29,101 - 1,024 28,764 18,391 6,171 4,260 2,125 1,347 92 379 - 5,580 - Female 158,640 82,578 76,062 - 4,919 53,758 29,920 24,571 9,232 2,815 5,393 2,388 1,103 55 24,348 107 30-34 Total 126,267 75,350 50,917 - 3,337																	
Female 202,800 85,407 117,393 - 4,083 74,741 55,402 23,681 10,278 5,895 2,970 1,437 802 55 23,005 451 25-29 Total 226,773 121,609 105,164 - 5,943 82,522 48,342 30,741 13,492 4,941 6,740 2,479 1,482 55 29,928 107 Male 68,133 39,031 29,101 - 1,024 28,764 18,391 6,171 4,260 2,125 1,347 92 379 - 5,580 - Female 158,640 82,578 76,062 - 4,919 53,758 29,950 24,571 9,232 2,815 5,393 2,388 1,103 55 24,348 107 30-34 Total 126,267 75,350 50,917 - 3,337 43,552 27,028 14,671 8,678 4,291 1,988 1,943 1,179 - 19,6				,	-					•					55		
25-29 Total 226,773 121,609 105,164 - 5,943 82,522 48,342 30,741 13,492 4,941 6,740 2,479 1,482 55 29,928 107 Male 68,133 39,031 29,101 - 1,024 28,764 18,391 6,171 4,260 2,125 1,347 92 379 - 5,580 - Female 158,640 82,578 76,062 - 4,919 53,758 29,950 24,571 9,232 2,815 5,393 2,388 1,103 55 24,348 107 30-34 - - 5,50,917 - 3,337 43,552 27,028 14,671 8,678 4,291 1,988 1,943 1,179 - 19,600 - Male 42,168 31,664 10,504 - 1,433 18,518 10,086 4,974 2,266 1,255 - 380 - - 3,258 - Female 84,099 43,686 40,413 - 1,904 25,034					-		•								-		
Total 226,773 121,609 105,164 - 5,943 82,522 48,342 30,741 13,492 4,941 6,740 2,479 1,482 55 29,928 107 Male 68,133 39,031 29,101 - 1,024 28,764 18,391 6,171 4,260 2,125 1,347 92 379 - 5,580 - Female 158,640 82,578 76,062 - 4,919 53,758 29,950 24,571 9,232 2,815 5,393 2,388 1,103 55 24,348 107 30-34 - - 3,337 43,552 27,028 14,671 8,678 4,291 1,988 1,943 1,179 - 19,600 - Male 42,168 31,664 10,504 - 1,433 18,518 10,086 4,974 2,266 1,255 - 380 - - 3,258 - Female 84,099 43,686 40,413 - 1,904 25,034 16,943 9,697 6,412 3,036<		202,800	85,407	117,393	-	4,083	74,741	55,402	23,681	10,278	5,895	2,970	1,437	802	55	23,005	451
Male 68,133 39,031 29,101 - 1,024 28,764 18,391 6,171 4,260 2,125 1,347 92 379 - 5,580 - Female 158,640 82,578 76,062 - 4,919 53,758 29,950 24,571 9,232 2,815 5,393 2,388 1,103 55 24,348 107 30-34 70,03 75,350 50,917 - 3,337 43,552 27,028 14,671 8,678 4,291 1,988 1,943 1,179 - 19,600 - Male 42,168 31,664 10,504 - 1,433 18,518 10,086 4,974 2,266 1,255 - 380 - - 3,258 - Female 84,099 43,686 40,413 - 1,904 25,034 16,943 9,697 6,412 3,036 1,988 1,563 1,179 - 16,342 - 35-39 7 5 16,323 11,922 4,460 3,458 1,325 2,651																	
Female 158,640 82,578 76,062 - 4,919 53,758 29,950 24,571 9,232 2,815 5,393 2,388 1,103 55 24,348 107 30-34 10 126,267 75,350 50,917 - 3,337 43,552 27,028 14,671 8,678 4,291 1,988 1,943 1,179 - 19,600 - Male 42,168 31,664 10,504 - 1,433 18,518 10,086 4,974 2,266 1,255 - 380 - - 3,258 - Female 84,099 43,686 40,413 - 1,904 25,034 16,943 9,697 6,412 3,036 1,988 1,563 1,179 - 16,342 - Sold 26,782 19,703 7,079 - 309 11,895 6,400 3,946 1,395 1,040 344 - - 87 1,365 - Female 67,776 37,965 29,625 186 2,321 25,860 9,924		•			-										55		107
30-34 Total 126,267 75,350 50,917 - 3,337 43,552 27,028 14,671 8,678 4,291 1,988 1,943 1,179 - 19,600 - Male 42,168 31,664 10,504 - 1,433 18,518 10,086 4,974 2,266 1,255 - 380 - - 3,258 - Female 84,099 43,686 40,413 - 1,904 25,034 16,943 9,697 6,412 3,036 1,988 1,563 1,179 - 16,342 - 35-39 - - 36,704 186 2,630 37,755 16,323 11,922 4,460 3,458 1,325 2,651 661 514 12,481 378 Male 26,782 19,703 7,079 - 309 11,895 6,400 3,946 1,395 1,040 344 - - 87 1,365 - Female 67,776 37,965 29,625 186 2,321 25,860 9,924 7,					-										-		-
Total126,26775,35050,917-3,33743,55227,02814,6718,6784,2911,9881,9431,179-19,600-Male42,16831,66410,504-1,43318,51810,0864,9742,2661,255-3803,258-Female84,09943,68640,413-1,90425,03416,9439,6976,4123,0361,9881,5631,179-16,342-35-3936,7041862,63037,75516,32311,9224,4603,4581,3252,65166151412,481378Male26,78219,7037,079-30911,8956,4003,9461,3951,040344871,365-Female67,77637,96529,6251862,32125,8609,9247,9763,0652,4189802,65166142711,11637840-4481416,4159,9036,9522,8422,1321,5969434677315,080-Male17,18013,1404,040-3357,2994,0881,9845889312873641,304-		158,640	82,578	76,062	-	4,919	53,758	29,950	24,571	9,232	2,815	5,393	2,388	1,103	55	24,348	107
Male 42,168 31,664 10,504 - 1,433 18,518 10,086 4,974 2,266 1,255 - 380 - - 3,258 - Female 84,099 43,686 40,413 - 1,904 25,034 16,943 9,697 6,412 3,036 1,988 1,563 1,179 - 16,342 - 35-39 Total 94,558 57,667 36,704 186 2,630 37,755 16,323 11,922 4,460 3,458 1,325 2,651 661 514 12,481 378 Male 26,782 19,703 7,079 - 309 11,895 6,400 3,946 1,395 1,040 344 - - 87 1,365 - Female 67,776 37,965 29,625 186 2,321 25,860 9,924 7,976 3,065 2,418 980 2,651 661 427 11,116 378 40-44 Total 57,218 35,362 21,856 814 16,415 9,903																	
Female 35-3984,09943,68640,413-1,90425,03416,9439,6976,4123,0361,9881,5631,179-16,342-Total94,55857,66736,7041862,63037,75516,32311,9224,4603,4581,3252,65166151412,481378Male26,78219,7037,079-30911,8956,4003,9461,3951,040344871,365-Female 40-4467,77637,96529,6251862,32125,8609,9247,9763,0652,4189802,65166142711,116378Total Male57,21835,36221,856-81416,4159,9036,9522,8422,1321,5969434677315,080-Male17,18013,1404,040-3357,2994,0881,9845889312873641,304-				,	-							1,988		1,179	-		-
35-39 Total 94,558 57,667 36,704 186 2,630 37,755 16,323 11,922 4,460 3,458 1,325 2,651 661 514 12,481 378 Male 26,782 19,703 7,079 - 309 11,895 6,400 3,946 1,395 1,040 344 - - 87 1,365 - Female 67,776 37,965 29,625 186 2,321 25,860 9,924 7,976 3,065 2,418 980 2,651 661 427 11,116 378 40-44 - - 814 16,415 9,903 6,952 2,842 2,132 1,596 943 467 73 15,080 - Male 17,180 13,140 4,040 - 335 7,299 4,088 1,984 588 931 287 364 - 1,304 - 1,304 - 1,304 - 1,304 - 1,304 - 1,304 - 1,304 - 1,304 -					-				,			-		-	-		-
Total94,55857,66736,7041862,63037,75516,32311,9224,4603,4581,3252,65166151412,481378Male26,78219,7037,079-30911,8956,4003,9461,3951,040344871,365-Female67,77637,96529,6251862,32125,8609,9247,9763,0652,4189802,65166142711,11637840-44770135,36221,856-81416,4159,9036,9522,8422,1321,5969434677315,080-Male17,18013,1404,040-3357,2994,0881,9845889312873641,304-		84,099	43,686	40,413	-	1,904	25,034	16,943	9,697	6,412	3,036	1,988	1,563	1,179	-	16,342	-
Male 26,782 19,703 7,079 - 309 11,895 6,400 3,946 1,395 1,040 344 - - 87 1,365 - Female 67,776 37,965 29,625 186 2,321 25,860 9,924 7,976 3,065 2,418 980 2,651 661 427 11,116 378 40-44 Total 57,218 35,362 21,856 - 814 16,415 9,903 6,952 2,842 2,132 1,596 943 467 73 15,080 - Male 17,180 13,140 4,040 - 335 7,299 4,088 1,984 588 931 287 364 - - 1,304 -																	
Female 67,776 37,965 29,625 186 2,321 25,860 9,924 7,976 3,065 2,418 980 2,651 661 427 11,116 378 40-44 Total 57,218 35,362 21,856 - 814 16,415 9,903 6,952 2,842 2,132 1,596 943 467 73 15,080 - Male 17,180 13,140 4,040 - 335 7,299 4,088 1,984 588 931 287 364 - 1,304 -				,	186					,			2,651	661	-		378
40-44 Total 57,218 35,362 21,856 - 814 16,415 9,903 6,952 2,842 2,132 1,596 943 467 73 15,080 - Male 17,180 13,140 4,040 - 335 7,299 4,088 1,984 588 931 287 364 1,304 -					-								-	-	-		-
Total 57,218 35,362 21,856 - 814 16,415 9,903 6,952 2,842 2,132 1,596 943 467 73 15,080 - Male 17,180 13,140 4,040 - 335 7,299 4,088 1,984 588 931 287 364 - - 1,304 -		67,776	37,965	29,625	186	2,321	25,860	9,924	7,976	3,065	2,418	980	2,651	661	427	11,116	378
Male 17,180 13,140 4,040 - 335 7,299 4,088 1,984 588 931 287 364 1,304 -																	
				,	-	-						,		467	73		-
					-		•					-			-		-
Female 40,038 22,222 17,816 - 479 9,116 5,816 4,968 2,253 1,202 1,309 579 467 73 13,776 -	Female	40,038	22,222	17,816	-	479	9,116	5,816	4,968	2,253	1,202	1,309	579	467	73	13,776	-

Table 6.10 Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Age Group, Sex, Previous Work Experience and Duration of Unemployment - Country Total: 2011

Table 6.10 Cont'd

		Unemplo	yed					Durat	ion of Ur	nemployr	nent(In I	Months)				
Age Group and Sex	Total Unemployed Persons	Unemployed Without Work Experience	Unemployed With Work Experience	Not Stated	< 1	1-6	7-12	13-24	25-36	37-48	49-60	61-72	73-84	85-95	96+	No sta d
45-49	1 6130113	Experience	Experience	Olaleu		T⁼U	1-12	10-24	23-30	<u>07</u> - 1 0	4 3-00	01-72	75-04	00-90	30+	<u> u</u>
Total	44,271	29,960	14,311	-	1,416	13,029	5,606	5,838	3,058	2,417	2,585	1,313	-	-	8,632	37
Male	10,464	9,389	1,076	-	663	3,485	1,366	2,192	275	458	439	452	-	-	1,135	0.
Female	33,807	20,571	13,236	-	753	9,544	4,241	3,646	2,782	1,960	2,146	862	-	-	7,497	37
50-54																
Total	34,995	24,998	9,998	-	533	10,265	6,060	5,657	1,400	628	1,600	355	-	-	8,499	
Male	10,971	8,814	2,157	-	129	3,524	2,847	1,925	314	146	336	-	-	-	1,750	
Female	24,024	16,183	7,841	-	404	6,740	3,212	3,731	1,086	482	1,264	355	-	-	6,750	
55-59																
Total	19,201	13,986	5,215	-	115	7,346	2,154	1,058	1,402	598	1,187	888	-	-	4,452	
Male	6,794	5,587	1,208	-	75	2,924	676	993	513	234	110	757	-	-	514	
Female	12,407	8,400	4,007	-	40	4,422	1,479	66	890	364	1,078	131	-	-	3,938	
60-64																
Total	16,508	13,536	2,973	-	1,668	5,810	1,754	1,223	2,133	482	371	387	287	-	2,394	
Male	7,265	6,527	739	-	319	2,864	1,063	314	1,750	482	-	100	-	-	374	
Female	9,243	7,009	2,234	-	1,349	2,946	691	909	383	-	371	287	287	-	2,020	
65+																
Total	21,220	18,529	2,691	-	176	6,897	3,889	2,483	935	363	1,000	985	369	-	4,122	
Male	14,743	13,615	1,128	-	161	3,981	2,510	1,790	935	363	687	615	66	-	3,635	
Female	6,477	4,914	1,563	-	15	2,916	1,379	693	-	-	313	371	303	-	486	

		Unemploy						Dura	tion of U	nemployr	nent (In I	Months)				
Region and Sex	Total Unemployed Persons	Unemployed Without Work Experience	Unemployed With Work Experience	Not Stated	< 1	1-6	7-12	13-24	25-36	37-48	49-60	61-72	73-84	85-95	96+	Not stated
COUNTRY TOTAL	1 0100110	Experience	Lipoliolioo													
Total	1,125,536	582,277	543,073	186	35,465	405,895	246,242	141,018	64,175	30,539	23,698	14,269	6,438	697	155,681	1,420
Male	375,585	216,446	159,139	-	11,879	150,699	87,837	47,543	21,744	10,507	5,609	3,216	1,516	87	34,843	107
Female	749,950	365,831	383,934	186	23,586	255,197	158,405	93,475	42,432	20,032	18,089	11,053	4,922	610	120,838	1,313
TIGRAY REGION			,				,			.,					.,	
Total	74,758	31,594	43,163	-	5,570	24,956	19,703	11,315	6,136	1,137	539	820	-	-	4,582	-
Male	21,057	12,529	8,528	-	1,931	9,096	4,519	2,855	2,055	-	83	-	-	-	519	-
Female	53,701	19,065	34,635	-	3,640	15,860	15,184	8,460	4,081	1,137	455	820	-	-	4,063	-
AFFAR REGION																
Total	15,066	6,088	8,978	-	10	4,009	5,547	2,378	551	209	218	73	59	73	1,940	-
Male	4,378	2,134	2,244	-	-	1,197	1,507	780	358	-	136	-	-	-	400	-
Female	10,688	3,953	6,734	-	10	2,812	4,040	1,598	193	209	83	73	59	73	1,540	-
AMHARA REGION																
Total	220,336	111,239	109,097	-	7,570	87,449	47,851	24,585	13,419	4,202	4,897	1,663	954	-	27,531	214
Male	73,128	41,008	32,120	-	2,395	31,533	19,062	6,640	3,849	1,241	1,379	377	46	-	6,497	107
Female	147,208	70,232	76,977	-	5,175	55,916	28,789	17,945	9,570	2,961	3,518	1,286	908	-	21,034	107
OROMIA REGION																
Total	264,769	126,146	138,437	186	9,860	94,579	60,204	26,749	13,027	6,779	4,540	1,923	2,158	537	43,962	451
Male	84,133	47,334	36,799	-	3,531	38,046	17,777	8,803	3,304	1,670	1,260	43	1,117	-	8,582	-
Female	180,636	78,812	101,638	186	6,329	56,533	42,427	17,946	9,723	5,109	3,280	1,880	1,041	537	35,380	451
SOMALI REGION																
Total	34,678	15,565	19,114	-	1,514	10,381	7,376	7,606	3,853	894	559	80	120	-	2,297	-
Male	13,735	7,105	6,630	-	593	4,667	3,383	2,420	1,864	393	333	-	-	-	81	-
Female	20,943	8,460	12,484	-	921	5,713	3,993	5,185	1,988	501	226	80	120	-	2,216	-
BENISHANGUL GUMU																
Total	5,470	2,727	2,742	-	112	2,190	764	898	686	280	44	139	76	-	279	-
Male	1,326	1,149	176	-	-	654	249	140	166	74	17	-	-	-	26	-
Female	4,144	1,578	2,566	-	112	1,536	516	758	520	206	27	139	76	-	253	-
SNNP REGION																
Total	90,711	48,066	42,644	-	947	41,901	24,597	7,240	3,093	1,585	2,146	62	277	-	8,864	-
Male	27,660	15,464	12,196	-	234	11,491	9,580	1,879	1,331	869	70	-	66	-	2,141	-
Female	63,050	32,602	30,448	-	713	30,410	15,016	5,361	1,762	716	2,076	62	212	-	6,723	-

Table 6.11 Currently Unemployed Populations of Urban Areas Aged Ten Years and Above by Region, Sex, Previous Work Experience And Duration of Unemployment (In Months) : 2011

Table 6.11 Cont'd

		Unemploy	yed					Dura	ation of U	nemployi	ment(In M	Nonths)				
Region and Sex	Total Unemployed Persons	Unemployed Without Work Experience	Unemployed With Work Experience	Not Stated	< 1	1-6	7-12	13-24	25-36	37-48	49-60	61-72	73-84	85-95	96+	Not stated
GAMBELLA REGION	1 0100110	Exponence	Exponence	otatou			,	10 2 1	20 00	0, 10		0.72		00 /0	,	otatou
Total	2,785	1,338	1,446	-	139	1,026	615	378	134	25	56	-	-	-	412	-
Male	1,368	695	672	-	56	452	383	178	77	-	-	-	-	-	221	-
Female	1,417	643	774	-	82	575	232	200	56	25	56	-	-	-	191	-
HARARI REGION																
Total	7,624	4,780	2,843	-	339	2,844	1,867	543	686	300	151	125	-	-	769	-
Male	2,267	1,494	773	-	70	1,108	473	140	222	38	-	85	-	-	132	-
Female	5,357	3,286	2,071	-	269	1,736	1,394	403	464	262	151	40	-	-	637	-
ADDIS ABABA CITY A	OMINISTRATIO	Ν														
Total	384,317	217,279	167,038	-	8,030	125,956	73,901	54,975	21,018	14,558	10,168	9,141	2,793	-	63,021	755
Male	139,334	81,694	57,641	-	2,680	49,793	29,442	22,027	8,353	6,153	2,330	2,548	287	-	15,721	-
Female	244,982	135,585	109,397	-	5,350	76,162	44,459	32,949	12,665	8,406	7,838	6,593	2,506	-	47,300	755
DIRE DAWA ADMINIST	RATION															
Total	25,024	17,453	7,570	-	1,372	10,605	3,817	4,352	1,573	572	379	243	-	87	2,024	-
Male	7,200	5,839	1,360	-	388	2,661	1,462	1,683	163	69	-	163	-	87	523	-
Female	17,824	11,614	6,210	-	984	7,944	2,355	2,669	1,410	503	379	80	-	-	1,501	-

			1ess: 201	-	Droble	me Face	d to Stor	Own Bus	inoss				
-					FIODIE	пь гасе	Lack of		111622				
						Lack of	working	Lack					
		I have	Shortage		Problem	Finance	Place	/absence	Shortage/				
Region and Sex		no	of	Lack of	of work	and	and	of	absence of	Don't	Lack of		Not
	Total	problem	finance	Training	place	Training	finance	License	equipment	Know	information	Other	Stated
COUNTRY TOTAL					-				• • •		•	•	
Total	316,808	21,553	175,522	2,696	26,890	10,761	34,422	1,549	1,547	943	512	38,980	1,434
Male	78,778	6,398	43,122	962	4,635	2,198	9,101	351	1,115	132	-	10,128	636
Female	238,030	15,155	132,400	1,733	22,255	8,562	25,321	1,198	432	811	512	28,852	798
TIGRAY REGION													
Total	22,809	2,152	16,741	-	726	-	931	83	-	-	-	2,175	-
Male	3,779	264	3,342	-	-	-	-	83	-	-	-	90	-
Female	19,030	1,889	13,399	-	726	-	931	-	-	-	-	2,085	-
AFFAR REGION													
Total	3,710	72	2,075	48	8	73	8	-	-	-	-	1,426	-
Male	910	-	715	19	-	-	-	-	-	-	-	176	-
Female	2,800	72	1,360	30	8	73	8	-	-	-	-	1,250	-
AMHARA REGION													
Total	75,459	4,900	42,689	509	8,349	635	6,581	328	597	51	69	9,758	992
Male	19,195	1,568	12,957	-	1,169	-	838	75	597	-	-	1,797	194
Female	56,264	3,333	29,732	509	7,179	635	5,743	253	-	51	69	7,961	798
OROMIA REGION													
Total	72,989	5,106	39,449	571	2,978	3,344	10,210	-	878	79	443	9,489	442
Male	18,586	1,008	9,607	-	291	446	2,557	-	446	39	-	3,749	442
Female	54,404	4,098	29,842	571	2,687	2,898	7,652	-	432	39	443	5,740	-
SOMALI REGION													
Total	11,844	1,311	6,310	49	84	362	1,273	-	-	-	-	2,456	-
Male	2,516	553	577	-	40	-	461	-	-	-	-	885	-
Female	9,329	758	5,734	49	44	362	812	-	-	-	-	1,571	-
BENISHANGUL GUMUZ	REGION												
Total	747	62	278	-	-	-	27	60	-	21	-	300	-
Male	83	62	-	-	-	-	-	-	-	21	-	-	-
Female	665	-	278	-	-	-	27	60	-	-	-	300	-
SNNP REGION													
Total	29,689	2,253	17,582	67	1,945	1,357	3,637	-	-	141	-	2,707	-
Male	6,783	1,262	3,252	67	428	557	316	-	-	-	-	902	-
Female	22,906	991	14,330	-	1,518	800	3,321	-	-	141	-	1,805	-

Table 6.12Currently Unemployed Population of Urban Areas Aged Ten Years and Above by Region, Sex, Types of Problem Faced
to Establish Own Business: 2011

Table	6.12	Cont'd

					Proble	ems Face	d to Star	t Own Bus	iness				
							Lack of						
						Lack of	working	Lack					
Region and Sex		I have	Shortage		Problem	Finance	Place	/absence	Shortage/				
		no	of	Lack of	of work	and	and	of	absence of	Don't	Lack of		Not
	Total	problem	finance	Training	place	Training	finance	License	equipment	Know	information	Other	Stated
GAMBELLA REGION													
Total	590	25	351	-	10	-	146	17	-	-	-	41	-
Male	202	-	55	-	-	-	146	-	-	-	-	-	-
Female	389	25	296	-	10	-	-	17	-	-	-	41	-
HARARI REGION													
Total	2,639	157	1,301	-	244	41	207	40	-	-	-	648	-
Male	470	-	309	-	-	-	41	40	-	-	-	80	-
Female	2,169	157	992	-	244	41	166	-	-	-	-	568	-
ADDIS ABABA CITY ADMINISTRATION													
Total	87,505	5,067	43,080	1,451	11,708	4,869	10,583	691	-	579	-	9,477	-
Male	25,296	1,682	11,861	877	2,707	1,196	4,595	-	-	-	-	2,380	-
Female	62,209	3,385	31,219	574	9,001	3,674	5,987	691	-	579	-	7,098	-
DIRE DAWA													
ADMINISTRATION													
Total	8,826	447	5,665	-	838	80	821	331	72	72	-	501	-
Male	961	-	448	-	-	-	147	153	72	72	-	69	-
Female	7,865	447	5,218	-	838	80	673	178	-	-	-	432	-

Annex I- III

- Annex I Survey Questionnaire
- Annex II Estimation Procedures of Total Ratio and Sampling Errors
- Annex II Estimates of CV's for selected Tables

SECTION 2: SOCIO-DEMOGRAPHIC CHARACTERSTICS OF HOUSEHOLD

SECTION 3: ECONOMIC ACTIVITY DURING THE LAST 7 DAYS (For members aged 5 years and above) Last 7 days refers to the Seven Days Prior to the Date of Interview.

201	202	203	204	205	206	207	208	209	210	301	302	303	304	305	306	307
				1	For mem	bers aged 5	For membe	ers aged 10 yea	irs	ACTIVITY STATU	s		•	•		-
Serial	For all memeber	rs of the Househol	d	1		ars and above	and o		MARITAL	During the last 7 days was a second in	did	<u>.</u>	e is	Emmenter	Even through one of the	If code 1 - 3 in col.306
Serial No.	Full name of usual household members	RELATIONSHIP	SEX	AGE	EDUCATIO	ONAL STATUS	т	RAINING	STATUS	During the last 7 days were you engaged in any kind of productive activities such as work for payment, for family gain or	hours ne)	ary job?	s? If ther	Enumerator:- Check in Col.302 total	Even though you were not working during the last 7 days, did you	What was the main reason for not
	List names of usual household members in the following order: -Head -Spouse -Never married children according	What is Your Relationship to the Head of Household? 0= Head	= Female	year. If under	Can you read and write? 1= Yes	What was the highest grade you have completed?	Have you received any professional or Vocational/	In what type of profession (Vocational/ technical) are	1=Never married 2=Married	profit, produce for own consumption? <u>Enumerator:</u> Ask respondents "What they have been doing in the last 7 days" then mark code "1" if the respondent was engaged in at least one of the activities mentioned below.	d for how many ncluding overtin	ur main/primary job?	additional job/s?	hours worked in the last 7 days? 1= Less than	have a job, business or holding which you will return to?	working during the last 7 days? 1. Sick, Injury 2. Holiday/ vacation
household member	-never manned cinited according to seniority in age -Married children and their household members who live in this housing unit -Other relatives - Non relatives	1= Spouse 2= Son/daughter of head and spouse 3= Son/daughter of head 4= Son/daughter of	1= Male 2=	age in completed olumn)	$\frac{1}{2} = No$	list of answers and codes	Technical training ? 1= Yes 2= No	you trained?	3= Divorced 4= Widowed 5= Separated	engages in an tests one on the activities memorate below. Mark code 2 if participated in none of the activities. - Work as paid employee for Government/ gov1- parastatal , domestic, NGO, Private organization? - Run a business, big or small, for self or with partners (e.g burber, shoe shining, trade.etc.)? - Work in Agriculture privatelyslaafied (e.g	journey time, in total for how in the last 7 days? (Including o	did you spend in your	in your	4 hours or not asked in col.302 2= 4 hours and above 7	 1= Yes, I have own business/enterprise 2= Yes, paid for duration of absence 3= Yes, with assurance or agreement for 	 Profiday values Training/ Education Off season No work temporarily Other/Specify
number of		spouse 5= Mother/father of head/spouse 6= Sister/brother of head/spouse		your age? (enter enter '00' in the	Go to Col. 208	are provided	Go to Col. 210	and codes are provided		ploughing and Cattle rearing, poultry,etc.)? - Produce goods for sale (e.g 'injera', 'Tella''Sefied'etc.)? - Produce permanent goods for own family? - Help in family farm or buinnes without payment ? eg. In agriculture,trade or in other enterprises.	Excluding lunch and . you work at all jobs i	How many Hours did	How many Hours did you spend no additional Job fill in code "98	Go to Col . 308	4= No Go to Col.401	
Serial		7 = Paid Domestic Workers 8= Other relatives 9= Non-relatives		What is you one year						Engage in construction or major repair for own/family use? other productive activity not mentioned above for own or family gain? 1. Yes 2. No →Col. 306	Week's Hours in all Productiv Activities	Week's Hours e in primary/ main job	Week's Hours in additional job/s/	_		
0 1																
0 2																
03																
04																
0 5																
06																
0 7			-							┃━━━━━□						
0 8																
09																
1 0																
1 1																

L

SECTION 3: LAST 7 DAYS ...(cont'd) (For members aged 5 years and above)

202	201	308		309		310			311	312	313	314	315		316	317	318	319
Full Name		For those who v	worked j			code 2 in col. 1 to(code 1-3			ose who did not work but		ho answered in Col.311	d code 01 - 06	If cod	le 1 - 4 is	filled in Col.314	If code 01 - 06 in Col.311	UNDEREMPL	Kind of additional
(Transfer from page1 , Col. 202)	(10	OCCUPATIO	N INDI	STRV EMPL	WMENT	C STATUS AND	WORK	DI AC	CE FOR THE MAIN JOB	Terms of Employment		EARNI	NGS FR	OM PA	ID EMPLOYM	ENT	Were you seeking and available to work	work sought or available for?
All household members	Line number (Transfer from page 1, Col.201)	What was your main occupation?		Where did you engage your 00= Business ho 01= Office	u work?	What was the product or service of th organization?	najor	s	What was your employment status in your main job? 01= Employee-government 02= Employee - Government parastatal	What are your terms	Are you paid in	What was the period	Based on What wa		Based on Col.314 what was the	What was the total amount paid in your main occupation	for additional hours in the last 7 days? For those not worked	 1=More hours at present job 2=Other job(s) in addition to the
column; out Col.308 - 320 hall be marked "" for members aged below 10 years.	I (Transfer)	<u>Note:-</u> If a person was not engaged at work in the last 7 days, then the occupation before that period		02= At home 03= On street 04= 'Gulit'/Open 05= Farm area/fi 06= Factory 07= Quarrying/n	n market ield	<u>Note:-</u> If a person was no engaged at work i the last 7 days, the the product/servic the organization	n en e of	() () ()	03= Employee - Private Orgn. 04= Employee- NGO or (Int.Organization) 05=Employee - domestic 06= Other employees 07= Member of	of employment in your main occupation? 1. Permanent employee	cash? or in kind? For your main paid emplo- yment?	of payment for the main job? 1= Hourly 2= Daily	the num of () y were paid during th last mon	ou 1 for 1e	amountpaid in your main occupation during that last pay period?	during the last month?	Were you seeking and available to work in the last 7 days?	present job 3= Other job(s) with more hours to replace present one
		will be asked		08=Any where a 09= Other /speci	as found	before that period asked.		0 0 1 1	Co-operatives 08= Self employed 09= Unpaid family worker 10= Employer 11= Apprentice 12=Others(specify) Go to Col.318	 Temporary employee Comployee Casual worker Other /specify 	1= In cash 2= In kind 3= Both	3= Weekly 4= Half month 5= Monthly 6= Yearly 7= Other/ Spec <u>ify</u> Go to Col.31	7				1=Yes 2= No Go to Col.501	Col.501
								-	60 10 C01.518			60 10 C01.51	<u>'</u>					
	01																	
	02																	
	03																	
	04																	
	05																	
	06																	
	07																	
	08																	
	10																	
	11																	

SECTION 4: UNEMPLOYM	ENT AND CHA	RACTE	RISTICS	OF UNEMPLOYED	D PERSO	NS (For those ag	ed 10 years a	and above)		SECTION 5: ECONOMIC ACTIVIT	Y DURING T	HE LA	ST 6 M(ONTHS	(For members a	ge 5 years and above)
202 201 40	402	403	404	405	406	407	408	409	410	501	502	503	504	505	506	507
	ded 4 you taken mainly in search of work or to start your own business?		404 If opportunities to work exist in the coming	405 Are you willing to take up work for wage or salary on locally prevailing terms or ready to undertake self-employment activity given the necessary resources and facility? If the answer is 'Yes' where do you prefer the place of job to be? 1. Not available 2. Within this dwelling. 3. Within residence village town only in Urban areas of this country					410 For how many mo- nths have you been unem- ployed?		502		504		506	
										1. Yes →Go to 503 2 .No		worked	unemployed	1 inactive		END
0 1																
0 2																
		//////												/////		
0 3										I						
0 4																
0 4																
0 5																
										I						
0 6																
										<u> </u>						
0 7																
0 8																
0.9										ll ——						
0 9										{} 						
1 0										11						
10										I						
1 1																
	\sim				1 -					, <u> </u>						
G 1 492			ımn 403				Column 40			4						
Column 402	01. Pregnancy/deliv	ery				I have no problem		07. Shortage/abs	ence of							
1. Searching vacancy advertising boards 2. Through News paper, Radio and TV	02. Illness/Injury 03. Personal/family	responsibility	previous v 09 Thought r			Shortage of finance Lack of training		equipment 08. Don't know								
3. I have unemployment card			•	io work available rivate work, I thought there will		Problem of working pla	ce/land	09. Lack of infor	mation							
4. Seeking assistance of friends, relatives, etc.	05. Old age/Pension			ge of money, raw materialetc.		Lack of finance and tra		10. Others /Speci								
5. Trying to establish own enterprise	06. Education/Train		11. Too young			Lack of working place	0									
6. Direct application to employers	07. Already found/n		12. Remittanc			Lack/absence of Licens										
7. Checking at work sites 8. Others	arrangement for 08. Possiblity to rejo	work	13. Culture/be 14. Others/Sp	lieving that it is the role of Men						-						

Section 6: Informal Sectors/Filtering Q)uestion/
---	-----------

202	201	601	602	603	604	605			606		60	7
Full Name Fransfer from page 1, Col. 202) Il household members are	1, Col.201)	Enumerator Please correctly transfer the name of employment status and their codes from Col 311	Under those transferred from Col.311, please ask only the following codes: 03,06,08,09,10,11 and 12.	How many people including yourself participate in the activities in this enterprise	How many times usually	What are your employment status in your main job? 1= Written agreement n with fixed duratio 2= Written agreement with undetermined duration 3= verbal agreement	What is the are doing th			face when you	Do your	e have
ransferred in this column. eave with "-" for those hildren aged less than	Line number(Transfer from page		 Yes, with detailed accounting system Yes, but not detailed accounting system No 		₩- Casualiy	 3- verbal agreement with fixed duration 4= verbal agreement with undetermined duration 						
years	Lii		4 = I do not know			5=Noting at all	1st Prpblem	2nd P	pblem	3rd Prpblem	_	
							ast i poteili		Porcin	ora i i polem		
	01							_				
	0 2											
	03											
	04											
	05											
	06											
	07											
	08											
	09											
	10											
	11											
			Connormation S				02= Lack of working place 08= Shor 03= Shortage of finance/Capital 09= Lack 04= Lack of working place and finance 10= Shor 05= Shortage of market 11= Lack 06= Shortage of training 12= Pres		ble to get license ortage of credit i ck of raw mater ortage of equipm ck of information ssure from gover ters	facilities ials nent m		

CENTRAL STATISTICAL AGENCY URBAN EMPLOYMENT - UNEMPLOYMENT SURVEY - 2011

101 Kill	102 Zone	103 Wereda	104 Town	105 Keftegna/Wereda/ K/Ketema	106 Kebele	107 Enumeration Area code	108 Household selection Ser.No	109 Name of Head of HH	110 Branch Office	111- Serial No of Enumeration Area	112 Is there continued questionnaire?
											1. Yes 2. No

- - -

	HOUR	N	AINUTI	E
INTERVIEW BEGIN				
INTERVIEW END				

CONFIRMATION

	Name	Signature	Date		Code
Enumerator			/	/	
supervisor			/	/	
Statistician			/	/	

THE CENTRAL STATISTICAL AGENCY URBAN CONTINUING EMPLOYMENT - UNEMPLOYMENT SURVEY - 2011

SECTION	1. AREA IDENTIF	ICATION O	F THE SELE	CTED HOUSEHO	DLD					page 1 of 4 (For Office Use Only)	
101 Kilil	102 Zone	103 Wereda	104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele	107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	111- Serial No of Enumeration Area	112 Is there continued questionnaire?
											1. Yes 2. No
										page 2 of 4 pages	

SECTION											
101 Kilil	102 Zone	103 Wereda	104 Town	105 Keftegna/wereda/	106 Kebele	107 Enumeration Area code 108 Household selection ser.nc 109 Name of Head of HH 110 Branch Office	112- Serial No of				

										page 3 of 4 pages
SECTION 1.	AREA IDENTIFICA	ATION OI	F THE SELEC	TED HOUSEHO	LD					(For Office Use Only)
				105 Keftegna/wereda/						112- Serial No of
101 Kilil	102 Zone	103 Wereda	104 Town	K/Ketema	106 Kebele	107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	Enumeration Area

									-	page 4 of 4 pages
SECTION 1.	AREA IDENTIFICA	TION OF	THE SELEC	FED HOUSEHO	LD					(For Office Use Only)
				105 Keftegna/wereda/						112- Serial No of
101 Kilil	102 Zone	103 Wereda	104 Town	K/Ketema	106 Kebele	107 Enumeration Area code	108 Household selection ser.no	109 Name of Head of HH	110 Branch Office	Enumeration Area
								1		

THE CENTRAL STATISTICAL AGENCY URBAN EMPLOYMENT - UNEMPLOYMENT SURVEY - 2011

SECTION 1. AREA	A IDENTIFI	CATION OF	THE SEL	ECTED HOUSEH	OLD			(For Office Use Only)
101 Kilil	102 Zone	103 Wereda	104 Town	105 Keftegna/wereda/ K/Ketema	106 Kebele	107 Enumeration Area code	110 - Branch Office	111- Serial No of Enumeration Area

For office work only (Branch Office)

	NAME	SIGNATURE	DATE	CODE
Enumerator				
supervisor				
Statistician				
Branch Head				

For office work only (Head Office)

	NAME	SIGNATURE	DATE	CO	DE
Editor					
Editor-verifier					
Encoder					
Encoder-verifier					

For office work only (Branch Office)

Number of Questionnaire

	Name	Signature	Date
Enumerator			
supervisor			
Statistician			

For office work only (Head Office)

Number of Questionnaire

	Name	Signature Date
Editor		
Editor-verifyer		
Encoder		
Encoder-verifyer		

Date

ANNEX II

Estimation Procedures of Total, Ratio and Sampling Errors

The following formulas were used to estimate the required variables by reporting levels.

1. Estimate of Total \hat{Y}_h in Major Urban Domain (Category I)

$$\hat{Y}_{h} = \sum_{i=1}^{n_{h}} \sum_{j=1}^{h_{hi}} W_{hi} Y_{hij}$$

Where,

$$W_{hi} = \frac{M_h H_{hi}}{n_h M_{hi} h_{hi}}$$
 is the basic sampling weight

2. Estimate of Total \hat{Y}_h in Other Urban Domain (Category II)

$$\hat{Y}_{h} = \sum_{i=1}^{n_{h}} \frac{M_{h}}{n_{h} n_{hi}} \sum_{j=1}^{n_{hi}} \frac{H_{hij}}{M_{hij} h_{hij}} \sum_{k=1}^{h_{hij}} Y_{hijk}$$
$$= \sum_{i=1}^{n_{h}} \sum_{j=1}^{n_{hi}} \sum_{k=1}^{h_{hij}} W_{hij} Y_{hijk}$$

Where,

$$W_{hij} = \frac{M_h H_{hij}}{n_h n_{hi} M_{hij} h_{hij}}$$
 is basic sampling weight

The following notations were used in the formula:

 M_{h} = Total number of households in stratum h obtained from the sampling frame.

- M_{hi} = Total number of households in EA/PSU i for major urban domain or in urban center/PSU i for other urban domain, stratum h obtained from the sampling frame.
- H_{hi} = Total number of households obtained from the survey listing in sample EA/PSU i stratum h for major urban domain.

- h_{hi} = Total number of households successfully covered in EA/PSU i stratum h for major urban domain.
- M_{h} = Total number of households of the domain in stratum h obtained from the sampling frame
- n_h = Number of successfully covered urban centers for other urban domain / covered EAs for major urban domain in stratum h.
- M_{hi} = Total number of households in EA/PSUi in stratum h obtained from the sampling frame
- $h_{hi=}$ = Total number of sampled and covered households in sampled EA/PSU i , and stratum h
- \hat{Y}_{hij} = The observed value of characteristic y for household j, in EA/PSUi, stratum h of Major Urban Domain

$$M_{hij}$$
 = Total number of households in EA/SSU j, urban centers/PSU i and stratum h obtained from the sampling frame for other urban center domain.

- n_{hi} = Number of sample EAs successfully covered in urban center/PSU i and stratum h for other urban center domain.
- H_{hij} = Total number of households obtained from the survey listing in EA/SSU j, urban center/PSU i and stratum h for other urban center domain
- h_{hij} = Number of sample households successfully covered in EA /SSU j, urban center/PSU i and stratum h for other urban center domain.
- Y_{hijk} = The observed value of a characteristic y for household k in EA/SSUj, urban center/PSU i and stratum h for other urban domain.
- Note: Estimate of total at country level, \hat{Y} , is obtained by summing up stratum/domain total estimates.

$$\hat{Y} = \sum_{h} \hat{Y}_{h}$$

4. Sampling Variance of the Estimates:

Sampling variance of estimate of stratum total are given by the following formulas: The variance of domain or reporting total estimate is:

$$Var(\hat{Y}_{h}) = (1 - f_{h}) \frac{n_{h}}{n_{h} - 1} \sum_{i=1}^{n_{h}} \left(\hat{Y}_{hi} - \frac{\hat{Y}_{h}}{n_{h}} \right)^{2} + f_{h} \sum_{i=1}^{n_{h}} (1 - f_{hi}) \left(\frac{h_{hi}}{h_{hi} - 1} \right) \sum_{j=1}^{h_{hi}} \left(\hat{Y}_{hij} - \frac{\hat{Y}_{hi}}{h_{hi}} \right)^{2}$$

in which $\hat{Y}_{hi} = W_{hi} \sum_{j=1}^{h_{hi}} Y_{hij}$ for major urban centers domains,

and
$$\hat{Y}_{hi} = \sum_{j=1}^{n_{hi}} W_{hij} \sum_{k=1}^{h_{hij}} y_{hijk}$$
 for other urban center domain.
 $V(\hat{Y}) = \sum_{h} V(\hat{Y}_{h})$

$$SE(\hat{Y}_h) = \sqrt{Var(\hat{Y}_h)}$$

5. Coefficient of Variation (CV) and Confidence Interval (CI)

The following formulas were used to calculate CV and CI of the domain (reporting level) total.

The coefficient of variation (CV) of domain total in percentage is:

$$CV(\hat{Y}_h) = \frac{\sqrt{VAR(\hat{Y}_h)}}{\hat{Y}_h} *100$$

and Ninety-five percent confidence interval (CI) of domain total:

 $\hat{Y}_h \pm 1.96 * SE(\hat{Y}_h)$

6. Ratio Estimates:

$$\hat{R}_h = \frac{\hat{Y}_h}{\hat{X}_h} and \ \hat{R} = \frac{\hat{Y}}{\hat{X}}$$

Where the numerator and the denominator are estimates of domain totals of characteristic y and x, respectively.

$$Var\left(\hat{R}_{h}\right) = \frac{1}{\hat{X}_{h}^{2}} \left[Var\left(\hat{Y}_{h}\right) + \hat{R}_{h}^{2} Var\left(\hat{X}_{h}\right) - 2\hat{R}_{h} Cov\left(\hat{Y}_{h}, \hat{X}_{h}\right) \right]$$

In which

$$Cov(\hat{Y}_{h}, \hat{X}_{h}) = (1 - f_{h})\frac{n_{h}}{n_{h} - 1}\sum_{i=1}^{n_{h}} \left(\hat{Y}_{hi} - \frac{\hat{Y}_{h}}{n_{h}}\right) \left(\hat{X}_{hi} - \frac{\hat{X}_{h}}{n_{h}}\right) + f_{h}\sum_{i=1}^{n_{h}} (1 - f_{hi}) \left(\frac{h_{hi}}{h_{hi} - 1}\right) \sum_{j=1}^{h_{hi}} \left(\hat{Y}_{hij} - \frac{\hat{Y}_{hi}}{h_{hi}}\right) \left(\hat{X}_{hij} - \frac{\hat{X}_{hi}}{h_{hi}}\right) \left(\hat{X}_{hij} - \frac{\hat{X}_{hi}}{h_{hi}}\right)$$

	(Current Status Approach) – COUNTRY TOTAL: 2011												
Age Group			•••		Non								
and Sex	All Persons	CV	Active	CV	Active	CV							
All Ages													
Total	10397080	1.41	6265367	1.57	4131249	1.84							
Male	4866065	1.52	3304538	1.64	1561317	2.21							
Female	5531015	1.47	2960829	1.79	2569931	1.96							
10-14													
Total	1513511	2.23	163037	7.66	1350474	2.31							
Male	713106	2.72	79007	9.78	634099	2.74							
Female	800405	2.51	84029	7.96	716375	2.72							
15-19													
Total	1735484	2.00	587006	3.16	1148479	2.41							
Male	751452	2.57	250346	4.47	501106	3.09							
Female	984032	2.24	336660	3.57	647372	2.73							
20-24													
Total	1556477	1.86	1106287	2.29	450029	3.10							
Male	679531	2.53	510306	3.03	169166	4.37							
Female	876947	2.20	595982	2.70	280863	3.53							
25-29													
Total	1470711	1.81	1236396	1.92	234315	4.49							
Male	699615	2.25	647935	2.37	51680	8.23							
Female	771096	2.26	588461	2.52	182634	4.77							
30-34													
Total	957508	2.61	845564	2.74	111944	5.84							
Male	498411	3.23	481734	3.34	16677	1.06							
Female	459097	3.00	363830	3.27	95267	6.38							
35-39													
Total	850679	2.49	751307	2.52	99372	6.59							
Male	401785	3.18	390114	3.19	11672	0.68							
Female	448894	2.86	361194	3.13	87700	6.62							
40-44													
Total	542611	2.55	467174	2.67	75394	6.65							
Male	298454	3.36	286581	3.37	11830	0.61							
Female	244158	3.34	180594	3.86	63564	6.75							
45-49													
Total	440017	2.62	370591	2.83	69426	6.44							
Male	215355	3.67	207346	3.76	8009	0.46							
Female	224662	3.65	163246	4.29	61416	6.80							
50-54			0/7554	~ • • •									
Total	347848	3.07	267551	3.46	80250	5.77							
Male	161285	4.23	149792	4.39	11493	0.60							
Female	186562	4.35	117760	5.51	68757	6.12							
55-59	0.45504	0.05	470400	4 5 0	70040								
Total	245521	3.85	172102	4.50	73312	6.34							
Male	113124	5.08	99621	5.48	13503	0.76							
Female	132397	4.88	72481	6.80	59809	7.02							
60-64	224//0	2 55	104700	4.00	100044	F 4F							
Total	234669	3.55	124728	4.89	109941	5.15							
Male	104252	5.68	80140	6.38	24112	1.30							
Female	130417	4.45	44589	7.77	85829	5.80							
65+	F00044	0.40	170/00		000045	2.00							
Total	502044	3.18	173622	4.64	328315	3.98							
Male	229695	3.90	121618	5.02	107970	6.04							
Female	272349	4.13	52004	8.00	220345	4.57							

ANNEX TABLE 1. Urban Population Aged Ten Years And Over By Age Group, Sex and Activity Status during the Last Seven Days (Current Status Approach) – COUNTRY TOTAL: 2011

	Not During the Last Seven Days (Current Status Approach): 201										
Age Group	Economically		-								
and Sex	Active	CV	Employed	CV	Unemployed	CV					
All Ages											
Total	6,265,367	1.57	5,139,831	1.60	1,125,536	2.99					
Male	3,304,538	1.64	2,928,953	1.66	375,585	4.07					
Female	2,960,829	1.79	2,210,879	1.96	749,950	3.12					
10-14											
Total	163,037	7.66	144,811	8.43	18,226	15.51					
Male	79,007	9.78	71,614	10.46	7,394	20.23					
Female	84,029	7.96	73,197	8.78	10,832	20.46					
15-19											
Total	587,006	3.16	429,517	3.66	157,489	5.46					
Male	250,346	4.47	192,665	5.04	57,681	8.03					
Female	336,660	3.57	236,852	4.45	99,808	6.26					
20-24	,										
Total	1,106,287	2.29	797,478	2.50	308,809	3.81					
Male	510,306	3.03	404,296	3.36	106,010	5.70					
Female	595,982	2.70	393,182	3.07	202,800	4.30					
25-29	575,702	2.70	575,102	5.07	202,000	ч. 5 0					
Z5-29 Total	1,236,396	1.92	1,009,623	2.09	226,773	3.73					
Male	647,935	2.37	579,802	2.09	68,133	6.23					
Female	588,461	2.52	429,821	2.80	158,640	4.58					
30-34		0.74	710 007	2.05	10/ 0/7	F 00					
Total	845,564	2.74	719,297	2.95	126,267	5.23					
Male	481,734	3.34	439,566	3.46	42,168	8.92					
Female	363,830	3.27	279,731	3.68	84,099	6.15					
35-39											
Total	751,307	2.52	656,750	2.65	94,558	5.96					
Male	390,114	3.19	363,332	3.27	26,782	12.04					
Female	361,194	3.13	293,418	3.49	67,776	6.86					
40-44											
Total	467,174	2.67	409,957	2.80	57,218	7.49					
Male	286,581	3.37	269,401	3.43	17,180	13.13					
Female	180,594	3.86	140,556	4.39	40,038	8.61					
45-49											
Total	370,591	2.83	326,320	3.03	44,271	8.09					
Male	207,346	3.76	196,881	3.77	10,464	16.49					
Female	163,246	4.29	129,439	4.78	33,807	9.40					
50-54											
Total	267,551	3.46	232,556	3.59	34,995	11.21					
Male	149,792	4.39	138,821	4.53	10,971	16.04					
Female	117,760	5.51	93,735	5.88	24,024	12.78					
55-59	,	0101	/0//00	0.00	2.,02.						
Total	172,102	4.50	152,901	4.70	19,201	11.82					
Male	99,621	5.48	92,826	5.74	6,794	20.18					
Female	72,481	6.80	60,074	7.19	12,407	15.13					
	72,401	0.00	00,074	7.17	12,407	15.15					
60-64	101 700	1 00	100 220	E 01	16 500	10 40					
Total Mala	124,728	4.89	108,220	5.21	16,508	13.69					
Male	80,140	6.38	72,874	6.47	7,265	20.74					
Female	44,589	7.77	35,346	8.41	9,243	19.65					
65+	470 /00		450.400		04.000	10.07					
Total	173,622	4.64	152,402	4.78	21,220	12.86					
Male	121,618	5.02	106,875	5.09	14,743	15.29					
Female	52,004	8.00	45,527	8.56	6,477	19.80					

ANNEX TABLE 2. Economically Active Urban Population of Major Towns Aged Ten Years And Over By Age Group, Sex, Whether Employed or Not During the Last Seven Days (Current Status Approach): 2011

	Amount of Total Payment (in Birr)																	
Period of Payment	Total Paid Employees	CV	<50	CV	50-99	CV	100-199	CV	200-399	CV	400-699	CV	700-999	CV	1000+	CV	Not Stated	CV
Total																		
Total	2,544,615	1.95	16,746	16.86	51,204	8.62	144,560	5.32	374,906	3.36	535,600	3.62	395,370	3.59	1,010,204	2.84	16,025	17.13
Male	1,537,512	1.97	2,625	35.39	11,687	16.15	36,681	10.55	138,215	5.22	316,242	4.00	268,832	4.15	754,454	2.78	8,775	21.37
Female	1,007,104	2.62	14,121	17.63	39,517	9.60	107,879	6.15	236,691	3.81	219,358	4.81	126,537	5.83	255,750	4.45	7,250	23.79
Hourly																		
Total	24,082	16.13	39	100.71	1,548	51.94	2,257	36.47	2,836	38.14	6,041	26.38	4,719	27.99	6,642		-	-
Male	18,213	15.40	-	-	720	58.19	1,202	50.10	2,517	42.57	3,648	25.90	4,082	29.33	6,045	24.32	-	-
Female	5,869	33.81	39	100.71	828	65.94	1,056	53.18	319	46.55	2,392	47.28	637	87.59	597	59.69	-	-
Daily																		
Total	126,225	6.50	2,659	30.03	2,901	29.02	6,284	23.78	18,991	13.66	39,579	12.03	29,112		26,626	10.71	73	99.71
Male	100,525	6.59	788	42.31	2,352	31.96	4,752	25.51	11,136	15.52	29,486	12.70	26,073	12.86	25,865	10.95	73	99.71
Female	25,699	13.60	1,871	38.80	549	64.83	1,532	38.41	7,856	24.07	10,093	20.99	3,039	35.90	760	47.12	-	-
Weekly																		
Total	106,716	9.45	3,058	59.25	667	69.83	5,934	23.37	17,081	13.90	32,945	13.93	21,750	17.19	24,659	12.22	622	-
Male	87,146	9.46	1,070	70.95	37	101.40	3,074	33.80	10,879	16.66	27,260	14.39	20,176	18.19	24,100	12.42	550	-
Female	19,569	15.10	1,988	67.13	630	73.67	2,860	27.39	6,201	19.24	5,685	23.39	1,574	39.78	559	74.39	72	-
Fortnight																		
Total	87,174	7.12	493	92.03	2,382	41.37	4,014	27.33	13,681	13.85	28,989	11.61	18,049	13.90	19,246	12.54	320	93.00
Male	61,553	8.43	41	98.55	527	86.78	301	77.02	4,577	24.75	21,439	13.02	15,795	15.19	18,552	12.97	320	93.00
Female	25,622	11.30	452	99.89	1,854	32.08	3,713	28.00	9,104	16.03	7,550	20.75	2,254	32.90	693	26.08	-	-
Monthly																		
Total	2,151,919	2.12	8,181	19.99	39,297	8.66	121,445	6.07	314,939	3.73	419,660	3.81	312,786	3.72	923,406	3.04	12,205	19.23
Male	1,233,109	2.14	446	69.25	5,888	21.84	24,826	13.09	104,090	6.20	226,853	4.24	194,535	4.35	670,815	3.01	5,656	24.59
Female	918,810	2.78	7,735	20.77	33,409	9.69	96,619	6.62	210,849	4.11	192,808	5.03	118,251	5.93	252,591	4.48	6,548	25.44
Yearly																		
Total	9,297	20.20	704	70.63	2,116	44.37	1,142	46.95	2,047	38.52	946	53.85	-	-	2,342	31.98	-	-
Male	7,499	23.16	-	-	1,744	51.14	973	53.40	1,647	41.23	794	61.24	-	-	2,342	31.98	-	-
Female	1,798	40.70	704	70.63	373	79.00	169	78.38	400	100.22	152	99.67	-	-	-	-	-	-
Other/specify																		
Total	36,880	16.44	1,611	60.45	2,293	51.11	3,485	25.68	5,331	25.44	7,440	26.46	8,953	39.12	7,284	19.61	482	99.90
Male	27,772	20.36	280	99.82	420	70.69	1,554	30.49	3,369	36.51	6,762	28.46	8,171	42.49	6,734	21.37	481.88	99.90
Female	9,108	22.99	1,331	70.09	1,874	60.52	1,931	39.33	1,962	29.76		61.03	782	58.03		52.21	-	-
Not Stated			•															
Total	2,323	46.59	-	-	-	-	-	-	-	-	-	-	-	-	-	-	2,323	46.59
Male	1,694	50.96	-	-	-	-	-	-	-	-	-	-	-	-	-	-	1,694	50.96
Female	629	70.60	-	-	-	-	-	-	-	-		-	-	-	-	-	629	70.60

ANNEX TABLE 3 Paid Employees of Urban Population Urban Areas Aged Ten Years and Over by Sex, Period of Payment and Amount Of Total Payment – COUNTRY TOTAL: 2011